Performance-oriented Integration of Combined Transport into Supply Chain Concepts

D I S S E R T A T I O N of the University of St.Gallen, School of Management, Economics, Law, Social Sciences and International Affairs to obtain the title of Doctor of Philosophy in Management

submitted by

Julia Christine Bendul

from

Germany

Approved on the application of

Prof. Dr. Wolfgang Stölzle

and

Prof. Dr. Karl Frauendorfer

Dissertation Nr. 3947

KDD Kompetenzzentrum Digital-Druck GmbH, 2012

The University of St.Gallen, School of Management, Economics, Law, Social Sciences and International Affairs hereby consents to the printing of the present dissertation, without hereby expressing any opinion on the views herein expressed.

St.Gallen, October 26, 2011

The President:

Prof. Dr. Thomas Bieger

Abstract

The thesis in hand develops a conceptual framework and offers practical recommendations on the performance-oriented integration of combined transport (CT) into supply chain (SC) concepts to overcome recent challenges in unimodal road transport, including sustainability and performance orientation.

Recent SC concepts are characterised by attempts to reduce inventory and to simultaneously increase flexibility and reliability. Therefore, the procurement, production and distribution concepts of shippers, suppliers and third party logistics service providers are increasingly connected and synchronised to meet the demands of end customers. At the same time, companies focus on financial key figures and increasing corporate value. SC activities are evaluated according to their effects on company value, too. The analysis of recent SC concepts shows that transport actors are often not integrated members of these SC concepts. The thesis in hand exploits this imperfection for the integration of CT into SC concepts. It is assumed that the tight integration of material and information flows opens up time buffers and increases coordination flexibility necessary for CT integration.

Current developments make the combination of rail and road transport a promising possibility to solve the conflict between recent forces, such as sustainability, problems in unimodal road transport and changing SC performance (SCP) requirements – even for distances below 100 km. Nevertheless, CT means the involvement of additional SC actors and additional processes.

A conceptual framework based on configuration theory provides propositions on the integration of CT into different SC concept types. The framework proposes an innovative perspective on the integration problem. The central assumption is that aiming at SCP an adaptation of the SC concept oriented to the characteristics of CT is necessary. The thesis provides practical recommendations on concrete adaptation points to improve the fit between SC and CT concept and the usage of integrative measures and instruments.

A simulation study based on the SC concept of a Swiss retailer enhances and validates the findings of the conceptual framework. The effect on SCP of the integration of regular line CT into a lean SC concept is analysed using an object-oriented, discrete event simulation. Specifically, the simulation study analyses the influence of adaptations on the distribution concept and production concept to reveal time buffers and to reduce coordination effort between SC actors.

Zusammenfassung

Moderne Supply Chain (SC)-Konzepte sind durch den Versuch gekennzeichnet, Lagerbestände weitestgehend zu minieren und gleichzeitig Flexibilität und Zuverlässigkeit zu erhöhen. Dabei werden die Beschaffungs-, Produktions- und Distributionskonzepte verladender Unternehmen sowie von Zulieferern und Logistikdienstleistungsunternehmen immer stärker integriert und synchronisiert. Gleichzeitig konzentrieren sich die Unternehmen zunehmend auf finanzielle Kennzahlen und richten ihre Aktivitäten auf eine kontininuierliche Steigerung des Unternehmenswertes aus. Auch SC-Aktivitäten werden hinsichtlich ihrer so genannten Performance-Wirkung analysiert. Die Analyse aktueller SC-Konzepte zeigt, dass Transportakteure oftmals keine integrierten Mitglieder dieser SCs sind. Die vorliegende Arbeit erschliesst diese Lücke für die Integration Kombinierter Verkehre (KV) in die SC-Konzepte, um so Nachhaltigkeitsforderugen und Problemen im Strassenverkehr zu begegnen. Zwar erfordert die Integration von KV auch die Einbindung weiterer SC-Akteure sowie zusätzliche Prozesse. Jedoch wird angenommen, dass die enge Integration der Material- und Informationsflüsse benötigte Zeitkontingente eröffnet und die unternehmensübergreifende Koordinationsflexibilität auf das erforderliche Mass erlaubt.

Der auf der Konfigurationstheorie basierende konzeptionelle Rahmen zeigt eine Reihe von Propositionen zur Integration unterschiedlicher KV-Konzepte in verschiedene SC-Konzepttypen hinsichtlich der Wirkung auf unternehmensübergreifende SC-Performance. Mit der zentralen Annahme, dass eine gezielte Ausrichtung des SC-Konzepts auf die Charakteristika des KV zu einer Erhöhung der SC-Performance führt, wird in der vorliegenden Arbeit eine innovative Sichtweise auf das Integrationsproblem präsentiert. Dazu liefert die Arbeit konkrete Handlungsempfehlungen und zeigt Anpassungspunkte auf, um den so genannter "Fit" zwischen SC-Konzept, KV-Konzept und konkreten, situationsspezifischen Massnahmen und Instrumenten zu verbessern.

Eine auf dem SC-Konzept eines Schweizerischen Handelsunternehmens basierende Simulationsstudie erweitert und validiert die Ergebnisse des entwickelten konzeptionellen Rahmens. Mittels einer objektorientierten, ereignisdiskreten Materialflusssimulation wird die SC-Performance-Wirkung der Integration eines KV-Konzeptes, das auf regelmässigen und terminierten Zugverkehren basiert, auf eine leane SC analysiert. Dabei werden insbesondere der Einfluss bestimmter Anpassungen von Beschaffungs-, Produktions- und Distributionskonzepten untersucht, um so für die KV-Integration erforderliche Zeitkontingente zu erschliessen und den Koordinationsaufwand zwischen den SC-Akteuren zu reduzieren. **Table of Contents**

Ab	stract	Ι
Zus	sammenfassung	П
Tal	ole of Contents	Ш
Lis	t of Figures	VI
Lis	t of Tables	VIII
Lis	t of Abbroviations	v III V
LIS	t of ADDI eviations	Λ
1	Integration of Combined Transport into Supply Chain Concepts from a Derformance Perspective: Need for Action	1
	a renormance rerspective. Need for Action	I
1.1	Problem Formulation and Research Questions	1
1.2	Scientific-Theoretical Positioning and Research Design	5
1.3	Research Process and Thesis Outline	8
2	Problem Concretisation - SCP Orientation Encourages the Integration of CT into SC Concepts	12
2.1		12
2.1	Needs, Challenges and Instruments for SC Integration	12
	2.1.1 Need for SC Integration	13
	2.1.2 Central Elements and the Fields of SC Integration 2.1.3 Normative Operative and Strategic Challenges of SC Integration	17
	2.1.3.1 Overview on SC Integration Challenges	18
	 2.1.3.2 Challenge 1: Main Objectives of Integrated SCs – Problems of SCM Conception 2.1.3.3 Challenge 2: Differing Operative Objectives and Problems of Shippers and CT 	19
	Actors	24 20
	2.1.5.4 Channenge 5. Focus on individual Company Value Growth 2.1.4 Vulnerability of Integrated SC Concepts	29
	2.1.5 Measures and Instruments for SC Integration	33
2.2	Coping with CT Complexity in SCs	41
	2.2.1 Application Fields of CT	41
	2.2.2 Service Complexity of Combined Transport Concepts	46
	2.2.2.1 Innerent Chanenges of Conformed Transport	40 47
	2.2.2.3 Diversity of CT Actors	49
	2.2.2.4 Processes and Structure of CT Services	53
	2.2.2.5 Service Character of CT Services	55
	2.2.3 External Factors Increasing the Complexity of CT	56
	2.2.3.1 Changing Shippers' Performance Requirements Challenging CT Services	50 58
	2.2.3.2 Competitive Situation of CT and Unimodal Road Transport	66
	2.2.4 Integrated Management of CT Concepts	68
2.3	SCP Orientation – Consideration of Strategic and Operational Key Figures	73
	2.3.1 Fundamentals of Performance Orientation	74
	2.3.2 Performance Understanding in a SC Context	77 91
	 2.3.5 Suitability of SCP as a Target System for the Integration of CT into SC Concepts 2.3.4 Development of a Target System for the Evaluation of CT Integration into SC Concepts 	81 87
2.4	Configuration Theory as Theoretical Research Approach for Integrating CT into SC Concepts	89
	2.4.1 Meaning of Theory Application for Application-Oriented Research	89
	2.4.2 Theoretical Solution Approaches towards SC Integration	90 02
	2.4.5 Selection of Theorem and Approaches with regard to the Research Froblem and Perspective	74

	 2.4.4 Configuration Theory as an Explanation Approach for the Performance-oriented CT Integration into SC Concepts 2.4.4.1 Basics of Situative Theory 2.4.4.2 Configuration Theory as an Advancement of Situative Theory 2.4.4.3 Application of Configuration Theory to the Problem of the Performance-Oriented Integration of CT into SC Concepts 	94 94 95 98
2.5	Intermediate Findings	99
3	CT as an Element of SC Concepts	102
3.1	SC Concepts as the Operationalisation of SC Strategies	102
	 3.1.1 SC Strategies as the Basis for SC Concept Configuration 3.1.2 Classification Approaches for SC Strategies 3.1.3 Impact of SC Strategy on the Configuration of SC Sub-Concepts 3.1.4 Elements and Structures of SC Concepts 3.1.4.1 SC Concept Configuration in a Lean SC 3.1.4.2 SC Concept Configuration in an Agile SC 3.1.4.3 SC Concept Configuration in a Leagile SC 	102 105 110 110 113 115 117
3.2	Development of a Classification of SCP Requirements regarding Transport Concepts	119
3.3	Configuration of SC Concepts	126
	 3.3.1 Transport Concepts as Linking and Embedded Elements of SC Concepts 3.3.2 Impact of Procurement Concepts on the Configuration of Transport Concepts 3.3.2.1 Meaning of the Procurement Concepts for SCP 3.3.2.2 Structures and Processes of Procurement Concepts 3.3.2.3 Classification and Configuration of Procurement Concepts – Identification of 	126 131 131 133
	3.3.3 Impact of Production Concept Configuration on Transport Concepts 3.3.3.1 Meaning of the Production Concept for SCP 3.3.3.2 Structures and Processes of Production Concepts 3.3.3.3 Classification and Configuration of Production Concepts – Identification of Design	135 141 141 142
	 3.3.4 Impact of Distribution Concepts on the Configuration of Transport Concepts 3.3.4.1 Meaning of the Distribution Concept for SCP 3.3.4.2 Structures and Processes of Distribution Concepts 3.3.4.3 Classification and Configuration of Distribution Concepts – Identification of Design Variables relevant for CT Integration 	143 148 148 149 153
3.4	Development of a CT Concept Typology - Classification using SCP Profiles	156
3.5	Intermediate Findings	160
4	Conceptual Research Framework of Performance-oriented CT Integration into SC Concepts	163
4.1	Central Aspects of the Development of Research Propositions on the Performance-oriented CT Integration	163
4.2	Development of a Conceptual Research Framework and Deduction of Propositions on the Performance-oriented CT Integration into SC Concepts	165
	 4.2.1 Identification of Central Constructs and Framework Conditions 4.2.2 Middle Range Constructs — Elements and Dimensions of Central Constructs 4.2.3 Manifest, Observable Variables 4.2.4 Moderating Variables of CT Integration into SC Concepts 	165 167 169 172
4.3	Cause-and-Effect Relationships between the Elements of CT and SC Concepts - Deduction of Methodological and Instrumental Implications	174
	 4.3.1 Step 1: Identification of Integration Points for CT into SC Concepts 4.3.2 Step 2: Impact of SC Strategy, SC Concept Configuration and SCP Requirements Profiles on CT Integration 	174 176

	4.3.3	Step 3: Identification of Adaptation Points for CT Concept Integration	179
	4.3.4	Step 4: Situation-specific Choice of Integrative Measures and Instruments for Performance oriented CT Integration into SC Concepts	e- 183
4.4	Interm	ediate Findings	191
5	Douf	umanas suiseted Integration of Combined Line Transport into	
3	a Lea	in SC - A Simulation Study	194
5.1	Resear	ch Design for the Simulation Study	195
	5.1.1 5.1.2 5.1.3 5.1.4 5.1.5 5.1.6	Scientific-theoretical Positioning of the Simulation Methodology Advantages, Restrictions and Application Fields of Simulation Methodology Quality Criteria for Simulation Studies Structure and Classification of Simulation Models Application Fields of the Simulation Methodology in Transport and SCM Science Discrete Event Simulation as the Research Methodology for the Analysis of Performance-oriented Integration of CT into SC Concepts	195 196 199 201 203 205
5.2	Proces	s Model for the Simulation Model Development	207
5.3	Step 1:	Problem Formulation and Target Definition	209
	5.3.1 5.3.2 5.3.3 5.3.4 5.3.5	Scope and Target Definition Problem Formulation - Understanding the Initial SC Concept CT Integration Scenarios Performance Indicators for the Evaluation of Simulation Scenarios Development of Statements on Expected Simulation Results	209 210 212 215 217
5.4	Step 2:	Data Gathering and Editing for Model Configuration and Validation	220
5.5	Step 3:	Development, Validation and Verification of the Simulation Model	229
	5.5.1 5.5.2 5.5.3	Conceptual Model Simulation Model Validation and Verification	230 235 238
5.6	Step 4:	Experimental Plan and Setup of Experiments	241
5.7	Step 5:	Description and Analysis of Simulation Results	242
	5.7.1 5.7.2 5.7.3 5.7.4 5.7.5 5.7.6	Basic Scenario – Modelling the 'as-is' Situation CT Scenario I - Integration without Adaptation to the SC Concept CT Scenario II – Production Concept Adaptation CT Scenario III – Distribution Concept Adaptation Discussion of Simulation Results Limitations, Validity and Generalisability of Findings	242 252 262 270 277 285
5.8	Interm	ediate Findings	288
6	Impl Integ	ications for Science and Practice on the Performance-oriented ration of CT into SC Concepts	291
6.1	Manag	ement Implications	291
6.2	Scienti	fic Implications	294
6.3	Limita	tions and Further Research	296
Ref	ference	es	302

List of Figures

Figure 1 - 1: Working definitions of central constructs.	4
Figure 1 - 2: Application-oriented research process and thesis outline.	11
Figure 2 - 1: Linking of formal and content targets of SCM.	22
Figure 2 - 2: Differences in the operative challenges of selected SC actors.	25
Figure 2 - 3: Road infrastructure shortages in Germany.	31
Figure 2 - 4: Management components of SC integration.	33
Figure 2 - 5: Overview of the technical and organisational management components derived from	
the literature.	34
Figure 2 - 7: Alpine transit by unaccompanied CT.	45
Figure 2 - 8: Schematic depiction of CT material (black) and information flow and additional planning and control processes (blue).	53
Figure 2 - 9: Overview of recent and expected CT terminal capacity shortages in Europe.	64
Figure 2 - 10: Schematic costs for unimodal, combined and multimodal road transport concepts.	67
Figure 2 - 11: Efficiency and effectiveness dimensions of performance.	76
Figure 2 - 12: Characterisation and ideal cause-and-effect relationships between the targets of SCP management.	80
Figure 2 - 13: Performance perspective for the integration of CT into SC concepts.	86
Figure 2 - 14: Target system of SCP and performance indicators for the integration of CT into SC concepts.	88
Figure 3 - 1: Generic SC strategies.	109
Figure 3 - 2: Transport structure for full and less than full truckloads.	129
Figure 3 - 3: SC structure for a packaged goods transport.	130
Figure 3 - 4: Generic procurement strategies according to supply risk and profit impact.	134
Figure 3 - 5: Overview of incoterms 2010.	139
Figure 3 - 6: Basic structures of distribution systems.	151
Figure 4 - 1: Central constructs of the CT integration into SC concepts.	167
Figure 4 - 2: Middle range constructs of the performance-oriented CT integration into SC concepts.	168
Figure 4 - 3: Manifest, observable variables of CT integration into SC concepts.	170
Figure 4 - 4: Conceptual research framework of CT integration into SC concepts.	172
Figure 4 - 5: Integrative measures and instruments as moderating variables for CT integration.	173
Figure 4 - 6: Schematic illustration of the integration points of CT concepts.	175
Figure 4 - 7: General applicability of CT concepts according to different SC strategies.	177
Figure 5 - 1: Schematic of the as-is situation – <i>Manor</i> distribution network (Western Switzerland).	212
Figure 5 - 2: Schematic scenario – Manor distribution network including CT (Western Switzerland).	214
Figure 5 - 3: Performance indicators for the evaluation of simulation scenarios.	216
Figure 5 - 4: Comparison of CO ₂ , NO _X and particle emissions for different transport modes (under consideration of all climate-related effects of air cargo).	222
Figure 5 - 5: Example demand history for delivery to the store in Geneva from DC Hochdorf for prioritised goods from January – September 2010 in pallet equivalents.	227
Figure 5 - 6: Demand in pallet equivalents per weekday for prioritised goods for the store in Yverdon from DC Möhlin.	228
Figure 5 - 7: Conceptual model simulation scenario with CT.	231
Figure 5 - 8: Data model for the simulation study (simplified UML notation).	232
Figure 5 - 9: Screenshot from the simulation model in Flexsim.	236
Figure 5 - 10: Example inventory course for delivery to the Chavannes store with goods from DC Hochdorf.	237
Figure 5 - 11: Conceptual model validation and verification.	239
Figure 5 - 12: Mean utilisation per primary transport route and weekday for the normal demand situation	243

Figure 5 - 13:	Mean utilisation per additional transport route and weekday for the normal demand situation.	244
Figure 5 - 14:	Average utilisation of primary transport vehicles for different transport routes and demand situations.	245
Figure 5 - 15:	Average utilisation of primary transport vehicles for different transport routes and demand situations.	245
Figure 5 - 16:	Transport time elements per store for DC Hochdorf.	246
Figure 5 - 17:	Transport time elements per store for DC Möhlin.	247
Figure 5 - 18:	Average lead time elements for different demand situations.	247
Figure 5 - 19:	Average inventory in the transport between DC and the stores.	248
Figure 5 - 20:	Average inventory for different demand situations.	249
Figure 5 - 21:	CO_2 emissions per transport route.	250
Figure 5 - 22:	Average deviation and delay for different stores and DCs.	251
Figure 5 - 23:	Utilisation of trains for CT scenario I.	255
Figure 5 - 24:	Transport time elements per store for DC Hochdorf for CT scenario I and the normal demand situation.	255
Figure 5 - 25:	Transport time elements per store for DC Möhlin for CT scenario I and the normal demand situation.	256
Figure 5 - 26:	Comparison between average lead time elements between basic and CT scenario I.	256
Figure 5 - 27:	Lead time comparison of basic and CT scenario I for different demand levels.	257
Figure 5 - 28:	Comparison of inventory levels per store for CT scenario I and the basic scenario.	258
Figure 5 - 29:	Comparison of inventory levels for CT scenario I for the different demand situations.	258
Figure 5 - 30:	CO ₂ emissions per transport route – comparison of basic scenario and CT scenarios.	259
Figure 5 - 31:	Deviation and delay of deliveries per store and DC for CT scenario I.	261
Figure 5 - 32:	Adherence to schedules regarding delays - comparison between basic and CT scenario I.	262
Figure 5 - 33:	Effect of production concept adaptation on train schedule and delivery windows.	264
Figure 5 - 34:	Utilisation of trains for CT scenario II.	265
Figure 5 - 35:	Transport time elements per store for CT scenario II for the normal demand situation.	266
Figure 5 - 36:	Comparison of lead time for different demand situations between CT scenario I and CT scenario II.	266
Figure 5 - 37:	Average inventory in the transport between DC and stores for CT scenario II in comparison with CT scenario I.	267
Figure 5 - 38:	Comparison of lead time elements for CT scenario II for different demand situations.	267
Figure 5 - 39:	Comparison of delay between CT scenario I and CT scenario II.	268
Figure 5 - 40:	Comparison of adherence to schedules for different demand situations.	269
Figure 5 - 41:	Effect of distribution concept adaptation to train schedule for transport route H 75 (CT scenario III).	271
Figure 5 - 42:	Comparison of lead times of CT scenario I and CT scenario III for different demand situations.	272
Figure 5 - 43:	Comparison of lead time elements between CT scenario I and CT scenario III.	273
Figure 5 - 44:	Average inventory in the transport between DC and stores coming from DC Hochdorf for 0 scenario III.	CT 273
Figure 5 - 45:	Inventory levels for different demand situations - CT scenario III.	274

Figure 5 - 46: Comparison of delay between CT scenario I and CT scenario III.

Figure 5 - 51: Comparison of adherence to schedules regarding delayed deliveries.

Figure 5 - 50: Comparison of lead time for different simulation scenarios.

Figure 5 - 48: Comparison of lead time elements for CT scenario III and CT scenario IIIa.

Figure 5 - 47: Comparison of adherence to schedules regarding delays for different demand situations.

Figure 5 - 52: Comparison of inventory levels for all scenarios and for different demand situations.

Figure 5 - 49: Comparison of inventory levels for CT scenario III and IIIa for different demand situations 277

274

275

276

278

279

281

List of Tables

Table 1 - 1: Scientific-theoretical positioning of the thesis in hand.	7
Table 2 - 1: Literature review of integrative measures from the field of I&C flow facility structure.	36
Table 2 - 2: Literature review of integrative measures from the field of workflow and activity structure.	38
Table 2 - 3: Literature review of integrative measures from the field of organisational structure.	39
Table 2 - 4: Literature review of integrative measures from the field of planning and control methods.	40
Table 2 - 5: Literature review of integrative measures from the field of product flow facility structure.	40
Table 2 - 6: Transport quantities in CT (unaccompanied) in 2009 in tons and TEU.	44
Table 2 - 7: Systematisation of CT.	48
Table 2 - 8: Comparison of shippers' performance expectations and CT service compliance.	58
Table 2 - 9: Classification criteria for CT business models.	70
Table 2 - 10: Selected performance indicators at the actor's level.	84
Table 2 - 11: Selected performance indicators at the dyadical level.	85
Table 2 - 12: Selected performance indicators at the concept level.	86
Table 2 - 13: Comparison of shippers' performance requirements and CT characteristics.	100
Table 3 - 1: Time-related SCP requirements depending on the SC strategy.	121
Table 3 - 2: Space-related SCP requirements depending on the SC strategy.	122
Table 3 - 3: Shipment size and transport quantity SCP requirements depending on the SC strategy.	122
Table 3 - 4: Product type-related SCP requirements depending on the SC strategy.	123
Table 3 - 5: Cost-related SCP requirements depending on the SC strategy.	124
Table 3 - 6: Flexibility-related SCP requirements depending on the SC strategy.	124
Table 3 - 7: Reliability-related SCP requirements depending on the SC strategy.	125
Table 3 - 8: SCP requirements profiles derived from SC strategy.	125
Table 3 - 9: Classification criteria for procurement concepts with regard to the impact on transport	
concepts.	136
Table 3 - 10: Impact of procurement concept configuration on SCP requirements.	141
Table 3 - 11: Classification criteria for production concepts with regard to the impact on transport concepts.	144
Table 3 - 12: Impact of production concept configuration on SCP requirements.	147
Table 3 - 13: Classification criteria for distribution concepts with regard to the impact on transport concepts.	153
Table 3 - 14: Impact of distribution concept configuration on SCP requirements.	156
Table 3 - 15: Characteristics of the CT concept 'line transport - regular booking'.	158
Table 3 - 16: Characteristics of the CT concept 'sporadic usage of line relation'.	159
Table 3 - 17: Characteristics of the CT concept 'network relation'.	160
Table 3 - 18: Overview on CT concept and suitable application fields.	161
Table 3 - 19: SCP requirements profiles of the three CT concept types.	162
Table 4 - 1: Identification of relevant concepts depending on the integration point.	176
Table 4 - 2: Applicability of CT types according to SC strategy and integration point.	179
Table 4 - 3: Example of contrasting lean SCP requirements profile and the performance profile of	
line transport concept with regular booking performance profile.	180
Table 4 - 4: Example of the misfit of SCP requirements and CT performance profile.	180
Table 4 - 5: Adaptation points for SCP requirements regarding time.	181
Table 4 - 6: Adaptation points according to performance dimension, criterion and involved	105
SC sub-concept.	182
I able 4 - /: Overview of analysed integrative measures and instruments and abbreviations. T 11 - 4 - 9 - Level + Git development	183
I able 4 - 8: Impact of integrative measures and instruments on time-related SCP requirements.	185
Table 4 - 9: Impact of integrative measures and instruments on space-related SCP requirements.	186

Table 4 - 10: Impact of integrative measures and instruments on shipment size and transport quantity related SCP requirements.	187
Table 4 - 11: Impact of integrative measures and instruments on cost related SCP requirements.	188
Table 4 - 12: Impact of integrative measures and instruments on product related SCP requirements.	189
Table 4 - 13: Impact of integrative measures and instruments on flexibility related SCP requirements.	190
Table 4 - 14: Impact of integrative measures and instruments on reliability related SCP requirements.	191
Table 4 - 15: Applicability of CT types according to SC strategy and integration point.	192
Table 4 - 16: Summary - impact of integrative measures and instruments on SCP requirements.	193
Table 5 - 1: Focus of simulation study.	209
Table 5 - 2: Sites modelled in the simulation.	221
Table 5 - 3: Emission factors for rail transport.	223
Table 5 - 4: Standardised inclination profile.	224
Table 5 - 5: Extract of route segments defined for the simulation with distance, street profile and transport time.	225
Table 5 - 6: Primary transport routes, transport resources and start times.	226
Table 5 - 7: Average demand for prioritised goods per weekday and store.	229
Table 5 - 8: Experimental plan for the simulation study.	242
Table 5 - 9: Average emissions for transport route H 60.	249
Table 5 - 10: Emissions per transport route.	250
Table 5 - 11: Emissions for a one-week simulation run.	250
Table 5 - 12: Overview of key figures regarding the adherence to schedules for the basic scenario.	252
Table 5 - 13: Extract of route plan for the CT scenario I (complete list in Appendix B-II).	253
Table 5 - 14: Train schedule between Härkingen terminal and Daillens terminal.	253
Table 5 - 15: Comparison of emissions for transport route H 60 for CT scenario I.	259
Table 5 - 16: Emissions for CT scenario I in comparison with the basic scenario - calculation of possible savings for a one-week simulation run.	260
Table 5 - 17: Adaptations to the production concept.	264
Table 5 - 18: Overview of key figures regarding the adherence to schedules for CT scenario II.	269
Table 5 - 19: Adaptations to the distribution concept for CT scenario III.	271
Table 5 - 20: Comparison of projected emission factors and possible savings for one year.	280

List of Abbreviations

3PL	Third Party Logistics Service Provider
4PL	Fourth Party Logistics Service Provider
AMMPL	Association Materials Management, Purchasing and Logistics
APS	Advanced Planning System
BSC	Balanced Scorecard
BOM	Bill of Material
B2B	Business-to-Business
B2C	Business-to-Customer
CIF	Cost, Insurance and Freight
ConWiP	Constant Work in Process
CPFR	Collaborative Planning, Forecasting and Replenishment
CRM	Customer Relationship Management
CSCMP	Council of Supply Chain Management Professionals
СТ	Combined Transport
DAF	delivered at frontier
DDP	Delivered Duty Paid
DC	Distribution Center
ECR	Efficient Consumer Response
ERP	Enterprise Resource Planning
EDI	Electronical Data Interchange
ETCS	European Train Control System
EVA	Economic Value Added
EXW	Ex Works'
FOB	Free On Board
ICC	International Chamber of Commerce
IT	Information Technology
I&C	Information and Communication
JIS	Just in Sequence
JIT	Just In Time
M&A	Mergers and Acquisitions
MRP	Material Requirements Planning
ODBC	Open Database Connectivity
OEM	Original Equipment Manufacturer
OPT	Optimized Production Technology
p.a.	per annum
p.c.	per capita
POS	Point Of Sale
UML	Unified Modeling Language
RDA	Resource Dependence Approach
RFID	Radio Frequency Identification
SC	Supply Chain
SCM	Supply Chain Management
SCP	Supply Chain Performance
SCOR	Supply Chain Operations Reference
SRM	Supplier Relationship Management
ТСО	Total Cost of Ownership

TEU	Twenty-foot Equivalent Unit
VMI	Vendor Managed Inventory
WMS	Warehouse Management System

Sites in Simulation Study

GEN	Geneva
LAU	Lausanne
MON	Montey
MOR	Morges
NYO	Nyon
PAY	Payerne
PF SIO	platform Sion
SIO	Sion
SIE	Sierre
T DAI	Daillens Terminal
T HAE	Härkingen Terminal
VES	Vesenaz
VEV	Vevey
YVE	Yverdon

1 Integration of Combined Transport into Supply Chain Concepts from a Performance Perspective: Need for Action

Combined transport (CT) is increasingly gaining the interest of policymakers, the general public, production and retailing companies as well as third party logistics providers (3PLs). As the combination of rail and road transport, CT is perceived as a feasible approach to deal with changing supply chain performance (SCP) requirements and the latest problems in road transport. Today, supply chain (SC) activities are evaluated not only according to cost, time and quality, but also performance requirements in terms of flexibility, reliability and sustainability. CT seems to be one alternative to unimodal road transport concepts to meet changing SCP requirements. However, CT underlies several inherent and market-related disadvantages. Additional transhipment processes and coordination activities usually require additional time and cost quotas. To reach an equivalent SCP the required time quotas must be gained by a targeted alignment of all upstream and downstream SC activities. This means that procurement, production and distribution concepts must be configured with regard to the CT integration.

The research objective of this thesis is the theory-guided development of practical guidance for the integration of CT into SC concepts. This overarching objective can be divided into three secondary objectives: (1) the identification and classification of shippers' requirements regarding transport concepts in general, (2) the identification of relevant elements, correlations and cause-and-effect relationships for the CT integration and (3) the development of specific recommendations on the instruments and measures supporting the integration. The following section concretises the problem description in the form of three research questions (section 1.1). The scientific-theoretical positioning and the resulting research design to answer these questions (section 1.2) as well as the thesis and research outline are introduced (section 1.3).

1.1 Problem Formulation and Research Questions

Problem Formulation

Recent SC concepts can be characterised by the attempt to reduce inventory and to simultaneously increase flexibility and reliability. SC concepts bind the production, pro-

curement and distribution processes of shippers, suppliers as well as the embedded and linking transport concepts.¹ 3PLs are increasingly integrated into the collaboratively synchronised value-adding processes.² These changes reflect the growing performance and value orientation of supply chain management (SCM), as expressed by the application of target systems that join key operative and strategic figures.

The material flow in SC concepts – especially in the last mile – is commonly accomplished by unimodal road transport concepts. Here, in practice the vision of crosscompany process linking meets its boundaries. Restrictions such as infrastructure shortages, increasing oil and energy prices, protectionism and ongoing market consolidation decrease reliability and speed.

As a result, CT becomes a central interest of policymakers, the general public, shippers and transport actors. CT seems to be a promising solution to overcome the latest problems in unimodal road transport and the vulnerability of SC concepts. CT combines the advantages of rail transport. It combines the advantages of cost efficiency of high transport volumes and weights with the low emission rates of rail transport with the high flexibility, continuous driver accompaniment and low costs over short distances of road transport. Next to usage for international or hinterland transport, there are first examples of the cost, time and quality neutral CT integration for short distance inland transport over transport distances below 100 km.³

For instance, in Switzerland, several big retail companies have integrated CT into their SC concepts. The main haulage on rail is in some cases below 100 km. The companies prove that the CT concepts meet cost, time and quality performance requirements and even exceed the performance level of unimodal road transport in terms of reliability.⁴ The CT concept increases the reliability level of store supply. Furthermore, CT improves the level of sustainability, expressed by the level of different emission factors (e.g., carbon dioxide (CO_2), nitric oxide (NO_x), particles). Shippers promote these activities to improve their 'green images'.⁵ For instance, Lidl and Migros use specific containers with advertisements of their latest CT activities.

¹ cf. Bowersox et al. (2002), p. 12. 2

See Fabbe-Costes et al. (2009); Shang (2009). 3

cf. Klaas-Wissing et al. (2010). 4

cf. Ibid.

In 2010, Coop bought the rail carrier Railcare to overcome the problems of process integration. cf. Railcare (2011).

Nevertheless, current literature and practice state that in most cases CT for inland and last mile transport is insufficient to meet recent SCP requirements.⁶ Experts state that the main task of CT actors is to strengthen shippers' trust in the performance of CT.⁷ The inherent disadvantages, especially the additional time quota caused by additional transhipment processes, the high coordination effort and the resulting high cost level are central aspects exacerbating a performance-oriented CT integration into SC concepts.

Transport concepts are central elements of SC concepts. They are embedded into the sub-concepts of procurement, production and distribution, connect SC actors and cover the 'last mile' distance to end customers. Therefore, they pass the performance characteristics of the entire SC to the end customer. Although transport actors connect all actors of the 'integrated' SC, they are often not integrated members themselves. This means that transport actors are usually not involved in SC planning activities, are not connected to IT systems and are often understood as the 'servants' to the procurement, production and distribution concepts.

The thesis in hand exploits this imperfection regarding the integration of CT into SC concepts. It is assumed that closing this 'integration gap' opens up time buffers and coordination flexibility, which allow the integration of CT while reaching the performance level of unimodal road transport. Therefore, a careful alignment of the SC and CT concepts is necessary.

To reach this target the thesis in hand suggests an innovative perspective on the integration problem. It is assumed that the adaptation of the SC concept, namely the subconcepts of procurement, production and distribution, to the characteristics of a specific CT concept is necessary to reach the required integration of material and information flow processes. This comprehension corresponds to a complete renunciation of the understanding of the transport concept as a servant. Instead, the transport concepts are interpreted as a guiding criterion for SC configuration.

Research Gap

Approaching this practical motivated problem formulation from a theoretical perspective, it can be stated that no published empirical work has yet dealt with the integration of CT into SC. Neither the SC integration literature nor SCP management research have so far considered the meaning of transport concepts and actors for SCP. There is also a

⁶ For instance Rapptrans (2005), p. 44; Klotz (2011).

⁷ cf. section 2.2.3.1; Klotz (2011).

shortage of literature analysing the interdependencies between the transport and SC concepts. This thesis contributes to close this research gap. For this purpose, the terms *'supply chain concept'*, *'combined transport concept'* and *'supply chain performance'* are defined to be the central constructs. The working definitions and links of the central constructs are visualised in Figure 1 - 1.



Figure 1 - 1: Working definitions of central constructs.⁸

A neutral and rational research perspective does justice to the high degree of novelty in both the practical and theoretical problem formulation. Thus, the focus is on the technical and organisational aspects of the CT integration problem. Further managerial aspects considering actor-specific perspectives are excluded. This reduces the complexity of the problem formulation and highlights the performance potential of CT integration into SC concepts.

The rough problem description proves the demand for further research. The following guiding research question is formulated to reflect the neutral research perspective:

Q₀: How can CT be integrated into SC concepts with regard to SCP?

Owing to the general character of the formulation, the research question is approached in three main steps.

⁸ Own illustration.

First, the structure and elements of SC concepts are examined. A typology of SC concepts is developed as well as an approach for the structured description of the SCP requirements regarding transport concepts.

Second, the meaning of process integration is highlighted. The SCP requirements of different procurement, production and distribution types in terms of the embedded and linking transport concepts are worked out and the impact on the integration of CT is discussed. For this purpose the procurement, production and distribution concepts are investigated in depth.

Third, the results are processed and enhanced for application in practice. A conceptual model to visualise the relations between the elements of SC and CT concepts is developed and the underlying cause-and-effect relationships are discussed. Finally, practical recommendations for the application on integrative measures and instruments to support CT integration are developed.

This threefold approach is reflected by three specific secondary research questions, aiming at different aspects of the guiding research question Q_0 .

 Q_1 : How can the requirements of SC concepts regarding transport concepts be described in accordance to SCP?

 Q_2 Which interdependencies between the elements of the SC and CT concepts can be utilised for the successful integration of the material and information flow processes of shippers and CT actors?

 Q_3 What practical recommendations can be derived for integrative measures and instruments to support CT integration with regard to SCP?

To understand how these research questions are approached in terms of research methodology and process, a general understanding of the thesis' positioning in research theory and philosophy is necessary.

1.2 Scientific-Theoretical Positioning and Research Design

The postulated research questions address different aspects of the same research problem. Thus, it is sensible to complement this rather general understanding by the scientific-theoretical positioning to support the choice of an appropriate research methodology.

Scientific-Theoretical Positioning

The thesis follows the principles of Punch (2005) and Saunders et al. (2007) by interpreting organisations as complex, open, social systems. Companies are influenced by a multiplicity of transformations.⁹ Following the considerations of Ulrich & Krieg (1974), organisations are assumed to be not completely controllable. From this perspective, business management is understood as applied science. The term *'applied'* means that the available theoretical knowledge is referred for application to the practice problem. To gain empiric knowledge is the pragmatic research objective. This objective aims not to test general theories and to explain reality, but to develop rules and models to create new realities.¹⁰

Ulrich (1995) distinguished between a theoretical and an applied research understanding.¹¹ In an applied research understanding, the problem formulation results from practice rather than from research. The problem formulation in the thesis in hand is identified in the SCM and CT practice. Ulrich (1995) claimed that in applied research understanding knowledge generation is rather inductive, whereas in theoretical research understanding it is deductive. The integration of CT into SC concepts seizes a complex and, up to now, practically and scientifically inadequately discussed problem formulation derived from economic practice. Recently, there has been no suitable approach for performance-oriented CT integration into SC concepts in practice. Thus, a purely inductive approach is insufficient for knowledge generation. There is no sustainable theoretical foundation for performance-oriented SC integration, either. Thus, a purely deductive approach for knowledge generation is insufficient, too. Hence, in the thesis in hand a combined inductive-deductive approach is applied. It is enhanced by a model-oriented simulative approach.¹² This supports the validation and improvement of the deductive results and supports the discovery of working relationships and principles.¹³

According to Ulrich (1995), the aim of applied research is to identify typical problems and to develop and test recommendations for a system's design rather than theory development.¹⁴ The thesis in hand aims at theoretical and practical research objectives. The theoretical research objective is the challenge to deductively identify and analyse

⁹ cf. Punch (2005), pp. 25; Saunders et al. (2007), pp. 136.

¹⁰ cf. Ulrich (1981), p. 168; {Ulrich, 1995 #976, pp. 165.

¹¹ cf. {Ulrich, 1995 #976}, pp. 165.

cf. Axelrod (1997a) defined simulation as a 'third way of doing science' (Axelrod (1997a), p. 24). Simulations start with a set of explicit assumptions(as deduction does), but do not prove theorems. It generates data that can be analyzed inductively.

¹³ cf. Axelrod (2005); p. 4.

¹⁴ cf. Ulrich (1995).

the relevant cause-and-effect relationships between the elements of SC and CT concepts for the performance-oriented CT integration using suitable theoretical approaches. Furthermore, configuration theory is transferred to the new application field of CT. The results will be compressed into specific design advice and practical recommendations for the performance-oriented integration of CT into SC concepts. For this purpose, the conceptual research framework is the major instrument.

In a theoretical research understanding the quality criterion is the universality of the findings. In applied research, the criteria are the reliability and applicability of the developed methods and rules for practical problems. Thus, the thesis in hand aims at the closeness of the developed conceptual model and, furthermore, at the applicability of the developed recommendations and their relevancy to the SCM practice. Table 1 - 1 summarises the considerations of the scientific-theoretical positioning.

	theoretical research understanding	applied research understanding	characteristics of this research
problem formulation	in research	in practice	in the practice of SCM and CT
knowledge generation	deductive	inductive	deductive model oriented-simulative** inductive
research target	theory development / testing: explanation of existing reality	identification of typical prob- lems and testing design suggestions	development of practical recommen- dation on the performance-oriented integration of CT into SC concepts
research criteria	universality	reliability and applicability of methods and rules; practical problem solving power	closeness of developed conceptual model; applicability of recommendations; relevancy to SCM practice

Table 1 - 1: Scientific-theoretical positioning of the thesis in hand.¹⁵

The research design is a consequence of the scientific-theoretical positioning.

Research Design

The thesis in hand applies a combined inductive-deductive approach enhanced by a model-oriented simulative research approach. Within this framework, statements are deductively gained based on configuration theory and are complemented by empirical knowledge.¹⁶

¹⁵ cf. Ibid.

¹⁶ The research design can be described as the summary of research, including data gathering, data analysis and interpretation. cf. Punch (2005), pp. 22, 133; Sarantakos (2004), p.14. The research design contains the positioning within the empirical research and acts as link between research questions as well as data basis. cf. Saunders et al. (2007), pp. 5. The design process includes relevant research strategic and methodological decisions: research philosophy, research approach, research strategy, chronological direction as well as the data gathering methods cf. Sarantakos (2004), p. 23.

The research design can be divided into two main steps:

(1) The first step contains the theory-guided development of a conceptual research framework and of propositions for the performance-oriented integration of CT into SC concepts.

The data basis for the deductive part of the research process is the preconception of the research problem developed in literature reviews, explorative interviews and explorative case studies in the fields of (a) CT, (b) SC integration, (c) SCP and (d) SCM. The conceptual research framework visualises the main argumentation line of the thesis.¹⁷ The framework includes the preliminary understanding (preconception) of the research problem and reflects both the practical and theoretical perspectives of the research problem.¹⁸ It serves as an explanatory model and as an orientation of the practical problem formulation. It connects the important constructs and shows the visualisation.¹⁹ (2) The second step is the simulation study for the inductive specification of deduced propositions and for the validation of the conceptual research framework.

The data basis for the inductive part of the research process is the simulation model based on empirical data on a Swiss retailer's distribution network. The simulation study analyses the effect of CT integration a distribution network with two distribution centres (DCs) and eleven stores. A discrete event-oriented material flow simulation shows the performance effect in comparison with unimodal road transport. Furthermore, the cause-and-effect relationships between the elements of the SC concept and the CT concept are analysed.

1.3 Research Process and Thesis Outline

Based on the research questions and the scientific-theoretical positioning, the research process and thesis outline are presented.

The research process reflects the tight connection of research and practice in application-oriented research.²⁰ In the first step, the practical research problem of CT integration into SC concepts is described comprehensively. Theoretical approaches from adjacent research fields, such as SC cooperation research, are reviewed for their applicabil-

¹⁷ cf. Wolf (2000), p. 37; Raffée (1989), pp. 37; Macharzina et al. (1991), p. 27.

¹⁸ cf. Ulrich (1981); Wolf (2008).

¹⁹ cf. Kubicek (1977), pp. 37; Wolf (2008), p. 3, 27.

²⁰ cf. Ulrich (1995), pp. 166.

ity. These theoretical solutions are analysed and a sophisticated model of CT integration, evaluation criteria and design recommendations presented. Finally, the design recommendations are validated and enhanced by means of a simulation approach.

The thesis is structured into six main parts. After this general introduction, the second chapter concretises the problem formulation and discusses different theoretical approaches to support the problem solving. First, the practical problem formulation of SC integration is analysed in detail (section 2.1). Second, the problem formulation is complemented with insights from specific fields of SCM and transport research. The specific inherent problems of CT are pointed out and are interpreted with regard to the integration into SC concepts and to SCP requirements (section 2.2). SCP is introduced as a system of objectives for the value- and finance-oriented management of SCs (section 2.3). Based on this compilation of challenges, theoretical explanation approaches suitable for the given problem formulation are introduced (section 2.4). Namely, the suitability of contingency theory for the integration of CT into SC concepts is shown. This theoretical approach supports the explanation of cause-and-effect relationships between the SC concepts' elements and the CT concept elements influencing SCP (section 2.4).

The third chapter addresses the meaning of a CT concept as an element of a SC concept. It clarifies the position and meaning of the CT concept on SCP. Therefore, the relationship between the procurement, production and distribution concepts and the embedded and linking transport concept is worked out. In particular, the relevant cause-and-effect relationships between the elements of the sub-concepts and the elements of the CT concept are identified. Initially, the third chapter analyses the construct of SC concepts as the operationalisation and implementation of a specific SC strategy (section 3.1). For the precise description of the SCP requirements regarding transport concepts a structured description framework with seven performance dimensions is developed (section 3.2). Based on this classification, the SC sub-concepts, namely procurement, production and distribution, concepts are analysed regarding their requirements for the embedded and linking transport concepts (section 3.3). In particular, the meaning of the SC concept configuration on the integration of the CT concept is worked out (section 3.3). Finally, a typology for the CT concept and corresponding performance profiles are developed (section 3.4).

The fourth chapter aims at the development of a conceptual research framework and the development of propositions on the performance integrated integration of CT into SC concepts. The conceptual framework illustrates the relevant interdependencies between the elements of the SC concepts, the CT concepts as well as integrative measures and

instruments with regard to SCP. In a first step, the main research idea is developed on the basis of the previous considerations. The section results in the description of the main proposition for the thesis in hand (section 4.1). The conceptual research framework builds on these considerations and follows the argumentation line of the configuration theory (section 4.2). In a step-wise approach, the central constructs of the research problem are segmented into middle range constructs as well as manifest and observable variables. The basic relationships between the main constructs and variables are summarised in first, rather general propositions. These propositions are specified in section 4.3. Here, the relevant cause-and-effect relationships between the elements of the SC and CT concepts as well as integrative measures and instruments are discussed in depth.

The fifth chapter aims at the specification and validation of the chosen aspects of the conceptual model by means of a simulation study. Therefore, the problem formulation is specified through the analysis of a lean SC concept and a combined CT concept with regular line trains.²¹ First, the research design of the simulation study is introduced. The suitability of the simulation method for the given research problem, central quality criteria and restrictions of the chosen methodology are discussed. Furthermore, the choice of a specific simulation approach and a simulation tool are discussed (section 5.1). Second, the development of the simulation model of the distribution network of a Swiss retailer is described (section 5.2). This includes the introduction of the simulation-specific problem and target definition (section 5.3), of the data gathering and editing process (section 5.4), the development of the experimental plan and the setup of experiments (section 5.6) and the discussion of simulation runs and results (section 5.7).

The sixth section highlights the central managerial (section 6.1) and practical implications (section 6.2) of the thesis in hand. Additionally, the section discusses the limitations of the findings and works out future research perspectives (section 6.3).

Figure 1 - 2 gives an overview of the application-oriented research process and thesis outline.²²

²¹ The different CT types are introduced in section 3.4.

²² Based on Ulrich (1995), p. 167.

1 Integration of CT into SC Concepts from a SCP Perspective: Need for Act 1.1 Problem Formulation, Research Questions and Research Objectives	
1.1 Problem Formulation, Research Questions and Research Objectives	tion
1.2 Scientific-Theoretical Positioning and Research Design	
1.3 Research Process and Thesis Outline	
Preconception and Concretisation of the Research Problem	
structuring of 2 Problem Concretisation - SCP Orientation encourages CT Integration	
problem into SC Concepts	
2.1 Need, Challenges and Instruments for SC Integration	
2.2 Coping with C1 Complexity in SCs	
rendition of 2.4 Configuration Theory as Theoretical Research Approach for Integrating CT into	SC
relevant concepts	se
2.5 Intermediate Findings	
Analysing Relevant Factors for CT Integration	
3 CT as Element of SC Concepts	
identification of 3.1 SC Concepts as the Operationalisation of SC Strategies	
relevant elements for 3.2 Development of a Classification of SCP Requirements regarding Transport Cond	cepts
the conceptual 3.3 Configuration of SC Concepts	
research framework 3.4 Development of a CT Concept Typology – Classification using SCP Profiles	
3.5 Intermediate Findings	
4 Concentual Research Framework of Performance-oriented CT Integration	
into SC Concepts	
4.1 Central Aspects for the Performance-Oriented CT Integration into SC Concepts	-
4.2 Development of a Conceptual Research Framework and Deduction of Proposition	ons on the
deduction of Performance-Oriented CT Integration into SC Concepts	
practical 4.3 Cause-and-Effect Relationships between Elements of CT and SC Concepts - Deduction of Methodological and Instrumental Implications	
recommendations 4.4 Intermediate Findings	
5 Performance-oriented Integration of Combined a Combined Line Transpor	rt
5 Performance-oriented Integration of Combined a Combined Line Transpor into a Lean SC Concept - A Simulation Study 5.1 Research Design for the Simulation Study	rt
Sector Sector Performance-oriented Integration of Combined a Combined Line Transpor into a Lean SC Concept - A Simulation Study 5.1 Research Design for the Simulation Study 5.2 Process Model for the Simulation Model	rt
5 Performance-oriented Integration of Combined a Combined Line Transportinto a Lean SC Concept - A Simulation Study 5.1 Research Design for the Simulation Study 5.2 Process Model for the Simulation Model 5.3 Step 1: Problem Formulation and Target Definition	rt
5 Performance-oriented Integration of Combined a Combined Line Transportinto a Lean SC Concept - A Simulation Study 5.1 Research Design for the Simulation Study 5.2 Process Model for the Simulation Model 5.3 Step 1: Problem Formulation and Target Definition 5.4 Step 2: Data Gathering and Editing for Model Configuration and Validation	rt
Sector and contract of the sector and contrecontract of the sector and contract of the sector and	rt
validation and enhancement of validation and enhancement of	rt
Validation and enhancement of conceptual research Step 2: Data Gathering and Editing for Model Configuration and Validation 5.7 Step 1: Problem Formulation and Verification of the Simulation Model 5.3 Step 1: Problem Formulation and Target Definition 5.4 Step 2: Data Gathering and Editing for Model Configuration and Validation 5.5 Step 3: Development, Validation and Verification of the Simulation Model 5.6 Step 4: Experimental Plan and Setup of Experiments 5.7 Step 5: Description and Analysis of Simulation Results	rt
Solution and enhancement of conceptual research frameworkSolution for the second	rt
validation and enhancement of conceptual research framework Step 1: Problem Formulation and Target Definition 5.1 Research Design for the Simulation Study 5.2 Process Model for the Simulation Model 5.3 Step 1: Problem Formulation and Target Definition 5.4 Step 2: Data Gathering and Editing for Model Configuration and Validation 5.5 Step 3: Development, Validation and Verification of the Simulation Model 5.6 Step 4: Experimental Plan and Setup of Experiments 5.7 Step 5: Description and Analysis of Simulation Results 5.8 Intermediate Findings	rt
validation and enhancement of framework Step 1: Problem Formulation and Verification of the Simulation Model 5.1 Research Design for the Simulation Study 5.2 Process Model for the Simulation Model 5.3 Step 1: Problem Formulation and Target Definition 5.4 Step 2: Data Gathering and Editing for Model Configuration and Validation 5.5 Step 3: Development, Validation and Verification of the Simulation Model 5.6 Step 4: Experimental Plan and Setup of Experiments 5.7 Step 5: Description and Analysis of Simulation Results 5.8 Intermediate Findings	rt ation
validation and enhancement of framework 6 Implications for Science and Practice on the Performance-oriented Integration 6 Implications for Science and Practice on the Performance-oriented Integration	rt ation
validation and enhancement of conceptual research framework 5 Performance-oriented Integration of Combined a Combined Line Transpor- into a Lean SC Concept - A Simulation Study 5.1 Research Design for the Simulation Study 5.2 Process Model for the Simulation Model 5.3 Step 1: Problem Formulation and Target Definition 5.4 Step 2: Data Gathering and Editing for Model Configuration and Validation 5.5 Step 3: Development, Validation and Verification of the Simulation Model 5.6 Step 4: Experimental Plan and Setup of Experiments 5.7 Step 5: Description and Analysis of Simulation Results 5.8 Intermediate Findings 6 Implications for Science and Practice on the Performance-oriented Integra of CT into SC Concepts 6.1 Management Implications 6.2 0.1	rt ation
statistic energy 5 Performance-oriented Integration of Combined a Combined Line Transporint oa Lean SC Concept - A Simulation Study 5.1 Research Design for the Simulation Study 5.2 Process Model for the Simulation Model 5.3 Step 1: Problem Formulation and Target Definition 5.4 Step 2: Data Gathering and Editing for Model Configuration and Validation 5.5 Step 3: Development, Validation and Verification of the Simulation Model 5.6 Step 4: Experimental Plan and Setup of Experiments 5.7 Step 5: Description and Analysis of Simulation Results 5.8 Intermediate Findings 6 Implications for Science and Practice on the Performance-oriented Integrations 6.1 Management Implications 6.2 Scientific Implications 6.3 Limitotiones	rt ation



²³ Own illustration in accordance with Ulrich (1995), p. 176.

2 Problem Concretisation - SCP Orientation Encourages the Integration of CT into SC Concepts

The second chapter concretises the problem formulation and discusses different theoretical approaches to support the problem solving. First, the practical problem formulation of SC integration is analysed in detail (section 2.1). Second, the problem formulation is complemented with insights from specific fields of SCM and transport research. The specific inherent problems of CT are pointed out and are interpreted with regard to the integration into SC concepts and to SCP requirements (section 2.2). SCP is introduced as a system of objectives for the value- and finance-oriented management of SCs (section 2.3). Based on this compilation of challenges, theoretical explanation approaches suitable for the given problem formulation are introduced (section 2.4). Namely, the suitability of contingency theory for the integration of CT into SC concepts is shown. This theoretical approach supports the explanation of cause-and-effect relationships between the SC concepts' elements and the CT concept elements influencing SCP (section 2.4). The main results of the chapter are then summarised in the form of intermediate findings (section 2.5).

2.1 Needs, Challenges and Instruments for SC Integration

This section points out the needs, the challenges and instruments of SC integration. The research stream on SC integration usually focuses on the integration of shippers (retailers and manufacturing companies) in buyer–supplier relationships. Lately, research activities have focused on the integration of 3PLs, too.¹ Although, there are no specific publications on the integration of (combined) transport actors into SCs, the research stream of SC integration is the basis for the further considerations of the CT integration into SC concepts. In a first step, the framework conditions, which force companies to closer integrate material and information flows with their SC partners, are discussed. Furthermore, the internal and external challenges of the integration are identified and analyzed with regard to the problem of integration of CT into SC concepts. Finally, central measures and instruments to reach SC integration are introduced.

e.g., Fabbe-Costes et al. (2009).

2.1.1 Need for SC Integration

Changing framework conditions force companies to increasingly focus their activities on end customers' needs. In times of decreasing customer loyalty, customers choose the product that best fits their needs in terms of cost, availability and customisation. Thus, *customer orientation* is perceived as the key factor for competition among SCs. To create a customer-oriented SC a close cross-company integration of material and information flow processes is necessary.

Framework Conditions Require the Close Integration of Material and

Information Flow Processes

A quantity of framework conditions exacerbates reaching customer orientation for companies and SCs. These framework conditions increase the performance requirements of SCs and thereby imply a close cross-company integration of material and information flows.

SCs have to deal with *various challenges*, such as the demand for individualised products, contracting product life cycles, shorter delivery times and high quality levels. A strong price competition among retailers implies low inventory levels and punishes out-of-stock situations. Retail and manufacturing companies promise availability, innovativeness, uniqueness and low prices to end customers. This encompasses the availability of seasonal, cooled, frozen and hazardous goods. These special products have specific requirements in terms of storage, handling and transport. For instance, groceries and pharmaceuticals must be consistently temperature-controlled and perishable goods require comparably short lead times.²

Society, policy and companies evaluate SC activities increasingly with regard to their *sustainability and environmental impact*. This means that not only the economic impact of an SC activity is taken into consideration.³ The orientation towards sustainability supports the stronger integration of material and information flow processes, since better planning and stronger bundling may reduce the waste of materials, transport capacities and thereby energy.

Safety and security activities claim to increase requirements in terms of tracking and

² cf. Anckar et al. (2002); Fernie (1995); Kent et al. (2001); Makinwa (2010).

³ A complete research stream has emerged for the topic 'sustainable SCM. See for instance: Craig et al. (2008); Linton et al. (2007); Markley et al. (2007); Pagell et al. (2008); Pagell et al. (2009); Seuring et al. (2008a); Seuring et al. (2008b); Svensson (2007).

tracing systems and the interruption-free monitoring throughout the SC. Production characteristics as well as transport conditions must be tracked and the data stored. In particular, cross-company transport must do justice to security regulations.⁴ To do so, the transport actors need specific knowledge on the transported goods and rely on close information flow integration with the shipper and all other SC actors.

The *Internet* leads to further changes in the structure of the global flows of goods. Customers can receive information on the best quality, prices, competitive products and alternate brands anywhere and at any time.⁵ For instance, social networks and message boards spread positive and negative information and may cause a rapid image decline or promotion of a specific product, brand or company.⁶ The Internet opens up new sales channels.⁷ Not only books, clothing and electronics are sold via the Internet, but recently shops for pharmaceuticals, groceries and photographs are increasingly gaining customers' interests.⁸ Thus, transport units are growing smaller and more time-critical. This effect is called the *'cargo structure effect'*.⁹ Often this goes along with the end customer's claim to be well informed on recent shipment statuses. Thus, tracking and tracing systems are required to closer integrate the information flow.

Finally, there is an *ongoing internationalisation of production and sales activities*. On one hand, the reason for this development is the reduction of production and personnel costs by shifting manufacturing sites or outsourcing production and logistics processes. On the other hand, companies open up new markets, such as the so-called BRIC and Next 11 countries.¹⁰ Internationalisation leads to global transport and logistics networks, ensuring end customers are supplied with the demanded goods. The effect of this ongoing internationalisation is also-called the *'logistics effect'*.¹¹

⁴ For instance, transport from and to the US must send announcements to the destination 24 hours in advance.

 $^{^{5}}$ For the meaning of the Internet on SCM see for instance, Lancioni et al. (2000).

⁶ cf. Peters (2010).

⁷ See, for instance, Gimenez et al. (2008); Gundlach et al. (2006)

In Switzerland, for instance, the two biggest retailers introduced online shops for groceries. See for instance: http://www.coopathome.ch and http://www.migros.ch/de/online-shops.html. See furthermore, Anckar et al. (2002), Baen (2001), Barua et al. (2000), Gundlach et al. (2006).

⁹ cf. Kille et al. (2008).

¹⁰ FinanceContent (2011). The terms go back on Jim O'Neill (Goldman Sachs). BRIC = Brazil, Russia, India and China. Next 11: Bangladesh, Egypt, Indonesia, Iran, Mexico, Nigeria, Pakistan, Philippines, South Korea, Turkey and Vietnam.

¹¹ cf. Kille et al. (2008).

Consequences for the Configuration of SC Concepts

The presented framework conditions require a close cross-company integration of material and information flow processes. Namely the procurement, the production, the distribution and the embedded logistics and transport processes are designed to meet the growing SCP requirements. For instance, production processes are distributed worldwide, raw materials, supplies and components are sourced globally and multilevel distribution concepts are introduced to reach the required level of shelf availability by means of early morning and multiple deliveries.¹²

A quantity of standardised logistics and SC concepts improving SC integration is applied in practice in order to deal with operative challenges in a complex and dynamic SC environment. For example, Stölzle (1999) discussed the meaning of these concepts on the integration aspect.¹³ On the demand side, for instance the CPFR concept (collaborative planning, forecasting and replenishment), the VMI concept (vendor-managed inventory) and the ECR concept (efficient consumer response) can be applied. From a supply point of view, supplier and industry parks as well as just-in-time (JIT) and just-in-sequence (JIS) concepts provide the framework for the integration of material and information flows.

What is SC Integration?

The term 'SC integration' means the integration of material and information flows between SC actors. According to Cooper et al. (1997), SC integration activities encompass:¹⁴

- *SC structure integration* (SC relationships, inter-organisational business units)¹⁵
- *SC process integration* (business processes, material and information flow process integration)¹⁶ and
- *SC management integration* (network culture, leadership, incentive schemes, SC controlling).¹⁷

 ¹² cf. Geir et al. (2006); Gerbens-Leenes et al. (2003); Henning et al. (2009); Kempainen et al. (2003); Linton et al. (2007); McKinnon (2009); Mortensen et al. (2008); Webb (1994).
 ¹³ f. Giri L. (1000). 147

¹³ cf. Stölzle (1999), pp. 147.

¹⁴ In accordance with the three main elements and key decisions of the SC conception.

¹⁵ SC structure integration deals with the development of SC relationships (e.g., trust building activities) and the introduction of cross-company organisational units (e.g., project teams). Cooper et al. (1997), pp. 70.

SC process integration means the cross-company integration of business processes. These efforts can span all value-adding levels, for example research and development, demand planning, disposition or capacity utilisation processes. Ibid., pp. 70.

A tight cross-company integration of material and information flow processes means that at the interfaces between different companies, time buffers are not required and information is not lost. The material flow processes are harmonised and synchronised between SC actors. Therefore, the production, transport and storage quantities are balanced and arrival and delivery frequencies, quantities and unit sizes are cooperatively planned. The controlling of sourcing, production and distribution processes is also accomplished cooperatively. This reduces, for instance, required warehouse capacities. So-called *'buffers'* gather production materials and goods for the next operation.¹⁸

An example of a tight integration of material and information flow processes is the JIT delivery concept, which can be typically found in the automotive industry.¹⁹ Here, the information flows of suppliers, buyers and usually carriers are tightly connected by common information systems or platforms. The original equipment manufacturer (OEM)²⁰ provides the supplier with forecasts on the planned production quantities and production program for a defined time horizon in advance. These forecasts are the basis for the suppliers' sourcing processes and capacity planning. On a daily basis, the OEM specifies and updates the demanded parts in a so-called '*delivery schedule*'.²¹ Therefore, standardised communication protocols are applied and usually used by all OEMs and suppliers.²² The supplier uses the delivery schedule to initiate its own production processes, often not more than 8 to 48 hours in advance. Here, a so-called *'pull control logic'* is applied. The material and information flow processes are *'synchronised'*. A frequent, regular line transport concept with low transport quantities connects suppliers and OEMs without warehousing processes.²³

This type of synchronisation of production processes between the OEM and upstream suppliers as well as the production processes and end customers' demands are also core aspects of the Toyota Production System. Harmonisation and levelling balance fluctuating customer orders. This means that production is not adapted to the orders

SCM integration means benefits for all involved SC members. Companies can access rare resources and new markets. They can share risks and reach synergy effects, e.g., developing new products and services, economies of scale and knowledge. Ibid., p. 70.

 $^{^{18}}$ cf. Ihme (2006), p 11.

¹⁹ See, for instance, Aigbedo (2007); Nielsen et al. (1995).

Original Equipment Manufacturer = 'A producer that provides a product to its customers, who proceed to modify or bundle it before distributing it to their customers.' Investorwords (2011).
 ²¹ of thms (2006) m 250. Emergen (1980)

²¹ cf. Ihme (2006), pp. 250; Jünemann (1989).

²² cf. Ihme (2006) pp. 169.

²³ For more details on control logics in production systems, see section 3.2.3.

on a daily basis, but rather that production quantities and programs are planned for some time in advance, for instance a week. This weekly program level harmonises customer orders and leads to a stable and continuous production flow.

The tight integration of material and information flow processes has several advantages.²⁴ Inventory levels can be kept low. Short-term changes can be realised because of the close coupling of the OEM's and suppliers' processes. High frequent JIT delivery concepts with small transport quantities serve as small inventory buffers. This combination of individualised products and short-term changes leads to a high level of customer orientation.

But the cross-company integration of material and information flow processes has several requirements and disadvantages. There must be a high level of trust among SC actors. These actors must allow the other insight into forecasts and planned production quantities. Hence, partners risk that this information is used opportunistically. In this case, trust and information sharing can mean economic disadvantages. The close coupling of material flow can lead to delivery disruptions to the end customer if there are problems in the upstream production or transport processes.²⁵

2.1.2 Central Elements and the Fields of SC Integration

SC configuration, coordination and cooperation can be understood as the basis of SC integration. The general agreement of SC actors to coordinate their connections and to determine the nodes within a SC network is the requirement for SC integration.

SC configuration comprises the configuration of material flows in the network of SC actors. It contains the definition of the focal company's position within the SC and the choice of strategic partners. *SC coordination* aims at the development of long-term and stable supplier relationships. These relationships allow the realisation of process, transaction, production and research and development cost savings. Furthermore, SC coordination activities aim at the increase of quality and service levels and the reduction of supplier risks. According to Lambert & Pohlen (2001), coordination activities can be distinguished according to the characteristics of the linked proc-

²⁴ See for instance Mentzer et al. (2001a); Min et al. (2007).

²⁵ For further details on vulnerability, see the following section.

esses.²⁶ They depend on the specific SC configuration.²⁷ *SC cooperation* activities depend on the qualitative and quantitative capability of SC partners. The intensity of cooperation activities must be distinguished according to the meaning of the partners as well as the involved processes.²⁸ *SC process integration* involves the cross-company integration of business processes at all value-adding levels.

The terms of SC configuration, coordination, cooperation and integration are tightly connected. It can be summarised that these constructs are the basis to implement a cross-company SC strategy and to improve SCP.²⁹ SCP emerged to solve the conflict between the performance-orientation of operative business processes at the SC level³⁰ and the primary target of company value increase. It can be reflected by different dimensions of shareholder value.³¹ The meaning of SCP orientation is analysed in depth in section 2.3.

2.1.3 Normative, Operative and Strategic Challenges of SC Integration

This section introduces the general challenges of SC integration. Also the problem of CT integration into CT into SC concepts is considered on three different levels.

2.1.3.1 Overview on SC Integration Challenges

Karrer (2006) identified three main impediments that prevent the systematic and performance-oriented integration of CT into SC concepts:³²

(1) *Main objectives of the integrated SC* are not valid for all SC actors (normative challenge),

cf. Lambert et al. (2001), p. 112 Managed' process links are critical for the focal company and thus, actively managed. 'Monitored' process links are important, but not critical and finally 'non-managed' process links. Here the focal company is not directly involved and thus, the managing effort is not justified. Furthermore, process links with non-actors of the SC can be distinguished. These are links where the SC is influenced by decisions or other SCs.
 27

²⁷ cf. Rudberg et al. (2003), p. 36.

²⁸ cf. Trumpfheller et al. (2004), p. 81. ²⁹ See for instance Duffy et al. (2006).

See for instance Duffy et al. (2006), pp. 57; Hsu et al. (2007), p. 296; Vereecke et al. (2006), p. 1176; Weber et al. (2000), p. 54; Fabbe-Costes et al. (2008).
 ³⁰ K. (2006) - (2006) - (2007

³⁰ cf. Karrer (2006), p. 213.

³¹ cf. Ibid., p. 213. The introduction of the SCP indicates that SC actors do not only concentrate on the content-related arrangement and adjustment of SC elements according to the corporate requirements, but on the measurability and controllability of the SCP (Sennheiser et al. (2008), p. 1). SCP can be concretised by the characteristics of efficiency and effectiveness orientation, multidimensionality and the future and potential orientation (Stölzle et al. (2004b), p. 125).

³² cf. Karrer (2006), p. 2.

- (2) Operational objectives and *challenges of shippers and CT actors* significantly differ because of the different core businesses in either production, retail or service delivery (operative challenge),³³
- (3) SC and CT actors focus their activities on their *own company value growth* (strate-gic challenge).

These impediments are reflected by the differing operative and strategic target systems and key figures to measure the target achievement of processes, people, business units, divisions or companies.³⁴ According to Wilke (2001), these impediments lead to so-called control gaps.³⁵ These gaps represent the general discussion on control scepticism on the control of complex social systems and thus, of SC processes.

To develop practical recommendations on the integration of CT into SC concepts the introduced challenges must be taken into deeper consideration. Thus, the following section analyses in depth the three main challenges of SC integration and interprets them in terms of CT integration.

2.1.3.2 Challenge 1: Main Objectives of Integrated SCs – Problems of SCM Conception

The first challenge of the performance-oriented integration of CT into SC concepts is the unsophisticated design of SCM conception. As a result, the corporate system of objectives of SC actors does not usually correspond to the main objectives of SCM.

According to Mentzer et al. (2001a), SCs can be defined as '[...] set of three or more entities (organisations or individuals) directly involved in the upstream and downstream flows of products, services, finances, and/or information from a source to a customer' (Mentzer et al. (2001a), p. 1). The SCM conception can accordingly be characterised by three main issues:

³³ In the given context, 3PLs are understood in a double role. On one hand, the central CT actors (e.g., operators, carriers, tractioners) are 3PLs. On the other hand, 3PLs can act as shipping companies and principals for intermodal transport concepts. Thus, in the applied understanding the term 'shippers' includes production and retail companies as well as 3PLs.

³⁴ cf. Sennheiser et al. (2008), pp. 186; Schönsleben (2002).

³⁵ cf. Wilke (2001), pp. 1.

(1) SCM must be understood as a system approach. This means that a SC must be considered as one unit and thus, managed as a whole.

The consideration of a SC as a system is confirmed by several authors.³⁶ In this interpretation, companies can be understood as system elements and with trade-off relationships.³⁷ For instance, in Wohlgemuth (2002) showed that the performance potential in a SC can only be disclosed when a whole value-adding network and not only dyadic partners and bilateral relationships are considered.³⁸ For the performance-oriented integration of CT into SC concepts this means that all CT actors must be included in the SCM and in planning activities.

(2) SCM is a strategic as well as a cooperative approach that spends resources for synchronising and converging activities between SC members.

This statement highlights the different aspects of SCM. First, it is assumed to be strategic. This means that SCM has a long-term character and it is understood as an instrument to reach competitiveness. Second, SCM is supposed to be cooperative. Thus, SC actors cooperatively work together to synchronise their material and information flow processes. Third, the term *'converging'* implies that SC actors need a common target system.³⁹ Transferred to the given research problem, this means that the performance-oriented integration of CT into SC concepts has strategic meaning for the involved actors. Furthermore, the integration can only be accomplished cooperatively with a common target system, which is valid for all SC actors.

(3) SCM addresses customer orientation. The authors point out that an individualised source of customer value must be established.

This statement highlights the SCM customer orientation.⁴⁰ As shown in the previous section, customer orientation is perceived as the key factor for competition among SCs. The integration of material and information flow processes is pointed out as the prerequisite for customer orientation. Customer orientation is expressed by the meeting of end customers' demands, for instance, regarding high delivery service, low costs and individualised products, despite obstructive framework conditions, such as infrastructure shortages or political regulations. For the integration of CT into SC concepts this means that CT actors have to align their activities towards the end customer structure.

³⁶ cf. Mentzer et al. (2001b), p. 1.

³⁷ cf. Karrer (2006), p. 86.

³⁸ cf. Wohlgemuth (2002).

³⁹ cf. Mentzer et al. (2001b), p. 7.

⁴⁰ cf. Min et al. (2007).

tomer orientation, too.

Although the importance of SCM in practice has been proved by several surveys,⁴¹ the initial ideas of the SCM concept often conflict with the applied systems of objectives. Furthermore, recent management approaches are not sufficient to deal with cross-company control problems. Hierarchical control approaches, differentiating between lead and operative instances, cannot be applied within a SC context.⁴² Thus, the interests of multiple stakeholders (e.g., shippers, terminal operators, CT operators, tractioners, carriers and end customers) have to be adjusted and conflicts of interests have to be considered. To realise advantages for all stakeholders, appropriate measures and instruments have to be applied. Whereas key figures of strategic management are appropriate for a post evaluation of functional divisions, processoriented and cross-company key figures (such as lead times, capacity utilisation, adherence to schedules and inventory levels) are not included in strategic, financial key figure portfolios.

Target System of SCM

There are five main SCM objectives. According to Gladen (2003), formal and content objectives of SCM are distinguished. The increase in end customers' benefits leads to an increase in the value of the SC. In order to increase the SC's value, companies attempt to decrease costs, realise time advantages and increase quality. Figure 2 -1 provides an overview of the linking between formal and content targets nd provides examples for associated target dimensions.

⁴¹ See, for instance, Accenture et al. (2003), Straube et al. (2008).

⁴² cf. Wildemann (2004b), Wildemann (2000), p. 8.


Figure 2 - 1: Linking of formal and content targets of SCM.⁴³

End customer orientation can be understood as the primary target of SCM. All other target dimensions can be derived from this overarching target.⁴⁴ End customers' benefits can be substantiated by the classical service objectives of logistics.⁴⁵ According to Pfohl (2010), these are the delivery time, delivery availability, delivery quality and delivery flexibility.⁴⁶ However, in SCM end customer benefit means the consequent orientation of all processes towards the end customers' needs.⁴⁷ The information about the specific state or status of the delivery can be interpreted as a benefit to the end customer, too.

With the increasing value orientation of management, value orientation principles are also integrated into the SCM conception.⁴⁸ This new way of thinking means that all value-adding activities in a SC must be aligned with the long-term target of value increase. Within a single company, this means that in SCM, the cause-and-effect relationship between often operative value drivers and corporate financial performance must be analysed.⁴⁹ Taking a cross-company perspective, value orientation means that the partial interests of single SC actors must be shifted towards the long-term maximisation of SC value. However, this target usually conflicts with the interests of

⁴³ With adaptations and enhancements, according to Ibid., p. 31.

⁴⁴ cf. Stefansson et al. (2000), p. 258; Heusler (2004), p. 20.

⁴⁵ cf. Pfohl (2010), pp. 36.

⁴⁶ cf. Ibid., pp. 43.

⁴⁷ cf. La Londe et al. (1994), pp. 46.

⁴⁸ For a value increase in SCM, see Otto et al. (2002), pp. 126; Möller (2003); Neher (2003); Hofmann et al. (2004).

⁴⁹ cf. Weber et al. (2001).

individual SC actors, who wish to increase their own values.⁵⁰

The decrease in costs, realisation of time advantages and increase in quality are the content targets leading to the formal targets of end customer orientation and value increase in the SC.

The idea of decreasing costs transfers the logistics principle of the 'total cost of ownership⁵¹ to SCM. SCM cost categories include the costs of the SC structure (e.g., fleet and warehouse costs, locations of production facilities, warehouses and transhipment points), SC processes (e.g., transport, handling, storage costs) as well as SC floating assets (e.g., inventory costs, introduction of pull philosophy, safety stocks).⁵² There are several cost accounting techniques to monitor SC costs.⁵³

The realisation of time advantages assumes that the end customer is generally interested in the fast delivery of the demanded product. Thus, the reduction in lead times and increase in adherence to schedules are sought. Different time spans are of special interest. The so-called 'time to market' means the time between product development and market introduction. The 'delivery time' means the time from order to delivery.⁵⁴ There are further time measures, such as 'time to react' to describe the duration required to react to short-term changes or to adapt the SC after disturbances.⁵⁵ For instance, the so-called 'Quick Response' concept gives the time dimension a specific meaning.56

The increase in quality means both the product and process quality. Benchmark is the subjective quality evaluation of the end customer.⁵⁷ For instance, the Total Quality Management approach focuses on a cross-company establishment of quality standards for process planning and operation.⁵⁸ Recently, the quality dimension has experienced an enhancement. Sustainability aspects are integrated into the quality understanding. Furthermore, knowledge management is encompassed, since the bundling of knowledge and resources can lead to learning effects in the SC.⁵⁹

⁵⁰ cf. Karrer (2006), pp. 54. 51

cf. Pfohl (2010), pp. 30.

⁵² cf. for instance Christopher (2005b), pp. 80, Bowersox et al. (1999), Callioni et al. (2005), pp. 135.

⁵³ For process cost accounting, transaction cost accounting or the total cost approach as the developments of accounting in logistics see Bowersox et al. (1999), pp. 91, Weber (2002), pp. 45. 54

cf. Christopher (2005a), p. 158.

⁵⁵ Background of the problem fields of complexity, intransparency and dynamics in SCs see Stölzle et al. (2001), pp. 75; Karrer (2006), pp. 54.

⁵⁶ The quick response concept interprets responsiveness, flexibility, the shortening of process times and real-time control as the main SCM target dimensions. For details see section 3.3.3. 57

cf. Pfohl (2006), pp. 260.

⁵⁸ See Iakovou et al. (2008); Linton et al. (2007); Simatupang et al. (2002); Seuring et al. (2008b); Taplin et al. (2006);.

⁵⁹ cf. Galgenmüller et al. (2004), pp. 84.

There are difficulties in the operationalisation of the introduced target dimensions. Additionally, mutual interdependencies between the objectives must be taken into consideration.⁶⁰ Owing to the number of SC actors – each of them with its own target systems and key figures – these interdependencies cannot be comprehensively modelled.⁶¹ However, the consideration of the different operative challenges of SC actors provides first insights into the conflicting goals in a SC including CT.

2.1.3.3 Challenge 2: Differing Operative Objectives and Problems of Shippers and CT Actors

There are multiple actors and operative challenges in one SC. However, between shippers and CT actors these challenges distinctively differ. Usually each SC actor focuses on his or her own field of operation and core competencies, although often the operative challenges have cross-divisional and cross-company characteristics. Nevertheless, the understanding of other SC actors' challenges and needs is usually limited. The integration of material and information flow processes requires encompassing knowledge on the operative challenges of all other actors.

In a first step, the different operative challenges of manufacturing, retail and transport actors are briefly discussed. Figure 2 - 2 gives an example of a SC including different modes of transportation. Exemplarily the figure provides an overview of the current operative challenges and focuses of different SC actors.

⁶⁰ cf. Karrer (2006), p. 22; Pfohl et al. (1997), pp. 88.

⁶¹ cf. Heusler (2004), p. 20, Karrer (2006), p. 22; Pfohl et al. (1997), pp. 89.



Figure 2 - 2: Differences in the operative challenges of selected SC actors.⁶²

SC managers of *retailers and manufacturers* aim to connect and synchronise the processes from the supplier's supplier to the customer's customer.⁶³ They try to connect the processes of the retail, service and producing companies from various industries. Each affected business unit deals with specific questions and sub-problems.⁶⁴

Manufacturers usually focus on the optimisation of quality and the resource efficiency of the production and assembly processes. Therefore, one central aspect is production planning and control (PPC).⁶⁵ For instance, the automotive industry claims PPC (next to the car assembly) is its core competency. According to the type of produced goods, manufacturers focus on either the realisation of economies of scale (reducing the marginal costs of mass production) or economies of scope (bundling effects through the common production of more than two products). They focus on the reduction of safety stocks, the utilisation of capacities (of machines, robots

⁶² Own illustration.

⁶³ See Cooper et al. (1997), p. 173.

 ⁶⁴ For instance, these operative challenges include supplier choice and supplier management, PPC, purchase prices, forecasting, marketing, customer management, shelf-availability and personnel effort.
 ⁶⁵ Example 1 and 1 an

⁶⁵ For further details on the tasks and objectives of PPC, see section 3.2.3.

etc.) and the acceleration of production, transport and logistics processes at a high quality level. This quality includes product quality, defective goods and adherence to schedules.⁶⁶

Retail faces the challenge of balancing assortment width,⁶⁷ marketing initiatives, fluctuating demand, missing storage room and personnel shortages because of low margins.⁶⁸ *Store management* focuses on optimal shelf availability, shelf space and the short replacement times accomplished by early morning and multiple deliveries.⁶⁹ The upstream DCs ensure stable and customised store delivery. Here, the focus is on efficient storage and order-picking systems as well as the provision and planning of the distribution transport and logistics processes.⁷⁰

The *3PLs and carriers* are influenced by the service character of transport and logistics.⁷¹ The main challenge is the immateriality of the service and the need to integrate the external factor. The customer of the logistics or transport service is the so-called external factor. This means that the performance of a service provider depends on the customer since the customer's behaviour, for instance at a good's provision, cannot be predicted.⁷² More and more 3PLs are integrated into the value-adding processes of shippers. For instance, 3PLs accomplish simple preparing production processes or value-adding processes, such as order picking and labelling.⁷³

Transport carriers must provide sufficient transport capacities and must economically utilise these.⁷⁴ *Road carriers* focus on the utilisation of transport capacities and the adherence to schedule dates and delivery windows for the delivery and collection of goods. Congestions exacerbate the punctual delivery of transported goods. The realisation of bundling effects and paired transport by appropriate route planning is essential in order to overcome the high level of fixed costs.⁷⁵ Further operative challenges include small delivery windows and rising transport costs because of tolls and energy costs. Increasingly road carriers apply the latest information and communication (I&C) systems, data loggers and temperature and state sensors to reduce transport

- ⁶⁷ For instance, hanging goods (clothes), cooled, frozen, perishable and hazardous goods.
- See, for instance, Credite-Suisse (2007), Gudehus (2005), p. 950; Kotzab (1997).
 A hush dama et al. (2010). David human et al. (2005)
- ⁶⁹ cf. Almehdawe et al. (2010); Baen (2001); Bretthauer et al. (2005)

⁶⁶ cf. Bessant et al. (1994); Gupta (1995); Kaijie et al. (2007); Lo et al. (2009); Sroufe et al. (2008); Stanley et al. (2001).

⁷⁰ See for instance Gudehus (2005), pp. 968.

For details on the service character of transportation processes see Bendul (2009); Corsten et al. (1992), pp. 181.

⁷² For a discussion of the meaning of the service character of CT see section 2.2.2.5.

For the scope of services provided by 3PLs see, for instance, Sheffi (1990); Forslund (2009); Carbone et al. (2005); Jayaram et al. (2011).
 G. D. L. (2000)

⁷⁴ cf. Bendul (2009).

⁷⁵ cf. Corsten et al. (1992); Stuhlmann (2000).

damages and loss, to increase transparency and to meet the information demand of shipments' recipients.⁷⁶ Furthermore, political regulations, such as emissions regulations, driving bans (for instance night, weekend and product driving bans), restrictive driving times and cabotage regulation influence road carriers' target systems.⁷⁷

Rail carriers as well as *CT operators* depend on the realisation of bundling and scheduling effects to deal with the high fixed cost level caused by specific investments. The availability and adherence of train paths, clearance profiles,⁷⁸ handling material and personnel are further challenges for CT actors. In particular, political aspects such as the infrastructure building as well as the dominance of passenger transport over freight transport (for instance, in Switzerland)⁷⁹ and regulation regarding noise emissions are central influencing parameters.⁸⁰

Inland or ocean vessel carriers operate on long-term schedules and depend on bundling effects as well as the realisation of economies of scale. The main advantage is the high transport volume and thus, the comparably low costs per unit and distance.⁸¹ Specific challenges are the comparable low speed, a strong weather dependency and for inland vessels the restricted inland water network.⁸² Infrastructure shortages in the European hinterland and the utilised capacities in ports can unexpectedly delay the delivery of goods. Lately, ocean vessel carriers are endangered off the East African coast because of pirate attacks.⁸³

Terminal operators respectively seaports are characterised by their locations and types. However, both inland and marine terminals focus on the capacity utilisation of investment-intensive handling equipment, such as cranes, reachstackers, waiting and storage spaces.⁸⁴ Increasingly, the offering of additional value-adding services, such as long-term storage, is central. At country borders as well as marine sites, short customs processes and fast transfers to the hinterland transport processes are the central evaluation factors for customers to choose ports and terminals.⁸⁵ Before the global

For technology usage in transport concepts see Skorna et al. (2009); Grimm et al. (2010).
 An anomaly a second discussion on the manning of transport policy impact on SC and CT see

An encompassing discussion on the meaning of transport policy impact on SC and CT see section 2.2.3.2.

A clearance profile is defined framework line characterised by the vertical across level of a driveway of roads and railways. For details see Thillmann (2006); N.N. (2004a); Siegmann (2011a).

⁷⁹ See Gehrken (2011).

⁸⁰ For details on the scope and impact of transport policy on transport concepts see section 2.2.3.2.

The Emma Maersk is the recently (July 2011) the bigest occar vessel. In sum 14.500 20' container TEU including 1000 containers for cooled goods can be loaded. For details see Mearsk (2011).
 For problems in water transport and for instance. N.N. (2011d). Even (2011a).

For problems in water transport see, for instance, N.N. (2011d); Jörgl (2011), N.N. (2011a).

⁸³ See Zamponi (2011); N.N. (2011c).

⁸⁴ cf. Seidelmann (1997); Seidelmann (2010).

⁸⁵ cf. Agersekov et al. (1983); Ballis et al. (2004); Wang et al. (2006)

economic crisis, many European inland and seaports and terminals had already reached their capacity limits. For instance, the port of Rotterdam, the Netherlands had lost transport volumes to smaller ports, such as Bremerhaven, Germany, since the customs processes and hinterland infrastructure connections allow shorter total lead times and higher adherence to schedules.⁸⁶ These examples show that the latest transhipment technologies and service orientation are central for terminal operators and ports.

Of these, several more actors indirectly influence SCP. They participate in the material flow processes themselves, but indirectly influence these *infrastructure providers, and authorities* provide the infrastructure framework for SC activities. The infrastructure providers are reliable for the enhancement and maintenance of the existing infrastructure of roads, railways, waterways and airports.⁸⁷

Authority and society further influence the enhancement of freight and people infrastructure. Examples are the reconstruction of the Stuttgart railway station⁸⁸ that was stopped by public protest and is still (August 2011) pending. The construction of the third runway at Frankfurt Airport⁸⁹ and the deepening of the Elbe River in Hamburg⁹⁰ are two more examples of projects temporarily delayed by non-governmental organisations.

Finally, *equipment, vehicle, wagon and locomotive manufacturers* influence the speed and performance of material, information flow and transhipment processes as well as noise and pollutant emissions.⁹¹

Each of the introduced actors focuses on its own viability, liquidity and individual company value growth. The different operative challenges are reflected by specific target systems and key figures. These key figures allow SC actors to evaluate corporate activities according to their impact on corporate value.

⁸⁶ cf. Stölzle et al. (2010); Stölzle et al. (2009).

⁸⁷ For details regarding the impact of transport policy on the integration of CT into SC concepts see section 2.3.

See for instance web pages of NGOs http://www.stuttgart21-ja-bitte.de/ or http://www.kopfbahnhof-21.de/ and the official web page http://www.das-neue-herz-europas.de/default.aspx.
See for instance propert (2011)

⁸⁹ See for instance Fraport (2011).

⁹⁰ cf. N.N. (2009).

⁹¹ See for instance VÖV (2008), pp. 38.

2.1.3.4 Challenge 3: Focus on Individual Company Value Growth

Companies increasingly focus on financial key figures and company value growth. Value-oriented management or the 'value-based view' is the most popular stream of strategic management.⁹² Owing to its service character, the quantification of logistics and the economic value added (EVA) of SCM is a special challenge.⁹³

There are different approaches to identify the EVA of SCM. Basically, the parameters influencing company value are connected with the driving parameters anchored in operative SC processes. The target dimensions mostly applied are yield, operating expenditure, floating capital and invested capital.⁹⁴

The differing operative challenges and the value orientation of single SC actors and CT actors are reflected by the applied target systems and their operationalisation in terms of key figures. These target systems include several conflicts. One example is that procurement activities aiming at a cost reduction are contrary to quality or supplier flexibility. In manufacturing the utilisation of production capacities does prejudice efficient storage and distribution because of set-up costs and lot size questions.⁹⁵ Several other target conflicts have been broadly discussed in practice and research.⁹⁶ These trade-off situations have to be considered when shortage costs are compared with storage costs or when transport speed is assessed with respect to transport costs (transport mode choice).⁹⁷

2.1.4 **Vulnerability of Integrated SC Concepts**

The cross-company integration of material and information flow processes is the basis of SC actors meeting customer demand, but it also causes vulnerability and mutual dependencies. As shown in the previous section, product availability is one key element for customer orientation. Interruptions in the material and information flow can quickly cause delivery disruption at the customer interface because of low inven-

⁹² cf. Horváth et al. (2001), pp. 273; Karrer (2006), pp. 156; Möller (2003), p. 58.

⁹³ cf. Karrer (2006), p 2; Pfohl (2010), pp 24. There are different scientific publications with the quantification and measurement of logistics and SC activities to the corporate value and at the development of recommendations for SCM based on value-oriented key figure hierarchies. See Alicke (2003); Arnold (2002); Hofmann et al. (2009); Jehle (2004); Marbacher (2001), p. 251. 94 cf. Karrer (2006), pp. 163; Lambert et al. (2001), p. 12.

⁹⁵

See publications considering the operations scheduling dilemma. e.g., Nyhuis et al. (2008), pp. 4; Pfohl (2010), pp. 33. 96

Such as Pfohl (2010), pp. 33. 97

Ssuch as Ibid., pp. 33.

tory levels. This increases the dependency among SC members and increases SC vulnerability.

The *infrastructure shortages* in European hinterland and *delays in the port processes* endanger time critical, global SC concepts. Most imported or exported goods enter and leave Europe via North Sea ports. Imported goods arrive at a few main ports in Europe, such as Rotterdam, Antwerp, Hamburg and Bremerhaven. At these main hubs, the goods are either bundled for sea haulage or distributed to their destinations in Europe.

Figure 2 - 3 shows recent infrastructure shortages in Germany. The red points out the rate of delays and congestions in road transport. The situation in Germany is representative of further European regions.⁹⁸ The figure shows that the goods for southbound transport can be affected by infrastructure shortages. Plain to see are congestions in the Ruhr area. All transport coming from Europe's biggest port in Rotterdam must pass this region.⁹⁹

Not only European roads, but rail and water hinterland infrastructure suffer significant capacity shortages. However, in comparison, the rail and water infrastructure still has free capacity depending on the transport relation and time.

⁹⁸ cf. EK (2005).

⁹⁹ cf. Planco (2007), pp. 5.



Figure 2 - 3: Road infrastructure shortages in Germany.¹⁰⁰

The previous paragraph showed the meaning of European infrastructure availability on the performance of global SCs. Thus, *transport policy* is a further central aspect for the design of global SCs. The development of transport policy and implementation take place at all political levels – local, regional, national and EU. National transport policies are crucial for infrastructure development. This applies to rail, road and waterway infrastructure as well as to interconnection points such as ports, terminals, logistics parks, maintenance and traffic management. However, transport policy influences global SCs through instruments such as regulations, traffic-specific taxes and duties, infrastructure development as well as international harmonisation and standardisation¹⁰¹.

Although the integration of material and information flow processes is necessary to meet customer demand and improve SCP, there are several examples of negative consequences for upstream and downstream SC actors caused by one company. These secondary failures and disruptions of material flow processes for downstream actors

cf. Ibid., p. 7. For details on the effect of infrastructure shortages see cf. Stölzle et al. (2009). For the impact of transport policy see section 2.2.3.2.
 cf. Ibid., p. 7. For details on the effect of infrastructure shortages see cf. Stölzle et al. (2009). For the impact of transport policy see section 2.2.3.2.

¹⁰¹ For the further consideration of the impact of transport policy on SCs, see section 2.2.3.2

are defined as 'vulnerability'. Such a situation requires flexible reactions and fast adaptations by all SC actors. To avoid growing storage volume, the production and procurement processes of all suppliers and sub-suppliers as well as distribution processes are to be suspended. The specific transport processes must remain available. At the same time, all SC actors have to prepare for the following ramp up situation.¹⁰² The following example shows the vulnerability of SCs caused by snowball effects. In 2001, a fire in a Mexican *Philips* plant led to a three-week production disruption at the Swedish company *Ericsson*.¹⁰³ The onset of foot-and-mouth disease in Great Britain in 2001 led to production stops at *Volvo* and *Ford*¹⁰⁴ and quality variations because of constructive changes at the diesel pumps for direct injection systems at *Bosch* led to breakdowns in nearly all German automotive factories in 2005.¹⁰⁵ Such series of reactions are also called 'snowball effects'.¹⁰⁶

SC integration faces several further challenges. Companies are members of more than one SC (*'multi-memberships'*). The comprehensive integration of all business partners is from an economic point of view not always reasonable because some relationships are only short-term. The variety of publications on dyadic cooperation proves that the management of one-to-one relationships is already a demanding task.¹⁰⁷ Trust and power aspects contradict close partnerships and integration. In practice, these problems lead to an integration of only the immediate vertical and horizontal SC partners of the focal company. Diseconomies of SC integration also arise from rising managing complexity, decreasing flexibility, financial risks and the loss of specific competencies. Furthermore, companies run risks by getting into dependencies, loss of freedom and draining knowhow.

There is a number of measures and instruments to support integration activities. The following section presents the results of a comprehensive literature review on integrative measures and instruments with regard to technical and organisational integration aspects.

¹⁰² Further examples are the onset of foot-and-mouth-disease in Great Britain in 2001 leading to production stops at Volvo and Ford (cf. Pfohl (2002)). In 2005 nearly all German automotive factories broke down due quality variations caused by the constructive change in the diesel pump for direct injection systems at Bosch (cf. Kersten et al. (2007), p. 13). Such series of reactions are also called 'snow ball effect'. cf. Sheffi (2005), pp. 143.

¹⁰³ cf. Norman et al. (2004), p. 441; Svensson (2004), p. 729.

¹⁰⁴ cf. Pfohl (2002).

¹⁰⁵ cf. Kersten et al. (2007), p. 13.

¹⁰⁶ cf. Sheffi (2005), p. 143.

¹⁰⁷ See for example Fabbe-Costes et al. (2007), p. 843; Sahin et al. (2005), pp. 579; Vachon et al. (2006), pp. 795.

2.1.5 Measures and Instruments for SC Integration

The literature provides a number of concrete measures and instruments to support customer orientation and material and information flow integration in SCs. Cooper (1997) distinguished integration activities into managerial and behavioural as well as technical and organisational management components.¹⁰⁸ Managerial and behavioural measures cover employee behaviour (e.g., election, advanced training, hospitalisation, incentive systems). The technical and organisational management components focus on material and information flow processes, structures and organisational aspects.¹⁰⁹

According to Cooper et al. (1997), the technical and organisational management components encompass management components regarding the I&C flow facility structure, workflow and activity structure, organisational structure, planning and control methods and product flow facility structures. Furthermore, definitions of architecture and components of I&C systems as well as on the material flow systems are considered. Figure 2 - 4 shows these considerations and illustrates allocating the management components into two main categories.

Technical and organizational management components	M m
I&C flow facility structure	n
work flow / activity structure	p
organization structure	r
planning and control methods	c
product flow facility structure	

Managerial and behavioral management components	
managementmethods	
power and leadership structure	
risk and reward structure	
culture and attitude	

Figure 2 - 4: Management components of SC integration.¹¹⁰

The thesis in hand excludes managerial and behavioural management components from deeper consideration. This focus reduces the complexity of the given problem formulation and corresponds to the neutral and non-actor-specific research perspective. The exclusion of problems regarding behavioural aspects, culture and leadership supports the perspective on the potential and chances of CT integration.

cf. Cooper et al. (1997), p. 12; Häusler (2001) distinguished between integration measures on a socio-emotional and formal-rational level. cf. Häusler (2001), p. 225.
 ¹⁰⁹

¹⁰⁹ cf. Cooper et al. (1997), pp. 12; Heusler (2004), p. 225.

¹¹⁰ With adaptations in accordance with Cooper et al. (1997), p. 12.

Following Cooper's definition of physical and technical management components (cf. section 2.1.2), an encompassing literature review was conducted. Several subcategories of specific technical and organisational measures and instruments of the five main categories could be identified.

- (1) I&C flow facility structure,
- (2) Workflow/activity structure,
- (3) Organisational structure,
- (4) Planning and control methods and
- (5) Product flow facility structure.

The developed catalogue of integrative measures and instruments serves as the basis for the examination of suitable integration measures for the performance-oriented integration of CT into SC concepts (cf. section 4.3).

There are measures that cannot be clearly allocated to one group. Furthermore, it must be considered that integrative measures do not work in isolation. Technical and organisational as well as managerial and behavioural measures work hand in hand. Usually groups of integrative measures are applied together. Thus, for an encompassing integration, managerial and behavioural approaches have to be considered, too. Figure 2 -5 provides an overview of the categorisation of technical and organisational management components for the integration of CT into SC concepts.



Figure 2 - 5: Overview of the technical and organisational management components derived from the literature.¹¹¹

¹¹¹ Own illustration based on literature review.

(1) I&C flow facility structure

Information technology (IT) enables the sharing of large amounts of data between companies. Several authors have discussed the critical role of IT on SCM activities.¹¹² Cooper et al. (1997), for instance, stated that the type and frequency of information passed and updated among SC actors strongly influences SC efficiency.¹¹³ I&C flow facility structure is suggested as the first measure for a SC integration project.¹¹⁴ The integration of I&C systems reduces technical barriers and incompatibilities and thus, improves communication between SC actors.¹¹⁵ Often, transaction cost theory serves as an explanation of the effect of IT systems' integrations. For instance, Johnson et al. (2007) confirmed the relationship among the availability of IT, SC applications and decreasing transaction costs.¹¹⁶ However, some authors have pointed out that it is not sensible to integrate all SC members in terms of I&C technologies.¹¹⁷ The availability of a well-functioning I&C flow facility structure serves as the basis of a number of further integrative measures, especially for the setup of joint planning and control measures.

The literature review identified four main groups of integrative measures and instruments regarding I&C flow facility: the application of (1) integrated I&C methods, technologies and systems, the usage of (2) online platforms and online data provision, the use of (3) automatically identification technologies and (4) the standardisation of I&C technologies.

Table 2 - 1 provides an overview of the publications dealing with integrative measures suitable from the field of I&C flow facility structure classified according to these four categories. If possible, publications are classified according to relevant sub-topics.

¹¹² See for instance Kearns et al. (2003); Vickery et al. (2003); Vanpoucke et al. (2009); Welker et al. (2008).

¹¹³ cf. Cooper et al. (1997).

¹¹⁴ cf. Ibid., p. 78.

¹¹⁵ cf. Bowersox et al. (2002); Li-Ling et al. (2009), p. 101; Sanders (2007); Vickery et al. (2003).

¹¹⁶ cf. Johnson et al. (2007).

¹¹⁷ See for instance Das et al. (2006); Vanpoucke et al. (2009), p. 1217; Williamson (1996).

integrated I&C	latest I&C sharing systems, methods and technologies: e.g., Bagchi & Skjoett-Larsen (2002); Murphy &
methods, tech-	Farris (1993); Chen & Paulraj (2004); Handfield & Bechtel (2002); Sanders (2007); Stefansson
nologies and	(2002)
systems	integrated databases, warehouses and collaboration systems: e.g., Murphy & Farris (1993); Chen &
	Paulraj (2004); Cheng (2010); Dehning et al. (2007); Fiala (2005); Frohlich & Westbrook (2002);
	Handfield & Bechtel (2002); Ke & Wei (2007); Kobayashi et al. (2003); Li et al. (2006a); Sanders
	(2007); Zhu & Kraemer (2002)
	joint measure, control and sharing of process data: e.g.,Murphy & Farris (1993); Chen & Paulraj
	(2004); Das et al. (2006); Frohlich & Westbrook (2002); Vanpoucke et al. (2009); Welker et al.
	(2008); Zhu & Kraemer (2002)
online platforms	e.g., Devaraj et al. (2007); Frohlich & Westbrook (2002); Hill & Scudder (2002); Jayanth & Keah-
and online data	Choon (2010); Mukhopadhyay & Kekre (2002); Zhu & Kraemer (2002)
provision	
automatically	bar coding, RFID-technology etc.: e.g., Amador & Emond (2010); Chao et al. (2007); Chao et al.
identification	(2007); Delen et al. (2007); Klein & Thomas (2009); Lee et al. (2011); Lin (2009); Anand et al.
technologies	(2009; Lin (2009); López (2011); Ngai & Riggins (2008); Sarac et al. (2010); Shin et al. (2011);
	Strassner & Fleisch (2005); Sundaram et al. (2010); Tajima (2007); Véronneau & Roy).
	sensor technology: e.g., Amador & Emond (2010); Anderson & Hong (2008); Ilgin & Gupta (2010);
	López (2011); Makinwa (2010); Shin et al. (2011); Wada et al. (2011); Yang et al. (2010)
standardisation	e.g., Akkermans & van der Horst (2002); Carlson et al. (2001; Polo-Redondo & Cambra-Fierro); Gunasekaran &
of I&C tech-	Ngai (2004); Jagdev & Thoben (2001); Keil et al. (2001); Lancioni et al. (2000); Polo-Redondo & Cambra-Fierro
nologies	(2008); Sanders (2008); Shin & Leem (2002).

Table 2 - 1: Literature review of integrative measures from the field of I&C flow facility structure.

(2) Workflow activity structure

Work structure indicates how tasks and activities are performed in a SC.¹¹⁸ Integrative measures of this group concern process structure and thus, directly influence the types, structures and sequences of material and information flow processes.¹¹⁹

In this category, the literature review identified five types of integrative measures and instruments: (1) online or electronic data interchange, (2) latest material flow equipment and technologies, (3) standardisation activities, (4) SC-wide quality standards and (5) collaborative improvement techniques. Table 2 - 2 provides an overview of the publications on integrative measures from the field of workflow and activity structure classified according to these seven categories. If sensible, publications are classified according to relevant sub-topics.

¹¹⁸ cf. Andrews et al. (1994); Ellram et al. (1990); Hewitt (1994).

¹¹⁹ cf. Kannan et al. (2005); Kim et al. (2011); Kobayashi et al. (2003); Liu et al. (2005); Nielsen et al. (1995); Li et al. (2006b); Qiu et al. (2006); Salema et al. (2010); van der Aalst et al. (2007); van der Aalst et al. (2003); Vrijhoef et al. (2000).

online or elec-	information sharing for B2B or B2C online purchasing: e.g., Mukhopadhyay & Kekre (2002);			
tronic data inter-	Frohlich & Westbrook (2002); Zhu & Cote (2004); Kong et al. (2004); Li & Lin (2006)			
change (EDI)	customer configuration/customisation of designated goods: e.g.,Murphy & Farris (1993)			
	online auctions: e.g., Murphy & Farris (1993); Pflughoeft et al. (2003); Poirier & Quinn (2004);			
	Poirier & Quinn (2004); Ranganathan et al. (2004); Zhu & Kraemer (2002)			
	demand oriented web based information sharing: for instance sharing of real-time point-of-sales data			
	and sales forecast: e.g., Aviv (2001); Cachon & Lariviere (2001); Choi (1995); Christopher (2005a);			
	Jaber et al. (2010); Anand et al. (2009); Wu & Chen (2006)			
	customer profiling and customer relationship management (CRM): e.g.,Bönte (2008); Dupre & Gruen			
	(2004); Frohlich & Westbrook (2002)			
	supply-oriented information sharing: e.g., provision of inventory ordering policies, inventory levels:			
	e.g.,Murphy & Farris (1993); Frohlich & Westbrook (2001); Krajewski & Wei (2001); Yang et al.			
	(2009); Zhu & Kraemer (2002)			
	master production schedules: e.g., Barua et al. (2000); Das et al. (2006); Frohlich & Westbrook (2001);			
	Krajewski & Wei (2001); Lancioni et al. (2000)			
	collaborations on net material requirements: e.g., Murphy & Farris (1993); Cachon & Lariviere			
	(2001); Krajewski & Wei (2001); Zhu & Kraemer (2002)			
latest material	e.g., Abacoumkin & Ballis (2004); Agersekov et al. (1983); Arnold et al. (2004); Ballis & Golias			
flow equipment /	(2002; Ballis & Golias (2004); Basu & Wright (2008); Chao et al. (2007); Dias et al. (2010); Erera et			
technologies	al. (2005); Evers et al. (1996); Greeff et al. (2004); Halevi (2001); Kreutzberger et al. (1999);			
	Macharis & Pekin); Meyer et al. (2009); Min & Zhou (2002); Ozbayrak et al. (2007); Panayides &			
	Song (2008); Sharma et al. (1999); Sheu (2002); Stefanovic et al. (2009); Thomas & Griffin (1996);			
	Walker et al. (2008); Wang & Cullinane (2006)			
standardisation	e.g., Lenzen et al. (2007); Akkermans et al. (2003); Akkermans & van der Horst (2002); Carlson et			
activities (material	al. (2001); Gunasekaran & Ngai (2004); Hatteland (2004); Keil et al. (2001); Lancioni et al. (2000);			
and information	Polo-Redondo & Cambra-Fierro (2008); Sanders (2008); Shin & Leem (2002)			
flow technologies)				
SC-wide quality	e.g., Anand et al. (2009); Basu (2008); Bessant et al. (1994); Buttermann et al. (2008); Foster Jr			
standards (e.g.,	(2008); González-Benito et al. (2003); Kannan & Tan (2005); Kaynak & Hartley (2008); Parast);			
DIN/ISO 9000 Six	Sroufe & Curkovic (2008); Truscott (2003); Sroufe & Curkovic (2008); Stanley & Wisner (2001);			
Sigma etc.)	Yeung (2008); Lo et al. (2009)			
collaborative	Collaborative continuous improvement techniques: e.g.,Choi (1995); Jaber et al. (2010); Anand et al.			
improvement	(2009); Wu & Chen (2006)			
techniques	poka yoke: e.g., Basu (2008); Barash); Harvey (1998); Lee & Lim (2005); Plenert (2007); Schippers			
	(2001); Tang & Tomlin (2008); Truscott (2003); Tsou & Chen (2008)			

Table 2 -	- 2:	Literature	review	of integrative	measures	from the	e field o	f workflo	w and a	octivity	structure.
				<u> </u>						•/	

(3) Organisational structure

According to Lambert & Cooper (2000), organisational measures influence both the individual firm and the SC.¹²⁰ The literature review showed up two groups of organisational measures and instruments: (1) cross-company operational activities, such as cross-functional teams account managers and dedicated planners for one buyer, and (2) strategic integrative activities, such as joint ventures and the creation of quasi-firms. Table 2 - 3 provides an overview of the publications on integrative measures from the field of organisational structure classified according to these two categories.

cross-company opera- tional activities (e.g., cross-functional teams / personnel; account man- agers etc.)	e.g., Alicke (2003); Cooper et al. (1997) Bagchi & Skjoett-Larsen (2002); Droge et al. (2004); Moffat (1998); Moore & Antill (2001); Lee-Kelley & Sankey (2008); Loo (2003); Panayides & Venus Lun (2009); Perona & Saccani (2004); Sako & Helper (1998); Thamhain (2004); Wang et al. (2005)
strategic integrative activities (e.g., Mergers &Acquisitions (M&A) activities; creation of quasi-firm; joint ventures etc.)	e.g., Alicke (2003); Cousins et al. (2006); Jayaram & Tan (2011); Kandemir & Hult (2005); Kroes & Ghosh); Ku et al. (2007); Petersen et al. (2005); Sarkis et al. (2011)

 Table 2 - 3: Literature review of integrative measures from the field of organisational structure.

(4) Planning and control methods

The planning and control of operations is central to a constant and stable crosscompany material and information flow. According to Lambert & Cooper (2000) and Alicke (2003), the joint planning of processes improves the quality of material flow processes.¹²¹ The literature review identified three main groups of planning and control measures: 'classic' (1) joint planning and control systems, (2) joint planning and control approaches and handling and the usage of (3) modelling and simulation techniques. Table 2 - 4 gives an overview of the integrative measures from the field of planning and control methods. If sensible, publications are classified according to relevant subtopics.

¹²⁰ cf. Lambert et al. (2000).

¹²¹ cf. Alicke (2003), p. 67; Lambert et al. (2000), p. 71.

• • • • • • • • • • • •	A C 1 in Manual international (2010). A list (2002). A second g. K. strang (2009). A set in terms of						
joint planning	e.g., Afshin Mansouri et al. (2010); Alicke (2003); Amaral & Kuettner (2008); Arshinder et al.						
and control	(2008); Barua et al. (2000); Bergström & Stehn (2005); Bose et al. (2008); Brun et al. (2006); Carr &						
systems	Kaynak (2007); Chang et al. (2008); Fleischmann et al. (2003); Frohlich & Westbrook (2002); Fugate						
(e.g., ERP, PPC	et al. (2006); Giunipero et al. (2008); Grawe et al. (2009); Henning et al. (2009); Hvolby & Steger-						
and APS (ad-	Jensen (2010); Gibson et al. (2005); Henning et al. (2009); Hvolby & Steger-Jensen (2010); Jonsson						
vanced planning	blanning et al. (2007); Jung & Lee (2010); Mabert et al. (2003); Newell et al. (2003); Poirier & Quinn (2004);						
and scheduling)	Rajagopal (2002); Rutner et al. (2003); Sanders (2007), Shobrys & White (2002); Soltani & Sadjadi						
systems and	(2010); Su & Yang (2010); Stadtler (2005); Verwijmeren (2004); Zachariassen (2008); Zoryk-Schalla						
modules etc.)	et al. (2004).						
joint planning	improved planning techniques, joint data mining, shared sales forecasts: e.g., Barua et al. (2000);						
and control	Cachon & Lariviere (2001); Devaraj et al. (2007); Frohlich & Westbrook (2001); Lee et al. (1997).						
approaches	joint capacity and demand planning through CPFR: e.g., Barua et al. (2000); Cao & Zhang (2010);						
	Chan & Zhang (2011); Esper & Williams (2003); Frohlich & Westbrook (2002); Hill & Scudder						
	(2002); Lee et al. (1997); Lee et al. (2011); Lyu et al. (2010); Mert et al. (2007); Pibernik et al.						
	(2007); Ramanathan & Muyldermans (2011); Sari (2008); Son & Sheu (2008); Whipple et al. (2010)						
	autonomous control approaches: e.g., Cimino & Marcelloni (2011); Hua et al. (2003); Lu & Chen						
	(2010); Scholz-Reiter et al. (2008); Siemieniuch & Sinclair (2002); Hülsmann & Windt (2007);						
	Windt & Bendul (2006); Windt et al. (2008)						
modelling and	reference models: e.g., Power (2005); Maier et al. (2002); Shin & Leem (2002); Röder & Tibken						
simulation tech-	(2006); Simon et al. (2007); Persona et al. (2007); Sarimveis et al. (2008), Trappey et al.); Verdouw et al.						
niques	(2010)						
	modelling and simulation approaches: e.g., Biethahn et al. (1999); Biswas & Narahari (2004); Biswas						
	& Narahari (2004); Caputo et al. (2003); Chan & Zhang (2011); Chatfield et al. (2009); Elmahdi et al.						
	(2006); Escudero et al. (1999); Georgiadis et al. (2005); Gunasekaran (2004); Hung et al. (2006); Li						
	et al. (2006b); Mahnam et al. (2009); Manuj & Mentzer (2008); Meixell & Norbis (2008); Mizgier et						
	al. (2010); Özbayrak et al. (2007); Rizzoli et al. (2002); Mele et al. (2006); Mizgier et al. (2010);						
	Neiro & Pinto (2004); Röder & Tibken (2006); Sinha et al. (2011); Rodriguez-Rodriguez et al.						
	(2011); Sarac et al. (2010); Sundaram et al. (2010); Trkman et al. (2007); Varma et al. (2007)						

Table 2 - 4: Literature review of integrative measures from the field of planning and control methods.

(5) Product flow facility structure

Product flow facility structure refers to the logistics-oriented product and load carrier design as well as packaging, which are approaches to improve handling, loading and capacity usage for the production, distribution and transport of goods, and thus, the interruption-free flow of information and material.¹²² Table 2 - 5 provides an overview of the integrative measures of product flow facility structure.

logistics-oriented	e.g., Diabat & Govindan (2011); Güngör et al. (2011); Hua et al. (2011); Kaijie et al. (2007); Ko et
product and load	al. (2011); Liu (2011); Rostamzadeh & Sofian (2011); Smith & Yen (2010); Sroufe & Curkovic
carrier design	(2008); Wang & Zhang (2010); Xia & Chen (2011)

Table 2 - 5: Literature review of integrative measures from the field of product flow facility structure.

¹²² Bretzke et al. (2010) introduced different approaches to a logistics-oriented product design, e.g., the reduction of bulkiness to reduce the required transport capacity, the reduction of product weight ('light weight construction') to diminish fuel consumption, the enhancement of the products' useful life to reduce the disposal processes, the modularisation of products and standardisation of products or product modules. cf. Bretzke et al. (2010), pp. 79.

The measures and instruments introduced in this section support the integration of material and information flow processes and thus, SCP. The complexity of CT exacerbates the integration of material and information flows because of several inherent and external challenges. The main characteristics, structures and elements of CT concepts are introduced in the following section.

2.2 Coping with CT Complexity in SCs

This section addresses CT's inherent challenges in the integration into SC concepts. CT underlies several market- and production-related disadvantages in comparison with unimodal road transport. The characteristics and challenges of CT concepts must be known to be successfully integrated. Thus, this section provides an overview of the application fields of CT concepts, discusses different aspects of CT service complexity and addresses the characteristics and challenges exacerbating the competitive situation with unimodal road transport.

2.2.1 Application Fields of CT

Economic and political players increasingly perceive CT as a promising solution to deal with problems in road transport. It is assumed that the combination of rail and road transport may prospectively even increase SCP improvements in comparison with unimodal road transport.¹²³ However, in comparison with unimodal road transport, CT has some inherent and market-related disadvantages.

Intermodal transport is defined as the *usage of one or multiple carriers for the transport of an empty or filled load carrier*. The main distance is covered by rail, ocean or inland vessel, whereas the comparably short pre- or ongoing transport is accomplished by road. CT describes a subset of intermodal transport and it is commonly applied to the carrier combination of road and rail transport.¹²⁴ However, often the terms combined, intermodal and multimodal transport¹²⁵ are synonymously used.¹²⁶ The thesis in hand focuses on the combination of rail and road transport. This CT concept has an inner-continental character, but it is often combined with previous inter-continental

¹²³ For instance, in the case of significantly rising energy prices as well as growing infrastructure shortages.

¹²⁴ cf. Hoffmann (2007), p. 14; Polzin (1999), p. 81.

¹²⁵ Usually the term ,multimodal transport' stands for the combination of at least three carriers.

¹²⁶ cf. Hoffmann (2007), p. 14. A combination with air freight is possible too, but is rather uncommon.

water transport.¹²⁷

The term '*CT concept*' determines all necessary technical, organisational, player, process and carrier definitions to ensure a stable time-, quality- and cost-defined transport of goods from a certain source to a delivery point by a combined rail and road transport with at least one transhipment process.

Advantages of CT

CT concepts allow the integration of rail -transport into SC concepts in regions without trackage systems or side tracks. CT joins the inherent advantages of the different carriers and, thus, means an affordable and environmental alternative to unimodal road transport. It is flexible as road transport is used to supply geographically distributed shippers and customers and it is characterised by the low emissions level and cost efficiency of rail transport for large transport quantities over long distances. In particular, the potential emissions reduction and the associated image effect is one main advantage for shippers and 3PLs to integrate CT into SC concepts. In 2008, the Association Materials Management, Purchasing and Logistics (AMMPL) published a survey on the reduction of CO_2 emissions.¹²⁸ In total, 170 German production and retail companies were surveyed according to their modal splits and the specific reasons for their transport carrier choices. The objective was to understand carrier choice and to analyse the relationship between CO_2 emissions and modal split. The survey showed that the main reason for shippers to switch from road transport to rail or CT is the improvement in CO_2 emissions balance.¹²⁹

CT has further advantages in comparison with unimodal rail transport. With the liberalisation of the European railway market, state rail carriers focus on CT rather than on single wagon freight transport. High fixed costs, the competitive situation and the resulting need for rationalisation are the reasons.

In Norway, Denmark, Spain and Great Britain, there is no single wagon freight transport because of the high level of fixed and availability costs.¹³⁰ The Italian, French, Austrian and Swiss state railways have significantly minimised their single wagon transport nets in the past decade.¹³¹ In 2002, the *DB AG* reduced operated railway sid-

¹²⁷ cf. Wang et al. (2010).

¹²⁸ cf. Wittenbrink (2008).

¹²⁹ cf. Ibid., p. 10.

¹³⁰ cf. DBAG (2007), p. 7.

¹³¹ cf. VAP (2010); Progtrans (2007), pp. 30, 50.

ings from approximately 2100 by 1200 (Mora C)¹³² to make the system more economical. According to de Jong (2011), single wagon freight transport at DB AG was dependent on few big shippers, but 70% of the customers generated only 5% of the revenues. The *DB* AG aimed for a cost reduction of approximately 250 Mio. Euro.¹³³

Today, the Swiss and French state railways perceive CT as central aspects for the financial reorganisation of freight transport divisions.¹³⁴ In Germany, the *DB AG* is increasing activities in the field of CT, too.¹³⁵ Furthermore, the traditional unimodal rail transport specific wagons are necessary. Thus, round trips are difficult to realise. According to an expert of the German consulting company *KombiConsult* (2011), in CT 85–100% of transport is paired.¹³⁶

Several transport political regulations have improved the competitive situation of CT in comparison with unimodal road transport. In Germany, transport vehicles, restricted to pre- and ongoing haulage of CT, are exempt from motor vehicle tax.¹³⁷ Furthermore, there are exemptions from driving bans for the pre- and ongoing haulage to the next closest shipping or receiving terminal. For below 200 km pre- or ongoing haulage, the exemption also encompasses Sundays and holidays.¹³⁸ Furthermore, the maximum weight for pre- and ongoing transport has been increased to 44 tons.¹³⁹ The Swiss government subsidises the transalpine CT, namely the terminal building and shipments.¹⁴⁰

Typical Products for CT

Generally, goods with stable and high transport volumes in both directions are affine for CT. Today, especially, dry, free-flowing and fluid hazardous goods are transported via CT.¹⁴¹ For such transport, special load carriers are used. Thus, unpaired transport is accepted since cleaning costs often exceed idle costs. Increasingly bulk cargo is transported in boxes and so-called 'bulk containers'.¹⁴² Furthermore, heavy goods, such as

¹³² cf. VDV (2000).

¹³³ cf. De Jong et al. (2007). ¹³⁴ of Heinriei (2011) ≈ 22

¹³⁴ cf. Heinrici (2011), p. 33. ¹³⁵ of Da Jang et al. (2007)

¹³⁵ cf. De Jong et al. (2007).

¹³⁶ cf. Heinrici (2011), p. 33. ¹³⁷ cf. § 3 No. 9 a KraftStG.

¹³⁷ cf. § 3 No. 9 a KraftStG. (Online available under: http://www.gesetze-im-internet.de/kraftstg/__3.html).

¹³⁸ cf. § 30 StVO. (Online available under http://bundesrecht.juris.de/stvo/__30.html).

 ¹³⁹ Kombiverkehr provides a comprehensive calculation example pointing out the economical advantages of this. cf. Kombiverkehr (2011).
 ¹⁴⁰ For further details of transport policy see section 2.2.3.2

¹⁴⁰ For further details of transport policy see section 2.2.3.2. ¹⁴¹ of Saidelmann (2010) pp. 31

¹⁴¹ cf. Seidelmann (2010), pp. 31. ¹⁴² of Müller (2011)

¹⁴² cf. Müller (2011).

metals or building materials, are commonly transported by CT.¹⁴³ Lately, also specific trailers for the CT of glass have been offered.¹⁴⁴

Owing to the high fixed costs level, CT is commonly applied to long and international distances. Several publications have stated these minimum transport distances. However, authors have stated that CT is not economically possible on distances below 300 km.¹⁴⁵ However, latest examples show that CT can also be integrated into SCs for short distances.¹⁴⁶

CT Quantities in Europe

Table 2 - 6 gives an overview of transport quantities in tons and TEU in Europe for domestic and international transport. The table shows that domestic inland transport comprises approximately two thirds in comparison with domestic continental transport. Likewise, in international transport continental transport comprises approximately two thirds and container hinterland transport the other third.

CT market segment	continental tra	nsport	container hinterland transport		total CT	
domestic (in tons)	30 199 059	33.0%	61 313 241	67.0%	91 512 300	100%
(in TEU)	2 984 189	32.0%	6 341 401	68.0%	9 325 590	100%
international	39 803 680	63.2%	23 226 120	36.8%	63 029 800	100%
(in tons)						
(in TEU)	3 707 550	60.5%	2 415 730	39.5%	6 123 280	100%
all services	70 002 739	45.3%	84 539 361	54.7%	154 542 100	100%
in TEU	6 691 739	43.3%	8 757 131	56.7%	15 448 870	100%

Table 2 - 6: Transport quantities in CT (unaccompanied) in 2009 in tons and TEU.¹⁴⁷

As Table 2 - 6 shows, in Europe CT are predominantly applied to the so-called *'hinter-land'*-distance from the North or Mediterranean Sea into central Europe and for alpine transit.¹⁴⁸ Figure 2 - 6 illustrates the distribution of European transport quantities in CT. Restrictive regulations support the transalpine CT.¹⁴⁹ For instance, Switzerland has set up a programme to reduce road transport volumes by two thirds between 2001 and 2018.¹⁵⁰ In 2009, Austria enacted a law that progressively restricts the transport of

¹⁴³ cf. Part 3 § 34 Absatz 6 Nummer 6 StVZO. (online available under http://www.stvzo.de/stvzo/B3.htm#34 and http://www. kombiverkehr.de/neptun/neptun.php/oktopus/ download/58).

¹⁴⁴ cf. Soller (2011), p. 18.

¹⁴⁵ cf. among others Seidelmann (1997).

¹⁴⁶ cf. Klaas-Wissing et al. (2010).

¹⁴⁷ cf. UIC (2010), pp. 26.

¹⁴⁸ cf. Ibid., p. 21.

¹⁴⁹ cf. Polzin (1999), p. 94.

¹⁵⁰ cf. BAV (2011b).

certain goods.¹⁵¹ For example, the unimodal road transport of wood and ceramics is prohibited and must thus, be shifted to rail.¹⁵²



Figure 2 - 6: Alpine transit by unaccompanied CT.¹⁵³

Driven by the problems in unimodal road transport, CT is increasingly integrated into retail SCs.¹⁵⁴ There are first examples of the cost, time and quality neutral integration of CT for the inland transport of consumer goods. The main haulage is accomplished by rail transport and this seems to be economical, even for distances below 100 km.¹⁵⁵

These '*innovative*' domestic and short distance CT concepts are the central interest of the thesis in hand. In comparison with 'traditional' transnational and long distance CT, the requirements for performance-oriented integration into SC concepts are particularly high. Thus, considerations and recommendations on the integration of short distance CT into SC concepts can easily be transferred to 'classical' CT services, since the competition between CT and unimodal transport over short distances is even more

¹⁵¹ cf. WKO (2011).

¹⁵² For details on the impact of European and national transport policy see section 2.2.3.2 ¹⁵³ of Sciellmann (2010), p. 46

¹⁵³ cf. Seidelmann (2010), p. 46. ¹⁵⁴ cf. Klass Wissing et al. (2010)

¹⁵⁴ cf. Klaas-Wissing et al. (2010).

¹⁵⁵ For details see ibid.

distinctive.

For the performance-oriented integration of CT into SC concepts the characteristics of CT services as well as recent SC concepts must be known. Thus, the following section addresses the service complexity of the CT concepts.

2.2.2 Service Complexity of Combined Transport Concepts

This section addresses the characteristics of CT concepts leading to a comparably high level of service complexity. This complexity exacerbates the competitive situation to the unimodal road transport. It results from the inherent characteristics, namely the variety of different CT concepts, of CT actors, processes and structures and the external factors namely shippers' requirements and transport policy.

2.2.2.1 Inherent Challenges of Combined Transport

CT is characterised by several inherent challenges. CT requires additional processes and actors in comparison with unimodal road transport. Thus, the costs of CT services and the coordination effort for the transport service are comparably high.

In a CT concept, the transported goods are at least once transhipped at a specific terminal from rail to road or vice versa. Thus, in comparison with unimodal road transport *additional handling, planning and control processes* as well as *additional transport actors* (e.g., terminal operators, rail carriers, infrastructure operators etc.) are necessary for the transport of the same good. This increases *planning, control and coordination efforts*, requires additional time amounts and increases costs.

The *integration of rail transport into SC concepts* in general bears specific challenges. In 2008, the AMMPL analysed the specific disadvantages of integrating rail transport into SC concepts. In particular, its low speed and flexibility, low service level and lacking service customisation were criticised.¹⁵⁶ Furthermore, shippers claim that transport quantities are too small to fill at least one wagon.¹⁵⁷

CT concepts must deal with several *organisational and IT challenges*. Usually CT actors pass information only *dyadically*. The information is usually restricted to own performance. The *operative, shipment-related, service offer- and demand-related* in-

¹⁵⁶ cf. Wittenbrink (2008), pp. 7.

¹⁵⁷ cf. Ibid., pp. 7.

formation is usually actor-specific.¹⁵⁸ Integrated I&C systems are still missing. Media discontinuity inhibits the SC-wide information interchange and transparency about transportation processes. Reasons are the incompatibility of the used IT systems and the number of involved actors. The cross-company availability of information is the prerequisite for common planning and control processes and thus, for the integration of material and information flow processes.¹⁵⁹ For instance, delays and interruptions in information flows may cause bottleneck situations and additional waiting times, for instance because of a wrong capacity planning at a CT terminal. These delays may affect overall SCP in terms of utilisation, lead times and costs.

The missing IT systems are reflected by *missing organisational structures*. Generally, CT concepts lack a missing coordination element, coordinating and designing the shippers' costs and performance requirements. Shippers and 3PLs as the customers of CT services usually claim to have only one counterpart.¹⁶⁰ This coordinating role and interface can be accomplished by each of the CT actors. Owing to these often dynamically changing role allocations, the marketing and quality management are complicated. Often there is *no planning and control entity* in the CT chain at all.¹⁶¹

Next to these general inherent challenges, the diversity of CT services exacerbates the dealing with CT in comparison with unimodal road transport. In particular, shippers are put off by this – at first glance – confusing variety of different concepts.

2.2.2.2 Diversity of CT Services

This section addresses the different classifications of CT services. In particular, the different types of service production are of interest.¹⁶²

There are different classification approaches for CT services. For instance, the classification into seaport hinterland and overland transport is commonly applied. Surface transport means the transport of swap bodies¹⁶³ or semi-trailers. The seaport hinterland transport is dominated by the transport of ocean containers.¹⁶⁴ For this transport cranable load carriers are required. Today, the share of these cranable load carriers is only

¹⁶⁴ cf. VÖV (2008), p. 34.

¹⁵⁸ cf. Taylor et al. (2000), p. 15; Hoffmann (2007), p. 72.

¹⁵⁹ cf. section 2.2.1.

¹⁶⁰ cf. UIC (2010), pp. 13.

¹⁶¹ cf. Hoffmann (2007), pp. 72.

For the development of specific practical recommendations on the integration of SC concepts a specific classification is necessary.

¹⁶³ There are other names, such as swap container, interchangeable container, etc.

15% (container (14%), cranable swap body and semi-trailers (1%), not cranable swap body and semi-trailers 45%, other trucks 40%).¹⁶⁵ For transhipment with a crane or reachstacker¹⁶⁶ the transport unit is grasped from above (vertical transhipment). 'Horizontal' transhipment technologies move the good horizontally, without vertically raising it. Here, either the truck or the wagon is equipped with transhipment equipment such as tracking and hydraulic systems to autonomously shift the transport.¹⁶⁷

The market segments of CT are often classified according to the type of load carriers. Thus, container and piggyback transport can be distinguished. Piggyback transport includes roll-on/roll-off, semi-trailer and swap-body transport. ¹⁶⁸ Roll-on/roll-off transport means the loading of transport units with self-propelled goods. A special type of roll-on/roll-off transport is the so-called 'rolling road' (horizontal transhipment).¹⁶⁹ Trucks with or without trailers are transported on specific low floor or so-called pocket wagons while the driver accompanies the truck in a specific wagon and can have the statutory break.¹⁷⁰ Table 2 - 7 gives an overview of the possible classification approaches of CT concepts.

transport type	surface transport			sea port hinterland transport		
type of load carrier	piggyback transport			load piggyback transport container transport		ort
	roll-on/ roll-off	semi-trailer	swap-body	land container	overseas container	
transhipment type	horizontal transhipment	vertical transhipment				
accompaniment	accompanied transport	unaccompanied transport				

 Table 2 - 7: Systematisation of CT.¹⁷¹

A multiplicity of actors accomplishes the different CT services. In comparison with unimodal road transport, more actors are involved in the service production process. The diversity of CT actors increases the complexity of service production and must be considered for the integration of CT into SC concepts.

cf. Cargobeamer (2011b). The conversion of a swap body or semitrailer costs approximately 5000 Euro and reduces the loading capacity. cf. Cargobeamer (2011a).
 VOV (2000) - 24

¹⁶⁶ cf. VÖV (2008), p. 34.

¹⁶⁷ cf. Ibid., p. 34. For instance the ACTS and the Modahlohr system.

¹⁶⁸ cf. UIRR (2011).The most common load carriers in the CT are the standardised swap body, class A (principally 13.6 m) and C (7.15 m, 7.45 m or 7.82 m), the ISO container of 20' (6.1 m), 30' (9.15 m) and 40' (12.2 m) as well as cranable semi-trailers (13.6 m).

¹⁶⁹ cf. Seidelmann (2010), pp. 29; Stölzle et al. (2008).

¹⁷⁰ cf. Seidelmann (2010), pp. 29. VÖV (2008), pp. 42.

¹⁷¹ cf. Böse et al. (2000).

2.2.2.3 Diversity of CT Actors

The production of CT services is accomplished by a multiplicity of legally and economically independent actors providing an overall marketable service.

Primary CT Actors

The primary actors of CT production are directly involved in the material and information flow processes.

Manufacturers and retailers are the shippers and principals of cargo transport. Many shippers charge 3PLs with the transport from a defined pick-up point to a destination point.¹⁷² 3PLs increasingly accomplish the entire logistics fulfilment independently and on its own responsibility.¹⁷³ The share of outsourced logistics and transport activities grows, but there are still many shippers with their own transport departments and capacities.¹⁷⁴ In many cases, manufacturers are also the brand owners. Separately or together with the retailing companies, they advertise their products. Often, this advertisement includes promises not only regarding the product quality, but also regarding availability and costs. Manufacturers act as the source of the flow of goods in the SC. Depending on the production concept, the manufacturer provides certain quantities of goods in certain time periods to be picked up by the carrier. The retailer acts as the sink for the flow of goods.

The core competencies of *3PLs* are transport (carriers) and warehousing services, but they also offer more complex value-adding services. Thus, 3PLs are often deeply integrated into the principle's value-adding processes. In contrast to *fourth party logistics providers* (4PLs), 3PLs provide their own assets, such as trucks, transhipment or warehouse capacities and knowhow. 4PLs focus on the coordination of the logistics service in one company or for an entire SC. 4PLs only provide knowhow and personnel, but do not own assets.¹⁷⁵

Carriers or freight forwarders (e.g., road carriers, rail carriers, shipping companies, air cargo carriers) accomplish mode-specific transport services on their own responsibility. The carriers are assigned by shippers, 3PLs or CT operators. Small and medium

¹⁷² cf. Gudehus (2005), pp. 970.

¹⁷³ cf. Sheffi (1990); Forslund (2009), Klaas-Wissing et al. (2010), Peters (2010).

¹⁷⁴ cf. Stölzle et al. (2011), p.72.

¹⁷⁵ For details on the 4PL concept and the meaning for the performance-oriented CT integration into SC concepts see section 3.3.

sized shippers usually do not influence the carriers transport mode choice. Thus, 3PLs are usually not bound to a specific transport carrier.¹⁷⁶ Big shippers with own transport capacities also demand CT services themselves. These shippers furthermore, transfer their sustainability strategies to the commissioned 3PL and determine the transport mode. They force 3PLs to increase the share of rail transport. Thus, shippers directly and indirectly demand CT services.

Road carriers accomplish the pre- and ongoing haulage in CT. A shipper or a 3PL assign the road carrier with the transport service in a certain time and for certain costs. In most cases, the road carrier can autonomously decide whether to use a CT service or fulfil the transport via unimodal road transport. Furthermore, road carriers are assigned by CT operators and offered door-to-door-services.¹⁷⁷

Rail carriers provide transport services via rail. They use rail infrastructure capacities, such as trackage and rail ports. Rail carriers require knowledge of rail transport specifies, such as electrical systems, track gauges and clearance diagrams as well as braking and security distances.¹⁷⁸ *Tractioners* provide the locomotives and the personnel for the train's drive. Usually, the rail carrier is the tractioner, too. Often rail carriers offer wagons for rental and corresponding services, such as maintenance and cleaning. Increasingly, rail carriers offer comprehensive logistics services to their customers.¹⁷⁹

Terminal operators offer transhipment services at the interface between carriers. Sometimes they offer additional services, such as warehousing, packaging and distribution services, repair, maintenance and wagon or load carrier rental.¹⁸⁰ Often, terminals are held by associations founded by CT actors.¹⁸¹

CT operators act as the interface between 3PLs, road and rail carriers, shippers and infrastructure operators and authorities. They purchase trains or part-trains from rail carriers and, to some extent, separate the traction service. CT operators provide complete transport services between consignment and receiving terminals. They, furthermore, offer door-to-door services for freight forwarders, 3PLs or shippers.¹⁸² CT operators take the risk for the utilisation of trains' capacities. Often, they offer further services, e.g., the provision of wagons, the physical handling and loading as well as

¹⁷⁶ cf. Bendul et al. (2009); Janz (2003), p. 15;.

¹⁷⁷ cf. VÖV (2008).

¹⁷⁸ cf. Hoffmann (2007), p. 71.

 $^{^{179}}$ cf. VÖV (2008).

¹⁸⁰ cf. Ibid.; Hoffmann (2007), pp. 24.

¹⁸¹ For instance DUSS (Deutsche Umschlagsgesellschaft Schiene-Strasse GmbH; www.duss-terminal.de).

¹⁸² cf. UIC (2010), pp. 47.

handling of the accompanying information such as delivery notes and customs documents.

Secondary CT Actors

The secondary actors of CT production are not directly involved in the material and information flow processes. They provide infrastructure, additional services and transport equipment.

Infrastructure operators and authorities provide the essential transport infrastructure (e.g., waterways, railways). Operating companies for roads plan, build and operate the road network in coordination with specific ministries. The EU claims that railway infrastructure operators must be legally independent from rail operations.¹⁸³ The aim of this separation is to grant transparent and neutral access to train paths for all interested rail carriers. Train paths can be understood as easements to the rail infrastructure, thereby enabling transport services. The infrastructure operator is responsible for the construction, maintenance and operation of the infrastructure. This includes process control and safety technique maintenance and repair as well as the development of train schedules. Furthermore, the infrastructure operator can be in charge of marketing railway lines.¹⁸⁴

In Germany, the infrastructure network is managed by the *DB Netz AG*. The *DB Station&Service AG* is reliable for the construction, maintenance and operation of stations and railports. Train path management is a central task of these organisations. Following payment, the train paths are assigned for a certain period of time. Germany has been criticised by the European Union, which claimed that market access to train paths free from discrimination is not possible, since the *DB AG* is both the rail carrier and infrastructure operator. The maintenance and operation of the railway network is supposed to be financed by the train path fees of specific trains. The stations are financed by station fees, to be paid for train stops at specific rail stations. In Germany, some tracks are not owned by the Deutsche Bah AG, e.g., between Elbe River and Weser River. Here, the owners and operators are reliable for maintenance.¹⁸⁵

cf. the passed guideline 91/440 from July 7, 1991 for the development of railway companies in the European Union (online available under: http://eur-lex.europa.eu/Notice.do?mode=dbl&lang=en&ihmlang=en&lng1=en,de&lng2=da,de,en,es,fi,fr,it,nl,pt, sv,&val= 241707:cs&page=)

¹⁸⁴ cf. Polzin (1999), pp. 96.

¹⁸⁵ cf. Weserbahn (2011).

The *Swiss Train Path Ltd* or *trasse.ch*¹⁸⁶ is the train path allocation authority for the Swiss Federal Railways (SBB), the BLS AG and the Südostbahn AG (SOB) [Swiss South-Eastern Railway]. In Switzerland, the outsourcing of the train path allocation to a legal independent decision making body is a prerequisite for reaching fair competition on rail. In cooperation with rail carriers, the infrastructure operator *trasse.ch* is also reliable for timetable development.

There are additional service providers, for instance, for *IT services, cleaning services* or renting wagons and locomotives. Special wagons, such as for the transport of hazardous goods, are often owned by the shippers themselves. For instance, the Transwaggon group owns 1800 flat wagons and 10,200 covered wagons for long-term and short-term rental.¹⁸⁷ The VTG AG owns more than 50,000 wagons with more than 1000 wagon types for rent.¹⁸⁸ Both companies also offer additional rail logistics and transport services. The availability of wagons and locomotives is central to the constant flow of goods. In 2008, before the economic crisis, the capacities for construction were used to the full. However, even after the crisis there is a huge demand for the latest wagons.¹⁸⁹ This includes the so-called *'whispering brakes'*, a special, noise-reduced brake type, and the reduction of infrastructure deterioration.¹⁹⁰ Thus, the manufacturers of wagons, locomotives and handling equipment such as cranes and reachstackers are important for a SC including CT.

An additional important actor is the *transport policy*. The national and European transport policy significantly influences the environmental factors for CT actors. Details on the impact of transport policy on CT integration into SC concepts are separately discussed in section 2.2.3.2.

Each SC actor accomplishes different processes. The following section introduces the number of partial processes and the structure of CT concepts.

¹⁸⁶ http://www.trasse.ch/.

¹⁸⁷ cf. Transwaggon (2011).

¹⁸⁸ cf. VTG (2011).

¹⁸⁹ cf. Stölzle et al. (2008); Weserems (2010). ¹⁹⁰ af DPAC (2011): Haister (2005): N.N. (2

²⁰ cf. DBAG (2011); Heister (2005); N.N. (2011b). Locomotive manufacturer must provide technical solutions for the raising mineral oil prices, for instance hybrid-locomotives and diesel-locomotives for the new transport relations without electric systems, such as the connection between China and Europe. cf. Riedl (2011), p. 46

2.2.2.4 Processes and Structure of CT Services

This section provides an overview of the material and information flow processes in CT concepts. Commonly, in comparison with unimodal road transport, CT processes seem to be much more complicated and less flexible, especially since unimodal road transport requires neither transhipment processes nor siding tracks.

CT concepts can, independent of the type of carrier, be subdivided into five main processes.

(1) Pre-haulage by road transport (collection),

(2) Transhipment to a consignment terminal,

(3) Main haulage in a rail, inland or seagoing vessel,

(4) Transhipment to a receiving terminal and

(5) Ongoing haulage by road transport (distribution).¹⁹¹

These CT processes are usually part of a cross-company SC.

Figure 2 - 7 illustrates the material and information flow processes in more detail by means of an example of a global SC.



Figure 2 - 7: Schematic depiction of CT material (black) and information flow and additional planning and control processes (blue).¹⁹²

¹⁹¹ cf. Polzin (1999), p. 90 f.

¹⁹² Own illustration.

The physical transport processes (black) are accompanied by information flows, which require several planning and control processes (blue). Information includes shipment-specific delivery documents, tracking and tracing information, temperature regulations or corresponding monitoring information.

The *manufacturer* stores, picks, packages and provides the produced goods for carriage. Depending on the specific SC concept, the goods are stored in a warehouse for a certain time period and provided here. Freight documents are issued by the shipper, carrier or CT operator. *Carriers* of all transport modes load and deload the goods and transport them to their destinations. Carriers sometimes accomplish further activities such as tracking and tracing, customs processes or freight document issuing. The *terminal operator* uses cranes or mobile handling equipment for the transhipment, loading and unloading processes. If necessary, the goods are buffered and picked according to customer orders. The *CT operator* provides transport capacities for the main haulage (train, plane and vessel), transhipment processes, train composition and trackage. The DC of the retailer stores and buffers the transported goods for onward transport to stores. The carriage to the store can be accomplished by CT, too (not depicted in the figure). In the store, the goods are buffered and shelves are replenished.

For all actors, *information* about orders and order forecasts are central for planning for instance equipment, vehicles, personnel and overheads. The retailer has a demand planning process on which it bases the order making process. These orders are processed by the manufacturer. The shipper or corresponding 3PL initiates the transport of goods.

A *multiplicity of CT actors* integrates a multiplicity of partial activities for CT service production. For only a few of these activities, there is an existing market. The partial CT services must be offered in total to the demanders (shippers, 3PLs), for instance in the form of terminal-to-terminal or door-to-door transport.¹⁹³

As in any service production, CT services are produced in two steps. Demand and service production usually take place simultaneously. This means for any CT service offers there must be a concrete demand. The shipper must provide information about

¹⁹³ Hoffmann (2009) provided an overview on different partial services according to provider and demander. cf. Hoffmann Hoffmann (2007), pp. 173.

the specific demand to initiate the disposition for the pre- and ongoing haulage and for the main haulage. Terminal capacities must be reserved. Furthermore, information on the irregularities, tracking and tracing, advice and invoicing must be provided.¹⁹⁴ The service character of CT is discussed in depth in the following section.

2.2.2.5 Service Character of CT Services

This section addresses the service character of CT services. The service character is a central aspect of the performance-oriented integration of CT into SC concepts. The integration of the shipper as the external factor complicates CT service production and thus, increases the need to align the CT concept and the SC concept.

Service production is a two-stage process. In the first stage, the so-called *pre*combination builds up the performance potential. The CT service provider, usually the CT operator, bundles internal input parameters and generates a marketable service. This means that the CT operator must hold sufficient capacities of transport vehicles, terminals, personnel, train paths, overheads and knowhow before offering the CT service on the market.¹⁹⁵

In the second stage, the end combination complements and concretises the performance potential by adding further internal and external production factors. Service production takes place when there is specific demand for the CT service.¹⁹⁶ Thus, the demand is the initiator of service production.¹⁹⁷ For instance, a 3PL places an order with a certain shipper for the terminal-to-terminal transport to the CT operator at a certain time point. The CT operator accepts the order and initiates the end combination of production factors. This means that the shipment is received at the terminal, carried to the destination terminal and, finally, handed over to the 3PL or the shipper.

The production of CT services is characterised by immateriality.¹⁹⁸ Neither the performance potential nor capability is storable or tangible. Owing to the immateriality and synchronicity of production and demand CT services are unique.¹⁹⁹ These characteristics of CT services mean that the evaluation of the characteristics of transport

¹⁹⁴ cf. Ibid., pp. 22. cf. Bendul (2009). 195

¹⁹⁶

cf. Corsten et al. (1992), pp. 180. 197

cf. Stuhlmann (2000), pp. 97. 198 cf. Corsten et al. (1992), pp. 180.

¹⁹⁹ cf. Hoffmann (2007), p. 55.

services is impossible at the forefront of service production.

Transport services are often standardised. Carriers can realise economies of scale by meeting the requirements of several similar shippers from different industries. However, these service offers also include customised components. For instance, the usage of specific equipment and implementation of individualised processes (e.g., for temperature-controlled transport) enhances the standard service. The accomplishment of the transport service becomes obvious by the higher price of the transported good.

Next to these inherent characteristics, the complexity of CT is increased further by the external factors introduced in the following section.

2.2.3 External Factors Increasing the Complexity of CT

CT is strongly influenced by factors that cannot or can only be slightly influenced by the involved actors. The development of CT services regarding processes, technologies and business models is driven by a number of external factors and stakeholders.

As shown in section 2.1.1, *shippers* have high performance requirements in terms of flexibility, speed, quality and reliability to meet the end customer's demand. The CT actors have to deal with strong fluctuations in demand and unmatched relations. These factors complicate the capacity management and cause high attendance costs.

Legislature also influences CT, especially transport policy. Here, modal shift and subsidy policies, regulation policy, traffic-specific taxes and duties, infrastructure and technology policy as well as the activities regarding international harmonisation and standards must be taken into consideration.

Finally, the section discusses the *competitive situation with unimodal road transport*, which challenges the providers of CT services. CT requires specific investment in handling equipment, time quotas for additional handling processes and an increased coordination effort. At the same time, switching barriers back to unimodal road transport are very low.

2.2.3.1 Changing Shippers' Performance Requirements Challenging CT Services

This section addresses the meaning of the shipper's performance requirements on the production of CT. It resumes the aspects presented in section 2.1 and interprets the

general challenges for SCs including CT.

Shippers require customised logistics and transport services reflecting their demanding procurement, production and distribution concepts. This includes value-added services as well as transport-specific services, such the provision of tracking and tracing information, customs clearance and insurance services.²⁰⁰ Among the quality criteria, flexibility, speed and reliability are shippers' central SCP requirements.²⁰¹ Owing to the direct competitive situation with road transport, the market-oriented price of the CT service is the central decision criterion for shippers.²⁰²

The *customisation and flexibility* of CT services requires a coordination of CT actors and an alignment of CT processes, for instance to deal with short-term changes in time windows or fluctuations in transport quantities.

In 2011, an online market screening showed that CT does not meet shippers' expectations.²⁰³ This survey analysed shippers' expectations and opinions about CT, but not the performance of CT specifically. It showed that CT actors have to strengthen trust in CT performance and improve CT image. Table 2 - 8 compares the shippers' expectations on different performance criteria and the compliance on a scale from 1 to 15 (1 for the highest meaning of the performance criterion and 15 for the lowest meaning). The table shows that reliability regarding time, predictability and price are the most important factors regarding the expectations of CT services. Distance to the terminal (rank 10), the offer of tracking and tracing services (rank 9) and short lead times (rank 8) were rated with the lowest values. The comparison with the compliance of specific performance criteria shows significant differences between the expectations of shippers and the compliance of the CT service. It turns out that especially for central decision criteria such as price (difference -2) and late booking (difference -4) the difference is comparably low. As shown above for integration into SC concepts, flexibility and cost requirements are of central importance. However, the survey showed that CT often meets or exceeds SCP requirements regarding safety (+4), hazardous goods transport (+4), distance to the terminals (+6) and short transport times (difference +2). Reliability (-7), predictability (-5) and customer service (-11) are critical aspects. Here, the expectation and compliance values are

²⁰⁰ cf. Ibid., p. 68.

²⁰¹ cf. Evers et al. (2000).p. 28; 2000; Hoffmann (2007), p. 68.

For details on the competitive situation between unimodal road and combined transport see section 2.2.3.3.

²⁰³ cf. Klotz (2011).

performance criteria	expectation (rank)	compliance (rank)
reliability (regular, punctual)	1	8
predictability	2	7
price	3	5
customer service	4	15
safety	5	1
late acceptance/booking	6	10
offer hazardous good'	7	3
short transport times	8	6
tracking and tracing	9	12
short distance to terminal	10	4

significantly different.

According to the authors, the survey proves that CT has to increase service quality as well image.²⁰⁵

Next to the impact of shippers' performance requirements, CT service production is significantly influenced by transport policy.

2.2.3.2 Impact of Transport Policy on CT

The *legislator* is an important external factor on the production of CT services. On one hand, transport policy supports CT because of regulations against unimodal road transport (especially modal shift and regulation policy). On the other hand, transport policy incites the increasing share of combined and rail transport (especially through subsidy policies). Nevertheless, transport policy positively and negatively influences CT service production in terms of costs, speed and flexibility (especially through traffic-specific taxes and duties, technology and infrastructure policy, international harmonisation and standards).

Transport policy influencing CT takes place at all political levels – local, regional, national and EU. Responsibilities are shared subsidiarily.²⁰⁶ International transactions of EU policy are highly important for international CT. The main tasks are to promote the coordination of national policies and to coordinate actions in other fields such as target setting for emissions and standardisation.

National transport policies are key players in infrastructure development. This ap-

Table 2 - 8: Comparison of shippers' performance expectations and CT service compliance.²⁰⁴

²⁰⁴ cf. Ibid.

²⁰⁵ cf. Ibid.

²⁰⁶ The highest political level acts as soon as the stated objectives cannot be reached by the lower levels.
plies to rail, road and waterway infrastructure as well as to interconnection points such as ports, terminals, logistics parks, maintenance and traffic management. Furthermore, the promotion of intermodality and innovations in transport depends on national governmental decisions.

Political instruments and measures provide the framework for the transport market, but also govern and control market developments. Economic reasons for regulatory policies are the high market share of governmental enterprises and specific shift policies to reduce external effects or balance capacity usage.

Modal Shift Policies and CT Promotion

The EU and several European governments promote the modal shift from road to rail or CT. The reasons are growing infrastructure shortages, rising transport volumes, sustainability and rising energy prices.

Transportation accounts for around 18% of total greenhouse gas emissions in Germany. About 41% can be attributed to goods traffic.²⁰⁷ Modal shift policies and promotion activities aim to reduce emissions and the environmental effects of unimodal road transport. The goal is to set up a sustainable and equitably utilised infrastructure network to avoid capacity bottlenecks for all transport modes. Until 2015, the main growth is expected for air and road transport, which produce the highest levels of emissions.²⁰⁸ Therefore, the shift to rail and waterways, which are more resource saving and cost-efficient, is in the central interest of current subsidy policies at an EU and federal level.

Today, European transport policy is nearly as expansive as are national transport policies. It contains business and social laws for traffic and transport safety as well as environmental protection. Both the EU and the individual states support rail and CT – even transnational – by means of different supporting programs.²⁰⁹ The main objective of these actions is the sustainable modal shift. The recent focus of subsidy policies has been infrastructure development and maintenance. However, investments in infrastructure in most EU countries are less than 1% of GDP (gross domestic product).²¹⁰ Furthermore, regulation policy is central for CT promotion.

²⁰⁷ cf. McKinsey & Company (2007), p. 11. ²⁰⁸ of PAEU (2010) Page (2005); PMU (20

²⁰⁸ cf. BAFU (2010), Beco (2005); BMU (2011).

For instance, the support program 'Marco Polo' for intermodal transport. cf. EU (2011).

²¹⁰ cf. BMVBS (2009).

Regulation and CT Promotion Policy

Regulation policy controls competition within transport markets. Usually, regulations are transport mode-specific. Although there are different activities promoting CT, restrictions for unimodal road transport by regulation policy are the most efficient.

Regulations that confine unimodal road transport include the enforcement of more frequent road controls and driving bans at certain times or in urban areas.²¹¹ Furthermore, the introduction of the digital tachograph and regulations of driving and rest shift competitive advantages towards CT. In particular, these regulations influence the cost and price levels and thus, shift the competitive situation with the unimodal road transport for the benefit of CT.²¹²

Recently, the organisation of railway companies and their regulation regarding the fair conduct of inter- and intramodal competitive conditions have been discussed. However, the liberalisation of the railway system in Europe is partially incomplete and it has stagnated. Lately, the European Union has criticised the liberalisation progress in certain EU member states.²¹³.

Moreover, the realisation of the interoperability of railway technology and safety systems, namely the ETCS, the European standard for railway safety systems, European driving licenses for locomotives and the implementation of environmental standards for traction vehicles and wagons are central aspects for the promotion of CT.²¹⁴

Traffic-specific Taxes, Duties and Subsidies

Traffic-specific taxes and duties affect the costs and competitive situation of CT actors. Therefore, tax harmonisation and subsidy policies within the European Union have strategic relevance. The different free trade agreements²¹⁵ and typical crossnational transport require harmonisation. Traffic-specific duties aim at an internalisa-

 214 cf. Ojeda et al. (2002).

²¹¹ For instance, the introduction of environmental zones, speed limitations and sectoral driving bans.

cf. Stölzle et al. (2010)

²¹³ cf. N.N. (2011e). Germany, Spain and France were asked by the European Union to proceed with the liberalisation of the railways. In Germany, for example, the DB AG has been made to separate railway infrastructure and operations. In particular, the fact that the revenues of the infrastructure network are not entirely reinvested into the enhancement and maintenance of the railway network has been criticized.

²¹⁵ cf. SECO (2011).

tion of resource usage, finance assurance and management of transport demands.

The latest topics are the EU emissions classification and the *internalisation of external costs* because of the change within the EU road tax directive. The internalisation of external costs aims to shift long-term costs to the causing transport actors. These include costs for accidents, infrastructure damage, noise and light emissions as well as effects on health, climate, nature and the landscape.²¹⁶ The internalisation of all costs, including external costs, corresponds to causative principles. In 2005 in Switzerland, the external costs caused by road transport reached CHF 8.1 Bio., whereas the external costs for rail transport for the same year were 0.5 Bio. The main parts were accident costs (CHF 2.0 Bio vs. 0.03 Bio), health costs (CHF 1.8 Bio vs. 0.1 Bio) and climate costs (CHF 0.9 Bio vs. 0.1 Bio).²¹⁷ This comparison of external costs shows that internalisation would have significantly increased the costs for unimodal road transport and shifted competitive advantages to CT.²¹⁸

The integrated cost and price management of CT actors has to consider the rising costs for taxes and duties. This is necessary to reach the required level of transparency and to meet shippers' performance requirements. This means that the toll systems of all the crossed countries and the prospective developments of these systems must be taken into consideration for the integration of CT. Uncertainties in transport policy,²¹⁹ such as the '*Alps transit exchange*' – expected for the time after 2020 – may negatively impact CT integration because of necessary CT-specific investment.²²⁰

Subsidy policies for rail and water transport (for instance, for siding tracks, terminals, etc.) are often not clearly communicated and presented for easy understanding. Often, CT actors are poorly informed of the possibilities to profit from subsidies.²²¹ However, a subsidy to cover investment costs can significantly influence the investment decision.

The *subsidiy policy for transalpine CT in Switzerland* during the economic crisis in 2008 and 2009 can serve as a positive example of incentivisation. While road transport providers offered services even below their differential costs, unaccompanied

- ²¹⁸ cf. Dahm (2011). ²¹⁹ of PAV (2011b) Pandy
- ²¹⁹ cf. BAV (2011b), Bendul (2011).

²¹⁶ cf. BfS (2005).

²¹⁷ cf. Ibid.

²²⁰ cf. Stölzle et al. (2008); Stölzle et al. (2009). ²²¹ of Stälzle et al. (2008); Stälzle et al. (2009).

²²¹ cf. Stölzle et al. (2008); Stölzle et al. (2009).

CT had to deal with an even greater decrease in transport volumes. To prevent the long-term return to unimodal road transport, the Swiss transport department decided to enhance subsidies per CT shipment within a time limit. This action affected the cost ratio between unimodal and CT in favour of CT and influenced shippers' decisions by affecting transport costs. These examples prove that a resilient transport policy and a planning and investment environment significantly influences the shift from unimodal road transport to CT.

However, subsidies for shipments can also shift the cost ratio between CT and unimodal road transport in favour of CT.²²²

For instance, in Switzerland, the government subsidises up to 50% of the construction costs of a CT terminal. In 2011, the Swiss government has subsidised each transalpine shipment 75 Euro. The CT operator also obtains up to 1900 Euro per transalpine train. The amount applies to shipments from and to Italy and depends on the source and destination terminals.²²³ For each transalpine shipment of the rolling road, the subsidy is 175 CHF. A train is subsidised by 3080 CHF.²²⁴ The subsidy for unaccompanied CT is paid in Euros, for the rolling road in Swiss Francs.

Infrastructure and Technology Policy

The availability of infrastructure is a central resource for all transport concepts. Recently, there have been infrastructure shortages on all carriers, where road shortages are most significant. Increasingly, practice and science claim that these shortages endanger frequent transport concepts and thus, flexible and lean SC concepts.

The government finances road, rail and waterway infrastructure by public means and toll returns. These services build the basis of infrastructure policy and result in an infrastructure network connecting consignees and 3PLs. All further infrastructure capacities such as siding tracks, handling areas and connections on company areas need to be provided by users.²²⁵

Infrastructure development and maintenance implies the construction of roads and railways considering *priorities with respect to capacity bottlenecks* as well as an in-

For details see for instance Bendul (2011).

cf. BAV (2011a). There are different tariffs from Italy to either the Netherlands (a), France (b), Great Britain, Belgium or Luxembourg (c), Denmark, North Germany, Scandinavia (d), Rhine-, Ruhr-, Main-area (Germany) (e), South-West-Germany or Switzerland with an amount of up to 1900 Euro. For instance, for a CT from Antwerp, the Netherlands to Novara, Italy with 28 shipments (75 €*28) + 1600 € = 3700 € is paid.

²²⁴ cf. Ibid.

²²⁵ This is an important overlapping of infrastructure and subsidy policies.

crease in reinvestments to maintain these. Furthermore, the optimisation and coordination of construction sites is part of the infrastructure policy. In addition, *technology policy* supports the feasibility of automotive engineering in the context of vehicle heights, weights and lengths, which are often interconnected with infrastructure capacities. For instance, there is an intensive discussion on the introduction of so-called 'gigaliners'. Moreover, telematics systems give opportunities to control transport flows by the provision of traffic information and suitable actions for network users.

CT actors benefit from the European infrastructure extension project TEN-T. The socalled TEN-T (Trans-European Transport Network) spans all transport carriers and all regions of the European Union.²²⁶ By adding missing traffic connections and by remedying bottlenecks, the traffic and mobility of people and freight can be upgraded. In addition, the TEN-T includes technological projects such as the European satellite navigation system 'Galileo', the so-called maritime motorway²²⁷ and the European railway traffic management system.

In particular, the availability and allocation of the capacities of the commonly used resources (e.g., terminals, carriers) cause problems for CT.²²⁸ Unbalanced capacities, particularly terminal capacities, are often caused by insufficient subsidy policies. Shortages in terminal capacities especially affect transport time and thereby overall SCP. The distances for pre- and ongoing haulage to the central major terminals are very long. Experts here see in the dispatch of terminal capacities a major need for action.²²⁹ From a shipper's point of view, the rigid time windows and time schedules especially cause problems: exceedingly late or early delivery and pick up times can only seldom be realised. Hence, this means significant quality and flexibility disadvantages for the image of CT.

Figure 2 - 8 predicts the shortages in CT terminals expected until 2015 in Europe. According to Seidelmann (2010), an additional transhipment capacity of 3.4 Mio transport units per year is necessary.²³⁰

²²⁶ cf. EK (2005).

²²⁷ Also called the 'motorway of the sea' or 'traffic separation zone'. ²²⁸ Also called the 'motorway of the sea' or 'traffic separation zone'.

²²⁸ cf. Clausen et al. (2006), p. 57.

²²⁹ cf. Cordes (2011).

²³⁰ cf. UIC (2010), p. 74.



Figure 2 - 8: Overview of recent and expected CT terminal capacity shortages in Europe.²³¹

Unbalanced capacities result from the distortion of *unaligned subsidy policies*. There are several examples of economic projects that cannot survive without financial support. At a European level, this missing alignment leads to less efficient isolated solutions.²³²

Technology policy supports corporate activities in the field of CT by supporting technology development and innovation subsidies. In particular, in the field of innovation and new technologies, a smooth transition of corporate activities, processes and actors must be assured by the introduction of technological standards. The meaning of harmonisation and standardisation processes is discussed in the following section.

²³¹ cf. Ibid., p. 75.

²³² cf. Frindik (2004), Stölzle et al. (2008). For instance, in France the innovative technological solution of the so-called 'Modalohr' system was introduced. This system allows lateral truck loading by the panning of wagon bottoms. The time-consuming frontal loading of the train can be avoided. Furthermore, the traction unit and the trailer can be independently transported. Thus, the pig-gyback transport and rolling road transport concepts can be combined. Owing to this innovative loading, there are specific requirements on the terminal layout and technique. Thus, these terminals require increased investments in comparison with the classical technique, but increase flexibility and operating costs. For instance, the transport units can be (de)loaded separately and at different terminals. The combination of the rolling road systems of France and the surrounding countries (e.g., Germany, Switzerland and Austria) would be sensible under network aspects. However, a combination with the classical terminal is not possible and cannot be accomplished without the specific investment in terminals and rolling stock.

International Harmonisation and Standards

International harmonisation and standards are central aspects for the production of cross-border CT services. In practice, international policies are only applicable in air and sea transport, whereas all other directives and implementation guidelines are substituted by EU policies.

For instance, there are standardisation activities in Europe regarding environmental standards for traction vehicles and wagons, general tax harmonisation within the EU and the European-wide harmonisation of *track prices*.²³³

Technical system differences challenge the management of SCs including CT. In particular, border-crossing transport faces problems caused by different types of loading units, wagons and locomotives as well as handling equipment. Furthermore, different transhipment technologies and the reducing number of railway side tracks restrict the net-ability of CT.²³⁴

Recently, European transport policy has focused on the standardisation of electrical systems, railway language, track gauges and clearance diagrams as well as on the dimensions of trains and trucks, load carriers and pallets. Recently, different train lengths have caused significant capacity losses for international rail and CT transport.²³⁵ Thus, there are activities to unify the train length to 750 m. In Germany, the maximum freight train length is 700 m.²³⁶ In Switzerland, the train length is principally 750 m²³⁷ whereas in Italy it is only 550 m.²³⁸ Thus, transnational CT and rail transport services between Germany, Switzerland and Austria cannot use capacities to the full.²³⁹ Thus, there have been recent efforts to standardise the train lengths on European freight transport corridors from North to South to 750 m.²⁴⁰ This additional time and costs cause significant competitive disadvantages for CT and rail transport in comparison with unimodal road transport.

Scientific studies and practical tests have been carried out to check the applicability of trains 1000 m long to increase the relation capacities. In 2009, the first tests were

²³³ cf. Klotz (2011).

²³⁴ cf. Böse et al. (2000), p. 279; Hoffmann (2007), p. 72. 235

cf. Siegmann (2011a); Siegmann (2011b). 236

cf. Fiedler (2005), p. 22. 237

relation specific its only 600 m, 238 cf. N.N. (2004b), p. 9.

²³⁹

cf. Wagner et al. (1999), p. 39; Consultants et al. (2004), pp. 46. 240

There are comparable capacity losses in East-West relations. In France, the maximum train length is 750 m. In Spain, it is only 450 m. The shorter siding tracks also cause additional shunting effort for the separation and joining of trains. Train rebuilding as well as regauging at the French-Spanish border (from normal to broad gauge) takes two to three hours. cf. Siegmann (2011a).

accomplished on the '*Betuweroute*' between Rotterdam, the Netherlands, and Oberhausen, Germany.²⁴¹

As shown, transport policy significantly influences the time-, space-, flexibility-, reliability and cost-oriented performance of CT services and thus, influences the competitive situation between CT and unimodal road transport. This competitive situation is clarified in the following section.

2.2.3.3 Competitive Situation of CT and Unimodal Road Transport

There is fierce competition between CT and unimodal road transport. The different levels of efficiency resulting from the inherent advantages and disadvantages of the carriers are central to the competitiveness of CT. Despite recent sustainability, energy shortages and green logistics considerations, costs remain the central selection criterion for transport carrier choice. Furthermore, the goods' and shipments' structures (type, size, quantity, frequency, lead time) as well as the available capacities of transport vehicles and load carriers are relevant decision criteria.²⁴²

The additional handling processes and coordination and planning activities cause cost disadvantages for CT in comparison with unimodal road transport. In particular, transhipment costs are step-fixed. Figure 2 - 9 gives a schematic overview of the cost situation of different intermodal transport concepts in comparison with unimodal road and rail transport. Figure (a) shows the steeper slope of the road transport curve in comparison with rail transport curve, whereas the fixed cost level of road transport is below the fixed cost level of rail transport. Thus, theoretically there is a certain transport distance where rail transport costs are below road transport costs. Figures (b) to (d) show further cost curves for multi-modal transport concepts. Further step-fixed costs are added for each transhipment process.

²⁴¹ cf. DBAG (2009).

²⁴² cf. Klotz (2011); Stölzle et al. (2008).



Figure 2 - 9: Schematic costs for unimodal, combined and multimodal road transport concepts.²⁴³

In particular, rail carriers, tractioners and terminal operators have to invest in costintensive vehicles and handling equipment. Thus, the standby costs of a CT service are significantly higher than those for unimodal road transport.

The resulting problems from these different cost structures became obvious during the economic crisis in 2009. The positive economic development until 2008 and increasing global transport volumes led to a massive enhancement of transport capacities for all transport modes. This development was reflected by a strong demand for vessels, rolling stock and trucks. Owing to long delivery times, these investments take effect only over time and capacities increase only gradually. Thus, in the economic crisis of 2009 a lowered demand for transport services met an increase in transport capacity. The result was a price collapse, which intensified competition in transport markets.

This situation has highlighted the price sensitivity of CT services. In summer 2009, road carriers offered their services at a price level not covering their marginal costs to sustain liquidity. Owing to the lower fixed costs and an often thin capital base, road

²⁴³ cf. Bonomyoung et al. (2001), p. 666.

carriers operated with massive allowances in the market. Since rail and CT actors could not offer discount prices, they experienced a significantly stronger decrease in transport volumes than did road carriers.²⁴⁴

Recently, most domestic transport and last-mile transport in SC concepts are accomplished by unimodal road transport. As shown in section 2.2.2.1, shippers have the prejudice that CT cannot meet SCP requirements and thus, prefer unimodal road transport. In addition to the cost and time aspects, the flexibility and the constant accompaniment of each shipment are important. A truck driver can change the transport route at any time and can always personally assure the goods' quality.²⁴⁵

2.2.4 Integrated Management of CT Concepts

The integrated management of CT concepts has specific challenges. These refer to the described internal and external framework conditions. To offer CT services to the market, integrated management is necessary. Resch (2008) identified three different types of CT concepts following the four dimensions of the business model proposed by Müller-Stewens and Lechner (2005).²⁴⁶

The authors distinguished between (1) the performance, (2) the value-adding, (3) the marketing and (4) the yield models. The model excels with its cross-company character and the possibility of consideration of different corporate divisions. Following the argumentation line of Müller-Stewens and Lechner (2005) the target of each business model is the enhancement of a shipper's benefits and the protection of competitive advantage.²⁴⁷ The model encompasses different perspectives and mutual relations between different perspectives. Resch (2008) applied the business model to the field of CT business models.²⁴⁸

To set up a CT service a central actor, such as the CT operator, arranges a group of actors, such as CT operators, rail carriers, road carriers and terminal operators, and ensures the availability of the necessary infrastructure resources. Either the group is individually arranged according to the performance requirements of a certain shipper or the group is arranged to offer a rather standardised CT service for different ship-

²⁴⁴ cf. Pfohl Stölzle et al. (2010).

²⁴⁵ cf. Stölzle et al. (2008).

²⁴⁶ cf. Müller-Stewens et al. (2005).

²⁴⁷ cf. Ibid., pp. 55. ²⁴⁸ af Pasch (2000) p. 2

²⁴⁸ cf. Resch (2009), p. 31; Ihde (2001).

pers.

In both cases, the group of actors requires the same *performance understanding*. Every actor needs different capabilities to handle the *determined shipment size* (full load, part load, mixed cargo) of a certain *physical condition* (universal or special cargo, such as liquid goods). The actors must agree on a *spatial structure of the transport service* (e.g., a standard service on regular relations or flexible services on different relations).²⁴⁹

Based on this common performance understanding a *mutual value-added model* is built. All CT actors agree on th*e institutional structure* and configuration of the service production (vertically integrated, specialised and network configurations). The positions and tasks of all the CT actors for the service production are determined and documented. Furthermore, the CT actors determine *communication and coordination mechanisms* (coordination, hierarchical instructions, standardisation and market-oriented prices).²⁵⁰

The marketing model classifies the CT business model according to the configuration of the *integrated marketing* process. This means that different *modalities of business relations and thus, the stability of the relation* between shippers and providers of the CT concept are determined (long-term / short-term). CT actors have to agree on the *conservation of existing and setting up of new transport relations* aiming at the shipper's long-term obligation to the CT concept.²⁵¹

The yield model determines how CT actors equitably share costs and yields. For the offer of standardised services, actors agree on a *target customer group* (e.g., a specific company size, industry and region). An *appropriate cost and price management* (simple/complex, static/dynamic) is necessary to share costs as well as yields equitably. However, integrated approaches are not widely implemented in practice yet.²⁵²

Table 2 - 9 gives an overview of the four sub-models, the dimensions and the classification criteria for CT business models.

²⁴⁹ cf. Resch (2009), pp. 35. ²⁵⁰ of Ibid. pp. 37

²⁵⁰ cf. Ibid., pp. 37.

²⁵¹ cf. Ibid., pp. 33.

²⁵² cf. Ibid., pp. 40.

criteria	dimensions						
performance model	shipment size	mass good	full truck load		less than truc load	k	packaged goods
	physical condi- tion	specialised			standard		
	spatial structure	rules relation	no rul	les - relation	rules networl	¢	no rules net- work
marketing	customer group	small shippers			huge shippers		
model	stability of CT configuration	long-term			short-term		
value-adding model	coordination mechanism	coordination	direct instruction		standardisati	on	market-oriented prices
	configuration	vertically integrated specialised			netwo	ork	
yield model	complexity	simple			differentiated		
	time horizon	static			dynamic		

 Fable 2 - 9: Classification criteria for CT business models.

In practice, the business model differs regarding the configuration of the CT transport chain. Thus, for the identification of clear types the differentiation into vertically integrated, specialised and network configuration is the basis. Resch (2009) identified three generic CT business models: the (1) relation coordinator, (2) the relation integrator and (3) the network integrator.

The Relation Coordinator

The business model of the *'relation coordinator'* is characterised by the number of legal and economic independent actors, focusing on the production of specific partial services. All partial services are produced autonomously and are self-reliable.²⁵⁴ Commonly, a 3PL focusing on shippers' SCP requirements mediates between the CT actors and the shipper.²⁵⁵

By connecting the partial service, the 3PL can offer a comprehensive product to the shipper for one relation. Usually 3PLs operate for several shippers. Thus, this business model is focused on full truckloads of specific goods and industries to reach a high capacity utilisation.²⁵⁶

Owing to the high degree of independency, the actors use market-oriented prices for coordination.²⁵⁷ Often, this goes along with short-term business relationships.

²⁵³ Own illustration in accordance with Resch (2009), p. 43.

²⁵⁴ cf. Hoffmann (2007), p. 22.

²⁵⁵ cf. Trost (1999)1999, pp. 82; Pfohl (2003), pp. 7

²⁵⁶ cf. Brandenburg et al. (2008), p. 468.

²⁵⁷ cf. Hoffmann (2007), p. 47.

The Relation Integrator

The business model of the 'relation integrator' is characterised by a vertically integrated CT chain. This integration results from one central actor, who initiates service production by its own transaction and/or by taking a long-term risk by agreeing to a high share of service production. An actor, offering such a 'one-shop solution' is defined as an 'integrator'.²⁵⁸ In this function, the integrator has intense contact with the shipper.²⁵⁹ Few processes, for instance the traction, are accomplished by additional independent actors. However, the integrated actor has full responsibility for capacity utilisation as well as the service level.²⁶⁰ It is even possible that an actor accomplishes all partial services on its own.

For instance, in 2002 DB AG repurchased Schenker AG as a 3PL and bought Hangartner AG as a Swiss specialist of CT services between Skandinavia and Italy owning several terminals. DB AG was then able to offer a completely integrated CT service to shipping companies.²⁶¹ Nevertheless, all processes are accomplished by one company, and the service production is characterised by the division of labour. The challenges or process integration are thus still demanding.²⁶²

The high level of integration in this business model allows the usage of hierarchical coordination mechanisms because usually there is an asymmetric power distribution. Thus, service production is characterised by direct instructions and rules.²⁶³ In the case of economic and legal independent actors, steering by market mechanisms reflects the high bargaining power of the vertically integrated actor.²⁶⁴

The relation integrator focuses on big shippers and long-term business relationships to ensure high capacity utilisation.²⁶⁵ This is necessary because of the concentration of the availability costs and the high fixed cost share of only one actor.²⁶⁶ The relation integrator focuses on high, regular transport volumes²⁶⁷ and universal goods with

- 264 cf. Resch (2009), p. 249
- 265 cf. Benkenstein et al. (2007), pp. 51.
- 266 cf. Holderied (2005), p. 61, Gudehus (2005), 834.

²⁵⁸ cf. Knorr (2005), p. 77; Brandenburg et al. (2008), p. 468. 259

cf. Knorr (2005), p. 77; Brandenburg et al. (2008), p. 468. 260

cf. Brandenburg et al. (2008), p. 468. 261

cf. de Miroschedji (2002), p. 54. 262

cf. Resch (2009), p. 48. cf. de Miroschedji (2002), pp. 51 263

²⁶⁷ Thus, relation integrators focus on FTL.

high quality forecasts for spatial and temporal demand.²⁶⁸ These characteristics allow the setup of regular line transport for single shippers on specific relations.

One main advantage of the 'relation integrator' business model is the possibility to exclude competitors from the necessary infrastructure by the internalisation of activities. This operation of private terminals inhibits the free network access to the railway infrastructure for third parties.²⁶⁹

The Network Integrator

The 'network integrator' business model stands for an organisational network of voluntarily cooperating actors. The network structure covers a geographic- and industry-specific market.

The setup of network structures has several advantages for the involved actors. First, the market power of small and medium-sized companies is increased even if they have no huge or regular transport volumes. Second, cooperation allows the bundling of transport volumes and thus, the realisation of a higher capacity utilisations.²⁷⁰ Third, it allows a significant cost reduction in comparison with entirely autonomous CT service production.²⁷¹

However, the network structure leads to strong dependencies among actors and requires mutual trust and intense communication and coordination mechanisms. Therefore, supporting I&C systems are required. Direct coordination and the steady exchange of views and ideas allow the harmonisation and alignment of partial activities.

The realisation of bundling effects allows the usage of block trains and the carriage of packaged goods. Therefore, specific transhipment points are required.²⁷² Service production in the network is comparably complex and thus, requires demanding planning and control systems. To reduce this complexity, processes are slightly standard-ised by the introduction of long-term framework contracts with shippers and a focus on standardised goods.

Recently, for instance, *DB AB* together with *Hellmann* and *Schenker* aimed to realise such a network under the name '*X*-*Pressnet*'.

²⁶⁸ cf. Brandenburg et al. (2008), p. 295.

²⁶⁹ cf. Knorr (2005), p. 468. ²⁷⁰ af Pachmann (2008), p.

²⁷⁰ cf. Bachmann (2008), p. 20.

²⁷¹ cf. Bahrami (2001), pp. 52.

²⁷² cf. Brandenburg et al. (2008), p. 248; Pfohl et al. (2003), p. 9; Stölzle et al. (2011), p. 77.

In 2007, *X-Pressnet* was founded to develop a sophisticated network service for the transport of time-critical goods by rail. Starting from 2012, fast and punctual services (98% punctuality) working with delivery windows will be offered.²⁷³ However, the expected high costs will restrict the offer to time-critical and valuable products. Project partners are *Kombiverkehr GmbH & Co. KG* (CT operator), *DB Netz AG* (infrastructure operator), *DHL freight GmbH* (subsidiary of *DHL*; carrier offering unimodal road, rail and CT services for FTL, LTL and packaged good), *DUSS GmbH* (*Deutsche Umschlaggesellschaft Schiene-Strasse*; terminal operator), *Hellmann Worldwide Logistics* (road carrier, in 2010 bought by *DB AG*) and *DB Schenker Rail AG*.²⁷⁴

The introduced CT business models are generic. There is a multiplicity of further possible configurations and often CT concepts cannot be clearly allocated to one business model. In fact, most companies participate in more than one CT concept and thus, run more than one business model. However, for the thesis in hand a different classification focusing on the classification of material and information flows and the integration into different SC concepts is required. For this purpose, the performance profile of a CT concept is central, rather than the underlying business model (cf. section 3.4). Therefore, the following section develops a basic understanding of SCP and the associated key figures and performance indicators.

2.3 SCP Orientation – Consideration of Strategic and Operational Key Figures

SCP orientation means the consideration of strategic and operational key figures at a SC level. The target system of SCP allows SC actors to focus on both the impact of their own activities and a clear end customer orientation.²⁷⁵ SCP can evaluate the success of the information and material flow integration and the impact of the CT integration using financial key figures. Thus, SCP is chosen as the target system to evaluate the given problem of CT integration into SC concepts. According to Fabbe-Costes & Jahre (2008), SC integration can be interpreted as a basis for SCP manage-

²⁷³ cf. Vekehr (2011).

Planned rail relations are Hamburg-Munich, Hamburg-Nuremberg, Hamburg/Bremen/Hannover – Frankfurt/Basel, Hamburg/Bremen/Hannover – Nuremberg, Osnabrück/Hannover – Regensburg/Landshut. cf. Ibid.
 ²⁷⁵ of Korror (2006) p. 125

²⁷⁵ cf. Karrer (2006), p. 125.

ment.²⁷⁶

In this section, the concept of SCP and the basics of a SCP management are briefly introduced to understand the meaning of a SCP-oriented integration of CT into SC concepts. The section's aim is to understand how SCP can be used as the SC-wide target system considering all the requirements and challenges of all SC and CT actors.

2.3.1 Fundamentals of Performance Orientation

SCP measurement focuses on the development and maintenance of measurement systems as well as the measurement and control of the SC. SCP management focuses on strategic and operative control functions in an integrated management approach.²⁷⁷ In this integrative function, SCP management acts as a bridge between corporate leadership and the performance system.²⁷⁸ Nevertheless, SCP management does not include comprehensive leadership, but rather it is limited to the specific application field that is being modelled by the included key figures and performance indicators.²⁷⁹

Performance measurement was introduced in the 1990s to overcome the problems of traditional key figure systems.²⁸⁰ Usually, these key figure systems are designed with regard to accounting needs, but cannot evaluate the operations of SCM or be used for strategic management purposes.²⁸¹

The current literature still lacks research in different fields of performance measurement.²⁸² For instance, there are open fields in terms of the conceptual development of performance measurement, new adaptation fields and the generalisation of the implementation of performance measurement. The performance-oriented CT integration

²⁷⁶ cf. Fabbe-Costes et al. (2008), pp. 131.

²⁷⁷ cf. Stölzle et al. (2004b).

cf. Karrer (2006), p. 4; Keebler (2001), p. 428. Based on these practical needs a completely new research field emerged in the early 1990s.
 ²⁷⁹ a Complex et al. (1004). Compare et al. (1007), p. 10. Part processes of management, such as company, policy and management.

cf. Caplice et al. (1994), Cooper et al. (1997), p. 10. Part-processes of management, such as company policy and management development are thus, not influenced by the performance management.

²⁸⁰ Traditionally, systems of key figures have been oriented to companies' accounting systems. These systems of key figures are not capable of meeting the entire information demand of diversified and market-oriented organisations. In particular, the aggregation level, the short-term consideration period, delayed information availability, the lack of planning and strategy orientation and the disregard of external influencing parameters and stakeholders of these accounting oriented key figures have been criticised. See for instance Kaplan et al. (1992), p. 71; Hoffmann (2000), pp. 22; Hammer (2003), pp. 519; Müller-Stewens et al. (2005) ,pp. 697.

²⁸¹ cf. Gleich (2001), p. 11; Kaplan et al. (1992); Karrer (2006), p. 118; Lebas et al. (2002); Neely et al. (1995).

²⁸² cf. Karrer (2006), p. 120.

into SC concepts can be understood as a new application field. The thesis in hand examines the new application field and thus, a generalisation of the given implementation rules is developed.

Several authors have discussed the definition of 'performance'. The number of different performance definitions shows the need for a common performance understanding to reach a common goal of CT integration into SC concepts.²⁸³

The term *'performance'* implies that it is a ratio key figure that compares different input parameters.²⁸⁴ Stölzle & Karrer (2004b) comprehensively discuss the performance term joining operational and strategic aspects.²⁸⁵ To achieve this match the performance understanding is specified by the concepts of efficiency and effective-ness.²⁸⁶

Performance measurement includes both monetary and non-monetary key figures. Several empirical surveys have demonstrated the need for non-monetary information, such as customer satisfaction, productivity or innovation. Thus, performance measurement includes non-monetary key figures. For instance, the introduction of Total Quality Management led to an increased focus on internal process key figures, such as error rates, setup and lead times and product quality.²⁸⁷ Increasing competition and end customer orientation have also raised the focus on customer satisfaction, time to market or number of new products.²⁸⁸

Performance is defined by its two components of efficiency and effectiveness. The *efficiency and effectiveness orientation* implies that performance is no absolute performance category, but a ratio that compares different input parameters.²⁸⁹ Efficiency and effectiveness orientation aims at the integrated measurement of resource usage, of the value-adding process and of the company's achievement of objectives.²⁹⁰

For instance, production management follows the physical definition and interprets performance as the output as accomplish work unit per time unit. (cf. Küppler et al. (2004)); tightly connected with this understanding is the term of 'efficiency' - including a technical and economic component. Technical efficiency refers to the term of 'productivity'. It means the relation of real output and real input. cf. Corsten (2004), p. 43; Karrer (2006), p. 122. Based on this definition, economic efficiency means the evaluation of input and output by the ascription of costs and proceeds respectively with expense and revenues. cf. Ibid., p. 122; Gleich (2001), pp. 38. There are several more approaches for the measurement of efficiency. See for instance Pfohl (2010), p. 41. introducing the concept of social and ecological efficiency. The controlling unit usually interprets performance as a determinant for the costs or quantity components of proceeds.

²⁸⁴ cf. Lebas et al. (2002), pp. 34, 68, 82, Mentzer et al. (1991), p. 34; Neely et al. (1995), p. 240, 82; Reiner (2004), p. 227; Stölzle et al. (2004b), p. 227, 240; Seidl (2002), p. 29.

²⁸⁵ cf. Stölzle et al. (2004b) ,p. 68

²⁸⁶ cf. Karrer (2006), p. 125; Neely et al. (1995), pp. 80.

²⁸⁷ cf. Karrer (2006); Barker (1995); p. 31.

²⁸⁸ cf. Karrer (2006), p. 131.

²⁸⁹ See for efficiency and effectiveness aspects of performance, e. g, Mentzer et al. (1991), p. 34; Neely et al. (1995); Lebas et al. (2002), p. 68; Stölzle et al. (2004b), p. 240; Seidl (2002), pp. 29.

²⁹⁰ cf. Caplice et al. (1994), p. 19; Karrer (2006), pp. 126.

Efficiency is concretised by two ratios: the degree of utilisation (resource input) and the degree of productivity (value-adding process).²⁹¹ Effectiveness orientation enhances the efficiency understanding by the aspect of target relativity in terms of the comparison of actual and planned results.²⁹² Effectiveness is defined as the ratio of effective input and planned output (target achievement). Thus, the outputs that are suitable for the main target achievement must be identified and the degree of achievement must be measured and adapted if necessary.²⁹³ Figure 2 - 10 visualises these considerations.



Figure 2 - 10: Efficiency and effectiveness dimensions of performance.²⁹⁴

Several authors have analysed different measurement types and quality criteria for key figures of performance operationalisation. Of the traditional quality criteria of empirical measuring instruments, such as validity, reliability and objectivity, the key figures and performance indicators included in the SCP target system must do justice to the following criteria:²⁹⁵

- Integration potential: The key figures support the cross-interface coordination and can be integrated into existing planning and control systems.
- Early indicator and forecast ability: The key figures support the recognition of changes. This allows early reactions and improves the calculation of trends.
- Aggregation ability: Key figures require an appropriate level of detail. A logi-• cal structure of the key figure system has to ensure both the aggregation and categorisation of key figures.

²⁹¹

cf. Ahn (2003), pp. 90; Gladen (2003), p. 30; Karrer (2006), p. 125. cf. Stölzle et al. (2004b), p. 240; Caplice et al. (1994), p. 19; Karrer (2006), pp. 126. 292

²⁹³ cf. Otto et al. (2002), p. 279; Gladen (2003), p. 132; Ahn (2003), p. 92, Karrer (2006), p. 126.

²⁹⁴ cf. Caplice et al. (1994), p. 19, Karrer (2006), p. 127.

²⁹⁵ See for comparable lists Caplice et al. (1994), p. 14; Delfmann et al. (2003), p. 44, Erdmann (2003), pp. 160.

Incentive compatibility: The key figures support compliant behaviour. This ٠ means that they avoid unsolicited management effects or dysfunctional behaviour.²⁹⁶ Thus, each key figure must be clearly allocated to one scope of responsibility and be linked with specific incentive systems. This aspect includes that the key figure is suitable to reflect the company's value-oriented objectives.

76

- Profitability: It must be ensured that the benefits of a key figure's measurement • exceed the costs of enquiry, analysis and evaluation.²⁹⁷
- Application orientation: The results of performance measurement must support • the decision makers for the evaluation and selection of action alternatives. Whenever possible, the task of decision support should be based on the latest IT.

To transfer the performance term to the SC level several adaptations are necessary. The following section introduces the main ideas of performance understanding in a SC context.

2.3.2 **Performance Understanding in a SC Context**

Several authors have enhanced the general performance understanding and transferred it to the SC level. This section discusses the necessary adaptations and enhancements for transferring the performance definition from the corporate to the SC levels.

For instance, Karrer (2003a) suggested the enhancement of the performance understanding by the characteristics of *multidimensionality* as well as *future and potential* orientation.²⁹⁸

The *multidimensionality of SCP* is often put on one level with the consideration of non-monetary key figures in management control.²⁹⁹ This refers to the configuration of the balanced scorecard, differentiating between 'finance', 'customer', 'internal processes' and 'learning and development' as categories for the central key figures. Multidimensionality stands for the different categories of objectives influencing the term of performance. This corresponds to the target system of SCM. The SCP understand-

²⁹⁶ cf. Gladen (2003), p. 11.

²⁹⁷ cf. Karrer (2006), p. 132. 298

cf. Ibid., p. 132; Power (2005), pp. 196. 299

cf. Hoffmann (2000), p. 17; Schrank (2002), p. 69.

ing considers both formal targets and content targets to be relevant.³⁰⁰ The content targets must be ascribed by a certain value to do justice to the financial, formal targets. This provides a special challenge since the content targets do not only refer to the result of the value-adding processes, but also the development of competencies and (success) potential. These *'intangible assets'* often only indirectly affect financial performance indicators.³⁰¹

The *future and potential orientation* means that SCP is no punctual event. It describes to what extent a company is able to reach future objectives. For this purpose, a process-oriented performance understanding is suitable. It focuses on the current activities leading to future results for prospective performance. The focus of performance measurement is thereby shifted from period- and past-oriented performance indicators to early indicators and the identification of influencing variables.³⁰² The main challenges are the uncertainty about future action alternatives as well as the divergence of cause (recent activity) and effect (potential, prospective activity). To be more precise, in the field of SCM, future and potential orientation means the growing importance of the time and flexibility dimensions in the SCM target system.³⁰³

Impacted by the balanced scorecard,³⁰⁴ Karrer (2003a) also highlighted five necessary adaptations to the general performance understanding to transfer it to the SC context:

- (1) Balance of performance measurement,
- (2) Ability to reflect cause-and-effect relationships,
- (3) SC strategy connection,
- (4) User-oriented preparation and visualisation of metrics and
- (5) Process orientation.³⁰⁵

(1) The *balance of performance measurement* means the claim of SCP understanding for a balance between internal and external respectively integrated and cross-company integrated key figures. This means that it can be necessary to simplify the corporate performance measurement. Usually, the high level of detail is not necessary

³⁰⁰ cf. Gladen (2003), p. 31; Karrer (2006), p. 127.

³⁰¹ cf. Kaplan et al. (2004), p. 52; Müller-Stewens et al. (2005), p. 703.

³⁰² cf. Stölzle et al. (2004a), p. 174; Lebas et al. (2002), p. 72. ³⁰³ This show publications on process priorited time managem

This show publications on process-oriented time management and time base performance management, e.g. Karrer (2006), p. 129. $\frac{304}{100}$ of Stälede et al. (2002), p. 75. Stälede et al. (2001), p. 78.

³⁰⁴ cf. Stölzle et al. (2002), pp. 75; Stölzle et al. (2001), p. 78.

³⁰⁵ cf. Karrer (2006), p. 127.

for the performance measurement of cross-company relationships.

(2) The *ability to reflect cause-and-effect relationships* means that the causal connections between the activities of single SC actors and the potential consequences for the entire SC must be reflected by SCP.³⁰⁶

(3) *SC strategy connection* means that performance orientation in a SC context must be aligned according to the SC strategy. This is a specific challenge, since often there is no common SC strategy.³⁰⁷ Usually, SC actors commonly choose SCM targets and thus, implicitly formulate and adapt SC strategy.³⁰⁸

(4) The *user-oriented preparation and visualisation of metrics must consider* the increasing number of recipients and increasing complexity of measurement subjects. The visualisation of the SC structure and relevant material and information flow processes is the basis to achieve a common understanding of the SC in general as well as SCP.³⁰⁹

(5) *Process orientation* is central for SCP measurement.³¹⁰ The transfer of logistics process thinking to performance measurement implies that the objectives, metrics and measures for SC control must consider not only the corporate and SC strategies, but also the SC processes. For this purpose, a bottleneck-oriented process analysis of the SC is recommended, which can be supported by specific methodologies and instruments.³¹¹ This leads to a countercurrent process and strategy orientation of performance measurement.

SCP Management

SCP management enhances the measurement aspect and includes the measurement and control of the integrated decision making following the development and maintenance of integrated measurement systems.

The objectives of SCP management are defined with a focus on the SCM concept. Aiming for the formal target of the increase in SC value, the content targets of the increase in end customer benefit, decrease in costs, realisation of time advantages and increase in quality are central.

³⁰⁶ cf. Keebler (2001), p. 430

³⁰⁷ cf. Stölzle et al. (2001), p. 144; Bacher (2004), p. 144 and the here referred authors.

³⁰⁸ In section 3.1 the development of SC strategies is comprehensively discussed.

³⁰⁹ cf. Bacher (2004), p. 175.

³¹⁰ cf. Karrer (2006), p. 143; Reiner (2004), p. 218. For the conception of process-oriented performance measurement in the telecommunication industry see Binder (2003), pp. 46.

³¹¹ cf. Karrer (2006), p. 143.

As shown in section 2.1.1, the target system of SCM is characterised by several interdependencies. Through the application of the SCP dimensions, the impact on the SCP management concept can be visualised. The *efficiency dimension* of performance is influenced by process efficiency and resource efficiency. This affects the cost situation of each SC actor. The effectiveness dimension focuses on the realisation of time advantages and increase in quality in the SC. These affect the increase of end customer benefit and the SC's market success. The *efficacy dimension* of SCP is shaped by the target of the increase in corporate value. Figure 2 - 11 visualises the relationships between the target dimensions.



Figure 2 - 11: Characterisation and ideal cause-and-effect relationships between the targets of SCP management.³¹²

The central challenge of SCP management is the creation of an awareness of the validity and significance of these targets as well as connecting the cause-and-effect relationships.³¹³ In particular, the connections between cross-company and a company's internal targets are central. The content targets directly influence formal targets. These have to be coordinated among SC actors and analysed regarding their effect on market success, actor-specific costs and the increase in the value targets of actors.³¹⁴

Based on these considerations the following section shows that SCP is a suitable target system for the evaluation of the integration of CT into SC concepts.

³¹² cf. Ibid., p. 219.

cf. Ibid., p. 218; Schary et al. (2001), p. 442. The authors argument that the cause-and-effect relationship and especially the effect direction of internal and external targets is often unknown.
 ³¹⁴

³¹⁴ cf. Schönsleben et al. (2003), pp. 22.

2.3.3 Suitability of SCP as a Target System for the Integration of CT into SC Concepts

The considerations of the SCP measurement and management conceptions are important for solving the research problem in the thesis in hand.

SCP is a suitable target system for the evaluation of the effect of CT integration into a specific SC concept because of the inclusion of strategic and operational aspects as well as the consideration of value- and finance-oriented aspects.

The SCP as a target system reflects the multiplicity and variety of different actors in SCs including CT. The multidimensional performance understanding allows applying the target system for actors with different strategic and operative alignments. CT is characterised by an environment of high investment and thus, long planning horizons. The future and potential orientation of SCP reflects this. Furthermore, since the analysis of the potential of CT integration is central to the thesis in hand, the SCP concept seems to be appropriate.

This section briefly reviews the literature on performance understanding and performance measurement in the context of CT. The recent performance understanding in the context of CT serves as the basis for the development of suitable performance indicators to evaluate the integration of CT into SC concepts.

Meaning of Performance Orientation in the Context of CT Concepts

In practice, a central challenge of the cross-company performance measurement is the actors' *inhibition* to share confidential data.³¹⁵ According to Hoffmann (2006),³¹⁶ further challenges are the service and network character of CT. The immateriality of the CT service, the uniqueness of the service offer as well as the specific framework conditions of the single actors are central concerns. For a cross-company performance management, a SC-wide performance understanding is necessary.

In a first step, four different dimensions of performance measurement for CT services are introduced. Based on this, the challenge of the evaluation of partial and total CT services is discussed. Several performance indicators and application possibilities are exemplarily introduced.

³¹⁵ cf. Hoffmann (2007), p. 152, Weber (2002), pp. 113.

³¹⁶ cf. Hoffmann (2007), p. 152.

Dimensions for the Performance Measurement of CT Services

The literature states that the performance of CT services can be measured by four different performance dimensions: (1) potential dimension, (2) process dimension, (3) output dimension and (4) the effect/benefit dimension.

In the *potential dimension*, the performance of a CT concept is understood as a capability or the availability of resources for the fulfilment of a specific service. The performance potential of a CT concept can be measured by qualitative and quantitative indicators. Quantitative indicators include the number of locomotives, wagons or train paths. These indicators provide no information on the usage of the resources.³¹⁷ Thus, additional qualitative information for the evaluation of the potential is required, such as the width of the service portfolio and the knowhow of the employees.³¹⁸ This qualitative information can be gained by experienced employees and by consulting experienced third parties.³¹⁹

CT service production can be evaluated by the indicators of the *process dimension*.³²⁰ The shipper as the external factor of the CT service production is important for the process dimension.³²¹ The customers of CT services are not in the service production all the time, but sometimes. This means external factors, for instance CT actors, influence the measurement as well as the result.³²² The key figures for the *process dimension* are qualitative or quantitative. Quantitative key figures include the distance covered in kilometres. Qualitative key figures refer, for instance, to disposition services or the provision of traffic information. The operating times of the disposition system or the share of requested and provided traffic information can express this.

The *result- or output-oriented dimension* interprets the performance of CT concepts according to the realised shipments' space/time changes. These include the space/time change. The result can be measured by the quantity of transported goods over a certain distance in ton kilometres or by the number of transport containers per time unit.³²³

³¹⁷ cf. Pfohl (2010), p. 213. Thus, these key figures are supposed to be part of target and performance agreements, for instance in terms of quotients of as-is operating hours and the maximum operating hours.

³¹⁸ cf. Hoffmann (2007), p. 153. ³¹⁹ af Engellie (1007), p. 42

³¹⁹ cf. Engelke (1997), p. 42.

³²⁰ cf. Reckenfelderbäumer (1995), pp. 34; Weber (2002), p. 65. ³²¹ af section 2.2.2.5 to the service character of CT services

³²¹ cf. section 2.2.2.5 to the service character of CT services.

³²² cf. Hoffmann (2007), p. 155.

³²³ cf. Engelke (1997), 46.

The *effect- or outcome-oriented dimension defines the performance of a* CT concept as the level/degree of fulfilment of a certain CT task. This means performance is evaluated at a rather immaterial level. For instance, meeting the shipper's demand can be an indicator of the effect-oriented performance dimension.³²⁴ Owing to the immaterial character of the indicator, the evaluation is characterised by a high level of subjectivity. The evaluation is accomplished from a customer's perspective with regard to its own demand structures, the demand structures of its customers and the market and competitive situation.

The literature shows that next to the classification according to performance dimensions the performance of CT concepts is evaluated for different scopes. The chosen performance indicators can reflect either partial or total performance. However, for the thesis in hand performance indicators for the evaluation of the entire SC must be developed.

Measurement of the Total and Partial Performance of CT concepts

The production of CT services means the integration of several partial services (cf. section 2.2.2.4). Thus, the performance measurement can be accomplished at the level of partial and total services. Hoffmann (2006) suggested a consideration of the performance measurement of CT concepts at three main levels: the single actor level, the dyadic level and the network level (all CT actors). For the thesis in hand, this perspective must be enhanced to the SC level.

³²⁴ cf. Weber (2002), p. 64.

At the *single actor's level*, the performance measurement considers only the activities of one CT actor. The performance indicators can be classified according to the efficiency and effective dimensions as well as the strategic and operative indicators. For instance, the efficiency of a road carrier can be evaluated by the operative costs per truck (strategic dimension) or the truck's capacity utilisation for the pre- and ongoing haulage (operative dimension). These indicators do not allow any statements on the performance of the other CT actors, the integration of material and information flow processes or the entire SCP. Figure 2 - 9 gives an overview of the strategic and operative performance indicators at the actor's level for road carriers, terminal operators, rail carriers, CT operators and 3PLs.

strategic / operative performance indicators		indicators for strategic performance	indicators for operative performance		
road carrier	efficiency	 average operative costs per truck average personnel cost share per kilometre 	capacity utilisation of trucks in pre and ongo- ing haulage		
	internal effectiveness	 degree of flexibility turnover per year	 adherence to schedules per delivery turnover per order		
terminal operator	efficiency	 average capital commitment per crane average capacity utilisation per transhipment facility 	 transhipment frequency per hour buffer and storage space capacity 		
	internal effectiveness	 degree of flexibility (value-adding services) increase in turnover by value-adding services 	 availability of transhipment capacities lead times in terminal 		
rail carrier	efficiency	average capacity utilisation of entire fleet	 utilisation of train per order number of trains per day 		
	internal effectiveness	 average adherence to schedules market share [%] 	• adherence to schedules at the arrival at the destination terminal		
CT operator	efficiency	average utilisation of terminal-to- terminal services			
•	internal effectiveness	number of offered block trains per week	booking deadline for customers [days]		
3PL	efficiency	average number of employees per order	number of employees per orderdisposition time per order		
	internal effectiveness	 degree of flexibility against shippers turnover share of certain CT market segments 	adherence to schedules at pickup and delivery		

 Table 2 - 10: Selected performance indicators at the actor's level.³²⁵

At the '*dyadic level*', performance measurement focuses on the interface between two CT actors. As shown in section 2.1, SC actors often focus on dyadical relationships instead of taking an integrated SCM perspective. In CT chains, the focus on dyadical relationships is also common. For instance, the performance between a CT operator and a rail carrier can be measured by the degree of flexibility in the availability of specific wagons (strategic dimension) or waiting times at the terminal (operative dimension)

³²⁵ cf. Hoffmann (2007), p. 173.

sion). Figure 2 - 10 gives an overview of selected performance indicators for the production of CT services. These performance indicators are neither suitable to evaluate the performance of the entire CT concept nor to evaluate SCP.

strategic / operative per- formance indicators	indicators for strategic performance	indicators for operative performance
shipper - road carrier	 average loss ratio degree of flexibility (e.g., regarding additional services, for instance container cleaning) 	adherence to schedules for collection of transported good
CT operator - terminal operator	• average availability of areas for buffering	 waiting time at terminal number of (un)loaded containers per hour
CT operator - rail carrier	 degree of flexibility (e.g., regarding additional services, for instance availability of special wag- ons, provision of single transport services be- yond main transport relations) average train cancellation days per year 	 number of locomotive engineers changes on one relation delay of locomotive at the railroad yard
3PL - CT operator	 quality index for evaluation of CT operator as supplier of terminal-to-terminal services average adherence to schedules of block trains 	 availability of transport capacities transport loss ratio of terminal- terminal-services

Table 2 - 11: Selected performance indicators at the dyadical level.³²⁶

The evaluation of the entire CT concept is accomplished at the so-called 'concept level'. Hoffmann (2006) introduced the term 'network level', but the term is misleading, since this thesis takes in an even wider perspective. Although Hoffmann (2006) interpreted the CT concept as a SC itself, an encompassing SCM perspective is not taken. The term 'concept level' expresses that all CT actors are encompassed: road and rail carriers, terminal operators, 3PLs, CT operators and shippers. This understanding does not equal the SC understanding of the thesis in hand. Neither the upstream processes of suppliers nor downstream processes towards the end customer are included in the evaluation on this concept level. Hoffmann gathered the performance of the CT chain under this term. She suggested that, for instance, the performance of a terminal-to-terminal service can be expressed by the degree of the fulfilment of value-added services (strategic dimension) or by the adherence to schedules of regular train departures (operative dimension). Figure 2 - 11 gives an overview of the strategic and operative performance indicators for the entire CT concept.

³²⁶ cf. Ibid., p. 173.

strategic / operative performance indicators	indicators for strategic perform- ance	indicators for operative perform- ance
3PL - CT operator - rail carrier - terminal operator (terminal-terminal services)	degree of fulfilment of value-adding services degree of flexibility regarding fluctuations in demand or relations for occasional transport	adherence to schedules of regular train departures reliability of provision of transport goods at the destination terminal for the ongo- ing haulage
shipper - 3PL (entire CT service)	entire costs of CT service service level level of total quality	adherence to schedules and delivery of the transport goods

 Table 2 - 12: Selected performance indicators at the concept level.³²⁷

For the thesis in hand, an external perspective on the entire SC concept including the CT concept is taken. This exceeds the performance measurement levels introduced by Hoffmann (2007).

Figure 2 - 12 visualises the different consideration levels. It shows the difference between (1) the company-specific performance measurement, (2) the CT concept performance measurement and (3) the SCP measurement. The figure shows that the company-specific perspective means the focus on *internal* effectiveness. This means for the evaluation that the main objective is the competitiveness of the single company. The cross-company perspective of the CT concept and the SC concept focus on *external* effectiveness and efficacy. This expresses the focus on the main target of competitiveness as opposed to other CT or other SC concepts.



Figure 2 - 12: Performance perspective for the integration of CT into SC concepts.³²⁸

³²⁷ cf. Ibid., p. 176.

³²⁸ With significant adaptations and enhancements according to Hoffmann (2007), p. 163.

For the performance measurement on a SC concept level, specific key figures or performance indicators are required.³²⁹ These performance indicators must meet the requirements of the formal and content targets of the SCM as well as the value-oriented management (operative and strategic dimension of performance indicators) (cf. section 2.1), the requirements on performance indicators (cf. section 2.2) and the characteristics of CT (c.f. section 2.3). In the following section, a specific target system including performance indicators for CT integration into SC concepts is developed.

2.3.4 Development of a Target System for the Evaluation of CT Integration into SC Concepts

To evaluate the integration of CT into SC concepts a SC wide performance understanding is necessary. This section introduces a target system and performance indicators based on the previous considerations of the performance-oriented integration of CT into SC concepts.

Requirements on the Developed Target System

The target system must meet the requirements on performance indicators introduced in section 2.3.1. The performance indicators must be able to be integrated and aggregated. Furthermore, the indicators must be suitable to indicate and forecast prospective developments. Additionally, profitability and application orientation are central criteria.

The performance orientation implies that the developed target system includes strategic and operative aspects. Furthermore, the research focus of the thesis in hand demands a focus on technical and organisation aspects and material and information flow processes. The developed target system does not include statements on the weighting of the included performance indicators. However, recent examples show that prospectively the meaning of classical key figures, such as costs, lead times, utilisation and inventories fade into the background for performance indicators reflecting reliability, stability, flexibility and sustainability aspects. The weighting of key figures must be accomplished for company, SC and situation. Furthermore, the mutual interdependencies and

⁸⁶

³²⁹ cf. section 2.2.

conflicts between different performance indicators are not considered.³³⁰

Performance Indicators

'Costs' and 'lead time' are 'classical' key figures expressing the efficiency of a SC concept. This means that that companies seek to reduce the value of these key figures. Lead time expresses the throughput time of an order.

However, depending on the perspective on the SC different time spans can be classified as lead time. The cost key figure encompasses all costs connected with material and information flow processes. Furthermore, flexibility is included in the target system to do justice to changing SCP requirements or short-term changes necessary because of end customer changes or interruptions in material flow processes.

The 'utilisation of capacities', and 'inventory levels' are classical key figures with a direct impact on SC costs. These key figures can be allocated to the effectiveness dimensions since material flow processes are configured to reach a defined level of these key figures. 'Adherence to schedules' is also a 'relative' key figure. It contributes to the increased meaning of SC stability and reliability.

Taking regard to the ongoing sustainability discussion, next to these material and information flows, emissions are included in the target system. In practice, there are several approaches to calculate so-called carbon footprints. These footprints aim at the inclusion of CO2 emissions into the life cycles of single products. Furthermore, sustainability is a central aspect for companies to shift transport volumes from unimodal road to CT (cf. section 2.2.3.3). Figure 2 - 13 illustrates the SCP target system for the integration of CT into SC concepts.

regard to material and information flow	efficiency lead time flexibility	effectiveness adherence to schedules utilisation of capacities inventory level	
additional key figues	costs emissions		

The developed target system serves to evaluate the integration of CT into the SC con-

³³⁰ For instance, the so-called 'scheduling dilemma' describes the conflict between capacity utilisation and adherence to schedules and lead time. cf. Nyhuis et al. (2008).

cept. However, the following section introduces configuration theory to support the problem solution process.

2.4 Configuration Theory as Theoretical Research Approach for Integrating CT into SC Concepts

This section introduces configuration theory as the theoretical research approach to support the solution process for the problem of CT integration.

The first step clarifies the meaning of theory application for practice-oriented research. In particular, the importance of the research perspective for the appropriate theory choice is highlighted. The second step introduces common theoretical research approaches for the problem of SC integration. The third step describes the selection of configuration theory and its application to the given problem formulation.

2.4.1 Meaning of Theory Application for Application-Oriented Research

The explanation of a problem formulation is supposed to be theory-guided whenever possible. Following Wolf (2005), the theoretical foundation of application-oriented research is essential.³³¹ For the thesis in hand, the theoretical foundation supports the developmet of propositions on the integration of CT into SC concepts.³³²

Theories provide general arguments to understand real world phenomena by abstracting singular instances that support the neutral researcher's perspective. They can be understood as distillates of the multiplicity of previous research projects. By means of theories, the historical development of new ideas and content can be tracked and are thus, interpreted as condensed information on practical challenges at certain time points by analysing emerging theoretical approaches. The theory foundation supports the researcher by integrating the research results into the existing knowledge base within the research discipline. Thus, the researcher's risk, for instance, regarding redundancy and lacking connectivity can be reduced.³³³

'Theories serve to describe, to explain and predict functions and issues. Usually, in economic research theories consist of a cross-linking of established hypotheses or approved empirical laws.' (translated according to Bortz & Döring 2005, p. 17).

³³¹ cf. Wolf (2008), pp. 54.

³³² cf. Stölzle (1999), p. 196.

³³³ cf. Wolf (2008), pp. 54.

Consistently in literature there is no theoretical approach that can entirely explain the phenomenon of SC control.³³⁴ SCM can be characterised by its cooperative character. Hence, theoretical approaches to the generation and control of cooperative relationships are commonly used to explain SCM and integration issues.³³⁵ The choice of suitable theoretical approaches is limited because of the neutral research perspective.

The following section briefly introduces common theoretical approaches towards the problem of SC integration.

2.4.2 Theoretical Solution Approaches towards SC Integration

This section introduces different approaches to theoretically explain the generation and control of cooperative relationships. As shown in section 2.1.3, cooperation serves as a basis for SC integration. Thus, the introduction of CT transport into SC concepts can be understood as the generation of a cooperative relationship between CT actors and shippers. The advantages and disadvantages of each theoretical approach are briefly discussed with regard to the transferability of argumentation patterns to the problem of CT integration into SC concepts.

To explain the cooperative inter-organisational relationships in SCM research the approaches of *transaction cost and agency theory*, as two elements of *new institutional economics theory*, are commonly applied.³³⁶ All new institutional theory approaches can be characterised by their focus on individual characters (either people or institutions) and the assumption of opportunistic behaviour. The *property rights theory* usually delivers no explanatory contribution to the questions of SC cooperation.

Transaction cost theory is an appropriate decision support framework regarding the chosen degree of vertical integration (make-or-buy) and serves to explain cooperative behaviour. It is also applied to design cooperative relationships (e.g., supplier–customer relationships) and respective logistics systems.³³⁷ To derive design-oriented statements transaction cost theory is often supported by situative approaches.³³⁸

³³⁴ cf. Gareis (2002), p. 175; Schmitz et al. (2004), p. 712; Staber (2000), pp. 58; Stölzle (1999), p. 25; Wohlgemuth (2002), p. 48.

³³⁵ cf. Karrer (2006), p. 69; Stölzle (1999), p. 25.

cf. Erdmann (2003), pp. 33; Gareis (2002), p. 186; Heusler (2004), pp. 111.; Rothfuss (2005), pp. 42.; Wolf (2008), pp. 327.

³³⁷ cf. Höll (2008).

³³⁸ See for instance Aggarwal (2007); Meisner Rosen et al. (2000); Skjoett-Larsen (2000); Stölzle (1999), p. 26.

Agency theory focuses on the contractual design and behaviour control within dyadic relations in which one party constitutes the agent hired by a principal.³³⁹ The relevant environmental factor is the incomplete and asymmetric information. The approach is characterised by a hierarchical relational structure.³⁴⁰

Systems theory is commonly applied to support the understanding of cooperative relationships. It has been enhanced to new systems theory. Here, systems are made up of related and interdependent components, including boundaries, outputs, inputs, transformation mechanisms (ways of converting inputs to outputs) and interfaces interacting with their environments. A system can be characterised by complexity and autonomy and cannot be directly influenced by the environment. Only the scope for autonomous control can be generated from outside the system.³⁴¹

Decision theory is commonly applied to explain logistical problems. Application fields include the definition of decision time points, the structuring of a problem or the development of a decision process.³⁴² In this context, game theory is applied to understand and predict the behaviour of players in a cooperative relationship.

Stölzle (1999) suggested applying these approaches to organisational procurement behaviour for the analysis of interactions between different actors, especially for buyer-supplier relationships.³⁴³ The approaches distinguish between different types of interactions (dyadic-personnel, multi-personnel, dyadic-organisational, multi-organisational).344

If there is a strong power and dependence relationship between cooperating actors, the resource dependence approach (RDA) can offer an explanation. The restricted availability of resources forces companies to buy these. This creates a situation of dependency. The RDA is a behavioural approach that provides explanation patterns towards dependency analysis between SC players and their behaviour.³⁴⁵

The structure of relationships between organisations in SCs can be compared with a network. In contrast to systems theory, network research and interaction approaches not only focus on the elements and connections within a system, but also consider the

³³⁹ cf. Eisenhardt (1988), p. 489; Eisenhardt (1989), p. 60; Lassar et al. (1996), p. 613.

³⁴⁰ See Hochhold et al. (2009), pp. 131.

³⁴¹ See for instance Forrester (1995); Lane (2000); Luhmann (1999); Schwaninger et al. (2008), p. 286; Weisbuch (2000). 342

cf. Lassar et al. (1996); Richardson (1991); Sterman (2001); Stevenson et al. (2000), p. 27; Weisbuch (2000). 343

cf. Ekwall (2007), pp. 27.; Stölzle (1999), p. 27. 344

cf. Kirsch (1973), pp. 112.

³⁴⁵ cf. Brunner (2008), pp. 31; Stölzle (1999), pp. 28.

interactions between these. As a result, network theory is especially appropriate for behavioural problems.³⁴⁶ There are two main streams of network research aiming at (1) the generation of networks and (2) the management of networks. There are several overlaps between network approaches and other theories. Thus, network research has not yet the status of its own theory. Nevertheless, it is often applied together with systems theory to logistics problems.

Situative approaches are organisational theories. They are also called *contingency approaches*. They allow statements about the choice of different design alternatives for a given problem formulation with regard to a specific target system. Considering the specific framework conditions concrete problem solution approaches can be derived from these approaches.³⁴⁷

This brief overview of available theoretical solution approaches in the field of SC cooperation serves as the basis for the selection process described in the following section.

2.4.3 Selection of Theoretical Approaches with regard to the Research Problem and Perspective

This section describes the selection of configuration theory to provide a theoretical explanation to the given research problem. To select suitable theoretical approaches, the research perspective is the guiding criterion.

This thesis takes a neutral and rational research perspective. The focus is on the integration of material and information flow processes in SCs including CT concepts. Following the terminology of Cooper (1997), the thesis focuses on SC integration in terms of *SC process integration*³⁴⁸, rather than *SC structure* integration³⁴⁹ and *SCM integration*.³⁵⁰ Thus, the focus is on the technical, physical and organisational aspects of the integration problem, rather than managerial and behavioural approaches.³⁵¹ This focus

³⁴⁶ cf. Grant (1991), p. 29.

³⁴⁷ See for example Carter et al. (2008), Kannan et al. (2005), p. 30.

³⁴⁸ SC process integration means the cross-company integration of business processes. These efforts can span all value-adding levels, for example research and development, demand planning, disposition or capacity utilisation processes.
³⁴⁹ SC process integration means the cross-company integration of business processes.

 ⁽SC relationships, inter-organisational business units) SC structure integration deals with the development of SC relationships (e.g., trust building activities) and the introduction of cross-company organisational units (e.g., project teams).

c.f. section 2.2.1 Cooper et al. (1997), p. 70. SCM integration means benefits for all involved SC members. Companies can access rare resources and new markets. They can share risks and reach synergy effects, e.g., developing new products and services, economies of scale and gain knowledge.

³⁵¹ cf. section 2.1.6.

reduces the problem's complexity, because all actor-specific aspects of the management of SCs are excluded from consideration.

Several theoretical approaches either consider actor- and organisation-specific problems or provide an explanation for the linking of objects in a network (e.g., network theory). For the research problem of CT integration into SC concepts a theoretical approach is required that is actor-neutral and focuses on the identification of successful configurations of the CT and SC concepts as well as integrative measures to improve SCP.

Since the research focuses on technical and organisational management components, all theoretical approaches towards the behaviour of individuals or organisations are excluded, such as the approaches of *new institutional theory*, *decision theory*, *game theory* or *RDA*,³⁵² from further consideration.

Furthermore, all approaches for the explanation of relationships between people or organisations are excluded, such as *network research, organisational procurement be-haviour and interaction approaches*. To ensure a strong focus on the core problem of CT integration, the willingness of actors to cooperate and to contribute to the performance-oriented design of these cooperations is assumed.³⁵³ Systems theory also focuses on the explanation of cooperative relationships. Furthermore, it assumes that the environment can only indirectly influence the system.

The application of situative approaches allows consideration from an objective perspective. *Configuration theory* is a development of situative theory. It aims at the consideration of different internal and external context factors explaining the development of different organisational structures. In particular, configuration theory supports the research process by linking the results of different research fields in a consistent overall picture.³⁵⁴ The configuration approach aims at the identification of inherent consistent patterns of design and context variables, so-called *'configurations'*.³⁵⁵ Transferred to the given research problem, this means that that consistent configurations as combinations of SC concept types and CT concept types are defined. These configurations are the basis for the identification of situative successful solutions for the integration problem.³⁵⁶

Further behavioural approaches are e.g., stimulus-contribution theory, hierarchy of needs and coalition theory.

This also explains the only brief introduction to the motivation for CT integration. Walf (2008) np. 452

³⁵⁴ Wolf (2008), pp. 452.

³⁵⁵ Ibid., p. 455.

³⁵⁶ cf. Klaas (2002), p. 52.

The following section briefly introduces the main characteristics of configuration theory.

2.4.4 Configuration Theory as an Explanation Approach for the Performanceoriented CT Integration into SC Concepts³⁵⁷

This section discusses configuration theory as an explanation approach for the research problem of the performance-oriented CT integration into SC concepts. Configuration theory supports the identification of successful combinations of different CT and SC concepts, of environmental conditions as well as integrative measures.

Configuration theory is an advancement of situative theory. To understand how configuration theory supports the problem solving process, in a first step the basics of situative theory are briefly introduced.

2.4.4.1 Basics of Situative Theory

The research objective of situative theory is the relationship between a situation and a configuration as well as between a configuration and the success of an organisation. An organisation can be characterised by its unique, measureable, formal organisational structural characteristics. These characteristics are determined by the given context. The central assumption of situative theory is that if organisational structure³⁵⁸ is aligned to the specific context, the organisation is *'efficient'³⁵⁹*. Thus, situative theory denies the existence of universal efficient organisational structures. It rather searches for formal structures that lead to a situatively reasoned efficiency through a *'fit'* with specific context factors.³⁶⁰

The results of this analysis are applicable in all companies not only in the given context. However, there is no universality of characteristics of behaviour and configurations, but rather a universality of relationships between these.³⁶¹

³⁵⁷ Also called 'second order contingency theory' or the 'gestalt approach'.

The behaviour of the member depends on the organisational structure as well as other external factors. According to Kieser, the behaviour can be excluded since the impact of behaviour is empirically difficult to measure. Thus, a structure-conform behaviour is assumed. Kieser (1999), p. 176; Kieser et al. (1992), p. 206.

³⁵⁹ There is no differentiation between the economic basic terms of *'effectiveness'* and *'efficiency'*. In this context, efficiency means meeting of organisational targets or, in general, the surviving of an organisation. cf. Kieser et al. (1992), p. 57; Kieser (1999), pp. 175; Staehle (1976), p. 36.

³⁶⁰ cf. Scherer et al. (1998), p. 334; Staehle (1976), p. 38; .

³⁶¹ This is an important difference in comparison with systems theory (makes statements only for specific situations).
The central research question of situative theory is:

'Which context factors explain (the presence of) different configurations?' (own translation according to Wolf (2008), p. 200)

To answer this main question a second question must be answered:

'How can the situation and configurations of the organisation be operationalised?' (own translation according to Wolf (2008), p. 200)

Several publications have dealt with the questions of (1) to what extent different configurations affect the success of organisations and (2) to what extent different situation/configuration constellations differ from each other regarding their success. The term of success is operationalised in several different forms. Often, the target systems of organisations are applied as reference points for success measurement.³⁶²

Transferred to the problem of CT integration into SC concepts, these basic considerations of situative theory can be interpreted as follows. The term 'situation' encompasses the relevant environmental factors of an organisation as well as the organisation's characteristics. These organisational characteristics correspond to the term 'SC concept'. The environmental characteristics include 'SC strategy' and 'SC environment'. A 'configuration' means the combination of a specific 'SC concept' and a specific 'CT concept'. The connection of situation and configuration determines the success of the specific configuration.

To apply situative theory, the considered variables must first be determined. There are 'context variables', 'design variables' and 'success variables'. In a second step, the variables have to be specified and the relationships between the variables have to be defined.

The advantages and disadvantages of situative theory have been widely discussed.³⁶³ One development to overcome the disadvantages is configuration theory.

2.4.4.2 Configuration Theory as an Advancement of Situative Theory

Configuration theory sets up the disadvantages of situative theory.³⁶⁴ Configuration theory aims to analyse multiple variables and their characteristics to determine the spe-

³⁶² cf. Wolf (2008), p. 200.

³⁶³ See Ibid., pp. 215.

³⁶⁴ Several authors have focused their research on the deficiencies of this approach. cf. Sennheiser et al. (2008), p. 452. The main critical points of mono causality, uncreative retesting, un-differentiation, linear relationships, ignorance of equifinality, context determinism and missing longitudinal section led to several enhancements of the theory aiming at an integrated design-oriented

cific organisational situation. The target of configuration theory application is the analysis of typical variable configurations and the identification of successful variable configurations. These configurations can be derived either empirically or from the literature.³⁶⁵

In contrast to situative theory, configuration theory assumes a multidimensionality of cause-and-effect patterns.³⁶⁶ Furthermore, configuration theory suggests the existence of non-linear interdependencies between context and design variables.³⁶⁷ For the performance-oriented integration of CT into SC concepts, this means that there can be more than one successful configuration of design variable for a given configuration of context variables. Furthermore, there can diverse integrative measures suitable for the specific configuration.

The basic assumption of situative theory is that an organisation is successful if the context and design variables are congruent.³⁶⁸ According to the *'consistency efficiency hypothesis'* the

'[...] effective structuring requires a close fit between the contingency factors and the design parameters'. (Mintzberg (1979), p. 219)

Configuration theory enhances this understanding by using the so-called *'configuration hypothesis'*.³⁶⁹

'[...] effective structuring requires an internal consistency among the design parameters' (Mintzberg (1979), p. 219)

The *'enhanced configuration hypothesis'* summarises both hypotheses. This assumes that consistent combinations of design variables must correspond to consistent combinations of internal and external context factors.³⁷⁰

'[...] the characteristics of organizations fall into natural clusters, or configurations. When these characteristics are mismatched - when the wrong ones are put together - the organizations does not function effectively, does not achieve full harmony.' (Mintzberg (1981), p. 103)

analysis of phenomena. cf. Höll (2008), Wolf (2008); Miller et al. (1984); Macharzina et al. (1991). All these publications aim at the identification of complex context-design-success patterns.

³⁶⁵ cf. Macharzina et al. (2005); Wolf (2008), p. 457.

³⁶⁷ cf. Wolf (2008), p. 452.

³⁶⁹ cf. Klaas (2002), p. 104; Mintzberg (1979), p. 219.

³⁶⁶ cf. Wolf (2000), pp. 4; Miller (1981), pp. 3; Mintzberg (1983), pp. 60.

³⁶⁸ cf. Mintzberg (1979), p. 219; Staehle (1994), p. 59,

^{&#}x27;[...] effective structuring requires a consistency among the design parameters and contingency factors.' Mintzberg (1979), p. 220, cf. also Scherer et al. (1998), p. 336, Staehle (1994), p. 59, Wolf (2000), p. 25.

Thus, configuration theory assumes the existence of consistent organisational configurations that can be characterised by harmonic patterns between design variables and context variables.

Basic Principles of Configuration Theory

Configuration theory is based on four general principles: (1) fit concept, (2) equifinality thinking, (3) configurational thrift and (4) discontinuity of transition.³⁷¹

(1) The 'fit concept' postulates that the success of an organisation can be interpreted as the consequence of the fit of at least two variables.³⁷² This general concept is applied to the development of the conceptual research framework of the thesis in hand. It is assumed that the fit of the CT and SC concepts as well as the fit with specific integrative measures and instruments can lead to or improve SCP. These two 'fits' correspond to the idea of internal and external coherence.

The fit of the CT and SC concepts (internal coherence) and the fit of these with the environment (external coherence) are necessary to achieve SCP.³⁷³

(2) 'Equifinality' means that the same end state can be reached starting from different initial situations (functional equivalence of situations). Transferred to the problem of CT integration, this means that that there are different ways to improve SCP from any SC concept initial situation. Wolf (2000) identified the potential of information processing to be central. This encourages focusing on the informational and technical aspects.

(3) *'Configurational thrift'* means that in reality there are a limited number of successful configurations. This number of configurations is many times smaller than is the number of variable combinations resulting from a systematic combination of all characteristics.³⁷⁴ Thus, it can be assumed that groups of the CT and SC concepts can be identified that fit together with more or less effect on SCP. However, this element of configuration theory implies that despite the choice of situative approach general guidance can be derived in the thesis in hand.

Groups of the CT and SC concepts can be assumed to work together with more or less performance effect. However, this element of configuration theory implies that, al-

³⁷¹ cf. Wolf (2000), p. 465; Wolf (2008), pp. 454.

³⁷² cf. Wolf (2008), p. 465.

³⁷³ cf. Ibid., p. 465.

In total, there are seven reasons for this assumption. cf. Henselek (1996), p. 274.

though we choose a situative approach, we can derive general guidance from our research. In total, there are seven reasons for this assumption.³⁷⁵

(4) *Discontinuity of transition*: Regarding the development of configurations, the theory implies that transitions are not steady, but soaring. According to Wolf (2000), there is an alternating sequence of momentum periods and transition periods.³⁷⁶ Thus, configuration theory encourages longitudinal analysis. According to the integration of CT, specific adjustments have to be considered over time. The success of integration has to be evaluated over time since the effects of changes are time lagged.³⁷⁷

The following section addresses the application of configuration theory to the given research problem after this rather general outline.

2.4.4.3 Application of Configuration Theory to the Problem of the Performance-Oriented Integration of CT into SC Concepts

Configuration theory supports the development of a situative workable orientation framework for the integration of CT into SC concepts. This framework consists of a restricted number of configurations, which reflect the major part of real SC concepts including CT. This set of configurations can serve as a diagnostic instrument³⁷⁸ as well as a heuristic orientation framework for solving design problems in practice (*'design function'*). Furthermore, these SC concept configurations (including CT) can serve as a framework for the systematic exploration of integration problems in the field of transport and SCM (*'knowledge function'*). Thus, the set of SC concept configurations can be understood as conceptual guiding points for the theory-guided exploration as well as for the application-oriented design of SC design and CT.

Configuration theory can explain the success of specific CT concepts, specific SC concepts and specific integrative measures and instruments. More precisely, configuration theory supports:

(1) The identification of typical context variables configurations, namely different CT concepts that appear in practice and that have a comparable character in terms of CT integration.

³⁷⁵ cf. Ibid., p. 274.

For justification of these assumptions cf. for example Veliyath et al. (1995); Wolf (2008), pp. 23.

³⁷⁷ cf. Wolf (2008), pp. 23

³⁷⁸ cf. Klaas (2009).

- (2) The identification of typical and successful combinations of context variables, namely CT concepts and SC concepts. The identified successful combinations are called 'gestalts'.
- (3) The identification of typical and successful combinations of integrative measures and instruments for the identified configurations and gestalts.

In this context, success means the adherence to the developed target system of SCP (cf. section 2.3.3.2). In practice, SCP improved with a unimodal road transport serves as the benchmark for the evaluation of the CT integration.³⁷⁹

Thus, configuration theory supports the answering of the second and third research questions.

The following section summarises and joins the main findings on configuration theory with the central findings of chapter 2.

2.5 Intermediate Findings

This section joins the main findings of the second chapter of the thesis in hand. The second chapter concretises the research problem and introduces the state of the art as the basis for the application of a theoretical solution approach. Thus, chapter 2 lays the foundation for answering the research questions presented in chapter 1.

Concretisation of the Research Problem

The second chapter concretises the problem of performance-oriented CT integration . The chapter identifies associated challenges caused by CT integration into SC concepts in three steps. First, the challenge of SC integration in general, second the challenges caused by the complexity of CT and third the challenge of a SC-wide performance orientation are illuminated. Furthermore, the second chapter introduces configuration theory to support the problem solution process. The section highlights that the integration of material and information flow processes in SCs is generally challenging. The complexity of CT and the increasing performance orientation of SC actors further exacerbate this integration. Configuration theory allows the consideration of this quantity of different aspects. It also enables the development of a conceptual research framework, in which the SC concept is the design variable, the CT concept acts as the framework vari-

³⁷⁹ cf. section 2.2.2.

able and both are connected by a 'fit', namely the integration of material and information flow processes, leading to improved SCP.

Five categories of integrative measures and instruments for SC integration were identified in a comprehensive literature review. These findings are the basis for the analysis presented in chapter 4. Here, the instruments and measures are discussed regarding their suitability in the field of CT and regarding their impact on certain SCP requirements.

Additional Time Amounts and Coordination Effort in Comparison with Unimodal

Road Transport

The second chapter points out the competitive situation of CT and unimodal road transport. Additional transhipment, handling and overheads increase time requirements and the multiplicity of CT actors raises the necessary coordination effort. Both aspects result in an increased cost level in comparison with unimodal road transport. Shippers and policymakers increasingly perceive CT as a promising solution to overcome problems in unimodal road transport, such as rising transport costs, congestion, sustainability initiatives and transport political regulations. However, many shippers claim that CT does not meet SCP requirements. Table 2 - 13 summarises the main contradictions and contrasts shippers' performance requirements with CT characteristics.

Shi	ppers' performance requirements	CT characteristics		
•	Shippers aim to reduce lead times and imply short transport times and high adherence to schedules.	CT requires additional time quotas in comparison with unimodal road transport.		
•	Shippers aim to reduce costs and to reach a high level of flexibility.	CT requires special investments in specific load carriers and handling equipment. Therefore, CT actors require long planning horizons to redeem investments.		
•	Shippers aim to introduce lean structures and to avoid unneces- sary processes or the waste of resources.	CT requires additional processes (transhipment, rail transport, buffering for bundling purpose etc.).		
•	Shippers demand constant accompaniment of transport goods.	In a CT concept on rail transport the goods are unaccompa- nied.		
•	Shippers do usually not include transport actors in their plan- ning activities and deny access to information platforms and forecasts.	The multiplicity of actors and partial processes increase coordination efforts and exacerbate capacity planning.		

Table 2 - 13: Comparison of shippers' performance requirements and CT characteristics.

As the chapter points out, stable, predictable transport volumes with comparably long lead times and regular frequencies are particularly suitable for CT. These characteristics allow foe the realisation of bundling effects and the reduction of planning efforts and overheads. Furthermore, CT networks are suitable for the transport of packaged goods with comparably long lead times.

Changing Framework Conditions Require an Innovative Target System

SCP is introduced as a target system that is suitable to meet the changing framework conditions of recent SCs. The section points out that a classical target system must be

aligned with regard to financial- and value-oriented purposes. This target system reflects operational and strategic aspects and includes the performance indicators of the efficiency and effectiveness dimensions. This means that that in the context of CT, new performance indicators must be applied to the evaluation of SCP. The reasons for this include the increasing vulnerability of SCs, demanding SCP requirements and uncertainty over energy prices. Thus, especially, the aspect of sustainability must be included and a re-weighting of performance indicators in favour of adherence to schedules and flexibility is necessary.

The second chapter concretises the research problem and introduces the first aspects for the development of solution approaches to answer the research questions presented in chapter 1. Henceforth, chapter 3 introduces a structured classification approach to describe SCP requirements regarding the embedded and linking transport concepts. On this basis, the meaning of CT as an element of SC concepts is discussed. The differentiation into procurement, production, procurement and transport concepts serves as the basis for developing the conceptual research framework in chapter 4.

3 CT as an Element of SC Concepts

The third chapter addresses the meaning of CT as an element of SC concepts. It clarifies the position and the meaning of the CT concept for SCP. Therefore, the relationship between the procurement, production and distribution concepts and the embedded and linking transport concept is worked out. In particular, the relevant cause-and-effect relationships between the elements of the sub-concepts and the elements of the CT concept are identified.

Initially, the third chapter analyses the construct of SC concepts as the operationalisation and implementation of a specific SC strategy (section 3.1). The SC concept interprets and transfers the SC strategy to the defined material and information flow processes and builds up corresponding organisational structures as well as management processes. For the precise description of the SCP requirements regarding transport concepts, a structured description framework with seven performance dimensions is developed (section 3.2). Based on this classification the SC sub-concepts, namely procurement, production and distribution concepts, are analysed regarding their requirements on the embedded and linking transport concepts (section 3.3). In particular, the meaning of the SC concept configuration on the integration of CT concepts is worked out (section 3.3). Finally, a typology for CT concepts and corresponding performance profiles is developed (section 3.4). The intermediate findings are summarised in section 3.5.

3.1 SC Concepts as the Operationalisation of SC Strategies

SC strategy has a long-term character. The SC strategy itself does not provide information about the concrete design of the material and information flow level and the corresponding structures and management processes. In the thesis in hand, a SC concept is understood as this missing operationalisation of SC strategy.¹

On the basis of Heuermann (2002); Hilletofth (2009); Shewchuck (1998).

3.1.1 SC Strategies as the Basis for SC Concept Configuration

Actions and parameters are of strategic character if they lead to the creation and exploitation of success potential or if they influence the long-term development of a company.² A strategy ensures that short-term decisions correspond with the company's long-term objectives.³ Thus, strategic management research differentiates between levels of strategy. According to Hofmann (2010), four generic levels of firm strategies can be distinguished.⁴

(1) The 'SC strategy' deals with the strategic alignment of inter-organisational activities.⁵ Both structural and dynamic components are necessary for this. Structural activities determine relationships, network structure and focus on the alignment of target systems within a network, whereas dynamic components work on the combination of resources together with other companies in inter-organisational routines or projects.⁶

(2) The *'corporate strategy'* aims at the definition of target industries and the composition of business activities.⁷

(3) A *'business unit strategy'* focuses on the realisation of competitive advantages and the question of how a company should compete in the chosen market.⁸ This means that companies aim at the production of goods or services at a lower cost level than competitors do or to differentiate from the competitors by unique value-adding activities.⁹

(4) *'Functional strategies'* refer to either operational or supporting activities. Operational activities include sourcing, production, distribution and logistics activities. Supporting activities include human resources or IT. The functional strategy must be aligned with the other components of firms' strategies.

Companies increasingly realise the need to integrate information and material flow processes with upstream and downstream business partners.¹⁰ Westbrook and Frohlich (2001) described this as a change from a vertical alignment of activities to a horizon-

² cf. Sennheiser et al. (2008).

³ cf. Grant (2002), p. 793.

⁴ cf. Brown et al. (2005), pp. 256.

⁵ Hofmann uses the term 'network strategy'. cf. Hofmann (2010) pp. 256.

⁶ cf. Ibid., pp. 256.

⁷. cf. Ibid.; Bowman et al. (2001); Grant (2002).

 ⁸ cf. Hofmann (2010); Grant (2002).
 ⁹ cf. Hambrick (1980); Porter (2004).

¹⁰ cf. Thompson et al. (2005); Tan (2001); van Hoek (2001).

tal alignment of activities.¹¹ The chosen SC strategy is central to the way of treatment of business partners.¹² Since SC actors aim for a common objective, companies tend to build up competitive advantages by commonly focusing on this.¹³ By this the SC differs from others by establishing a common value-adding capacity.¹⁴ The introduction of the SC strategy proves this understanding. According to Campbell and Wilson (1996), the SC strategy can be understood as an enhancement of the operations strategy.¹⁵ While the operations strategy aims to ensure that competitive priorities are accomplished by one company, the SC strategy means the objectives of the focal company deal with the SC members. Several authors have stated that to choose a suitable SC strategy the SC environment has to be carefully analysed.¹⁶

Lo and Power (2010) interpreted the SC strategy as a set of choices made by companies to match the environmental contingencies. Several authors have followed this SC strategy understanding and defined the choice of an appropriate strategy as a positioning within an environment of fundamental trade-offs.¹⁷ However, recently the understanding has changed since several companies build capabilities in groups of interest.¹⁸

The SC strategy determines the alignment of an entire SC. According to Chopra and Meindl (2004), the term 'SC strategy' can be defined as follows:

'A supply chain strategy determines the nature of procurement of raw materials, transportation of materials to and from the company, manufacture of the product or operation to provide the service, and distribution of the product to the customer, along with any follow-up service. From a value chain perspective, SC strategy specifies what operations, distribution, and service will try to do particularly well.' (Chopra & Meindl (2004), p. 29)

The definition gives an overview of the contents of the SC strategy. The definition highlights the meaning of procurement, distribution, production as well as transport and logistics processes. However, for implementation purposes the definition of a SC strategy is not sufficient. For the implementation of material and information flow processes and the corresponding organisational and management structures more spe-

¹¹ cf. Webster (2002); Lo et al. (2010). ¹² of Erablish et al. (2001)

¹² cf. Frohlich et al. (2001).

 $^{^{13}}$ cf. Lo et al. (2010).

¹⁴ cf. Campbell et al. (1996). ¹⁵ of L₂ at al. (2010): Compl

¹⁵ cf. Lo et al. (2010); Campbell et al. (1996); Lo et al. (2010); Harland (1996); Harland et al. (1999). ¹⁶ of Harland et al. (2001); Harland et al. (2004); Lombort et al. (1998); Lo et al. (2010)

¹⁶ cf. Harland et al. (2001); Harland et al. (2004); Lambert et al. (1998); Lo et al. (2010).

¹⁷ cf. Ojasalo (2004); Hayes et al. (1978); Hayes et al. (1979).

¹⁸ cf. Lo et al. (2010); Ferdows et al. (1990); Lo et al. (2010).

cific determinations are needed. In the thesis in hand, this more 'operative' and situation-specific concretisation is named the 'SC concept'.

In particular, the corporate strategy influences the SC strategy.¹⁹ If the corporate strategy is focused on service aspects, the SC strategy must ensure that products and services are available at anytime and anywhere, namely they must have an outstanding delivery service without out-of-stock situations. Therefore, the SC strategy focuses on delivery speed and accuracy. If the corporate strategy is focused on quality aspects, the SC strategy must guarantee a high level of delivery reliability with a low error rate in SC processes, including reliable quality control processes (especially regarding product quality). If the corporate strategy focuses on flexibility, the SC strategy must ensure the fast and safe product launch and the possibility of meeting fast changing customer demands. This means the SC strategy supports the production of a high product variety and reacts to fluctuations in the demand of different product types. If the corporate strategy focuses on cost aspects, the SC strategy must focus on an efficient infrastructure to reduce overall SC costs.²⁰

In the thesis in hand, the SC concept is understood as the operationalisation of the SC strategy.²¹ The SC concept considers the necessary company, product and environmental contingency factors that influence the specific design and configuration of the material and information flow processes and the corresponding structures and management processes. The SC concept determines precisely the operationalisation using resources and processes and considers the specific internal and external contingencies (framework conditions) and capacities, such as resources (e.g., personnel, technical facilities and equipment and knowhow), product characteristics, regional SC structure and performance requirements of the end customer with regard to the SC strategy.²²

Owing to this close term understanding and the tight connection between SC strategy and the SC concept, the following section introduces the basics of SC strategy development.

 21 On the basis of Heuermann (2002); Hilletofth (2009); Shewchuck (1998).

²² cf. section 1.3.

¹⁹ cf. for instance Rosenzweig et al. (2004); Hofmann (2010), pp. 19.

²⁰ cf. Heuermann (2002); Hofmann (2010), pp. 19; Heuermann (2002).

3.1.2 Classification Approaches for SC Strategies

This section highlights the meaning of choosing the right SC strategy according to the specific demand and supply characteristics.²³ The section improves the understanding of different SC strategies and the resulting SC concepts.

As stated in the previous section, the SC strategy determines procurement, transport, service, manufacturing and distribution configurations to supply the end customer with the desired product. Shewchuck (1998) showed why different SC strategies are required for different types of products. He proved that products differ in their demand types and usually in their marketing and manufacturing environments, too.²⁴ The author distinguished between products with a *'compressed life cycle'*, *'compressed time-to-market'*, *'mass customisation via assembly'* and *'mass customisation via processing'*. Each product group has different marketing and manufacturing characteristics.²⁵

Fisher's Approach to SC Strategy Classification

Fisher (1997) suggested the assignment of suitable SC types according to different product types.²⁶ He stated that a mismatch of product type and SC strategy results in severe problems in business operations. The approach of Fisher has been the basis for several other SC strategy classification approaches.

Fisher (1997) assumed that functional products should be supplied by efficient (in terms of costs and productivity) 'lean' SCs. Functional products satisfy the basic needs of customers.²⁷ These usually do not change over time and are thus, stable and predictable in demand. This leads to comparably long life cycles and high competition, resulting in low profit margins. Innovative products should be supplied by responsive 'agile' SCs.²⁸ Companies aim at innovations to overcome low margins. Fashion and personal computers are obvious examples of innovative products. Fisher introduced examples of companies that aim to bring innovation to functional prod-

²³ cf. Porter (1980), p. 465.

²⁴ cf. Christopher et al. (2008).

²⁵ Mass customisation accomplished by processing, for instance for spectacle lenses, made-to-order shoes or clothes can be characterised by a high product variety and a single unit demand with very short lead times. The production environment enables the high product variety by on-the-spot processing and a shallow bill of material. The different product types impact the other SC activities. To get further information on the choice of a suitable SC strategy the information and the material flow de-coupling points have to be identified. cf. Shewchuck (1998), p. 461; Christopher et al. (2008); Mason-Jones et al. (1997).

²⁶ cf. Naylor et al. (1999). ²⁷ cf. Fisher (1997)

²⁷ cf. Fisher (1997).

²⁸ cf. Ibid.; Fearne et al. (2006).

ucts, such as Starbucks²⁹ and Ben & Jerry's³⁰ with new flavours and innovative shop concepts.³¹

Lean SCs emphasize efficiency and bring along the risk of mismatch between production and customer demand. Agile SCs focus on responsiveness and may cause a risk of a low level of production efficiency.³²

Following Fisher (1997), an efficient SC aims to '[...] supply predictable demand efficiently at the lowest possible cost' (Fisher (1997), p. 108), whereas a responsive SC focuses on the fast reaction on customers' needs and thus, aims '[...] to minimize stockouts, forced markdowns, and obsolete inventory'. (Fisher, 1997, p. 108) According to Fisher's understanding, a SC strategy includes several corporate operational strategies. These include a manufacturing focus, inventory strategy, corporate meaning of lead times and product design strategy. Furthermore, supplier selection criteria are included.

Works Extending Fisher's Model - Development of Christopher's Generic SC

Strategies

Several authors have since extended Fisher's model. However, others disagree with the approach of product nature as the only or main factor for SC strategy choice.

Naylor et al. (1999) combined Fisher's model with the manufacturing paradigms of lean and agile.³³ Setting cost and quality, lead time and service level as well as the stability of demand as classification elements, they distinguished between lean, agile and leagile SC strategies.³⁴ Leagility combines both leanness and agility objectives. Processes upstream of the customer decoupling point apply lean processes. Processes downstream of the customer decoupling point apply agile processes. A leagile system tends to find equilibrium between cost reduction and fast customer response.

Further researchers have reworked the categorisation of products and introduced differentiated categories (for instance, the 'innovative-unique' type³⁵ and 'hybrid' type). Among others, product complexity was suggested to be a significant aspect of SC

International retailer specialised in coffee products. Starbucks buys, roasts and sells coffee products in the owned and licensed shops. For further details see www.starbucks.com.
 US is a compared by Usiler and the details are further details are normal backgroup.

³⁰ US ice cream producer owned by Unilever. For further details see www.benjerry.com.

³¹ cf. Hilletofth (2009), p. 106.

³² cf. Fisher (1997), p. 107.

³³ cf. Ibid.

³⁴ Similar classification approaches were proposed by Naylor et al. (1999); Childerhouse et al. (2000); Christopher et al. (2001); Mason-Jones et al. (2000).

³⁵ cf. Startton et al. (2003).

strategy choice.³⁶

Some authors have supported Fisher's model, but claimed that product nature is not only influenced by demand aspects,³⁷ but that product architecture, life stages, customisation level and number of parts are also relevant characteristics. Vitasek et al. (2003), for instance, proposed the consideration of volume demand and variability.³⁸ Cigolini et al. (2003) developed a conceptual framework for the choice of a SC strategy according to product life cycle and production complexity.³⁹

Pagh and Cooper (1998) proposed product life cycle, product customisation, product variety, product value, relative delivery time, delivery frequency and uncertainty of demand as relevant classification variables.⁴⁰ The four types of SC strategies, namely full speculation, logistics postponement, manufacturing postponement and full postponement, go back to the authors. Lamming et al. (2000) set product innovation, product uniqueness and product complexity as classification variables to distinguish three different strategies.⁴¹

Another group of researchers have argued that sources of uncertainty (not only demand uncertainty) should be considered in the SC strategy since it deals with downstream and upstream partners. Sources for uncertainty include raw materials procurement,⁴² production uncertainties (caused by lead time variations)⁴³ and the capacity restrictions of focal companies and suppliers, which were not considered in Fisher's model.⁴⁴

Based on these considerations, Christopher et al. (2005) introduced a classification of SC strategies according to three parameters: (1) product type (standard vs. special), (2) demand type (stable vs. volatile) and (3) replenishment lead times (short vs. long).⁴⁵ Special products with low volume and irregular demand can be characterised by short life cycles and a high level of individualisation. A standardised product with stable demand and high volumes can be characterised by longer life cycles and only limited possibilities for customisation. Replenishment lead time is identified as the

³⁶ cf. Lamming et al. (2000); Christopher et al. (2002), Lamming et al. (2000); Lo et al. (2010). ³⁷ of Bomdos (2002); Catalan et al. (2002); Lo et al. (2010); Bomdos (2002)

³⁷ cf. Ramdas (2003); Catalan et al. (2003); Lo et al. (2010); Ramdas (2003).

³⁸ cf. Wong et al. (2005). ³⁹ cf. Vitasek et al. (2003)

 $^{^{39}}$ cf. Vitasek et al. (2003).

⁴⁰ cf. Cigolini et al. (2004). ⁴¹ of Page et al. (1008)

⁴¹ cf. Pagh et al. (1998). ⁴² af Lemmin et al. (2000).

⁴² cf. Lamming et al. (2000); Ho et al. (2005); Lee (2002). ⁴³ of La et al. (2010)

⁴³ cf. Lo et al. (2010). ⁴⁴ cf. Ho et al. (2005).

⁴⁴ cf. Ho et al. (2005).

⁴⁵ cf. Kim et al. (2002); Christopher (2005a), p. 283; Christopher et al. (2006), p. 463.

central parameter for the choice of a suitable SC strategy because it significantly influences the responsiveness of the SC. Product type and predictability are related, so that the derived taxonomy includes only two dimensions.



Figure 3 - 1: Generic SC strategies.⁴⁶

If demand is stable, and thus, predictable and lead times are short, a *lean strategy with a focus on continuous replenishment* (type I) is suggested. If lead times are long a *lean strategy* but *with a focus on planning and optimisation* (type II) is proposed. If demand is unpredictable and lead times are short, Christopher et al. (2006) suggested an *agile SC strategy*. If demand is unpredictable and lead times are rather long, the authors suggested a hybrid *'leagile' SC strategy*. Here, a decoupling of a lean and an agile part of the SC through postponement is the best approach. In literature, the terms for these three different strategies are also used and discussed for manufacturing systems.⁴⁷

The classification of SC strategies presented by Christopher et al. (2005) is the basis for the following considerations on the configuration of SC concepts. This classification apppears to be suitable for the research problem of CT integration into SC conccepts because lead time and demand characterisistics are central aspects for the CT concept choice.

⁴⁶ According to Christopher et al. (2008), p. 119.

⁴⁷ e.g., Christopher (2005a); Goldsby et al. (2006); Huang et al. (2002); Startton et al. (2003).

3.1.3 Impact of SC Strategy on the Configuration of SC Sub-Concepts

The SC concept can be understood as the operationalisation and implementation of the SC strategy. The SC strategy determines the SC concept configuration, namely the configuration of the procurement, production and distribution concepts as well as the embedded and linking transport concepts.⁴⁸

SC managers have to make several decisions about how to operationalise SC strategy. They have to decide whether to design processes, structures and management as cost-efficient or responsiveness-oriented.⁴⁹ Thus, the derived SC concept can differ significantly between SCs with the same strategy.

A SC concept can be defined as the interpretation and transfer of the SC strategy to concrete *material and information flow processes*.

A 'SC **concept'** defines all relevant technical and organizational structures, planning and control structures as well as management processes to ensure the stable time-, quality- and cost-defined material and information flow on all relevant value adding levels with a focus on meeting the end customer's demand. It is based on the SC strategy and encompasses the configuration of the procurement, production and distribution sub-concepts as well as the embedded and linking logistics and transport concepts.⁵⁰

The relationship between SC strategy and SC concept is analysed in depth in the following section.

3.1.4 Elements and Structures of SC Concepts

'SC concept' is a collective term for the procurement, production and distribution concepts as well as their embedded and linking transport concepts.⁵¹ The configuration of these sub-concepts determines the information and material flow.

Meyr und Stadtler (2005) published a classification of SCs according to functional and structural characteristics.⁵² They suggested classifying SCs according to the functional characteristics of the procurement type, production type, distribution type and sales type.⁵³ Furthermore, Hofmann (2010) suggested the implementation of the SC

⁴⁸ cf. Heuermann (2002); Hilletofth (2009); Shewchuck (1998).

⁴⁹ cf. Agarwal et al. (2006), p. 21; Hilletofth (2009); Hofmann (2011); Naylor et al. (1999).

⁵⁰ Definition on the basis of Heuermann (2002); Hilletofth (2009); Shewchuck (1998).

⁵¹ according to Hilletofth (2009); Hofmann (2011); Rosenzweig et al. (2003).

⁵² cf. Heuermann (2002).

⁵³ Sales type corresponds to the product and demand characteristics gathered in the SC strategy. cf. section 3.1.3.

strategy in the three functional areas of procurement, production and distribution.⁵⁴ The thesis follows these considerations. It is assumed that a SC concept can be described by the specification of the configuration of the procurement, production and distribution concepts.

The classification according to SC sub-concepts reveals a general problem. The classification of the distribution and procurement concepts implies an actor-specific perspective. The distribution concept of one actor corresponds to the procurement concept of the other actor. Since this aims at the consideration from a non-actor-specific research perspective, in the following section procurement and distribution concepts are analysed separately.

Logistics concepts are embedded and linking elements of the procurement, production and distribution concept. Following the definition of the Council of SCM Professionals (CSCMP), 'logistics' is defined as '[...] that part of SCM that plans, implements, and controls the efficient, effective forward and reverses flow and storage of goods, services and related information between the point of origin and the point of consumption in order to meet customers' requirements.⁵⁵ Following this definition, the transport concept is a part of the logistics concept focusing on the material flow aspect and the aspect of spatial transformation.

Production concepts have corresponding activities at the information and material flow level. Transport concepts supply and connect the different production sites. Procurement and distribution processes have a preliminary informational character. Logistics and transport processes concretize procurement and distribution concepts in the form of specific material and information flow processes.

A *transport concept* ensures the material flow between a certain source and a certain destination through the provision of the corresponding capacities of vehicles, personnel, knowhow and overheads.⁵⁶ Usually transport concepts are part of a logistics concept.⁵⁷

⁵⁴ cf. Hofmann (2010).

⁵⁵ cf. Hofmann (2011).

⁵⁶ A transport concept can cover more than one source and one destination. For more details on the structures and elements of transport concepts see section 3.3.1.
⁵⁷ In brief the logistic concepts are characterized by the artist SC. Logistics concepts are characterized by

⁵⁷ In brief, the *logistics concept* ensures material and information flow along the entire SC. Logistics concepts are characterised by their '*dyadical*' character. The logistics concept includes material flow as well as information flow processes, such as warehousing, transport and handling, sorting and order picking, packaging and signing (material flow) as well as order processing (information flow). The logistics concept distinguishes from the definition of the SC concept (cf. section 3.1) because of its dyadical character. This means that the logistics concept considers only two SC actors. cf. CSCMP (2011), p. 9.

A 'transport concept' defines all relevant technical and organizational structures, planning and control structures as well as management processes to ensure the stable time-, quality- and cost-defined material flow from a specific source to a specific destination.⁵⁸

Transport concepts can have company-internal as well as cross-company character and can link two or more sites in a SC.

The thesis in hand focuses on the performance-oriented integration of CT concepts. Thus, the following considerations focus on the impact of SC concept configuration on the embedded and linking transport concepts. Logistics processes are interpreted as integrative parts of the procurement, production and distribution concepts and are thus, not separately discussed.

There are several transport concepts in one SC. They are either embedded in the specific concepts or link these by connecting a source and a sink with a material flow. To fulfil this task the performance requirements of the procurement, production and distribution concepts regarding the transport concept must clearly be described. For the structured description of SCP requirements in section 3.2, a framework of seven performance dimensions is developed. The proposed performance dimensions in this framework support can be applied to develop so-called SCP requirements profiles.

There are a number of standardised practices for the design of SC concepts. These standardised practices can be distinguished according to different scopes, application fields and objectives. In the literature and in practice, these practices are usually termed 'logistics concepts', 'SC concepts' or 'delivery concepts'. These standard concepts mean that the same configuration of the procurement, production and distribution concepts or partial aspects of these can be applied to a SC. Thus, the term 'SC concept' must be distinguished between that used in the thesis in hand and that used in the literature and in practice.

The applied standard SC concepts have a direct impact on the configuration of material and information flow processes. For instance, the *concept of industry parks*⁵⁹ is an approach to create proximity between suppliers and buyers and thus, to positively

A 'logistics concept' defines all relevant technical and organisational structures, planning and control structures as well as management processes to ensure the stable time-, quality- and cost-defined material and information flow between two actors.

Definition developed on the basis of Pfohl (2010), pp. 9; ibid.

⁵⁸ Definition developed on the basis of From (2013), pp. 9, 104. Definition developed on the basis of Gudehus (2005); Heuermann (2002); Hilletofth (2009); Pfohl (2010), pp. 9.

⁵⁹ For details on the concept of industry parks see for instance Das (2010); Hao et al. (2008);Lambert et al. (2000); Patil et al. , p. 405.

influence the integration of material and information flow processes. Different practices aim at shifting responsibilities and focusing on the corporate core competencies. These are the *VMI concept*⁶⁰ and outsourcing activities, such as the *3PL*⁶¹ and *4PL concepts*⁶². In the field of procurement the alignment of suppliers and buyers' activities such as *Supplier Relationship Management (SRM)*⁶³ and *Customer Relationship Management (CRM)*⁶⁴ are applied. These approaches are closely related to the practices to improve planning and forecasting, such as the *CPFR* concept⁶⁵ as well as the *Efficient Consumer Response (ECR)* concept.⁶⁶ All standardised SC concepts can be represented by the configuration of the procurement, production and distribution concepts as shown in sections 3.2.2 to 3.2.4.⁶⁷

The following sections build on these considerations and discuss the SC concept configuration for the SC strategies presented by Christopher (2005).

3.1.4.1 SC Concept Configuration in a Lean SC

The term 'lean' first appeared in the field of manufacturing in 1990.⁶⁸ In 1996, it was enhanced to the concept of '*lean thinking*'.⁶⁹ The main idea of lean thinking was the elimination of all non-value-adding activities.⁷⁰ The original idea of lean thinking can be drawn back to the *Toyota Production System*. This focuses on the efficient use of resources by level scheduling⁷¹ and states that a value stream is developed that reduces all non-value-adding processes with regard to time.⁷² The measures include the decrease in inventory levels and lot sizes, the elimination of paperwork and redundant processes, the evaluation of suppliers on quality and delivery performance as

⁶⁰ For details on the concept of VMI see for instance Darwish et al. (2010); Kwak et al. (2009); Sari (2008), p. 399; Sennheiser et al. (2008).

⁶¹ For details on the concept of 3PL see for instance Carbone et al. (2005); Ellram et al. (1990); Jayaram et al. (2011), p. 411; Sennheiser et al. (2008); Yao et al. (2010). For the meaning of the 3PL concept for integration of CT concepts see for instance Sheffi (1990) and Bendul et al. (2009).

⁶² For details on the concept of 4PL see for instance Bretzke et al. (2010), p. 47; Bobel et al. (2009); Bobel et al. (2011), p. 414; Sennheiser et al. (2008), p. 413.
⁶³ Enclusion of the concept of the second s

⁶³ For details on the concept of CRM see for instance Win (2008), p. 391; Choy et al. (2002), p. 387.

⁶⁴ For details on the concept of SRM see for instance Fricker (2008); Koch et al. (2008), p. 387, 391: Sennheiser et al. (2008).

⁶⁵ For details on the concept of CPFR see for instance Esper et al. (2003); Sari (2008); Sennheiser et al. (2008), p. 385.

⁶⁶ For details on the concept of ECR see for instance Siebel (2002), p. 391.

⁶⁷ The postponement and mass customisation concepts are often termed SC or logistics concepts. However, in the thesis at hand the postponement strategy is discussed in the context of SC strategies (cf. section 3.1.3). The mass customisation concept, as a management approach based on the postponement considerations, is discussed in the SC strategy context, too.

⁶⁸ cf. Mason-Jones et al. (2000).

⁶⁹ cf. Womack et al. (1990).

 $^{^{70}}$ cf. Womack et al. (1996).

⁷¹ cf. Christopher et al. (2001); Ohno (1988).

⁷² cf. Hilletofth (2009); Riezebos et al. (2009).

well as the introduction of long-term contracts with a reduced supplier basis.⁷³ Lean principles are suitable for application in a stable and predictable market environment with low varieties of products.⁷⁴

If lead times are short, a *'continuous replenishment'* strategy can be suggested.⁷⁵ Huge production companies apply this concept for retailing customers to ensure a stable product flow based on cooperative and automated order systems. Deliveries are controlled by actual sales. Often, this includes the application of VMI concepts. Here the supplier is responsible for the retailer's inventory holding.⁷⁶ In production, the concept can be applied in the form of the Kanban concept.⁷⁷

If the lead times are long, lean strategies are applied that *'plan and optimize'* the SC most efficiently. Christopher (2008) provided an example from the UK retailer Woolworth. Woolworth sells several million Christmas trees every year. Woolworth sources these from China. Thus, orders are placed more than six months in advance. Nevertheless, because of stable demand, risk is evaluated to be low because of the given experience.⁷⁸

A lean SC strategy causes the configuration of SC sub-concepts with regard to efficiency and for rather high volumes.

The *procurement concept in a lean SC* is characterised by a comparably high share of bought goods.⁷⁹ SRM is characterised by regular, high volume transactions. Thus, supplier relationships are comparably close and are based on a high level of information sharing.⁸⁰ The demand management and forecasting in a lean SC is quantitative and data-driven. Forecasting in a lean SC is accomplished at the finished goods level and the control of variability is comparably easy.⁸¹

The *production concept in a lean SC* is characterised by the manufacturing policy of make-to-stock.⁸² Usually, the decision for a lean SC corresponds with standard products and stable and predictable demand. This means that production is based on forecasts rather than specific customer orders. Production processes are continuous and

 ⁷³ cf. Naylor et al. (1999); de Treville et al. (2004).
 ⁷⁴ of Uillet fb (2000)

 ⁷⁴ cf. Hilletofth (2009).
 ⁷⁵ cf. Christopher (2000), p. 464.

 ⁷⁶ For details on the concept of VMI see Christopher et al. (2008); Darwish et al. (2010); Kwak et al. (2009); Sari (2008), p. 399; Sennheiser et al. (2008).

⁷⁷ cf. Yao et al. (2010); Dickmann (2009), p. 389.

⁷⁸ cf. Sennheiser et al. (2008), p. 464.

 ⁷⁹ cf. Balakrishnan et al. (2004); Christopher et al. (2008); Mason-Jones et al. (2000); Naylor et al. (1999); Olhager (2003).
 ⁸⁰ of Sharland et al. (2004)

⁸⁰ cf. Sharland et al. (2003).

⁸¹ cf. Mason-Jones et al. (2000); Naylor et al. (1999); Olhager (2003); Stavrulaki et al. (2010). ⁸² af Balalwichnen et al. (2004); Chei et al. (2000); Naylor et al. (1909); Ohne (1989); Stavrula

⁸² cf. Balakrishnan et al. (2004); Choi et al. (2009); Naylor et al. (1999); Ohno (1988); Stavrulaki et al. (2010).

characterised by large volumes and batches.⁸³ Production and assembly are organised in lines and with regard to efficiency.⁸⁴ Product design can also be affected by the SC strategy. In a lean SC, product design is also focused on cost consciousness and efficiency (e.g., the integration of barcodes into the product design of discounters). Lean SCs aim at a balance between efficiency and flexibility.⁸⁵ Here, standardised products or packages are typical.⁸⁶

The distribution concept in a lean SC is characterised by the focus on costefficiency.⁸⁷ The configuration logistics process and order fulfilment is cost-driven and characterised by high volumes.⁸⁸ Thus, in a lean SC the distribution concept is configured with regard to high transport quantities and comparably low costs.⁸⁹ Demand management is based on quantitative data and algorithmic forecast mechanisms. The number of intermediaries between the manufacturer and end customer is comparably high.⁹⁰

3.1.4.2 SC Concept Configuration in an Agile SC

The term of agility is often connected with responsiveness and the ability to match demand and supply in unpredictable markets.⁹¹ According to Christopher (2000), agility can be defined as:

'Agility is a business-wide capability that embraces organizational structures, information systems, logistics processes and, in particular, mindsets. A key characteristic of an agile organization is flexibility.' (Christopher (2000), p. 37)

The basis of the agility concept can be found in the flexible manufacturing system.⁹² The focus of the agile SC strategy is information flow integration at all SC levels to create high market-responsiveness in order to deal with volatile demand and to avoid lost sales, markdowns and the obsolescence of inventories.⁹³ Thus, the focus of an agile SC strategy is on the informational aspects rather than on efficient physical

⁸³ cf. Agarwal et al. (2006); Sharland et al. (2003). 84

cf. Goldsby et al. (2006). 85

cf. Lee (2002); Mason-Jones et al. (2000). 86

cf. Mason-Jones et al. (2000); Naylor et al. (1999); Olhager (2003). 87

cf. Stavrulaki et al. (2010). 88

cf. Narasimhan et al. (2006). 89 cf. Stavrulaki et al. (2010).

⁹⁰

cf. Choi et al. (2009); Goldsby et al. (2006); Mason-Jones et al. (2000). 91

cf. Stavrulaki et al. (2010); Agarwal et al. (2006). 92

cf. Hilletofth (2009). 93

cf. Christopher et al. (2001).

product flows.⁹⁴ The target is the reduction process and information time at all SC levels.⁹⁵

Christopher (2008) introduced Zara, a Spanish fashion garment manufacturer and retailer, as an example of an agile SC strategy. Zara needs no more than four weeks from design until products are in stores. Cross-functional teams are used to reaching the required high level of flexibility.

An agile SC strategy causes the configuration of SC sub-concepts with regard to responsiveness and rather small volumes.⁹⁶

The *procurement concept in an agile SC* is characterised by the alignment of processes to responsiveness and flexibility.⁹⁷ The purchasing policy aims at the assignment of capacities. Supplier relationships are characterised by rather infrequent and low volume transactions. In comparison with a lean SC strategy, the supplier relationship is a more opportunistic collaboration with more collaborative barriers.⁹⁸ Demand management is based on experience. Forecasting is accomplished at the raw materials level.⁹⁹

The *production concept in an agile SC* is characterised by a manufacturing policy of assembled, made or designed depending on specific customer orders.¹⁰⁰ Thus, the predictability of demand and planning basis for transport concepts is restricted. The production concept in an agile SC is characterised by a make-to-order or design-to-order manufacturing policy. The production system in an agile SC focuses on responsiveness and flexibility and is thus characterised by job shop project and a rather jumbled material flow.¹⁰¹ The focus on flexibility is reflected by individualised and specialised product designs. In an agile SC, the focus on flexibility is reflected by individualised and specialised product designs.¹⁰²

The *distribution concept in an agile SC* focuses on responsiveness and flexibility.¹⁰³ Order fulfilment is time-driven and characterised by rather low volumes.¹⁰⁴ Therefore, the number of intermediaries between the manufacturer and end customer is

 $^{^{94}}$ cf. van Hoek (2001).

⁹⁵ cf. de Treville et al. (2004).

⁹⁶ cf. Heuermann (2002), p. 135, 146; Hilletofth (2009), pp. 19.

⁹⁷ cf. Balakrishnan et al. (2004); Mason-Jones et al. (2000); Olhager (2003); Stavrulaki et al. (2010).

⁹⁸ cf. Stavrulaki et al. (2010); Sharland et al. (2003).

⁹⁹ cf. Balakrishnan et al. (2004); Naylor et al. (1999).

¹⁰⁰ cf. Choi et al. (2009); Stavrulaki et al. (2010); Goldsby et al. (2006); Lee (2002); Sharland et al. (2003).

¹⁰¹ cf. Agarwal et al. (2006); Mason-Jones et al. (2000); Storey et al. (2005). ¹⁰² af Navler et al. (1000)

¹⁰² cf. Naylor et al. (1999).

¹⁰³ cf. Stavrulaki et al. (2010).

¹⁰⁴ cf. Swafford et al. (2006).

rather small.¹⁰⁵ The applied forecast mechanisms are rather consultative than being algorithm-based. In a rather agile SC, experience is central for forecasting and thus, for the procurement of raw materials.¹⁰⁶ Demand management generally depends on experience.¹⁰⁷

3.1.4.3 SC Concept Configuration in a Leagile SC

Goldsby et al. (2006) proposed the idea that a lean–agile hybrid can be applied to the provision of base and surplus demand. They suggest answering base demand by a lean SC, whereas demand peaks can be managed in an agile manner.¹⁰⁸

More common is the leagile understanding in terms of the *'postponement concept'*. The postponement concept has its roots in the linking of uncertainty costs and risks with product differentiation.¹⁰⁹ The main idea is that uncertainty costs can be avoided or reduced by postponing processes, such as logistics and manufacturing activities until concrete orders are available.¹¹⁰ For a special product is may be sensible to postpone manufacturing,¹¹¹ while for a standard product it may be sensible to postpone distribution processes.¹¹²

Christopher (2008) introduced the example of Hewlett Packard and its strategy for DeskJet printers. A semi-finished product is manufactured in North America and then shipped to four regional centres. These are operated by 3PLs. Here, the products are configured and delivered when customer orders are received.¹¹³

The so-called *'logistics postponement'* aims at the storage of a high inventory level at one or a few strategic locations to postpone the inventory location upstream of the SC as late as possible. This means distribution is delayed as long as possible.¹¹⁴

'Manufacturing postponement' tends to keep products in a neutral state as long as possible.¹¹⁵ The differentiation point is shifted to the latest possible point by delaying assembly, production, sourcing or even design until there are concrete orders. This

¹⁰⁵ cf. Stavrulaki et al. (2010). ¹⁰⁶ of Christopher (2000)

 $^{^{106}}$ cf. Christopher (2000).

¹⁰⁷ cf. Goldsby et al. (2006); Mason-Jones et al. (2000); Narasimhan et al. (2006).

 $^{^{108}}$ cf. Stavrulaki et al. (2010).

 $^{^{109}}$ cf. Goldsby et al. (2006).

¹¹⁰ cf. Bucklin (1965); Pagh et al. (1998); Swaminathan et al. (2003); van Hoek (2001).

¹¹¹ cf. Kisperska-Moron et al. (2010); Lin et al. (2010); Yeung et al. (2007).

¹¹² cf. Brun et al. (2009); Ma et al. (2002). For details see p. 464; Christopher et al. (2008); Huang et al. (2008); Pagh et al. (1998).

¹¹³ cf. Wong et al. (2010).

cf. Bowersox et al. (1996); Christopher et al. (2008); Swaminathan et al. (2003).

¹¹⁵ cf. van Hoek (2001).

means the customer decoupling point is moved downstream of the SC as far as possible. Before this point, the SC is forecast-driven, whereas after the decoupling point, the SC is order-driven.¹¹⁶

A leagile SC strategy causes the configuration of SC sub-concepts depending on the location of the customer decoupling point. Upstream of the decoupling point (typically in manufacturing or distribution), processes are designed with regard to efficiency; downstream, processes are designed with regard to responsiveness.¹¹⁷

The *procurement concept in a leagile SC* is characterised by the location of the customer's decoupling point.¹¹⁸ Depending on this, the leagile SC is characterised upstream by rather cost-efficient processes and downstream by rather responsive and flexible processes. Often, VMI concepts are applied.¹¹⁹ The supplier relationships upstream of the order decoupling point have a rather lean character. Thus, demand management, forecasting and variability control are data-driven. The relationships with suppliers are based on regular, high volume transactions.¹²⁰

The *production concept in a leagile SC* is characterised by the manufacture-to-order or assembly-to-order manufacturing strategy.¹²¹ Here, modular designed products or packages are typical. Depending on the customer's decoupling point, the production processes are designed either with regard to efficiency or to flexibility focus.¹²² Thus, in a leagile SC assembly lines and small batch jobs are common.¹²³

The *distribution concept in a leagile SC* depends on the customer's decoupling point.¹²⁴ Upstream of the customer decoupling point, the leagile SC has a rather lean character, whereas downstream of the customer decoupling point the leagile SC has a rather agile character.¹²⁵ Upstream, logistics processes are configured with regard to efficiency, cost and high volumes.¹²⁶ Downstream, processes are characterised by

¹¹⁶ Often, *the concept of mass customisation* is discussed in the context of postponement. The mass customisation concept is a hybrid management approach towards a coincident differentiation and cost leadership strategy based on the postponement concept. The target of the concept is the production of goods and services for a huge market by meeting the differing demands of customers at a cost level that is comparable to standardised products. Therefore, mass production on the basis of a generic product architecture and different product modules is necessary. End products only differ in the specific configuration. The manufacturer aims at the individualisation of products and the interaction with customers. This increases variety and complexity, but increases the quantity of sales. cf. for instance Aigbedo (2007); Bowersox et al. (1996); Dörflinger et al. (2001); Feitzinger et al. (1997).

¹¹⁷ cf. Heuermann (2002), p. 135, 146; Piller (2006), pp. 19.

cf. Stavrulaki et al. (2010); Balakrishnan et al. (2004); Mason-Jones et al. (2000); Naylor et al. (1999); Olhager (2003).

¹¹⁹ The VMI concept is in depth discussed in section 3.3.

¹²⁰ cf. Mason-Jones et al. (2000); Naylor et al. (1999); Sharland et al. (2003).

¹²¹ cf. Choi et al. (2009); Goldsby et al. (2006); Lee (2002); Olhager (2003).

¹²² cf. Stavrulaki et al. (2010); Mason-Jones et al. (2000); Olhager (2003).

¹²³ cf. Agarwal et al. (2006); Naylor et al. (1999). ¹²⁴ af Goldsby et al. (2006)

 $^{^{124}}$ cf. Goldsby et al. (2006).

¹²⁵ cf. Duclos et al. (2003); Stavrulaki et al. (2010).

¹²⁶ cf. Lummus et al. (2003)

flexibility and responsiveness. Order fulfilment is time-driven and characterised by rather low volumes.¹²⁷ The number of intermediaries between the manufacturer and end customer is rather small.¹²⁸

The following section introduces a classification approach for the description of the SCP requirements of SC sub-concepts regarding the embedded and linking transport concepts. This structure is the basis for the in-depth discussion of the impact of the SC concept configuration on CT integration.

3.2 Development of a Classification of SCP Requirements regarding Transport Concepts

The procurement, production and distribution concepts have specific performance requirements regarding embedded and linking transport concepts. These requirements result from the specific design of material and information flow processes, organisational structure and management processes. The integration of a CT concept (and any other transport concept) into a SC concept demands a clear description and communication of performance requirements. Thus, this section develops a structured description approach for SCP requirements with seven categories for the cross-company communication of SCP in terms of transport concepts. This description approach aims at answering the first research question.

Development of the Classification Approach

A precise description of SCP requirements regarding transport concepts is the basis for the integration of material and information flow processes in SCs. As shown in section 2.1, one problem for the close integration of material and information flows is the missing understanding of processes and operational challenges among SC actors. The operational problems of SC actors, especially including CT, significantly differ (cf. section 2.1.2.3). This section aims to generate clear and consistent performance requirements on transport processes to improve the mutual understanding between the SC and CT actors.

Few publications in the SCM literature deal with transport integration, transport mode and transport carrier choice.¹²⁹ Although several authors have discussed the

¹²⁷ cf. Mohammed et al. (2008).

¹²⁸ cf. Goldsby et al. (2006); Mason-Jones et al. (2000); Stavrulaki et al. (2010).

¹²⁹ cf. Stavrulaki et al. (2010); Meixell et al. (2008)

central attributes of transport mode and transport carrier decision, a general description structure is missing.¹³⁰ Few have discussed transportation mode and carrier choice in terms of SC integration.¹³¹ One central topic in the field is the effect of SC design on transport costs.¹³² However, no publications have covered the targeted design of SCs to enable the integration of a specific transport concept.

In order to understand the specific SCP requirements of shippers 23 interviews with experts from medium-sized and large companies in different industries (building, chemistry, mineral oil, food, retail, steel and mechanical engineering) were conducted. Either a member of the executive board or the chief logistics officer was interviewed. The experts were chosen because of their industry expertise and experience with CT.¹³³ The interview results were structured and concretised based on literature and Internet research.¹³⁴ The methodology does justice to the quality criteria for expert interviews. The interviews were documented, data were triangulated and the experts were integrated into a communicated validation of results.

Seven Categories to Describe SCP Requirements

Seven main categories of shippers' performance requirements were identified. Shippers specify their requirements on transport services primarily regarding (1) time, (2) space, (3) transport quantity and shipment size, (4) product type and (5) costs. Furthermore, there are secondary requirements regarding (6) flexibility and (7) reliability.¹³⁵

Time-related SCP Requirements

A '*time*' dimension gathers all aspects with regard to process and lead times, delivery and pickup dates and frequencies as well as time windows. In particular, transport time is an important key performance indicator for the evaluation of transport services.¹³⁶ Time-related performance requirements must address transport, transhipment, storage, handling and communication times as well as time amounts for associ-

 ¹³⁰ e.g. Goldsby et al. (2000); Abshire et al. (1991); Bardi et al. (1989); Crum et al. (1997); Danielis et al. (2005); Dobie (2005); Evers et al. (2000); Gibson et al. (2002); Gibson et al. (1993); Kent et al. (2001); Lu (2003); Maier et al. (2002); Naim et al. (2006); Premeaux (2002).
 ¹³¹ a G Configure et al. (2003); Carter et al. (1005); Evers et al. (2003); Lise et al. (2007); Miller et al. (2003); Shinghel et al. (2002).

¹³¹ cf. Caplice et al. (2003); Carter et al. (1995); Esper et al. (2003); Liao et al. (2007); Miller et al. (2003); Shinghal et al. (2002).

cf. Baumgarten et al. (2004); Kasiske (2004); Murphy et al. (1993) mainly focuses on strengthening the image of rail and water carriage.
 Sas Asmendia A. I for a list of interview conducted.

¹³³ See Appendix A-I for a list of interviews conducted.

For instance Bortz (2005) presented a compilation of quality criteria. cf. Bortz et al. (2006), pp. 308.

The term 'secondary' implies that flexibility and reliability can be defined with regard to each other and to the other criteria. cf. Nagel (2008); Chen et al. (2004).

ated logistics services. These requirements can be summarised in three main categories: (a) delivery windows,¹³⁷ (b) transport / lead times and (c) delivery frequencies. Depending on the SC strategy, the general characteristics of these criteria can generally be stated. A lean SC strategy is usually reflected by comparably short delivery windows to reduce the resource usage at the delivery site. In agile SCs, the delivery windows are longer since the availability of products in general has priority over the reduction of resource usage. Transport and lead times are long in comparison with agile SCs. A lean SC requires comparably low, but regular delivery frequencies to realise bundling effects. In an agile SC, the delivery frequency must ensure the responsiveness and is thus, more frequent, but irregular. A leagile SC can be characterised by a lean and an agile part depending on the customer's order decoupling point. Thus, the characterisation is indifferent – either short or long. Table 3 - 1 summarises these findings.

dimen- sion	performance criteria	lean	leagile	agile
time	delivery window	short	indifferent; depending on the integration point	long
	transport / lead time	long	indifferent; depending on the integration point	short
	delivery frequency	low / regular	indifferent; depending on the integration point	high / irregular

 Table 3 - 1: Time-related SCP requirements depending on the SC strategy.

Space- and Location-related SCP Requirements

All shippers' requirements on the source and destination of the material flow can be summarised by the *'space'* dimension. This can be specified by different dimensions. It contains requirements about transport distance between the different pickup and delivery sites (manufacturing, warehouses, transhipment and stores). Additionally, the number of these pickup and delivery sites as well as those of the sources and destinations must be considered. In particular, for the integration of rail, water and air transport the distance to the infrastructure connection is important. This distance can be evaluated by the share between the entire transport distance and the distance to the infrastructure connection.

A lean SC is characterised by comparably long transport distances and distance to infrastructure, whereas in an agile SC distances must be shorter to increase respon-

¹³⁷ The criterion 'delivery window' was also suggested by Frohlich et al. (2002).

siveness. The length of transport distances usually corresponds to the number of sources and destinations. In a lean SC, the number of sites is usually lower than that in an agile SC. These can be described, for instance, by the places of loading and unloading, specific capacities (for instance, for transhipment, buffering, storage), equipment and layout. Table 3 - 2 summarises the findings on space-related performance requirements.

dimension	performance criteria	lean	leagile	agile
space	transport distance	long	indifferent	short
			depending on the integration point	
	distance to infrastruc-	long	indifferent	short
	ture connection		depending on the integration point	
	no. of sources and	low	indifferent	high
	destinations		depending on the integration point	

 Table 3 - 2: Space-related SCP requirements depending on the SC strategy.

The 'transport quantity and shipment size' dimension is a central aspect of the design of a transport concept. In a lean SC, transport quantities and the sizes of the shipments are comparably high. In an agile SC, transport quantities and shipment sizes are rather small. Table 3 - 3 summarises the findings on shipment size and transport quantity SCP requirements.

dimension	performance criteria	lean	leagile	agile
transport		high	indifferent depending on the integration point	low
shipment size			depending on the integration point	

 Table 3 - 3: Shipment size and transport quantity SCP requirements depending on the SC strategy.

The '*product type*' dimension gathers all product-specific requirements. The central requirements of a transport service result from the type of goods (e.g., aggregate, chilled, hazardous, fragile etc.), goods mix and transport quantity. These criteria directly affect the requirements for load carriers (size, number, weight, design, etc.) as well as handling and storing equipment. Thus, the need for certain types and designs of loading ramps, vendor docks and sidings, handling and transhipment equipment, such as stackers, forklifts, reachstackers and cranes can be summarised under the 'product' category. The meaning of the '*product type*' dimension has been highlighted by several authors.¹³⁸ In a lean SC, products are usually standard with no specific requirements. In an agile SC, products can be rather special. In a leagile SC, product type depends on the point in the SC, but often products are of modular structure.

¹³⁸ cf. Christopher (2000); Devaraj et al. (2007); Frohlich et al. (2001); Jayanth et al. (2010).

Product type is a central requirement in terms of transport concepts. The decision about a certain product type is strategic in nature (cf. section 3.1). Thus, for the discussion of the impact of SC sub-concept configuration it is assumed that product type cannot be influenced by the adaptations to sub-concepts. Table 3 - 4 summarises the findings on product type-related SCP requirements.

dimension	performance criteria	lean	leagile	agile
product		standard	standard, special,	special
type			modular	

 Table 3 - 4: Product type-related SCP requirements depending on the SC strategy.

The '*cost*' dimension captures requirements regarding desired target costs, process costs and transaction costs. Shippers' requirements regarding cost performance criteria can be expressed, for instance, by transport costs, costs per transport or transport unit and emissions per transport unit or kilometre. Shippers have certain requirements around risk sharing, devolution of property, incoterms,¹³⁹ insurance and emergency aspects, which can be summarised in the '*cost*' category. Different authors have confirmed the meaning of costs from the shippers' points of view to be key performance indicators within SCs.¹⁴⁰ In particular, the meaning of inventory and inventory turns has been analysed in several publications.¹⁴¹ In a lean SC, the share of transport costs and entire turnover is comparably high. Thus, the cost pressure is distinctively higher than that in an agile SC. Here, the share of transport costs is comparably low and thus, transport costs are usually not central to the SC design. Table 3 - 5 summarises the findings on the cost-related SCP requirements.

dimension	performance criteria	lean	leagile	agile
cost	transport cost / turnover	high	medium	low

Table 3 - 5: Cost-related SCP requirements depending on the SC strategy.

'*Flexibility*' aspects can be discussed with regard to each of the previous categories. Describing the requirements for transport concepts, especially flexibility regarding time, space and transport quantity, is central. Flexibility requirements can be defined regarding spatial aspects such as the variety of final destination and origin. Flexibility can also be stated with regard to delivery times, delivery frequency and short-term changes as well as with regard to the handling of different product types and quanti-

¹³⁹ Incoterms are international rules for typical trade contracts. Following Rosenzweig et al. (2003).

¹⁴⁰ cf. Chen et al. (2004); Devaraj et al. (2007); Gabler (2011); Poirier et al. (2004).

¹⁴¹ See for instance Frohlich et al. (2001); Ranganathan et al. (2004); Rosenzweig et al. (2003).

ties. Often, the term *'responsiveness'* is used for external flexibility. It stands for flexibility regarding different product types, the product mix, volumes, processes and destinations.¹⁴² This is supported by several authors, for instance by and Chen & Huang (2004). Flexibility is necessary to deal with demand uncertainties,¹⁴³ variability,¹⁴⁴ product variety¹⁴⁵ and lead time compressions.¹⁴⁶ For the integration of CT into SC conceptss, three main flexibility dimensions can be distinguished. Each can be described by the variance and frequency of variances. In a lean SC, the flexibility of the criteria regarding time (e.g., shorter lead times, delivery times), quantity (e.g., more or less transport quantities) and space (e.g., different pickup/delivery sites) must be comparably low. In an agile SC, the flexibility in all dimensions must be high to ensure a high level of responsiveness. In a leagile SC, the flexibility requirements depend on the specific location in the SC. Table 3 - 6 summarises the findings on the flexibility-related SCP requirements.

dimension	performance criteria	lean	leagile	agile
flexibility	time	low	indifferent	high
	(variance/ frequency)		depending on the integration point	
	quantity (variance / fre-	low	indifferent	high
	quency)		depending on the integration point	
	space (variance /frequency)	low	indifferent	high
			depending on the integration point	

Table 3 - 6: Flexibility-related SCP requirements depending on the SC strategy.

'*Reliability'* levels can be defined regarding time (lead time, time window, etc.) and product quality. The meaning of delivery reliability has been discussed in several publications including Rosenzweig et al. (2003), Chen & Huang (2004), Devaraj et al. (2007) and Poirier & Quinn (2004). For the thesis in hand, adherence to schedules and product quality are defined as the relevant reliability requirement dimensions. Adherence to schedules can be expressed as the ratio of punctual and delayed transport. Product quality can be expressed as the ratio of transport without quality complaints and transport with quality complaints. In a lean SC, based on predictability and the reduction of inventories the reliability dimensions are of high importance. Although reliability is important for leagile and agile SCs too, the required level is comparably lower. Table 3 - 7 summarises the findings on the reliability-related SC requirements.

¹⁴² cf. Zhu et al. (2002), p. 1144.

¹⁴³ cf. Fisher et al. (1994); Griffiths et al. (2000); Harrison (1996); Reichhart et al. (2007).

¹⁴⁴ cf. Harrison (1996); Krajewski et al. (2005).

¹⁴⁵ cf. Berry et al. (1999); Inman et al. (1997); MacDuffie et al. (1996).

¹⁴⁶ cf. Randall et al. (2001); McCutcheon et al. (1994), p. 1144.

dimension	performance criteria	lean	leagile	agile
reliability	adherence to schedules	high	medium	medium
	product quality	high	medium	medium

 Table 3 - 7: Reliability-related SCP requirements depending on the SC strategy.

Table 3 - 8 summarises the considerations in the form of SCP requirements profiles for lean, leagile and agile SCs.

performance criteria	dimension	lean	leagile	agile
time	delivery window	short	indifferent	long
	transport time	long	indifferent	short
	delivery frequency	low / regular	indifferent	high/ irregular
space	transport distance	long	indifferent	short
	distance to infrastructure connec- tion	long	indifferent	short
	number of sources and destinations	low	indifferent	high
transport quantity / shipment size		high	medium	low
product type		standard	indifferent (standard, special, modular)	special
cost	transport cost / turnover	high	medium	low
flexibility	time (variance of lead time/ rate)	low	indifferent	high
	quantity (variance / rate)	low	indifferent	high
	space (variance / rate)	low	indifferent	medium
reliability	adherence to schedules (level / rate)	high	medium	medium
	quality (level / rate)	high	medium	medium

 Table 3 - 8: SCP requirements profiles derived from SC strategy.

The following section introduces the structures and elements of the procurement, production and distribution concepts for different SC strategies. These considerations serve as the basis to identify the '*designable*' aspects of the SC for the targeted adaptation of the SC concept to enable performance-oriented CT integration.

3.3 Configuration of SC Concepts

This section discusses the elements and structures of SC concepts and derives the specific performance requirements regarding the embedded and linking transport concepts. In a first step, the general structures and elements of SC sub-concepts are introduced. In a second step, a structured description of SCP requirements regarding the procurement, production and distribution concepts of the embedded and linking transport concepts is developed. This approach is the basis for the discussion of different SC sub-concept configurations and the specific impact on the applicability of CT concepts. Therefore, it is assumed that product- and demand-related criteria are not alterable to influence the integration of a certain CT concept.

3.3.1 Transport Concepts as Linking and Embedded Elements of SC Concepts

This section addresses the structures and elements of transport concepts as linking and embedded elements of the procurement, production and distribution concepts. In the thesis in hand, transport concepts are interpreted as the central elements of SC concepts. They significantly influence overall SCP. Transport concepts must meet the SCP requirements of the procurement, production and distribution concepts. Thus, this section provides a general understanding of transport concepts by introducing the main strategies and structures. Therefore, it is necessary to take the transport operator's perspective regarding the transport concept- Usually, 3PLs or carriers are responsible for the linking and embedded transport processes of procurement and distribution concepts.¹⁴⁷ Outsourcing production internal transport processes is rather uncommon.¹⁴⁸

Structure and Strategies of Transport Concepts

Transport concepts accomplish the material flows within procurement, production and distribution concepts. Often, the transport concept is part of a wider logistics concept (cf. section 3.1). Transport planning encompasses different sub-problems. Fleet planning means the optimal composition of the vehicle fleet regarding vehicle type, capacity and number. Vehicle action planning is based on a given vehicle fleet and considers the optimal allocation of vehicles and customers.

A *transport strategy* is the operation strategy of transport concepts. It determines how the transport control unit steers the waiting transport orders, guides the transport vehicles to the destinations and controls the processing at the different sites in a SC concept.¹⁴⁹ According to Gudehus (2005), the choice of the optimal transport strategy is often the best way to optimize a logistics concept.¹⁵⁰ Usually, the adaptation of the transport concept is faster and cheaper than is investing in new technological solutions. Transport strategies influence the complexity of the transport network, the performance of the existing transport system, the number of vehicles and the functionality and safety of the transport system.

¹⁴⁷ cf. Stölzle et al. (2011), pp. 223. For details on the concept of 3PLs see Reichhart et al. (2007); Carbone et al. (2005); Ellram et al. (1990); Jayaram et al. (2011), p. 411; Sennheiser et al. (2008). For the meaning of the 3PL concept for the integration of CT concepts see for instance Sheffi (1990) and Bendul et al. (2009).

¹⁴⁸ cf. Stölzle et al. (2011), pp. 223.

¹⁴⁹ cf. Bobel et al. (2009), p. 823.

¹⁵⁰ cf. Gudehus (2005), p. 823.

Gudehus (2005) suggested the classification of transport concepts according to different types of transport strategies.¹⁵¹ The *station strategy* determines the way to process waiting transport orders at different sites for loading and transport to the destination. Processing can be accomplished according to a defined sequence, for instance the first-come-first-served rule. This means that arriving transport orders are processed in their sequence or on arrival. Furthermore, the different priority rules (e.g., for certain products or destinations and rush orders) are applied. By applying a free processing sequence, all orders for one destination or driving direction are gathered (independent of their arrival sequence) until the capacity limit or a scheduled time of departure is reached.¹⁵²

The processing of transport orders according to a defined sequence, such as firstcome-first-served, is rather inappropriate for accomplishment by CT. The unforeseeable sequence of transport goods and destinations prevents the usage CT because of the missing time in advance for the dispatching of rail transport and transhipment processes. This means that in a free processing sequence the bundling effects support CT integration, especially in combination with a transport schedule, since here the necessary planning basis is given with defined destinations and transport times.

Loading Strategy

The bundling of loads improves capacity usage and thus, reduces costs, but it means additional waiting times.¹⁵³ There are two different *strategies for the loading of transport vehicles*: the one-destination and mixed-destination strategy. This means either the transport vehicle loads only goods for one destination or for multiple destinations on the same transport route. Furthermore, transport without and with multiple loadings are distinguished. Either only the completely empty transport vehicle is reloaded or the transport vehicle is reloaded on the transport vehicle is reloaded on the transport vehicle is reloaded on the transport vehicle.

Generally, the loading strategy does not influence the applicability of CT concepts as long as rail transport is accomplished with complete loading. However, depending on the situation-specific SC concept configuration and SCP requirements regarding the

¹⁵¹ cf. Ibid., p. 824. Gudehus uses the term of 'transport strategy'.

¹⁵² cf. Ibid., p. 823.

¹⁵³ cf. Ibid., p. 824.

¹⁵⁴ cf. Ibid., p. 824.

transport concept, it can be assumed that the fewer the stations on the transport route the more time buffers can be accessed for the integration of CT into the specific SC concept.

Transport Route Strategy

The *transport route strategy* determines the *sequence of destinations* for a certain transport vehicle and thus, sets the transport route. For instance, the transport route strategy depends on the shortest transport distance, the shortest transport time or lowest transport costs. In addition, strategies aiming at a high level of capacity usage are applied. Schedule strategies determine the transport processes according to a determined transport schedule. Usually, a transport schedule is configured to minimize transport distances and maximize capacity utilisation. For transport systems with only a few relations the driveways and roundtrips with the shortest distances, driving times or lowest costs can be found by full numeration and the comparison of all possible solutions. For complex transport networks, this optimisation is usually not possible. Operation research has developed different heuristic approaches for the approximation of the optimal solution.

Following the considerations above, the availability of transport schedules supports CT integration depending on the situation-specific SCP requirements regarding the transport concept. Transport concepts without a predefined sequence of transport sites prevent the integration of CT because of the missing planning time in advance. The transport structure can be understood as the counterpart of the loading strategy from a shipper's or from the good's perspective.

Transport Structure

Pfohl (2005)¹⁵⁵ classified different transport concepts according to the structure of the transport chain. An unbroken transport chain is accomplished without changing the transport vehicle and directly joins the source and destination. FTL means that the entire shipping space is filled. LTL means that parts of the shipping space are booked by different shippers.¹⁵⁶ Although this notation implies the reference to road transport, these considerations regarding full and partial loading can be transferred to other modes. Rail, water and air cargo transport can accomplish stopovers to load or

¹⁵⁵ cf. Ibid., pp. 164.

¹⁵⁶ This differentiation is common for all transport modes.

deload shipments. Figure 3 - 2 summarises the considerations in a schematic figure.

A so-called broken transport chain means that the shipper and consignee change the transport vehicle. According to Pfohl (2005), this can be termed CT in the broadest sense.157



Figure 3 - 2: Transport structure for full and less than full truckloads.¹⁵⁸

Relation transport is defined as direct FTL transport between shipper and consignee without transhipment processes. This type of delivery can, for instance, be for found in JIT and JIS concepts with high transport frequency and high transport vehicle utilisation.¹⁵⁹ Often, the relationship between returned empties and goods of 1:1 has defined delivery windows. A relation transport corresponds to a transport route strategy with only one destination.

The *milkrun concept* is a specific type of the relation transport concept for distribution and procurement concepts. Usually, within defined time windows partial loads are bundled on a defined transport route and delivered to the consignee without transhipment processes. Additionally, the concept includes the return of empty load carriers. Likewise, the milkrun concept can be applied to the distribution of goods starting from a central shipper. This allows the improvement of the capacity utilisation of transport vehicles and an increase in transport frequency. Thus, the concept supports the short-term transport optimisation and reduction of transport costs.¹⁶⁰ A milkrun concept corresponds to a transport route strategy with several destinations.

Packaged goods transport is a type of broken transport since here more than one transport vehicle is used. Figure 3 - 3 shows the schematic structure of a transport network for packaged goods transport.¹⁶¹

¹⁵⁷ cf. Pfohl (2010), p. 165.

¹⁵⁸ cf. Ibid., p. 11.

¹⁵⁹ cf. Müller et al. (2009), p. 29; Iskan (2009), p. 626, Schorb et al. (2007), p. 10. 160

cf. VDA (1997), p. 31; Iskan (2009), p. 2.

¹⁶¹ For details different freight or transport networks see for instance Gudehus (2005), pp. 956; Alicke (2003).



Figure 3 - 3: SC structure for a packaged goods transport.¹⁶²

Pfohl (2005) also distinguished between broken transport in the proper meaning of the word, namely including a change of load carrier and often including storing and further handling processes, and CT, namely without a load carrier change. ¹⁶³ However, the term 'CT' is in this publication not restricted to the combination of road and rail transport. Here, the author distinguishes between piggyback services and container services.¹⁶⁴

For the integration of CT services, the transport structure is central. Different structures require different CT concept types. CT concept types with regard to the structure of material and information flows are introduced in section 3.4.

3.3.2 Impact of Procurement Concepts on the Configuration of Transport Concepts

This section clarifies the elements and processes of procurement concepts to identify adaptation points for the performance-oriented integration of CT into SCs. In a first step, the meaning of the procurement concept for SCP is worked out. In a second step, the structures and processes of the procurement concept are described. Furthermore, the impact of the SC strategy on the design of the procurement concept is discussed. Finally, classification criteria for procurement concepts are discussed and criteria influencing the SCP requirements regarding the embedded and linking transport concepts are identified.

¹⁶² cf. Wildemann et al. (2011), p. 12.

¹⁶³ cf. Müller et al. (2009), pp. 164.

¹⁶⁴ For further details on the structure and characteristics of CT services see section 2.2.
3.3.2.1 Meaning of the Procurement Concepts for SCP

The procurement concept defines the planning, control, implementation and monitoring of all activities related to procurement and determines the corresponding structures, resources and management processes.¹⁶⁵ Recently, procurement processes have gained strategic meaning. This has resulted from the dependency of production and distribution concepts on the interruption-free supply with goods.¹⁶⁶

The procurement concept significantly influences the cost structures of manufacturing and retail companies. Often, the share of purchasing costs is more than 50% of a product's total costs.¹⁶⁷ *Strategic procurement management* aims at the development of capabilities as well as procurement activities by means of procurement market investigation, procurement controlling and benchmarking.¹⁶⁸ The thesis in hand focuses on *operative procurement management*.¹⁶⁹ This encompasses the recognition of demand, identification and choice of suppliers, order processing and controlling and accounting.¹⁷⁰

The *target system* of procurement concepts depends on the procurement object and the resulting requirements. General targets are cost reduction, differentiation and the security of supply.¹⁷¹ *Cost reduction* encompasses the reduction of acquisition prices, transport, warehouses, provisions, processes, overhead costs as well as material costs. This also includes debt service costs as well as out-of-stock costs.¹⁷² *Differentiation* means the improvement of product and process quality,¹⁷³ for instance by standards for suppliers, the improvement of service levels and lead times as well as the participation in the supplier's knowhow and image. The *security of supply* focuses on the increase of adherence to schedules, a high level of delivery availability,¹⁷⁴ distribution of risk and increase in the market share of the purchasing market. Increasingly, sustainability and environmental orientation influence the target system of procurement. As a result, companies increasingly choose suppliers that are ecofriendly.¹⁷⁵ Further targets include the supply-based diversification of the protection of inde-

- ¹⁷¹ cf. Ibid., pp. 66.
- ¹⁷² See Ibid., p. 32. ¹⁷³ See for instance Pi

¹⁶⁵ Following the definition of Pfohl (2010) and the definition of SC concepts (section 3.2.1).

¹⁶⁶ cf. Cooper et al. (1997); Ursel et al. (2010).

¹⁶⁷ cf. Koppelmann (2002).

¹⁶⁸ cf. Ursel et al. (2010), pp. 56; Arnold (1997), p. 32.

¹⁶⁹ cf. Piontek (1994), p. 96.

¹⁷⁰ cf. Hamm (1997), pp. 99.

¹⁷³ See for instance Piontek (1994), p. 32.

cf. Grimm et al. (2010); Peters (2010); Piontek (1994), p. 32.
 See fer instance Piontek (1004).

¹⁷⁵ See for instance Piontek (1994).

pendence and flexibility as well as the increase in vertical integration.¹⁷⁶

The procurement concept must be carefully adjusted and linked with the other distribution, production and procurement concepts in the SC. This ensures the stable and reliable supply of the production concept with raw materials and supplies, parts and externally manufactured goods. The procurement concept is central for each SC concept because it ensures the operability of the production systems and thus that end customers are supplied with goods.

The procurement concept is concretised at the material flow level by transport and logistics processes. Transport processes convert purchasing processes into specified material flows for the production concept's supply or direct distribution. Furthermore, warehousing, transhipment and additional logistics services have a special importance.

The following section builds on these general findings and introduces the basic structures and processes of procurement concepts. The section aims at a general understanding how the configuration of the procurement concept influences SCP requirements regarding the embedded and linking transport concepts.

3.3.2.2 Structures and Processes of Procurement Concepts

Procurement encompasses all corporate activities for the supply of goods necessary to realise corporate targets. Procurement markets obtain services, production facilities (fixed capital goods), materials (raw materials and supplies, semi-finished goods) and trade goods. Usually, these tasks are accomplished by a purchasing department.¹⁷⁷

Applied standardised concepts include the concept of industry parks to increase the proximity between buyer and supplier.¹⁷⁸ The SRM concept aims at the processoriented evaluation and choice of suppliers from a customer's point of view. The evaluation of the relationship is accomplished according to supplier lifetime value.¹⁷⁹ The CPFR concept focuses on the improvement of efficiency and effectiveness by means of standardisation and improved planning and forecasting.¹⁸⁰

¹⁷⁶ cf. Kudla et al. (2011), p. 32.

¹⁷⁷ cf. Piontek (1994), p. 343.

For details on the concept of industry parks see for instance Das (2010); Hao et al. (2008); Lechner et al. (2008); Patil et al. , p. 405.

For details on the concept of SRM see for instance Fricker (2008); Koch et al. (2008), p. 387, 391; Sennheiser et al. (2008).

¹⁸⁰ For details on the concept of CPFR see for instance Esper et al. (2003); Sari (2008); Sennheiser et al. (2008), p. 385.

The procurement process can be divided into seven sub-processes to provide the required materials and goods in the demanded quality and quantity at the desired site and time point and at competitive prices. (1) The production and distribution departments (in retail) give a material requisition to the purchasing department. After the (2) supplier makes a choice, an (3) order is made. The goods are (4) transported and (5) received by the purchasing company. The delivery is (6) confirmed and the accounting department (7) verifies the invoice and initiates the payment transaction.¹⁸¹

The *procurement strategy* classifies procurement goods according to supply risk. This risk definition encompasses the number of suppliers, substitute possibilities, risks of storage as well as possibilities of in-house production. Furthermore, an evaluation of the procurement volume, namely using an ABC analysis, is accomplished.¹⁸² Within a four-field matrix, four different procurement product types can be distinguished. From these a general procurement strategy can be derived.



Figure 3 - 4: Generic procurement strategies according to supply risk and profit impact.¹⁸³

Products with high purchase volumes and low supply risks, so-called '*leverage*' goods, are usually standardised objects purchased from a high number of suppliers. A supplier switch can be accomplished easily. To increase corporate success aggressive behaviour on the procurement market is typical to increase and stabilise demand

¹⁸¹ cf. Siebel (2002), p. 13.

¹⁸² cf. Arnold (1997), p. 133.

¹⁸³ cf. Ibid.; Gudehus (2005), p. 92.

power and to activate the competition on the procurement market (active procurement marketing).¹⁸⁴

The procurement of *goods with low volumes and low risks*, so-called *'non-critical'* goods, for instance standard parts, which are technically mature and standardised,¹⁸⁵ should be cost efficient and require low resources. Adaptation points for these types of procurement goods are the reduction of order processing, the centralisation of the purchasing function or the outsourcing of procurement activities.¹⁸⁶

Procurement goods, characterised by *low procurement volumes and high supply risks*, so-called *'bottleneck'* goods, should focus on the stabilisation of supply, either by warehousing or by the substitution of products or new suppliers. Usually, there is a restricted number of suppliers. However, the resource intensity to handle these types of goods should be restricted because of the low procurement level.¹⁸⁷

For products with *high procurement volumes and high supply risks*, so-called *'strate-gic' goods*, a strategic solution is required that focuses on the vertical and technology-based cooperations with selected suppliers. These selected suppliers are integrated into research and development activities, too. The resulting long-term partnership leads to a reduction in planning and disposition effort for the procurement of goods.¹⁸⁸

The integration of CT simplifies as the higher and more stable transportation volumes become. Thus, CT solutions are initially suitable for 'leverage' and 'strategic' goods. However, the procurement strategy for strategic products is characterised by diversification, exploitation and balancing activities in order to increase the independence on single suppliers and increase the security of supply. Thus, the required transport processes are rather unstable and change regarding relations and transport quantities. Thus, the suitability for the integration of CT for the procurement of strategic goods is restricted. 'Leverage' goods with high transport quantities and low supply risks, and thereby rather stable supplier relations, are most suitable for CT integration.

The specific configuration of the procurement concept is significantly influenced by product type, market demand and SC strategy. The following section clarifies the

¹⁸⁴ cf. Canie et al. (2005), p. 92.

e.g., connection elements, semi-finished products, stationary, paper etc.
 ¹⁸⁶ of Arroald (1007) p. 01

¹⁸⁶ cf. Arnold (1997), p. 91.

¹⁸⁷ cf. Ibid., p. 92.

¹⁸⁸ cf. Ibid., p. 92.

relationship between the SC strategy and procurement concept configuration.

3.3.2.3 Classification and Configuration of Procurement Concepts – Identification of Design Variables relevant for CT Integration

This section addresses the classification criteria for procurement concepts with regard to the integration of CT. In particular, the impact of different classification criteria on SCP requirements regarding the CT concept is discussed.

There are multiple approaches to the classification of procurement concepts.¹⁸⁹ However, for the thesis in hand it is critical to identify not only classification criteria, but initial points for the performance-oriented integration of CT into SC concepts. This means so-called *'design variables'* have to be identified. These criteria can be configured independent of product type and SC strategy. They are starting points to adapt the SC concept and to change SCP requirements with regard to the embedded and linking transport concepts.

Arnold (1997) identified procurement subject, procurement type, product characteristics, product value, aggregate phase, market size and demand predictability as criteria for the classification of different procurement concepts.¹⁹⁰ However, in the thesis in hand, it is assumed that product- and demand-related characteristics are not changeable and thus, not suitable for the adaptation of SC concepts with regard to CT integration.

Furthermore, Arnold (1997) identified supplier choice and number of sources, procurement area, procurement time and delivery agreements, namely the agreed incoterms¹⁹¹ as classification criteria for procurement concepts.¹⁹² These criteria are relevant for the integration of CT into SC concepts.

Table 3 - 9 gives an overview of the classification criteria that are relevant to the integration of CT into SC concepts.

¹⁸⁹ cf. Ibid., p. 92; Meyr et al. (2005).

¹⁹⁰ cf. Stadtler (2005), p. 93. The specific characteristics often correspond to one of the general procurement strategies.

¹⁹¹ cf. Arnold (1997); ICC (2010b).

¹⁹² cf. Gabler (2011), p. 96.

classification criteria	characteristics					
supplier choice	sole	single		dual		multiple
procurement area	local		domestic		global	
procurement time	stock sourcing		demand tailore	d sourcing	single	sourcing
incoterms	see Figure 3 - 5					

 Table 3 - 9: Classification criteria for procurement concepts with regard to the impact on transport concepts.¹⁹³

The impact of the different classification criteria on the embedded and linking transport concepts is discussed in the following. In a first step, the classification criteria and typical characteristics are introduced. In a second step, the impact of different characteristics on the seven identified SCP requirement categories developed in section 3.2 is clarified.

Supplier Choice

Supplier choice influences the number and locations that must be covered by the transport concept.¹⁹⁴ Furthermore, it influences the total inventory level and can also affect SCP depending on the reliability of the supplier. Supplier choice can be utilised to influence SCP requirements. Thus, it directly influences time-related performance aspects, namely delivery windows, lead times and delivery frequency depending on the location and reliability of the supplier. Total transport distance, distance to the infrastructure connection (e.g., the next highway or CT terminal) and number of sites on the transport route are also influenced by supplier choice. Depending on the location and delivery frequency, shipment size may be affected, too. These aspects also influence transport costs. Arnold (1997) stated the importance of the SC concepts' flexibility (regarding time, transport quantities and space) and reliability regarding adherence to schedules and product quality.¹⁹⁵

In general, the fewer suppliers there are in the procurement concept, the higher are transport volumes and bundling effects. Thus, depending on the situation-specific configuration of the SC concept a low number of suppliers supports the integration of CT. However, smaller and irregular transport volumes with rather long lead times could be delivered by CT networks.

¹⁹³ Compilation in accordance to Arnold (1997), pp. 93; Ibid., pp. 311.

¹⁹⁴ Supplier choice corresponds to the procurement strategy introduced in the previous section.

¹⁹⁵ cf. Arnold (2002), pp. 93; Arnold (1997), p. 311.

Procurement Area

The *procurement area* is of special meaning for transport mode choice. Global suppliers require long distance transport modes such as ocean carriers or air cargo. Usually, rail and road transport are common for local and domestic suppliers. Local sourcing means that a spatial proximity to the supplier is assumed. Typical examples are industry parks¹⁹⁶ or the in-house production of suppliers.

Economic advantages of CT can be realised the better, the longerlong transport distances are. Thus, domestic and global procurement areas (ongoing haulage after main haulage by sea or air transport) are in particular suitable for CT integration. The integration of CT into a distribution concept with local procurement area depends on the extent of the area and distance to the customer.

Procurement Time

Procurement time is central to the available advance of time for the physical procurement process. Stock sourcing aims at the realisation of economies of scale and security of supply and is especially suitable for parts with low value and high transport quantities. This type requires singular, but huge transport and logistics capacities. Demand tailored sourcing is driven by customer orders and thus, is often unspecific. It means that a coordination of supplier and buyer is necessary. The introduction of regular transport with huge transport volumes is often not possible. Singular sourcing is characterised by high and regular demand.¹⁹⁷ Here, rolling demand plannings are applied. Therefore, the customer-oriented product is ordered immediately before delivery, but the transport concept can be configured based on a stable basic demand.

A long procurement time and procurement from stock positively influence the applicability of CT. However, the variant of the product does not necessarily influence transport. Thus, in demand tailored sourcing the integration of CT is also possible if transport quantities and relation can be sufficiently predicted. In a JIT or JIS concept, the implementation of CT is complicated because of short lead times and the requirements for high transport frequencies.

¹⁹⁶ For details on the concept of industry parks see Arnold (2002); Das (2010); Hao et al. (2008); Patil et al., p. 405.

¹⁹⁷ Often here the standardised JIT concept is applied.

Incoterms

'Incoterms' is the abbreviation for 'international commercial terms'. The International Chamber of Commerce (ICC) publishes a series of predefined commercial terms for international commercial transactions.¹⁹⁸ The terms serve to clearly communicate the tasks, costs and risks of transport and delivery. Although incoterms were introduced for cross-country transport, the following considerations can be transferred to delivery agreements within one country.

The target of usage is the reduction of uncertainties resulting from different interpretation of such terms in different countries. Incoterms were first published in 1936. They are updated periodically. The eighth version was published in January 1, 2011. Figure 3 - 5 gives an overview of the latest incoterms regarding documentation, risks and cost agreements (version 2010).



Figure 3 - 5: Overview of incoterms 2010.¹⁹⁹

Recently, there have been 11 incoterms. Figure 3 - 5 shows these incoterms in ascending order of the responsibility of the vendor. The usage of incoterms is voluntary and requires no contractual agreement. In practice, most commonly used incoterms are 'ex works' (EXW), 'free on board' (FOB), 'cost, insurance and freight '(CIF) and

¹⁹⁸ cf. ICC (2010a).

¹⁹⁹ cf. ICC (2010b); (illustration according to http://www.rc-commodity.com/info-center/incoterms.html).

'delivered duty paid' (DDP).²⁰⁰

Incoterms directly influence the cost distribution in the SC. This includes the cost of risks, transport, handling and warehousing as well as customs, duties and taxes. Incoterms thus indirectly impact all other SCP requirements since the distribution of responsibilities includes the decision of the operator SC. Thus, it can be assumed that the change in incoterms may especially influence time flexibility as well as reliability regarding product quality, since the risk is variously distributed.

The agreed incoterms do not directly influence material flow processes, but rather the reliable actor for the transport, the property situation and the risk allocation, costs and documentation obligations. In general, depending on the SC concept configuration, an incoterm that shifts reliability for the transport processes to one actor, such as EXW (customer reliable for the transport) or DDP (supplier reliable for the transport) increases the predictability and the planning basis for the integration into SC concepts. However, depending on the SC structure and transport distance a CT concept can also be integrated for the pre- or ongoing haulage for the other incoterms with shared reliabilities, such as FOB (free on board; delivery to an agreed port) or DAF (delivered at frontier; delivery to an agreed delivery point on a frontier).

Table 3 - 10 summarises the findings on the impact of procurement concept configurations on SCP requirements developed in section 3.2. The product type is excluded in this illustration since it is assumed in a SC concept.

²⁰⁰ cf. ICC (2010a); Gabler (2011); ICC (2010a).

reading direction		impact of procur	ement concept config	guration	
		supplier choice	procurement area	procurement time	incoterms
SCP requirements					
time	delivery window	Х	Х	Х	
	transport time	Х		Х	
	delivery frequency	Х		Х	
space	transport distance	Х	Х		
	distance to infrastructure	Х			
	number of sources and destinations	Х			
shipment size and		Х		Х	
transport quantity					
cost	transport cost/ turn- over	Х	X	Х	Х
flexibility	time	Х	Х	Х	Х
	quantity	Х		Х	
	space	Х	Х		
reliability	adherence to schedules	Х	Х	Х	
	quality	X	X	Х	Х

 Table 3 - 10: Impact of procurement concept configuration on SCP requirements.

Next to the procurement concept, the production and distribution concepts have specific SCP requirements regarding transport concepts. The following section discusses the meaning of the production concept on the integration of CT.

3.3.3 Impact of Production Concept Configuration on Transport Concepts

This section clarifies the elements and processes of the production concept to identify adaptation points for the performance-oriented integration of CT into SC concepts. In a first step, the meaning of the production concept for SCP is worked out. In a second step, the structures and processes of production concepts are described. Furthermore, the impact of the SC strategy on production concept configuration is discussed. Finally, classification criteria for production concepts are identified and the SCP requirements regarding the embedded and linking transport concepts are worked out.

3.3.3.1 Meaning of the Production Concept for SCP

According to Wildemann (2004), modern production concepts can be characterised by the orientation towards the main objectives of the company and the SC. The segmentation of production and network organisation are the central ways to deal with the increasing complexity of products and production systems.²⁰¹ A production concept encompasses the definition of all planning, control, implementation and monitoring processes for the production and assembly as well as the determination of the corresponding structures, resources and management processes.²⁰²

Transport concepts are central to the operation of the production concept. They link the production concept with the upstream processes of the procurement concept and with the downstream processes of the distribution concept. Furthermore, the internal transport of raw materials, parts and modules to specific facilities and to buffer and warehouses is accomplished. These tasks can be defined as production logistics in a narrower (internal processes) and wider sense (external processes).²⁰³

The following section introduces the basic processes, elements and objectives of production concepts to understand how it influences SCP as well as the configuration of transport concepts.

3.3.3.2 Structures and Processes of Production Concepts

Production can be defined as the manufacturing and altering of goods and services by the aid of other goods and services.²⁰⁴ Thus, the production concept must meet the SCP requirements resulting from a changing market environment. A high adaptation speed to the environment is necessary to reach a high level of market and customer orientation.²⁰⁵ Hence, a production concept's target system focuses on the optimal use of resources with regard to the triad of costs, quality and time completed by the objective of flexibility. In brief, the general target of a production concept is to increase productivity, decrease lead and production times, reduce production costs (material cost, personnel, facility cost, etc.) and improve product quality.²⁰⁶

²⁰¹ cf. ICC (2010b), p. 1.

²⁰² Definition developed on the basis of Wildemann (2004,), pp. 295.

²⁰³ cf. Ibid., pp. 295.

²⁰⁴ cf. Arnold et al. (2008); Kern (1979), p. 5;

²⁰⁵ cf. Wildemann (2004a), p. 1.

²⁰⁶ cf. Ibid., p. 295.

The production process can be distinguished according into (1) PPC processes, (2) the internal transport of materials and work pieces (between different production steps), (3) the part/components or system manufacturing and (4) the component/system pre- and end assembly.²⁰⁷

PPC is a central element of the production concept. The optimal supply of production with resources according to time, quality and flexibility aspects is the main target of PPC. Usually, powerful software applications support PPC (so-called 'PPC systems').²⁰⁸

The specific configuration of the production concept is significantly influenced by product type, market demand and SC strategy. The following section clarifies the relationship between SC strategy and production concept configuration.

3.3.3.3 Classification and Configuration of Production Concepts – Identification of Design Variables relevant for CT Integration

This section introduces different classification criteria for production concepts. The criteria that influence the SCP requirements regarding transport concepts are identified and discussed.

Sames and Büdenbender (1997) proposed a morphological scheme to describe different order fulfilment strategies in production concepts.²⁰⁹ Usually, the morphological scheme is applied to a first analysis and documentation of a given production concept as well as for the description of a prospective future state. It is restricted to the manufacturing of piece goods.²¹⁰ The authors distinguished between classification criteria regarding the initialisation of order fulfilment activities (e.g., order release), product characteristics (e.g., product portfolio, product structure, product design), disposition characteristics (e.g., release of secondary order demand, stocking type) and manufacturing process characteristics (e.g., production control logic).

Several of these classification criteria depend on product type, market demand or SC strategy and are thus, not considered as design variables for the performance-oriented integration of CT into SC concepts.

²⁰⁷ cf. Arnold et al. (2008), pp. 27.

²⁰⁸ cf. Hackstein (1989), p. 262; Luczak et al. (1999), pp. 250; Wiendahl (2009), p. 1.

²⁰⁹ cf. Wiendahl (2009)`.

²¹⁰ In contrast to process manufacturing. cf. Sames et al. (1997), p. 77. For further considerations, process manufacturing is excluded.

Additionally, it is assumed that production $type^{211}$ and assembly type are not alterable, but mainly depend on product type and market demand for the purpose of CT integartion.²¹² Table 3 - 11 shows the classification criteria for production concepts that influence the transport concept configuration.

classification criteria	characteristics					
order release	production on demand (single orders)	n demand production on demand (frame orders)		customer anonymous pre-production		production on stock
secondary demand release	order oriented		partly order orien period-oriented	nted, partly	period-	oriented
stocking type	no stocking of de- manded goods	stocking of demands of low structure levels		stocking of demands on high structure levels		stocking of goods
production control logic	pull-control logic			push control log	ic	

 Table 3 - 11: Classification criteria for production concepts with regard to the impact on transport concepts.²¹³

Order Release

'Order release' characterises the linking of the production and sales market. The central dimension for differentiation is the release type of primary demand. This release can be initiated by customer orders (production on demand with single orders or framework orders), by sales expectations (production on stock) or by a combination of these two basic order release types. In the case of a combination, certain product components are pre-manufactured based on sales expectations, which are then completed or end assembled after customer orders occur.²¹⁴ Thus, the order release type influences time-related performance requirements as well as time-related flexibility aspects. Furthermore, the type of order release influences the shipment size and transport quantities as well as the requirements regarding the corresponding flexibility.

Transferring these considerations to CT integration implies that because of the high production quantities, stable production and resulting long lead times a good planning basis positively influences the applicability of CT into a SC concept.

The 'production type' characterizesises the frequency of the repetition of production processes. cf. Luczak et al. (1999).
 The 'sourcing type' determines the ratio of internal and external sources goods. The higher the number of outsourced production processes the higher the number, quantity and frequency of transport processes into the focal company location. cf. Sames et al. (1997); Luczak et al. (1999). The sourcing type is excluded from considerations here, since the field of procurement and sourcing comprehensively discusses the impact of procurement concept configuration on transport concepts (cf. section 3.2.2).

²¹³ own illustration in accordance to Arnold et al. (2008); Kern (1979); Luczak et al. (1999), p. 110; Sames et al. (1997), pp. 301.

²¹⁴ cf. Arnold et al. (2008); Kern (1979); Luczak et al. (1999), p. 110, Sames et al. (1997), pp. 301.

Release of Secondary Demand

The dimension of the 'release of secondary demand' is tightly connected with the demand determination of products and components. The dimension describes the connection to the procurement concept. The procurement concept must ensure the production concept's material demand. Thus, the more regularly the demand for the products can be determined with regard to time and quantity, the more regularly the secondary demand can be specified. The secondary demand determination can be order-oriented, period-oriented or a combination of both. Period-oriented demand determination means the secondary demand is gathered and determined for several orders for a defined time period. The combined and the period-oriented demand determination is especially applied in companies with a wide spectrum of product variants. Standard components in several orders can be gathered and commonly sourced. The applied transport concepts are influenced not only by the production quantities, but also by the available advance time to set up the necessary transport processes.²¹⁵ The release of secondary demand influences time-related SCP requirements as well as the required time flexibility. Owing to the impact of secondary order release on inventories, it influences capital commitment and warehouse costs.

The higher the share of period-oriented secondary order release the better the applicability of CT concepts. Owing to the periodic character, the required predictability for the integration of CT concepts is assumed. However, further situation-specific characteristics must be included in the consideration, especially transport planning for the delivery of orders. The character of the transport concept (periodic- or demand-oriented) can differ from the release of the secondary demand.

Stocking Type

'Stocking type' describes the scope of stocked parts, assembly groups and goods. For the production of customer-specific goods, for instance in special engineering or plant manufacturing, stocking is usually not possible. In SCs with a focus on short delivery times,²¹⁶ delivery times are often shorter than in a manufacturing process. Thus, there must be stocks for a customer anonymous pre-manufacturing of standard

²¹⁵ In the original scheme, two dimensions – the determination of product/component demand and the determination of secondary demand – are distinguished. The content is here summarised for the focus on transport and logistics processes. The way and the structural level of demand determination are used to distinguish five types of demand determination.

²¹⁶ cf. section 3.1.3: lean II - continuous replenishment or agile SCs.

components. This stocking can be restricted to single demand positions, such as parts and assembly groups or can encompass readily manufactured and assembled products.²¹⁷ The stocking type influences the requirements regarding transport time and delivery frequency. The lower the stock levels the higher the shorter transport times and delivery frequency and the smaller the shipment size as well as the corresponding flexibility aspects. Furthermore, there are requirements on reliability regarding adherence to schedules as well as product quality.

It can be summarised that stocking goods generally supports the integration of CT concepts. If goods are stocked, lead times are comparably long, bundling effects can be realised and thus, transport quantities are presumably high.

PPC Logics

There are two basic 'PPC logics'. In a so-called push system, the work is scheduled and order releases are based on demand. In a pull system, the work is authorised and order releases are based on the system's status.²¹⁸

Push systems are inherently due date-driven. Products are 'pushed' into production and are taken to the next work centre on schedule. 'Push' thereby releases materials into production as customers' orders are processed and the materials become available. The system has centralised planning and there is aligned information and material flows only between the predecessor and successor working centres. Material Requirement Planning or Manufacturing Resource Planning (MRP), Optimised Production Technology (OPT) and order release with load limitation are typical 'push' systems.²¹⁹

Pull systems are inherently rate-driven. Products are pulled through production. Thus, it is a control measure to release materials into production only when they are needed. Typical 'pull' systems' control strategies are ConWiP (Constant Work in Progress)²²⁰ and Kanban.²²¹ This pull principle is also used to coordinate supply relationships, such as the synchronised supply concepts of JIT and JIS.

²¹⁷ Common for simple standard products, which are customer anonymously sold.

²¹⁸ For the planning and control of production orders several more approaches and methods have been developed. For an comprehensive overview see for instance Sames et al. (1997). Some of them were implemented in commercial PPC-systems, but in practice simple methods like forward and backward scheduling as well as priority rules are commonly applied. cf. Teunis (2003), p. 33. 219 cf. Teunis (2003); p. 335; Wiendahl (2009), p. 65.

²²⁰

cf. Luczak et al. (1999) The strategy of ConWiP stabilises the inventory in a production system. The inventory is measured either by the number of parts or by the number of orders in the system. As soon as the inventory in a production system has reached the target inventory a new order is only 'pushed' into the system every time another order leaves it.

²²¹ cf. Lödding (2001), pp. 95, pp. 66; Luczak et al. (1999); pp. 342.

Relevant for the configuration of transport concepts are especially the impact of the PPC logic on the inventories, transport quantities, transport times as well as pickup and delivery times and the definition of the specific adherence to schedules. In production systems with a push control logic, transport concepts must be adapted to the hazard of huge stockings caused by planning defects. In a pull controlled production system, the immediate processing of orders requires low inventory levels, which require a high adherence to schedules and quality in transport processes. The production control logic especially influences the requirements on delivery windows. Furthermore, shipment size as well as flexibility and reliability are influenced.

The availability of high transport volumes and a high level of predictability support the applicability of CT into a SC concept. In a push system, the level of predictability is comparably higher than it is in a pull system. However, for the configuration of a transport concept the type of product (or variant) is not necessarily important. The predictability of transport quantity and transport time is of central interest. Thus, the applicability of CT cannot be evaluated separately based on the PPC logic, but the entire SC concept configuration must be considered.

reading direction		impact of produ	iction concept config	uration	
Ļ	1	order release	release of secon- dary demand	stocking type	production control logic
SCP requirements					
time	delivery window	Х	Х		Х
	transport time	Х	Х	Х	
	frequency	Х	Х	Х	
space	transport distance				
	distance to infrastructure				
	no. of sources and destinations				
shipment size and		Х		Х	Х
transport quantity					
cost	transport cost/ turn- over		Х		
flexibility	time	Х	Х	Х	Х
	quantity	Х	Х	Х	Х
	space				
reliability	adherence to schedules			Х	Х
	quality			Х	Х

Table 3 - 12 summarises the findings on the impact of production concept configurations on SCP requirements developed in section 3.2.

 Table 3 - 12: Impact of production concept configuration on SCP requirements.

After the discussion of the different configuration elements to influence the SCP requirement on transport concepts, the meaning of the production concept configuration on the integration of CT is discussed.

3.3.4 Impact of Distribution Concepts on the Configuration of Transport Concepts

This section clarifies the elements and processes of distribution concepts to identify adaptation points for the performance-oriented integration of CT into SCs. In a first step, the meaning of the distribution concept for SCP is worked out. In a second step, the structures and processes of the distribution concept are described. Furthermore, the impact of the SC strategy on the design of the distribution concept is discussed. Finally, classification criteria for distribution concepts are discussed and criteria influencing the SCP requirements regarding the embedded and linking transport concepts are identified.

3.3.4.1 Meaning of the Distribution Concept for SCP

The distribution concept encompasses the definition of the planning, control, implementation and monitoring of all distribution processes as well as the determination of the corresponding structures, resources and management processes. Distribution means the efficient physical delivery of goods from the manufacturer to the customer.²²² The distribution concept links the supplier and the buyer. Thus, distribution planning focuses on customer services and distribution costs.

Distribution can be defined as the allocation of products from the manufacturer to the end customer.²²³ In particular, the distance between manufacturer and sales markets, fluctuations in demand, the balancing of production lot sizes, transport units and sales units as well as the assortment for the end customer or retail are influencing factors. The term 'distribution logistics' has an even wider scope; it encompasses the planning and control distribution nets.²²⁴ The general target of distribution concepts is to increase customer service and to decrease distribution costs (transport, packaging,

²²² cf. Wiendahl (2009), p. 9.

²²³ cf. Pfohl (2010), p. 198.

²²⁴ cf. Wahl (1999), p. 9.

stocking, personnel, rent, etc.).

Nowadays, sales markets and production locations are often geographically remote from each other. Huge distances have to be covered to deliver the product to the end customer. Furthermore, companies classify the sales market according to customer groups or sales regions, which resemble their market-specific characteristics, such as frequency, quantity and type of demanded goods. To meet the demand of all sub-markets often companies set up complex distribution systems with decentralised, multi-level stocks. Thus, central problems are the choice of the optimal structure of the distribution net, the design of I&C systems and technical systems of information and material flow processes.²²⁵

The distribution concept must be carefully adjusted and linked with production as well as with the embedded and linking transport concepts. The distribution concept is central for each SC concept since it ensures the provision of goods for the end customer and thus, ensures financial flow.

Transport and logistics processes ensure the material flow of distribution concepts. Transportation processes accomplish the spatial balancing task and logistics processes, such as stocking and transhipment, and balance demand fluctuations and viable lot sizes of production.

In the following section, the basic processes, elements and objectives of distribution concepts are introduced to understand how the distribution concept configuration influences SCP as well as the design and configuration of the embedded and linking transport concepts. The discussion of the requirements on the distribution concept's requirements on transport concepts is central to gain further insights into the performance-oriented integration of CT into SC concepts.

3.3.4.2 Structures and Processes of Distribution Concepts

The distribution concept is the SC's interface with the (end) customer. Thus, it has to reflect the end customers' requirements in terms of SCP. SCP-oriented key figures become transparent to the end customer through distribution. As a result, the end customer has to be integrated into SC concepts. Central standardised concepts for this

¹⁴⁷

²²⁵ cf. Ibid., p. 9.

integration include CRM²²⁶ and ECR.²²⁷

Distribution concepts bridge the spatial and temporal distances between the production and consumption of goods by means of logistics and transport processes.²²⁸ According to Siller (2011), the main tasks of distribution logistics are demand forecasting, inventory planning, inventory usage, warehouse filling, order processing as well as physical delivery.²²⁹ The three basic elements of a distribution system at the material flow level are warehouses, transport and inventories (warehouse stock and transport stock). The distribution concepts encompasses five main processes to ensure products are delivered from manufacturer to customer: (1) planning and construction of warehouse locations,²³⁰ (2) warehousing, (3) packaging, (4) transport and (5) administrative processes, such as order processing, invoice processing and sales processes.²³¹ Often, these tasks are also defined as *'distribution logistics'* in the literature.²³²

Distribution Structures

To understand the specific challenges of CT integration from the perspective of the distribution concept, a general understanding of the structure of distribution concepts is necessary. For systematic consideration, the structures of distribution systems can be affiliated to four basic structures. Pfohl (2010) distinguished three basic types of material flow: direct, indirect and combined material flow (Figure 3 - 6).²³³

A distribution structure with direct material flow from the manufacturer to the customer is called a one-level system. One-level systems allow the delivery without additional stocking and/or handling processes. In practice, small companies in the capital goods industry or companies with few, huge customers have one-level distribution structures.²³⁴ Often, 3PLs are responsible for the distribution of goods or operation of DCs.

If the geographical distance between source and delivery points is too long to ensure the defined delivery time, the setup of multi-level distribution systems is necessary.

For details on the concept of CRM see Ibid., p. 391; Choy et al. (2002), p. 387.

For details on the concept of ECR see Sennheiser et al. (2008), Siebel (2002), p. 391.

²²⁸ cf. Sennheiser et al. (2008), p. 9.

²²⁹ cf. Wahl (1999), p. 9.

There is a number of publications on quantitative location planning. See for instance Siller (2011), p. 17. f_{231}^{231} = $f_{231}^{$

²³¹ cf. Pfohl (2010), pp. 198. ²³² See Webl (1000)

²³² See Wahl (1999).

²³³ cf. Siller (2011), p. 6.
²³⁴ cf. Pfohl (2010), p. 10.

According to Pfohl (2010), these multi-level concepts are characterised by indirect material flows.²³⁵ The material flow is disrupted by at least one additional distribution site. The activity of such a disruption point can be either bundling (concentration / bundling point)²³⁶ or allocating goods (distribution point).²³⁷

Additional storage sites are implemented to supply sub-markets, in which a high service level is a competitive factor. By establishing geographical proximity between DCs and customer sites huge distances and average delivery times can be reduced. Furthermore, there are legal restrictions, such as night and weekend driving bans that require the introduction of additional storage and distribution levels.²³⁸

To determine the location of the distribution and storage points of specific goods, demand regularity and customer preferences are central. According to the idea of an ABC analysis, products with a high inventory-to-sales ratio should be made available to the customer. Products with lower demand should be stocked at a higher distribution level in order to decrease so-called edge assortment problems ('selective storage').²³⁹

A combined material flow features both direct and indirect material flows. Through the combination, key accounts can be supplied directly to ensure a high service level, while small customers can be supplied via multi-levels to utilise transport capacities sufficiently.²⁴⁰

²³⁵ cf. Wahl (1999), p. 201.

Delivery of small lot sizes from different production sites, gathering at the concentration point and transhipments on a new transport vehicle, for instance a vessel or wagon.
 For instance a control warshows that requires products from a circle production site, huffers these and compliand downstream.

 ²³⁷ For instance a central warehouse that receives products from a single production site, buffers these and supplies downstream distribution warehouses.
 ²³⁸ of Peopl (2010) p. 202

²³⁸ cf. Pfohl (2010), p. 202. ²³⁹ of Ibid. p. 11

²³⁹ cf. Ibid., p. 11.

²⁴⁰ cf. Ibid., p. 203.



Figure 3 - 6: Basic structures of distribution systems.²⁴¹

The structure of a distribution concept can be visualised by a multi-level net of knots and edges. This structure can also be called *'distribution net'*.²⁴² According to Fleischmann (1993), a distribution net can be defined as the flow of end products from the production site to the customer.²⁴³ This flow consists of transport, storage and handling processes. The net consists of production sites, centralised warehouses, regional warehouses, DCs, transhipment points and customers.

Often, different distribution channels are chosen for different types of customers.²⁴⁴ Direct deliveries of key accounts from centralised warehouses and production sites are common, since these customers offer sufficient transport quantities. Smaller customers are served by DCs.²⁴⁵

There are two central criteria to distinguish between different distribution networks. A vertical distribution structure means a distribution network structure from production sites and suppliers to centralised warehouses and distribution sites to the customers. A horizontal distribution structure means the number of warehouse sites per storage level.²⁴⁶

Common weak points are historically grown distribution structures leading to high inventory and transport costs. The low level of the process integration of logistics service providers leads to high inventory levels that could be reduced by higher in-

²⁴¹ cf. Wahl (1999), p. 11.

²⁴² cf. Ibid., p. 11.

²⁴³ cf. Fleischmann (1993); Wahl (1999), p. 12.

²⁴⁴ cf. for instance Changxian et al. (2009); Frazier (1999); Hua et al. (2011); Parment (2008); Zimmermann et al. (2009).

²⁴⁵ cf. Siller (2011), pp. 18; Wahl (1999), pp. 10.

²⁴⁶ cf. Wahl (1999), p. 11.

formation availability. Still, the differentiation of logistics services, especially for ebusiness business-to-customer, is not sophisticated.²⁴⁷ At the I&C level, the virtualisation of inventories and the implementation of fleet management systems as well as customer information systems can send a delivery note by email.²⁴⁸

The specific configuration of the distribution concept is significantly influenced by product type, market demand and SC strategy. The following section clarifies the relationship between SC strategy and distribution concept configuration.

3.3.4.3 Classification and Configuration of Distribution Concepts – Identification of Design Variables relevant for CT Integration

This section introduces different classification criteria for distribution concepts. The section identifies the criteria that influence the SCP requirements regarding transport concepts.

Hoppe (2002) proposed a classification for different distribution concepts. The author suggested customer structure, assortment width, shelf life and typical products as classification criteria.²⁴⁹ However, for the thesis in hand it is assumed that productand demand-related criteria cannot be adapted to influence the integration of a CT concept. Furthermore, Hoppe (2002) presented four classification criteria that influence the SCP requirements of transport concepts: (1) regional structure, (2) echelon structure, (3) delivery service and (4) delivery time point. Table 3 - 13 gives an overview of the transport-relevant configuration criteria for distribution concepts.

classification	characteristics							
criteria								
regional	several DCs per country		one DC per country		several countries		one central DC	
structure					per DC			
echelon struc-	direct material fl	al flow; indire		material flow,	combined - dire	ct	indirect material flow,	
ture	one level		multi le	evel	and indirect		multi-le	evel
delivery	'same day'	'next da	ıy' or	24 - 48 hours	24 - 72 hours	> 72	2 h	> 1 week
service		'next m	orning'					
delivery time	determined		flexible		windows		free	
point								

 Table 3 - 13: Classification criteria for distribution concepts with regard to the impact on transport concepts.²⁵⁰

²⁴⁷ cf. Siller (2011), pp. 18.

²⁴⁸ cf. Banerjee et al. (2007); Bose et al. (2008); Fabbe-Costes et al. (2008); Ursel et al. (2010).

²⁴⁹ cf. Hoppe et al. (2002), pp. 24.

²⁵⁰ cf. illustration with significant adaptations in accordance to Ibid., pp. 24.

Regional Structure

The classification is oriented to the *regional structure* of the distribution concept, which can be of decentralised, national, regional and central character.²⁵¹ This classification corresponds to the availability of more or less DCs in the distribution network or in a sales region. Regional structure influences transport time and delivery frequency. Based on this, the shipment size is influenced. In particular, spatial aspects, such as transport distance, distance to infrastructure connection and number of sources and destinations as well as flexibility aspects, are influenced by the regional structure of a distribution concept. As a result, the regional structure influences the cost dimension, too.

In general, the centralisation of goods, and thereby a regional structure with only one or a few DCs per country, supports the integration of CT concepts. It can be assumed that the greater the number of DCs the more transport quantities are split up. However, the specific SC concept configuration must be considered for a comprehensive evaluation of CT applicability.

Echelon Structure

The *echelon structure* corresponds to the distribution structure introduced in section 3.3.4.2. This characterises the number of distribution levels and explains whether the material flow in the distribution concept is direct, indirect or a combination of both (Figure 3 - 6).

The echelon structure gives information about the structure of material flows in the distribution network. In general, the fewer distribution sites in the network there are the higher are the bundling potential and transport volumes and thus, the better the CT applicability. However, as for the evaluation of the regional structure the specific SC concept configuration, transport quantities, relations and times have to be considered. The echelon influences transport time, delivery frequency and thus, shipment size as well as corresponding flexibility. Furthermore, the echelon structure affects spatial dimension, namely transport distance, distance to infrastructure connections and number of sites. Thus, the echelon structure also influences cost.

²⁵¹ cf. Ibid., pp. 13, 24.

Delivery Service

The *delivery service* determines how far in advance the order for a specific delivery needs to be placed. For instance, in the medical and pharmacy industry, 'same day' delivery is common. 'Next day' and 'next morning' deliveries are usual in retail as well as for spare parts. Delivery times of one or more days, weeks or months are possible depending on the types of goods, the SC strategy and regional structure.²⁵² The delivery service especially influences transport time and delivery frequency as well as the corresponding flexibility and adherence to schedules. Thus, delivery time also affects costs.

As shown in section 2.3.8, long lead times support the integration of CT into SC concepts. Thus, in general, a longer delivery time positively influences the applicability of CT. However, depending on the SC concept configuration predictability of the transport processes an integration of CT for lead times longer than one day is possible. For same-day delivery, the applicability is restricted to special framework conditions (e.g., proximity of infrastructure connection, appropriate train departure).

Delivery Time Point

The *delivery time point* determines whether the delivery has to arrive at the destination at a determined (often on a schedule basis), at a flexible (with prior advice) or at a free time point. In practice, time window concepts are increasingly applied.²⁵³ The impact of the delivery time point type is restricted to temporal SCP aspects, namely it affects delivery window, transport time and delivery frequency.²⁵⁴

In general, a free delivery time point eases the integration of CT concepts into a SC concept. However, depending on the situation-specific CT concept configuration it is possible to meet defined delivery times and delivery windows.

Table 3 - 14 summarises the findings on the impact of distribution concept configurations on SCP requirements developed in section 3.2.

²⁵² cf. Changxian et al. (2009); Michaelraj et al. (2009); Siller (2011).

²⁵³ cf. Campbell et al. (2004); Quak et al. (2009); Zobel (2011).

²⁵⁴ cf. Hoppe et al. (2002), pp. 24.

reading direction		impact of distribu	tion concept config	uration	
		regional structure	echelon structure	delivery service	delivery time point
SCP requirements					
time	delivery window				Х
	transport time	Х	Х	Х	Х
	delivery frequency	Х	Х	Х	Х
space	transport distance	Х	Х		
	distance to infrastructure connection	X	Х		
	number of sources and destinations	Х	Х		
shipment size and transport quantity		Х	Х		
cost	transport cost/ turnover	Х	Х	Х	
flexibility	time	Х	Х	Х	
	quantity	Х	Х	Х	
	space	Х			
reliability	adherence to schedules			Х	
	quality				

 Table 3 - 14: Impact of distribution concept configuration on SCP requirements.

After the discussion of the different configuration elements that influence the SCP requirements regarding transport concepts, the meaning of the distribution concept configuration on the integration of CT concepts is finally discussed.

3.4 Development of a CT Concept Typology - Classification using SCP Profiles

This section introduces a classification approach for CT concepts focusing on material and information flows. Three different CT concept types are characterised by socalled 'performance profiles'. These profiles can be understood as the equivalent of the SCP performance profiles that express the SCP requirements of a certain SC concept configuration.

The CT business model classification introduced in section 2.2.2.2 serves as the basis, but the typology does not aim to cluster similar business models, but rather group similar SCP requirements regarding CT transport. Therefore, the typology of CT concepts follows the classification approach of SCP requirements (as developed in section 3.2), but focuses on the descreiption of performance capabilities rather than performance requirements.

Different CT concepts are classified according to spatial and time-related transport

performance characteristics. This means that the concept is viewed from a SC perspective.²⁵⁵ Furthermore, the performance characteristics are discussed based on the performance categories introduced in section 3.2. There are three different types of CT concepts focusing on the material and information flows: *(1) line transport with regular booking, (2) line transport with flexible booking and (3) packaged goods networks.*

There is no discussion on product type, reliability and costs. The performance characteristics in these categories must be determined situation specifically. Hereby, the three identified CT types do not significantly differ regarding these dimensions. However, in practice the usage of these dimensions supports the specification of the CT performance profile.

Line Transport – Regular Booking

The *'line transport - regular booking'* is a service for shippers that regularly book a certain transport on a regular train line.

This CT concept type is characterised by its comparably high speed (short transport time), punctuality and frequency. Owing to the comparably short transport times and high frequency, the CT concept can be applied to comparably short transport distances (cf. chapter 5). Because of the high transport frequency, the concept is suitable for comparably short distances from shipper and consignee to the infrastructure connection. Usually a line transport concept connects only a few sources and destinations (cf. section 3.2.1 relation transport). Mass goods or FTL are suitable for the application of line transport with regular bookings. Although the time flexibility regarding certain connections is low (train departures are usually not delayed if a certain shipment is missing), shippers have high flexibility regarding switching transport quantities to the next train or the next day. This flexibility regarding transport quantities and space is high, too. Table 3 - 1 summarises these findings.

The dark grey fields characterize the highest performance level that can be reached by the specific CT concept. Grey fields show lower performance levels that can obviously be met, too. White fields characterise that a CT concept can only restrictedly reach this performance level. A cross means that the CT is not suitable for this per-

²⁵⁵ The classification of CT services introduced in section 2.2.2.2 takes a CT service provider's point of view and aims at the determination of the underlying business model.

dimension	performance criteria	characteristi	cs				
time	delivery window	5 min	30 min	1 h		6 h	day
	transport time	< day	24 h		48 - 72 1	h	> 72 h
	delivery frequency	multiple per o	day 1 x day	/	1 x weel	k	sporadic
space	transport distance	short		medium		high	·
	distance to infrastructure connection	low		medium		high	\searrow
	number of sources and destinations	1 - 1	n - 1/1	- n	few - n	n - few	n - m
quantity	shipment size / transport quantity	> mass good	> FTL		LTL	\sim	packaged
flexibility	time	high		medium		low	
	quantity	high	<	medium		low	
	space	high		medium		low	

formance requirement.

Table 3 - 15: Characteristics of the CT concept 'line transport - regular booking'. (dark grey = suitable (highest performance level); grey = suitable (lower performance level); white = restrictedly suitable; X = not suitable)

For instance, *Swiss Post* provides a CT service up to eight times a day. This allows shippers to cancel advised transport in the short-term (e.g., because of congestion) and offers the ability to switch to the next train (usually 2–3 h later) without additional cost.²⁵⁶

Another example is *ERS Railways* (a subsidiary of *A.P. Møller*), which offers regular shuttle trains for seaport transport.²⁵⁷ The *Hamburger Hafen und Logistik AG* (*HHLA*) offers shuttle trains between the Hamburg ports and the terminals in Prague. *ERS* and the *HHLA* offer the booking of transport capacities on a regular basis, for instance daily, weekly or monthly.²⁵⁸

Line Transport – Flexible Booking

The CT concept of *'line transport - flexible booking'* is similar to the previous concept in terms of service production. The concept aims at a high level of flexibility level for the shipper. This means a shipper can flexibly and in the short-term book transport or can switch shipments on regularly offered lines without pre-booking, but there is no guarantee of free capacity. This type of CT is applicable for shippers for spontaneous congestion, weather changes and strongly varying transport volumes.²⁵⁹ CT actors have a high risk to fill their capacities, but they are able to increase price levels for the sporadic service. However, in combination with a dynamic pricing

²⁵⁶ For further details to the CT offer of the *Swiss Post*, see chapter 5.

²⁵⁷ cf. ERSRail (2011).

²⁵⁸ cf. Parment (2008), p. 33.

²⁵⁹ For instance, for the hinterland transport of globally sourced goods arriving in a sea port.

model the shipper can save money by early booking, whereas the CT operator gets a higher predictability of demand.

The time-related performance of a line transport with flexible booking is comparably high regarding the time window of delivery. Owing to the high punctuality, even short delivery windows can be met. The transport time is longer in comparison with the regular booking since the availability of transport capacities at the desired time point cannot be guaranteed. Because of the sporadic character of the demand, this transport concept is especially suitable for long distance transport and when infrastructure connections are comparably close. As in the previous concept, the line transport concept usually connects few sources and destinations. Owing to its sporadic character, it is suitable for FTL. The flexibility of the line transport with regard to booking is comparably high in terms of time, quantity and space. However, it must be considered that the shorter the time between booking and departure, the higher the risk that there is no sufficient capacity on the desired transport. Table 3 - 2 summarises these findings.

dimension	performance criteria	characteri	stics					
time	delivery window	5 min	30 min	1 h		6 h		day
	transport time	< day	24 h	48 - 72 1	h	> 72 h		•
	delivery frequency	multiply pe	er day 1 x day		1 x weel	κ.	s	poradic
	transport distance	short	\sim	medium	\sim		high	
space	distance to infrastructure connection	low		medium			high	
	number of sources and destinations	1 - 1	n - 1/1 ·	- n	few - n /	n - few	r	ı - m
quantity	shipment size / transport quantity	> mass goo	rd > FTL		LTL	\sim	F	backaged
flexibility	time	high		medium			low	
	quantity	high		medium			low	
	space	high		medium			low	

Table 3 - 16: Characteristics of the CT concept 'sporadic usage of line relation'.(dark grey = suitable (highest performance level); grey = suitable (lower performance level);white =restrictedly suitable; X = not suitable)

Packaged Goods Network

The CT concept of '*packaged goods network*' focuses on the transport of packaged goods and LTL. In the past, there have been several attempts to realise packaged

good networks via CT, e.g.,the Swiss *Cargo Domino*²⁶⁰ and *DHL Parcel Intercity* (PIC)²⁶¹ and the express cargo trains operated by *Hellmann Rail Solutions*.²⁶²

The lead time is comparably long to realise the different bundling and recombination processes. Usually a maximum lead time is offered. Thus, the delivery window for delivery and pickup is comparably long. The distance to the next terminal can be rather long, since usually there is a number of pickup sites and pre-haulage shipments are bundled. Thus, the packaged goods network is sensible for operation with multiple shippers as well as many loading and unloading sites. The flexibility of the concept is comparably high. The time, quantity and pickup and delivery places for shipments can be changed at short notice. Table 3 - 3 summarises these findings.

dimension	performance criteria	characteristics					
time	delivery window	5 min 30 i	min	<u> </u>	6 h		day
	transport time	< day	24 h	>	48 - 72 h		> 72 h
	frequency	multiply per day	1 x day	1	1 x week		sporadic
space	transport distance	short		medium		high	
-	distance to infra-	low		medium		high	
	structure connection						
	no. of sources and	1 - 1	n - 1/1	- n	few - n / n - fev	N	n - m
	destinations						
quantity	shipment size /	> mass good	>F¶-		LTL		packaged
	transport quantity						
flexibility	time (variance of	high		medium		low	
	lead time/ rate)						
	quantity	high		medium		low	
	(variance / rate)						
	space (variance /	high		medium		low	
	rate)						

Table 3 - 17: Characteristics of the CT concept 'network relation'. (dark grey = suitable (highest performance level); grey = suitable (lower performance level); white =restrictedly suitable; X = not suitable)

After the discussion of different CT concept types and performance profiles, the following section summarises the main results of chapter 3.

3.5 Intermediate Findings

Chapter 3 addresses the meaning of CT as an element of SC concepts. The section introduces the SC sub-concepts, namely the procurement, production and distribution

²⁶⁰ cf. SBBCargo (2011).

²⁶¹ Operated by *Railion* later *DB Schenker Rail*. cf. Heinrici (2011). A CT train between Hamburg and Munich operated by *DHL* introduced in 2001. Concentrated on the night jump and with more than 140 km/h and up to 99% punctuality. cf. Ernst (2011).

²⁶² Initiated 2004 between Hannover and Nuremberg. Recently eight railway stations (North: Hamburg, Bremen, Osnabrück, Hannover) and (South: Frankfurt, Nuremberg, Regensburg, Landshut, Basel). cf. Homeyer (2004), p. 14.

concepts based on the relationship with a specific SC strategy. Cause-and-effect relationships between the elements of the SC and CT concepts are worked out based on an empirically developed classification of SCP requirements. Therefore, chapter 3 answers the first sub-ordinated research question and contributes to answering the second.

Chapter 3 introduces the SC concepts as the operationalisation of SC strategies. A SC strategy depends on the product type as well as supply and demand-related characteristics. This means the SC strategy influences the configuration of the procurement, production and distribution as well as the embedded and linking transport concepts. The SCP requirements regarding transport concepts of specific sub-concept, namely the procurement, production and distribution concepts, are discussed. The SCP requirements can be summarised in a so-called 'SCP requirements profile'. This is the basis for the identification of adaptation points for the integration of CT into a specific SC concept. Similarly, each CT concept be characterised by a so-called 'performance profile'. These profiles are the focus of the service characteristics, rather than of service production.

The presented approach for the description of SC concepts and their requirements on transport concepts is universal. It is neither restricted to a specific type of transport concept nor to a specific type of SC concept. Moreover, the SCP requirements of standardised SC concepts can be described by the presented classification approach.

Table 3 - 18 briefly characterises the different CT types regarding suitable application fields.

	CT concept types	characteristics
А	line transport - regular booking	high transport quantities (mass good and FTL), short lead times, high frequencies and low variations, high level of predictability
В	line transport - flexible booking	high level of flexibility, comparably high level of adherence to schedules,
		medium lead time, comparably high costs
С	CT network	small transport volumes (packaged good) and packaged goods, long lead
		time, many locations, high costs

Table 3 - 18: Overview on CT concept and suitable application fields.

	CT concept type	Α	В	С
time	delivery window	high	medium	low
	transport time	high	high	low
	delivery frequency	high	low	medium
space	transport distance	short	short	medium
	distance to infrastructure connection	short	short	medium
	no. of sources and destinations	few-few	few-few	many-many
shipment size and transport quantity		medium - high	medium	low
flexibility	time	medium	high	high
	quantity	low	high	high
	space	medium	medium	high

Table 3 - 19 applies the developed structuring approach and states the performance profiles of the three different CT concept types.

 Table 3 - 19: SCP requirements profiles of the three CT concept types.

(A - line transport -regular booking; B - line transport - flexible booking; C - CT network).

Chapter 3 introduces an approach to describe the configuration of a SC concept with regard to the integration of CT. In particular, the impact of certain elements of the subconcept configurations on the applicability of CT services is concretised. Therefore, chapter 3 serves as the basis for the development of the conceptual research framework presented in the following section.

4 Conceptual Research Framework of Performance-oriented CT Integration into SC Concepts

The fourth chapter aims at the development of a conceptual research framework and the development of propositions on the performance integrated integration of CT into SC concepts. The conceptual framework illustrates the relevant interdependencies between the elements of the SC concepts, the CT concepts as well as integrative measures and instruments with regard to SCP. In a first step, the main research idea is developed based on previous considerations. The section results in the description of the main proposition for the thesis in hand (section 4.1). The conceptual research framework builds on these considerations and follows the argumentation line of configuration theory (section 4.2). In a step-wise approach the central constructs of the research problem are segmented into middle range constructs as well as manifest and observable variables. This procedure differentiates the complex research problem into smaller partial problems. The basic relationships between the main constructs and variables are summarised in first, rather general propositions. These propositions are specified in section 4.3. Here, the relevant cause-and-effect relationships between the elements of the SC and CT concepts as well as integrative measures and instruments are discussed in depth. Section 4.4 summarises the results in the form of intermediate findings.

4.1 Central Aspects of the Development of Research Propositions on the Performance-oriented CT Integration

This section aims at the development of a main research proposition for the performance-oriented integration of CT into SC concepts. The section joins the central findings of the previous chapters and bases the main research propositions on it.

All CT concepts have inherent challenges and general requirements in common. These characteristics restrict the configuration of SC concepts in which CT can be successfully embedded. As shown in section 2.2.1, high and predictable transport quantities with regular demand and comparably long lead times facilitate the integration of CT into SC concepts. However, as shown in section 3.4, these characteristics correspond to the CT concept type of line transport with regular booking (type A). Nevertheless, the introduction of CT business concepts (cf. section 2.2.2.2) and the service-oriented classification of CT concepts (cf. section 3.4) prove that more of

these 'typical' transport characteristics can be met. More flexible CT offers can meet the requirements of shorter lead times, fluctuating demand and transport quantities. Here, for instance, the line CT with flexible booking (type B) is suitable. CT networks (type C) can be applied to rather small transport volumes and comparably long lead times.

One central challenge of CT integration into SC concepts results from the competitive situation with unimodal road transport. The increased time quotas, high coordination efforts as well as additional transhipment, handling and overhead processes result in an increased cost level in comparison with unimodal road transport.

Nevertheless, the potential advantages of CT integration are obvious. Delays induced by congestion and infrastructure shortages in unimodal road transport can be avoided and thus, the transport concept's reliability level can be increased. Furthermore, a SC concept can gain more flexibility by CT integration. CT can be understood as an additional transport mode, which can easily be operated next to unimodal road transport.¹ Owing to the better emissions profile of rail transport in comparison with road transport, the total emissions level can be reduced, depending on the transport distance shifted from road to rail transport. If this advantage is specifically promoted, this effect can positively influence the shipper's image.²

The analysis of recent challenges of SC integration (cf. section 2.1) and CT (cf. section 2.3) shows that CT actors are so far not integrated members of SCs. The conceptual research framework builds on the idea of exploiting this imperfection for the integration of CT into SC concepts. The central argumentation line of the conceptual research framework is based on this consideration.

The tight integration of material and information flow processes can open up time buffers and increase coordination flexibility. There are two approaches to achieve this necessary integration of material and information flows: (1) the alignment of the SC and CT concepts and (2) the situation-specific usage of integrative measures and instruments.

This main research idea can be explained by two basic assumptions:

(1) The situation-specific adaptation of the SC concept opens up the time requirements for possible CT integration.

¹ cf. Stölzle et al. (2008).

² cf. section 2.3 for further motivations for shippers to integrate CT into their SC concepts.

(2) The SC and CT concept-specific applications of integrative measures generate coordination flexibility and cost savings to reach the SCP level of unimodal road transport.

A conceptual framework supports the specification of these generic assumptions into propositions. The following section describes the development, structures and elements of the framework.

4.2 Development of a Conceptual Research Framework and Deduction of **Propositions on the Performance-oriented CT Integration into SC** Concepts

A conceptual research framework supports the differentiation of the given complex research problem in partial problems. The development of a conceptual research framework is accomplished in a two-phase process.³ In the first step, all relevant constructs, variables and framework conditions of the research problem are identified (section 4.2). In the second step, the assumed relationships between the constructs and variables are determined (section 4.3). Assumptions on these relationships are concretised in the form of propositions.

4.2.1 **Identification of Central Constructs and Framework Conditions**

The conceptual research framework illustrates all relevant constructs, elements and dimensions of a given research problem⁴ and visualises the main argumentation line of the thesis in hand.

The conceptual research framework systematises structures, penetrates the problem formulation and supports the development of initial propositions.⁵ The framework joins the preconceptions from the literature review, expert interviews and explorative case studies with the chosen theoretical approach.⁶ The conceptual research framework shows the important elements of the research problem and connects the identified constructs relevant to the problem solution.⁷ The 'fit concept' of configuration

cf. Ibid. 6

cf. Rössl (1990), pp. 101. 3

⁴ cf. Kubicek (1977). 5

cf. Wolf (2008), p. 27. cf. Kubicek (1977), pp. 37; Wolf (2008), p. 3.

theory serves as the general structure for the framework.⁸ This corresponds to research theory literature on framework design.⁹

Figure 4 - 1 illustrates the relationships between the three main constructs. The design of the conceptual research framework is three-fold. The framework distinguishes context, design and success variables. According to the main research question, the main constructs are 'SC concept', 'CT concept' and 'SCP'. The process of 'integration' is the connecting process element leading to a 'fit'. Additionally 'integrative measures and instruments' are introduced as 'moderator variables'.¹⁰ It is assumed that each construct can be described more precisely by its elements, dimensions or subconcepts.

- (1) The SC concept is assumed to be the design variable.¹¹ The design variable is the element that is 'designed' to reach a fit with the framework variable. This assumption follows the considerations of different SC concept configurations in section 3.1.
- (2) The CT concept is assumed to be the framework variable. This assumption follows the considerations of the inherent characteristics of the CT concept presented in section 3.4. The framework variable is assumed to be unalterable. This means for the conceptual research framework that the performance profile of a CT concept is given and cannot be changed.
- (3) The SCP is assumed to be the performance variable. This assumption follows the considerations of the meaning of value and finance orientation in SCM in section 2.3. The performance variable evaluates the quality of the reached fit between the framework and design variables.

Furthermore, framework conditions influence the characteristics of the central constructs and thus, the goodness of fit. Building on the consideration in sections 2.2 and 3.1, it is assumed that the SC environment and SC strategy are relevant framework conditions that affect the configuration of SC and CT concept types, the integration process and thus, overall SCP. Figure 4 - 1 gives an overview of the relationship between the central constructs. This general description of the given research problem contributes to answering the main research question.

cf. section 2.4.

cf. Wolf (2000), p. 37. 10

cf. Bortz (2005), pp. 6, 636. cf. Wolf (2008), pp. 345. 11

As substantiated in section 2.1, the focus of the thesis in hand is to analyse the material and information flow processes and structures¹² rather than the management aspects.¹³ Thus, behavioural aspects and actor-specific problems are excluded from profound consideration. This focus allows the illumination of CT integration potentials rather than actor-specific challenges. The degree of novelty of the research topic, too, recommends this approach to reduce the complexity of the research problem.

	framework conditions SC strategy		SC environment
	design variable supply chain concept	FIT!	framework variable combined transport concept
ral tructs		performance variable supply chain performance	Q ₀

Figure 4 - 1: Central constructs of the CT integration into SC concepts.¹⁴

These considerations lead to the first proposition:

 P_1 : The better the fit between the SC sub-concepts' configurations and the CT concept types, the higher the level of SCP.

This rather generic framework and the proposition must be concretised. Thus, the following section introduces so-called *'middle range constructs'* for specification.

4.2.2 Middle Range Constructs — Elements and Dimensions of Central Constructs

The central constructs of the conceptual research framework are further concretised by *'middle range constructs'*. These middle range constructs are based on the typologies for SC and CT concepts. This specification is the basis for the deduction of detailed propositions. A selection of these propositions is analysed in depth in the simulation study presented in chapter 5.

Following the procedure and terminology of configuration theory (cf. section 2.4), the SC concept is defined as the design variable. It is concretised by its embedded sub-concepts (cf. section 3.3) of the:

 13 cf. Lambert et al. (2001).

¹² Following the deliberations of Cooper et al. (1997) and Lambert et al. (2001), both the SC concept and CT concept can be distinguished by concept structure, concept processes and concept management. cf. Cooper et al. (1997), p. 177; Lambert et al. (2001), p. 71. This understanding corresponds to the presented typologies of SC concepts (section 3.3), CT concepts (section 3.4) and SCP understanding (section 2.2).

¹⁴ Own illustration.
- (1) Procurement concept, (cf. section 3.3.3),
- (2) Production concept, (cf. section 3.2.4),
- (3) Distribution concept (cf. section 3.2.5) and
- (4) Embedded and linking transport concepts. (cf. section 3.2.2).¹⁵

It can be assumed that the configuration of these sub-concepts influences the applicability of CT and thus SCP.

The CT concept is assumed to be the *framework variable*. The classification of CT concepts was introduced in section 3.4. Three CT concept types were identified:

- (1) Regular line trains regular booking,
- (2) Regular line trains flexible booking and
- (3) CT networks.

It can be assumed that the applicability and effect on SCP differ according to the CT concept type.

SCP is chosen as the *performance variable* for the evaluation of the fit between SC and CT concepts. As shown in section 2.3.1, the SCP term can distinguish between efficiency and effectiveness. This means that that the effect of the CT integration into a SC concept is evaluated according to these two dimensions. This two-dimensionality does justice to the principle of equifinality (cf. section 2.4). There are different possible approaches for a given initial situation since SCP is situation- and actor-specifically defined.

Figure 4 - 2 provides an overview of the middle range constructs. Furthermore, the typologies of SC sub-concepts and CT concepts contribute to answering the second research question.



Figure 4 - 2: Middle range constructs of the performance-oriented CT integration into SC concepts.¹⁶

The middle range constructs must be concretised by manifest and observable vari-

¹⁶ Own illustration.

¹⁵ This means that a CT concept must be understood as a part of the specific SC concept itself. Furthermore, the impact of different sub-concept configuration on specific SCP requirements has been discussed.

ables. This is necessary to develop more detailed propositions.

4.2.3 Manifest, Observable Variables

The middle range constructs are further specified by manifest and observable variables. The measurability of these variables is the requirement for the development of first propositions.

To evaluate different configurations of sub-concepts six primary and two secondary dimensions of SCP requirements were empirically identified (the deductions of these requirements is presented in section 3.2). There are eight categories for the analysis of the SCP requirements of the sub-concepts, namely the procurement, production and distribution concept and the linking and embedded transport concepts. The different configurations can be understood as '*SCP requirements profiles*' (Figure 4 - 3, left). This structured description of SCP requirements contributes to answer the first research question.

Corresponding to the SCP requirements profiles, there are eight dimensions for the analysis of different CT concepts. Different configurations can be expressed by so-called *'CT performance profiles'* (Figure 4 - 3, right; deduction of the performance profiles in 3.4).

The evaluation of the SCP is accomplished by means of seven performance indicators. The performance indicators reflect the characteristics of performance measurement in a CT context and the concept of SCP management. (cf. deduction in section 2.3.5).

- (1) Adherence to schedules
- (2) Inventory levels
- (3) Utilisation of capacities
- (4) Lead time
- (5) Flexibility (on short-term changes)

These key figures above focus on the integration of material and information flows. However, to do justice to the SCP focus on financial key figures and the meaning of sustainable behaviour for the image of the SC actors, two more key figures are added:

- (6) Costs (e.g., process costs, capital commitment) and
- (7) Emissions (e.g., CO_2 , NO_x and particles)

Key figures (1) - (3) can be assigned to the effectiveness dimension. Key figures (4) -(7) can be assigned to the efficiency dimension (cf. Figure 4 - 3).

In practice, the weighting of these key figures must be accomplished for company, SC and situation. Generally, the thesis in hand does not evaluate and weight the importance of these key figures. However, the meaning of different key figures is discussed for the standardised logistics and SC concepts introduced in chapter 3.

manifest observable variables	iling Rey figu	ures		CT concept performance profile relie ^{bility}
time space shipment size / transpor product type cost	regard to material and information flow additional keyfigues	efficiency lead time flexibility costs emissions	effectiveness adherence to schedules utilization of capacities inventory level	time space shipment size / transport quantity product type cost

Figure 4 - 3: Manifest, observable variables of CT integration into SC concepts.¹⁷

These considerations lead to two more detailed propositions. These propositions take special regard of the developed key figures.

 P_2 : The better the fit between SC concept and CT concept, the higher SC efficiency in terms of (a) shorter SC lead times, (b) higher SC flexibility, (c) lower SC costs and (d) lower SC emissions.

 P_3 : The better the fit between SC concept and CT concept, the higher SC effectiveness in terms of a better achievement of the defined level of (a) adherence to schedules, (b) capacity utilisation and (c) inventory levels.

Figure 4 - 4 summarises the findings of the conceptual research framework. The following section builds on the illustration of dependencies between the central constructs on all three consideration levels.

¹⁷ Own illustration.



Figure 4 - 4: Conceptual research framework of CT integration into SC concepts.¹⁸

The following section introduces a moderating variable of the fit between SC and CT concept. More precisely, integrative measures and instruments are introduced as moderating variables.

4.2.4 Moderating Variables of CT Integration into SC Concepts

In this section, the meaning of integrative measures and instruments for CT integration into SC concepts is specified. Therefore, an additional element is appended to the conceptual research framework. The technical and organisational management components for SCM, introduced in section 2.1.6, are assumed to be moderating variables. This means that the integrative measures influence the fit between SC and CT concept. This contributes to answering the third research question. Figure 4 - 5 illustrates the considerations on the effect of integrative measures and instruments on the fit between SC and CT.

¹⁸ Own illustration.



Figure 4 - 5: Integrative measures and instruments as moderating variables for CT integration.¹⁹

The technical and organisational management components influence the integration process. If the integrative measures are suitable for the given configuration of SC and CT concept, the usage can improve SCP. The management components are integrative measures and instruments. There are five groups of integrative measures, introduced in section 2.1.6:

- (1) I&C facility structure,
- (2) Workflow & activity structure,
- (3) Organisational structure,
- (4) Planning and control methods and
- (5) Product flow facility structure.

Following the considerations in section 2.1.6, the integrative measures are assumed to impact all performance dimensions of CT integration. Thus, there are two propositions on the effect of integrative measures.

 P_3 : The better the fit of integrative measures and instruments and the configuration of SC concept and CT concept, the higher SC efficiency in terms of (a) shorter SC lead times, (b) higher SC flexibility, (c) lower SC costs and (d) lower SC emissions.

 P_4 : The better the fit of integrative measures and instruments and the configuration of SC concept and CT concept, the higher SC effectiveness in terms of the better the achievement of the defined level of (a) adherence to schedules,

¹⁹ Own illustration.

(b) capacity utilisation and (c) inventory levels.

Propositions P_1 to P_3 have rather general characters. They are statements on the general cause-and-effect relationships, but are not specific enough to be transferred to practical recommendations. Thus, the following section analyses the cause-and-effect relationships between the elements of the conceptual research framework in depth.

4.3 Cause-and-Effect Relationships between the Elements of CT and SC Concepts - Deduction of Methodological and Instrumental Implications

This section builds on the propositions P_1 to P_3 to analyse the cause-and-effect relationships between SC and CT concepts as well as the impact of integrative measures and instruments on the fit. This aims at the development of specified propositions and at the development of practice-oriented recommendations for CT integration into SC concepts. Thus, the section contributes to answering the third research question. Therefore, a four-step approach is applied.

The progressive approach includes:

- (1) The identification of integration points for CT into SC concepts,
- (2) The impact of SC strategy, SC concept configuration and SCP requirements profiles for CT integration,
- (3) The identification of adaptation points of SC sub-concepts to change SCP requirements profiles with regard to the typology of CT concepts and
- (4) The situation-specific effects of integrative measures and instruments on SCP.

The results of these considerations are summarised in the form of propositions.

4.3.1 Step 1: Identification of Integration Points for CT into SC Concepts

In a first step, the general integration points for CT into SC concepts are identified. The definition of specific integration points allows keeping the number of developed propositions low. These four types characterize different types of CT integration situations. Figure 4 - 6 gives a schematic illustration of the integration points.



Figure 4 - 6: Schematic illustration of the integration points of CT concepts.²⁰

I. supplier - manufacturer

CT concepts can be integrated for the material flow processes between the supplier's and manufacturer's production site. Therefore, the configuration of the production concepts, the manufacturers' procurement concepts and the supplier's distribution concept are relevant.

II. manufacturer - store

CT concepts can be integrated to ensure the material flow processes between the manufacturer's production site and the retailer's store. Therefore, the configuration of the manufacturer's production and procurement concepts as well as the retailer's distribution concept are relevant.

III. manufacturer - DC

CT concepts can connect the manufacturer' production site and the retail's DC. The differentiation to the previous integration point (II) is necessary, since store delivery and DC delivery can significantly differ in terms of SCP requirements regarding the embedded and linking logistics and transport concepts.

IV. DC - store

CT concepts can be integrated to connect the DC with the store. The configuration of the retailer's distribution concept is central for the integration.

The classification of possible integration points delivers information on the involved SC sub-concepts. The sub-concepts determine the specific SCP profiles to be met by the CT performance profiles. Thus, the targeted adaptation of the sub-concepts contributes to a performance-oriented CT integration into SC concepts. Table 4 - 1 summarises the considerations in a schematic overview of the relevant sub-concepts, which have to be taken into consideration for a performance-oriented CT integration into SC concepts.

²⁰ Own illustration.

Integration point	Ι	Π	III	IV
SC strategy	supplier - manufacturer	manufacturer- store	manufacturer - DC	DC - store
production concept supplier	Х			
distribution concept supplier	(X)			
procurement concept manufacturer	(X)			
production concept manufacturer	Х	Х	Х	
distribution concept manufacturer		(X)	(X)	
procurement concept retailer		(X)	(X)	
distribution concept retailer		Х	Х	Х

Table 4 - 1: Identification of relevant concepts depending on the integration point. (X = primary impact; (X) = secondary impact).

4.3.2 Step 2: Impact of SC Strategy, SC Concept Configuration and SCP Requirements Profiles on CT Integration

In a second step, general recommendations on the suitability of certain SC strategies for CT integration are developed. Since SC concepts are defined as the operationalisation of SC strategies, initially, the applicability is determined by demand and supply characteristics, namely predictability and lead times.

The SC strategy impacts the central supply and demand characteristics in a SC, which are relevant for CT integration, such as the *predictability* of transport volumes, times and frequencies as well as the *lead times*. Figure 4 - 7 visualises the general applicability of CT into SC concepts according to different SC strategies. The classification of SC strategies can be affiliated to two main criteria: predictability of demand and length of lead times.

Generally, SCs with lean strategies (type I – plan and optimize and type II – continuous replenishment) are suitable for the integration of CT because of the high level of demand predictability. The lean (type I) strategy and the leagile strategy are suitable for the integration of CT because of their comparably long lead times. The integration into SC concepts with an agile SC strategy seems to be rather difficult because the characteristics of the SC and CT concept configurations are rather contrary.



Figure 4 - 7: General applicability of CT concepts according to different SC strategies.²¹

These considerations give first guidance on the general applicability of CT concepts into a certain SC concept.

Next to the SC strategy, the integration point is central for the integration of a CT into a SC concept.

(I) The material flow processes *between supplier and manufacturer* in a lean and a leagile SC are characterised by rather high transport volumes as well as regular and long lead times.²² The integration of regular line transport is suitable if the daily, weekly or more frequently transport quantities reach a FTL level. In a lean (type I - plan and optimize) environment, CT networks can also be used for smaller shipment sizes. In an agile SC, depending on shipment size, transport relation and delivery time, flexible line trains can be applied to the rather fluctuating transport quantities.

(II) The material flow processes between manufacturer and stores is characterised by

²¹ With significant enhancements on the basis of Christopher (2005b).

²² Assuming the customer decoupling point is downstream of the SC.

rather small transport quantities. Stores usually have no warehouse capacities to hold transport goods. Shipping to the store directly from a production site is only common for predictable and medium-sized transport volumes, namely in a lean SC (type I - plan and optimize). This means the shipment size is usually LTL. Thus, the required transport volume for line transport is usually not reached. However, for comparably large stores with regular transport volumes (FTL) the integration of regular line transport can be possible. Furthermore, the sporadic integration, e.g., in the case of specific promotion activities, with flexible line concepts is applicable. For long lead times and packaged goods the usage of a CT network is appropriate.

In lean SC (type II - continuous replenishment) and leagile SCs²³ with short lead times, the integration of CT between manufacturer and store is generally not applicable. However, depending on the situation-specific framework conditions for comparably long lead times and smaller transport quantities the integration of CT networks can be possible.

(III) The material flow between *manufacturer and a retailer's DC is characterised by comparably* high transport volumes. In particular, in lean SCs the integration of regular line transport is conceivable depending on transport quantities and frequency. For smaller transport quantities with comparably long lead times (lean type I - plan and optimize), the usage of CT networks is also possible.

For this integration point the differentiation of leagile SCs into SCs with manufacturing and logistics postponement strategy is necessary. For the delivery between manufacturer and store a leagile SC with manufacturing postponement is characterised by rather fluctuating and smaller transport quantities. Thus, depending on the other framework conditions line transport with flexible booking and CT networks are applicable. In a logistics postponement environment, transport quantities have a rather lean character. Thus, the application of line transport with regular booking and for comparably long lead times and small shipment sizes the usage of CT networks is possible. In an agile SC depending on the specific transport quantities only the flexible line transport and the CT network may reflect the fluctuating transport quantities.

(IV) The material flow between *DC and store* is characterised by rather small transport quantities. In lean SCs, the introduction of regular line transport is sensible. In a

²³ Assuming the customer decoupling point is upstream of the SC.

lean SC (type I - plan and optimize) with comparably long lead times, the sporadic usage of flexible line trains, e.g., for promotion activities, is also conceivable. In leagile and agile SCs, the delivery between DC and store is characterised by responsiveness and speed. Thus, the integration of CT would contradict these ideas.

Table 4 - 2 summarises these considerations and provides general recommendations on the CT concept types into SCs with different SC strategies.

Integration point	Ι	Π	III	IV
SC strategy	supplier - manu- facturer	manufacturer- store	manufacturer - DC	DC - store
lean I (plan and optimize)	A, C	(A), (B), C	A, C	A, (B)
lean II (continuous replenishment)	А,	-	А	А
leagile I (manufacturing postponement)	А	-	B, C	-
leagile II (logistics postponement)	Α	-	В	-
agile	(B)	(C)	(B), (C)	-

Table 4 - 2: Applicability of CT types according to SC strategy and integration point.

A = line trains - regular offer; B = line trains - flexible offer; C = CT network; A applicable; (A) applicable in limited extend

These considerations are of rather generic character and do not regard specific framework conditions. The following section builds on these generic recommendations and identifies adaptation possibilities to improve the fit between a specific SC concept and a specific CT concept.

4.3.3 Step 3: Identification of Adaptation Points for CT Concept Integration

In the previous section, a general fit between different SC and CT concept types was discussed. However, to practically integrate CT into a SC concept the SCP requirements profile of the SC concept and the SCP profile of the CT concept must be compared. This comparison shows the fits or misfits regarding certain SCP requirements. Thus, in the following section, the adaptation points for influencing the SC concept configuration are introduced to overcome any identified *'misfits'*.

Example of the Identification of Adaptation Points

Table 4 - 3 provides an example of profile comparison. Here, the general SCP requirements profile of a lean SC (type I - plan and optimize) and the performance profile of a line transport concept with regular booking concept are contrasted. Table 4 -2 shows that the combination of a lean SCP requirements profile fits with the CT performance profile regarding all performance criteria.

dimension	performance criteria	lean I SC concept – SCP requirements	line trains – regular booking – CT performance profile	fit ?
time	delivery window	long	high	\checkmark
	transport time	long	high	\checkmark
	delivery frequency	low	high	\checkmark
space	transport distance	lon	short	\checkmark
•	distance to infrastructure connection	long	short	\checkmark
	number of sources and destinations	low	few-few	\checkmark
quantity	shipment size /transport quantity	high	medium - high	\checkmark
product type		standard	standard	\checkmark
costs	transport cost / turnover	high	medium	\checkmark
flexibility	time	low	medium	\checkmark
, i i i i i i i i i i i i i i i i i i i	quantity	low	low	\checkmark
	space	low	medium	
reliability	adherence to schedule	medium	high	\checkmark
	quality	medium	high	

Table 4 - 3: Example of contrasting lean SCP requirements profile and the performance profile of line transport concept with regular booking performance profile ($\sqrt{=}$ general fit; X = misfit).

Nevertheless, the comparison of the level of specific performance requirements may show misfits resulting from the situation-specific framework conditions. For instance, there may be misfits regarding the SCP requirements in the time dimension, which cannot be met by the chosen CT concept (Table 4 - 4). In this example, the SC concept requires short transport times, whereas the CT concept offers medium transport times. Furthermore, the SC concept requires more frequent deliveries.

dimension	performance criteria	SCP require- ments	CT performance	fit?
time	delivery window	long	high	\checkmark
	transport time	short	medium	Х
	delivery frequency	high	medium	Х

Table 4 - 4: Example of the misfit of SCP requirements and CT performance profile.

Following configuration theory, the time dimension of the CT concept is assumed to be the framework variable and thus unalterable. The SC concept is defined as the design variable. This means for the given example that the requirements of the default transport time and delivery frequency must be adapted if the integration of CT is desired.

In sections 3.3.2, 3.4.2 and 3.5.2, different adaptation points for the SC sub-concepts, namely the procurement, production and distribution concepts, are identified. Table 4 - 5 summarises the findings for time-related SCP requirements.

dimension	performance criteria	procurement concept	production concept	distribution concept
time	delivery win- dow	 supplier choice procurement area procurement time 	 order release release of secondary demand production control logic 	delivery time point
	transport time	procurement areaprocurement time	 order release release of secondary demand stocking type 	 regional structure echelon structure delivery service delivery time point
	delivery fre- quency	supplier choiceprocurement time	 order release release of secondary demand stocking type 	regional structureechelon structure

 Table 4 - 5: Adaptation points for SCP requirements regarding time.

 (developed in sections 3.3.2, 3.4.2 and 3.5.2).

This means that to adapt the performance requirements regarding transport time, depending on the integration point (cf. section 4.31.; Figure 4 - 6: (I) supplier – manufacturer, (II) manufacturer - store, (III) manufacturer – DC, (IV) DC – store), there are different adaptation points. For the delivery between supplier and manufacturer, the manufacturer's procurement concept can be adapted (cf. Table 4 - 1). This means for the alignment of transport times supplier choice, procurement area and procurement time can be altered (cf. section 3.3.2.3). Furthermore, the adaptation of the production concept in terms of order release, secondary order release and stocking type is possible (cf. section 3.4.2.3). Finally, the adaptation of the supplier's distribution concept is possible. For instance, delivery service, regional structure and echelon structure could be altered to influence the required transport and thus, the fit with the CT concept (cf. section 3.5.2.3).

Depending on the specific framework conditions there may be misfits regarding all identified SCP requirements. These misfits give advice about the alignment of performance requirements and characteristics. Specific adaptation points can be activated to counteract these misfits. In sections 3.3 to 3.5 for each SC sub-concept, adaptation possibilities and their effects on the specific SCP requirement dimensions were worked out. Figure 4 - 6 presents these findings structured according to the SC sub-concept and the different performance dimensions.

Product type is a central requirement for transport concepts. The decision over a certain product type is of strategic nature (cf. section 3.1). Thus, for the discussion of the impact of the SC sub-concept configuration it is assumed that product type cannot be influenced by the adaptations to sub-concepts (cf. section 3.2).

dimension	performance criteria	procurement concept	production concept	distribution concept
time	delivery window	 supplier choice procurement area procurement time 	 order release release of secondary demand production control logic 	delivery time point
	transport time	procurement areaprocurement time	 order release release of secondary demand stocking type 	 regional structure echelon structure delivery service delivery time point
	delivery frequency	supplier choiceprocurement time	 order release release of secondary demand stocking type 	regional structureechelon structure
space	transport distance	supplier choiceprocurement area		regional structureechelon structure
	distance to infra- structure connec- tion	supplier choice		regional structureechelon structure
	number of sources and destinations	supplier choice		regional structureechelon structure
quantity	shipment size and transport quantity	supplier choiceprocurement time	order releasestocking typeproduction control logic	regional structureechelon structure
cost	transport cost / turnover	 supplier choice procurement area procurement time incoterms 	 release of secondary demand 	regional structureechelon structuredelivery service
flexibility	time	 supplier choice procurement area procurement time incoterms 	 order release release of secondary demand stocking type production control logic 	 regional structure echelon structure delivery service
	quantity	supplier choiceprocurement time	 order release release of secondary demand stocking type production control logic 	delivery serviceregional structureechelon structure
	space	supplier choiceprocurement area		regional structure
reliability	adherence to schedules	supplier choiceprocurement areaprocurement time	stocking typeproduction control logic	delivery service
	quality	 supplier choice procurement area incoterms procurement time 	production control logic	

 Table 4 - 6: Adaptation points according to performance dimension, criterion and involved SC sub-concept. (developed in sections 3.3.2, 3.4.2 and 3.5.2).

Next to these adaptation points, integrative measures and instruments can be utilised to influence the fit between SC and CT concepts.

4.3.4 Step 4: Situation-specific Choice of Integrative Measures and Instruments for Performance-oriented CT Integration into SC Concepts

This section addresses the effect of integrative measures and instruments on the integration of CT into SC concepts. The integrative measures and instruments introduced in section 2.1.6 are analysed regarding their impact on specific SCP requirements. Thus, the section specifies the findings of the effects of integrative measures and instruments for the research problem on the integration of CT into SC concepts. The findings can be understood as practical recommendations for the targeted usage of integrative measures and instruments. The section provides a selection basis as well as an outlook on the expected effect of the measures. Thus, this section answers the third research question.

'Integrative measures and instruments' were introduced as moderating variables in the conceptual framework depicted in Figure 4 - 5. This means that the integrative measures cannot impact the fit or misfit in general, but improve the goodness of fit (cf. argumentation in section 4.1). The goodness of fit results from the impact of the integrative measures and instruments on single SCR requirements criteria.

Section 2.1.6 introduces 13 categories of integrative measures and instruments suitable for application in a SC concept including CT. Table 4 - 7 provides an overview of the integrative measures and instruments and introduces a numbering as an abbreviation for the following tables.

integrative measure and in- strument group	abbreviation	measures and instruments
I&C flow facility structure	1	integrated I&C methods, technologies and systems
	2	online platforms and online data provision
	3	automatically identification technologies
	4	standardisation of I&C technologies
workflow facility structure	1	online or electronic data interchange (EDI)
	2	latest material flow equipment / technologies
	3	standardisation activities
	4	SC-wide quality standards
	5	collaborative improvement techniques
organisational structure	1	cross-company operational activities
	2	strategic integrative activities
planning and control methods	1	joint planning and control systems
	2	joint planning and control approaches
	3	modelling and simulation techniques
product flow facility structure		product and load carrier design

 Table 4 - 7: Overview of analysed integrative measures and instruments and abbreviations.

 (developed in section 2.1.6).

Impact of Integrative Measures and Instruments on Time Aspects

Performance requirements of the 'time' dimension can be positively influenced by all types of integrative measures. The application of latest and *integrated I&C methods*, technologies and usage accelerates information exchange and communication times and serves as the basis for the application of cross-company planning and control approaches. These targeted collaborative activities reduce planning, handling and transport efforts in all SC parts and thus, positively influence the different lead time parts, such as transport time and production time. The usage of online platforms accelerates order placing and processing. Online data provision reduces waiting times and allows, even in the case of delays, the planning and control of delivery windows. The precise cross-company calculation of material flow processes allows the definition and meeting of short delivery windows²⁴ because of the better forecasting possibilities and online planning approaches.²⁵

Measures of the category 'workflow / activity structure' are the most obvious approach for the acceleration of lead and transport times. Practical examples include 'business process reengineering' or 'process optimisation' projects.²⁶ The introduction of online or EDI solutions accelerates information flow processes. As a result, crosscompany material flows can be closer coupled and lead times can be decreased. Furthermore, it is the basis of cross-company planning approaches and thus, positively influences the meeting of time windows. The usage of the latest and standardised material flow equipment and technologies, e.g., load carriers and conveyors, reduces waiting and transition times and thus, delays in material flow at the interfaces between SC actors. For instance, the application of the latest load carriers and transhipment technologies enables the handling of hazardous or frozen goods. In particular, the introduction of SC-wide quality standards, for instance for the EN ISO 9001:2008 or industry-specific quality norms such as the VDA 6.1 (automotive), harmonises the material flow and can even shorten lead times, since control processes can be reduced.²⁷ The continuous improvement of the cross-company process integration of material and information flows supports obtaining the long-term fit of CT and SC.

²⁴ cf. Zobel (2011); Quak et al. (2009).

²⁵ cf. Campbell et al. (2004); Chan et al. (2011); Wang et al. (2008). ²⁶ cf. Daymout (1993): Longson et al. (2007); Nairo et al. (2004); P.

²⁶ cf. Davenport (1993); Jonsson et al. (2007); Neiro et al. (2004); Papageorgiou (2009);Shobrys et al. (2002).

²⁷ cf. Bessant et al. (1994); Dowell et al. (2000); Jayaraman et al. (1999); Kaijie et al. (2007); Kim (2009);Parsons (2002).

Operative cross-company organisational measures, such as the introduction of project teams, decrease communication, reaction and decision times and thus, shorten and stabilise transport and total lead times.

Planning and control methods are central aspects to influence time-related SCP requirements. Based on the availability of EDI and general I&C systems, joint planning and control systems (e.g., ERP, PPC and APS systems and modules) as well as corresponding planning and control approaches significantly influence the design of delivery times and frequencies as well as lead times. Modelling and simulation techniques are used to improve the cross-company understanding of material and information flow processes. The encompassing understanding of the relevant cause-and-effect relationships is the basis for all optimising and aligning activities for the integration of material and information flows in a SC including CT. Thus, indirectly modelling and simulation techniques influence all time-related SCP requirements.

The 'product flow facility structure' is central for the time aspects of SCP requirements since the network structure impacts transport times and thus, lead times. Furthermore, this means the logistics-oriented design of products or load carriers can significantly reduce handling times. Table 4 - 8 summarises the considerations of the impact of integrative measures and instruments on time-related SCP requirements.

jbroit -		Imj	oact	of in	tegi	rative	meas	sures	and i	nstru	ment	s on				
	SCP requirements dimen-		C fac Ictui	:ility ∵e		worl stru	cflow cture	/ acti	vity		organ tiona struc	nisa- l ture	planning and con- trol methods			product flow facility structure
sion	1	2	3	4	1	2	3	4	5	1	2	1	2	3		
time			-	+				+			+		+			+
	window	+	+	0	0	+	0	0	+	0	0	0	+	+	(+)	0
	lead time + + + o		0	+	+	+	+	+	+	0	+	+	(+)	+		
	frequency	0	0	0	0	0	0	0	0	0	0	0	+	+	(+)	0

Table 4 - 8: Impact of integrative measures and instruments on time-related SCP requirements. (+ positive impact; - negative impact; +/- negative or positive impact to be weighed up (depending on framework conditions); o no relevant impact; () indirect impact)

Impact of Integrative Measures and Instruments on Space Aspects

Integrative measures of *organisational structure* and *product flow facility structure* are suitable to influence shippers' SCP requirements regarding the *'space'* dimension to allow the integration of CT into a SC concept.

The joint design of the SC network (for instance, the definition of locations for transhipment or storage facilities) improves the spatial responsiveness of the SC. To meet SCP requirements regarding location and distance for the freight transport integrative measure of the *product flow facility structure* category as well as *workflow / facility* structure measures are suitable. Through the joint design of the SC network (for instance, the definition of locations for transhipment or storage facilities) improves the spatial responsiveness of the SC. For instance, in Zermatt, Switzerland, two Swiss retailers and a private rail transport company collaboratively designed, planned, built and operate a CT terminal for the delivery of the Alp-city with goods.²⁸ Table 4 - 8 summarises the considerations of the impact of integrative measures and instruments on space-related SCP requirements.

		Im	Impact of integrative measures and instruments on													
SCP requ sion	SCP requirements dimen-		I&C facility structure				cflow vity st	/ ructu	ire		orgai tiona struc	nisa- .l :ture	plann trol m	ing and 1ethods	product flow facility structure	
		1	2	3	4	1	2	3	4	5	1	2	1	2	3	
space			1	0			0					+	0			0
	distance								\sim	0	+				\square	
	location												$>\!$		\mid \times	
	no. of sites											0 +			\checkmark	

Table 4 - 9: Impact of integrative measures and instruments on space-related SCP requirements. (+ positive impact; - negative impact; +/- negative or positive impact to be weighed up (depending on framework conditions); o no relevant impact; () indirect impact)

Impact of Integrative Measures and Instruments on Shipments Size and

Transport Quantity

The product and demand are central for the definition of shipment size and entire transport quantity. As discussed in section 3.1, the product itself as well as the demanded quantities can be influenced neither by the SC concept design nor by the usage of integrative measures and instruments. However, the integrative measure of product flow facility structure can influence shipment size and thus, transport quantity. For instance, the choice of specific load carriers and a logistics-oriented product design influence the utilisation of transport capacities and thus, shipment size. The usage of transport planning approaches for route planning, capacity planning and so on can improve the utilisation of vehicles. Thus, transport quantities and shipment size can indirectly be influenced. Table 4 - 10 summarises the considerations of the impact of integrative measures and instruments on shipment size and quantity-related SCP requirements.

²⁸ cf. 1815.ch (2011); Bahnonline (2011).

	Impact of integrative measures and instruments on														
SCP requirements	I I I&C facility structure 1 2 3 4				workflow / activity structure					orga tiona struc	nisa- l ture	planni trol m	ing and ethods	product flow facility structure	
	1	2	3	4	1	2	3	4	5	1	2	1	2	3	
shipment size / transport quantity			D			0				0		(+)		+	
		0									<	(+)	(+)	0	+

 Table 4 - 10: Impact of integrative measures and instruments on shipment size and transport quantity related SCP requirements.

(+ positive impact; - negative impact; +/- negative or positive impact to be weighed up (depending on framework conditions); o no relevant impact; () indirect impact)

Impact of Integrative Measures and Instruments on Costs

The 'cost' dimension is affected by all integrative measures. Often, transaction cost theory is applied to explain both the positive and negative impacts of measures on the cost dimension. Costs are generated before, during and after the application of integrative measures and instruments. These short-term costs must be balanced with the generated cost advantages and further positive effects on SCP in the mid- and long-term.

As shown in section 2.1.6, the integration of cross-company I&C approaches significantly influences SC costs. The planning, design, implementation and operation of I&C systems causes significant costs, but in the long-term I&C systems reduce process costs. As shown, the usage of cross-company I&C systems as well as online platforms and online data provision can reduce planning, control and monitoring effort and thus, reduce costs. The usage of auto ID technologies reduces handling processes, but causes costs for the necessary equipment (antennas, tags, etc.) and I&C systems. The standardisation of I&C technologies is the basis to reduce interface problems and to integrate the cross-company information flow between CT actors and shippers.

Integrative measures regarding the 'workflow facility structure' positively influence the cost dimension of SCP requirements. Usually, before a SC actor makes significant modifications to the process structure, the profitability and amortisation time for investments is analysed. Hence, a solely positive effect of this type of integrative measures can be assumed. Online and EDI as well as new material flow equipment / technologies (e.g., for transport and transhipment) cause initial costs to be balanced by cost savings in the long-term. The standardisation of technologies and quality management as well as the introduction of continuous improvement techniques reduce costs because of fewer interface problems and less process disturbances. The introduction of a SC-wide quality management with consistent standards improves the efficiency of business processes, reduces process disturbances and thus, reduces costs.

In the short-term, *organisational measures* increase costs. In the mid- and long-term, these costs can indirectly be saved, for instance, by reducing communication and planning times. Thus, organisational measures are thought to negatively affect the cost dimension.

Integrative measures from the field of *'planning and control methods'* simplify and accelerate material and information flow processes. This reduces overall SC costs. However, specific investments are necessary and operating costs must be included in the efficiency analysis. Owing to this conflicting impact, measures are classified as indifferent according to the specific framework variables of CT and SC concepts. Table 4 - 8 summarises the considerations of the impact of integrative measures and instruments on cost-related SCP requirements.

reading direction	Imp	oact	of in	tegr	ative	meas	ures	and i	nstru	ments	on				
SCP requirements	I&(stru	C fac Ictur	e ∵e		workflow / activity structure					organisa- tional structure		planning and con- trol methods			product flow facility structure
	1	2	3	4	1	2	3	4	5	1	2	1	2	3	
cost	+/-					+					-		+/-		+/-
	+/-	+/-	+/-	+	+/-	+/-	+	+	+	-	-	+/-	+/-	0	+/-

Table 4 - 11: Impact of integrative measures and instruments on cost related SCP requirements. (+ positive impact; - negative impact; +/- negative or positive impact to be weighed up (depending on framework conditions); o no relevant impact; () indirect impact)

Impact of Integrative Measures and Instruments on Product Type

In particular, the integrative measures of 'workflow facility structure' and 'product flow facility structure' influence the performance requirements of the 'product type' dimension. The usage of the latest material flow technologies and standardisation activities positively affect the good's quality. In particular, the application of latest load carriers and transhipment technologies enables special transport, such as that of hazardous or frozen goods. The standardisation of technologies reduces the handling effort and problems at the system interfaces and thus, improves quality. The introduction of SC-wide quality management improves the quality of products and services. Improvement techniques aim to develop product quality. Table 4 - 12 summarises the considerations of the impact of integrative measures and instruments on product-related SCP requirements.

reading direction	Imj	Impact of integrative measures and instruments on													
SCP requirements	I&C facility structure				workflow / activity structure					organisa- tional structure		planning and con- trol methods			product flow facility structure
	1	2	3	4	1	2	3	4	5	1	2	1	2	3	
product			D				+)		0		+
	$\langle \rangle$	>	<	\langle	0	+	+	+	+	\geq	<	\square	\succ	\langle	+

Table 4 - 12: Impact of integrative measures and instruments on product related SCP requirements. (+ positive impact; - negative impact; +/- negative or positive impact to be weighed up (depending on framework conditions); o no relevant impact; () indirect impact)

Impact of Integrative Measures and Instruments on Flexibility

The SCP requirements of the flexibility dimension can be influenced by all categories of integrative measures and instruments.

The measures of *I&C facility structure* improve cross-company information flow processes and thus, increases responsiveness and time-related flexibility. Cross-company I&C systems and online data provision affect the availability of information and thus, positively influence the responsiveness of the involved actors. Thus, the usage of online and joint planning and control systems and the specific technologies are the basis for the flexibility of transport services and thus, flexibility of the SC.

The measures of the category 'workflow / activity structure' positively and negatively affect SCP requirements regarding flexibility. Online and EDI accelerate communication processes and thus, increase flexibility. New and standardised material flow technologies are faster and thus, affect the flexibility of processes. Nevertheless, the higher the level of automation the more goods can be operated, the more the flexibility of the specific system is restricted. The introduction of principles of the Toyota Production System or further 'lean' logistics concepts may accelerate material flow for improved costs, but reduces the flexibility of the system (cf. section 3.1).

Cross-company 'organisational activities' increase the flexibility of cross-company material and information flow processes. The availability of operational cross-company activities, such as the integration of functional teams and key account managers, increases the knowledge of mutual processes and thus, flexibility. Strategic organisational measures can influence flexibility positively and negatively. Depending on the specific framework conditions, M&A activities and the creation of joint ventures may change the SC structure and this may influence the space and time-related flexibility dimension.

Planning and control systems and approaches allow for the flexible planning of arri-

val times and schedules and thus, improve capacity utilisation.

Product and load carrier design can influence flexibility regarding transport quantities. Table 4 - 13 summarises the considerations of the impact of integrative measures and instruments on flexibility-related SCP requirements.

reading direction Impact of integrative measures and instruments on																
SCP requirements			I&C facility structure				workflow / activity struc- ture					organisa- tional structure		ing and ethods	product flow facility structure	
		1	2	3	4	1	2	3	4	5	1	2	1	2	3	
flexibility	flexibility		+			+/-				+/-		+		+		
	time	+	+	0	0	+	+/-	+/-	+	0	+	+/-	+	+	0	+
	quantity	0	0	0	0	0	+/-	+/-	+	0	0	0	+	+	0	+
	space	0	0	0	0	0	0	0	0	0	0	+	0	0	0	0

Table 4 - 13: Impact of integrative measures and instruments on flexibility related SCP requirements. (+ positive impact; - negative impact; +/- negative or positive impact to be weighed up (depending on framework conditions); o no relevant impact; () indirect impact)

Impact of Integrative Measures and Instruments on Reliability

The performance dimension of reliability is positively influenced by the usage of latest, standardised, appropriate and proved planning and control methods based on I&C*technologies*. The availability of cross-company information because of joint I&C systems especially improves adherence to schedules. Increasingly auto ID technologies are applied to automated data gathering and processing. Often, auto ID systems are coupled with temperature sensors or location technologies and thus, reliability in terms of quality can be positively influenced.²⁹

Furthermore, adaptations in the '*workflow and activity structure*' can make material and information flows more reliable and stable. New but already standardised technologies can be assumed to be reliable and thus, improve reliability in terms of adherence of schedules as well as good quality. The introduction of quality standards is the central factor to ensure a high and cross-company level of good quality.

The choice of suitable planning, control and monitoring systems and approaches can improve adherence to schedules using sophisticated forecasting techniques. The usage of modelling and simulation techniques allows prior 'testing' of processes and thus, certain disturbance factors and problems can be forecasted and avoided.

Adaptations regarding product or load carrier design are central for the reliability of

²⁹ cf. Amador et al. (2010); López (2011); Makinwa (2010); Yang et al. (2010).

reading direction Impact of integrative measures and instruments on																
SCP requirements			I&C facility structure			workflow / activity struc- ture					organisa- tional structure		planning and con- trol methods			product flow facility structure
		1	2	3	4	1	2	3	4	5	1	2	1	2	3	
reliability				+				+				0		+		+
	adherence to schedules	+	+	0	0	+	+	+	+	0		0	+	+	+	+
	quality	0	0	+	0	+	+	+	+	0			0	0	0	0

good quality. Table 4 - 14 summarises the considerations of the impact of integrative measures and instruments on reliability-related SCP requirements.

Table 4 - 14: Impact of integrative measures and instruments on reliability related SCP requirements. (+ positive impact; - negative impact; +/- negative or positive impact to be weighed up (depending on framework conditions); o no relevant impact; () indirect impact)

The discussion of integrative measures and instruments is joined with the conceptual research framework in the following section.

4.4 Intermediate Findings

Chapter 4 applies configuration theory to develop a conceptual research framework and propositions on the performance-oriented integration of CT into SC concepts. Furthermore, the cause-and-effects between the SC and CT concept elements are specified. The measures and instruments for the integration of material and information flows identified in section 2.1.6 are discussed with regard to the applicability to the problem of CT integration and the effect on the specific SCP requirements.

Practical Recommendations on the Usage of Integrative Measures and

Instruments

The results of chapter 4 serve as practical recommendations of the performanceoriented integration of CT into SC concepts. The developed framework and underlying assumptions, propositions and cause-and-effect relationships deliver advice for practical application regarding the choice of a suitable CT concept, of adaptation points and for the usage of integrative measures and instruments. A process for the successful integration of CT into SC concepts with a focus on technical and organisational aspects is also provided.

Integration point	Ι	Π	III	IV
SC strategy	supplier - manufacturer	manufacturer- store	manufacturer - DC	DC - store
lean I (plan and optimize)	A, C	(A), (B), C	A, C	A, (B)
lean II (continuous replenishment)	А	-	А	А
leagile I (manufacturing postponement)	А	-	B, C	-
leagile II (logistics postponement)	А	-	В	-
agile	(B)	(C)	(B), (C)	-

Table 4 - 15 provides practical guidance for the identification of suitable CT concepts depending on the SC strategy and the desired integration point.

 Table 4 - 15: Applicability of CT types according to SC strategy and integration point.

 A = line trains - regular booking; B = line trains - flexible booking; C = CT network

Based on this prior choice of a generally suitable CT concept, the conceptual framework suggests the development of a so-called SCP requirements profile of the SC concept at the specific integration point as well as a performance profile of the corresponding CT concept. The section provides guidance for the comparison of requirements and performance profiles and presents a selection of theoretically derived adaptation approaches.

Table 4 - 16 summarises the findings in an overview table. The summary shows that especially the time dimension can be positively influenced by the application of integrative instruments and measures. Only organisational instruments and measures can affect the SCP requirements of the space dimension. Adaptations to the product flow facility structure positively influence requirements regarding shipment size and transport quantity. Furthermore, this type of integrative measure influences the requirements regarding product type. Adaptations in the workflow activity structure can also be applied here. The influence of integrative measures on the cost dimension has to be evaluated from both a short-term and a long-term view. In particular, integrative measures regarding the applied I&C structure, the utilised planning and control methods as well as the workflow structure impact the cost dimension. Flexibility and reliability can be positively influenced by measures of all categories. However, integrative measures of the organisational structure may cause additional costs, which have to be taken into consideration.

reading direction	impact of integrative measures and instruments on										
SCP requirements	I&C facility structure	workflow / activity structure	organisational structure	planning and control methods	product flow facility structure						
time	+	+	+	+	+						
space	0	0	+	0	0						
shipment size / transport quantity	0	0	0	(+)	+						
product type	0	+	0	0	+						
cost	+/-	+	-	+/-	+/-						
flexibility	+	+/-	+/-	+	+						
reliability	+	+	0	+	+						

Table 4 - 16: Summary - impact of integrative measures and instruments on SCP requirements. (+ positive impact; - negative impact; +/- negative or positive impact to be weighed up (depending on framework conditions); o no relevant impact; (indirect impact))

The developed conceptual framework, propositions and more specific cause-andeffect relationships must be empirically validated. Therewith, chapter 4 serves as the basis for the simulation study presented in chapter 5. The deeper evaluation of a lean SC concept and a CT concept with line transport and regular booking serves to validate and concretise the conceptual research framework presented in this chapter.

5 Performance-oriented Integration of Combined Line Transport into a Lean SC - A Simulation Study

According to McKelvey (1999), modelling and simulation can be understood as the primary scientific process if the given research problem is too complex or too costintensive for empirical observation or the application of mathematical modelling approaches.¹ In SCM, empirical research requires the observation and survey of all SC actors that are relevant to the given problem formulation. However, this requires the interruption of daily operations and material flow processes and is thus, often not possible.² Mathematical modelling approaches are usually restricted to dyadical relationships, for instance, between buyer and supplier or shipper and carrier. With every actor included in a mathematical model the complexity rises and thus, several assumptions are necessary.³ Thus, mathematical modelling seems to be insufficient for the analysis of the integration of CT into SC concepts. Simulation methodology is chosen for the in-depth analysis of the performance-oriented integration of CT into SC concepts for the thesis in hand.

The fifth chapter aims at the specification and validation of the chosen aspects of the conceptual model using a simulation study. Therefore, the problem formulation is specified to the analysis of a lean SC concept and a combined CT concept with regular line trains.⁴ First, the research design of the simulation study is introduced. The suitability of the simulation method for the given research problem, central quality criteria and restrictions of the chosen methodology are discussed. Furthermore, the choice of a specific simulation approach and a simulation tool are described (section 5.1). Second, the development of the simulation model of the distribution network of a Swiss retailer is described (section 5.2). This includes the introduction of the simulation-specific problem and target definition (section 5.3), the data gathering and editing process (section 5.4), the development of the experimental plan and setup of experiments (section 5.6) and the discussion of simulation runs and results (section 5.7). Section 5.8 joins the

¹ cf. McKelvey (1999).

² cf. Ibid. There are few publications on experiments; these are usually focused on human behaviour in SCM. $\frac{3}{2}$

³ cf. Grössler et al. (2005), p. 446.

⁴ The different CT types are introduced in section 3.4.

intermediate results with the findings from the previous chapters to formulate first practical recommendations for the performance-oriented integration of CT into SC concepts.

5.1 Research Design for the Simulation Study

This section gives an overview of the research design for the simulation study and points out the suitability of the approach for the given problem formulation. In a first step, the simulation methodology is classified in terms of research philosophy. In the second step, the advantages and restrictions of the simulation methodology are discussed. In this context, the general application fields for the simulation methodology are introduced. In the third step, different classification approaches and types of simulation models are discussed. This serves as the basis for the concluding choice of a suitable simulation approach and tool for the given problem formulation.

5.1.1 Scientific-theoretical Positioning of the Simulation Methodology

Simulation means conducting experiments with a model of a dynamic system. The simulation model is usually simplified and more abstract than is the real system, but it reflects the central characteristics of the system. The simulation allows the observation of system behaviour over time.⁵

According to Axelrod (2005), *'simulation is a third way of doing science'* (Axelrod (2005), p. 5). Inductive research approaches aim at the discovery of patterns of empirical data, whereas deductive research approaches aim at the development of assumptions and proving them. A simulation builds on specific assumptions as deductive research does. However, it does not prove these. A simulation creates data to be analysed that can be inductively compared with real life data.⁶ The generated data are not empirically gained, but rather they depend on the assumptions and rules of the simulation.⁷ Thus, it does not aim at the identification of patterns in these data (as induction does) and does not aim at the development of assumptions, but focuses on the analysis of the consequences caused by the assumptions made. Axelrod (2005) stated that simu-

⁵ cf. Zeigler (1984).

⁶ cf. Gilbert et al. (1999), p. 16.

⁷ cf. Axelrod (1997a).

lations can be understood as 'thought experiments' (Axelrod (2005), p. 5).⁸

Usually, the simulation is based on a model derived from theory and transferred to an executable simulation model.⁹ But simulations can also contribute to theory testing and development. Following Hannerman (1995), simulations *'provide an explicit and systematic way of deducing the implications of a theory as it operates under particular circumstances to make predictions about outcomes over time.* '(Hannerman (1995), p. 460).

The simulation methodology has several advantages and application fields for application in the field of SCM and CT. However, the following section introduces the restrictions of the simulation methodology that have to be taken into consideration.

5.1.2 Advantages, Restrictions and Application Fields of Simulation Methodology

This section clarifies the advantages and restrictions as well as the application fields of the simulation methodology. These considerations are the basis for the choice of the simulation methodology in general and the specific simulation approach.

Simulation can be understood as the *'middle way'* between formal modelling approaches and empirical methods, such as experiments and observations. The simulation offers, as experiments do, possibilities to examine the effect of altering one variable while holding all other variables constant.¹⁰ In general, there is better control over the experimental framework parameters.¹¹ Simulation models are formally defined. Nevertheless, no specific mathematical formulations are necessary for the analytical solving of the given problem formulation.¹² Furthermore, a simulation allows the analysis of a SC for a long time span in a compressed time.¹³

Thus, simulation can support the planning of non-existent systems. Visualisation often supports the analysis of complex cause-and-effect relationships. Often, the usage of simulation studies reduces planning costs and reveals cost risks.¹⁴

⁸ cf. Axelrod (2005), p. 5.

Simulation (involves constructing a testable model and the development process generally proceeds from theory to model to simulation. cf. Hannerman (1995), p. 458.
 of thid = 450.

¹⁰ cf. Ibid., p, 459.

¹¹ cf. Axelrod (2005), p. 8.

 $^{^{12}}$ cf. Kuhn et al. (1998), pp. 1.

 $^{^{13}}$ cf. Kot et al. (2009).

¹⁴ cf. Kuhn et al. (1998), pp. 1.

A 'trial-and-error' process is not advisable in the context of SCM and investmentintensive CT.15 Simulation methodology, in particular, allows the analysis of causeand-effect relationships and thus, of dynamic SC behaviour. It allows the systematic investigation of interdependencies between the elements of SC and CT concepts and the identification and quantification of cause-and-effect relationships.

Simulation methodology is a suitable methodology for the description and exploration of complex systems.¹⁶ Axelrod (1997)¹⁷ and (2005)¹⁸ analysed the advantages of simulation more in detail, identifying seven application fields: (1) prediction, (2) performance, (3) entertainment, (4) training, (5) education, (6) proof and (7) discovery.

(1) Prediction: Simulations can be used to calculate complex input data. These data are processed by means of hypothesised mechanisms to generate predictions about the consequences. These results can, for instance, be used to develop or refine a theory.¹⁹

(2) Performance: In particular, in the field of artificial intelligence simulations are used to accomplish specific tasks. Axelrod (2005) introduced medical diagnosis, speech recognition and function optimisation as examples. Artificial intelligence can be understood as the simulation of human decision making, perception or interaction.²⁰

(3) Training: There are several approaches of simulation systems for training purpose. Simulations train people by the provision of a dynamic, interactive and accurate environment (e.g., flight simulators).

(4) Entertainment: Some training applications are enhanced for entertainment purposes. Today, a number of simulations have complete imaginary worlds available in the computer games market.

(5) Education: The education function of simulation is closely related to training and entertainment. Axelrod (2005) introduced the computer game 'SimCity',²¹ which enables players to experiment with variables in a hypothetical city (e.g., tax rates). The education effect lies in users' ability to transfer relationships and principles to their real lives.²²

¹⁵ cf. Grössler et al. (2005), p. 447. 16

cf. Hannerman et al. (1997). 17 cf. Axelrod (1997a).

¹⁸ cf. Axelrod (2005).

¹⁹

For instance the 'digital factory'. For details see for instance Bracht et al. (2011), Kühn (2006). 20

cf. Axelrod (2005), p. 3. 21 See for instance www.simcity.ea.com.

²² cf. Axelrod (2005).

(6) Proof: Simulations can serve to provide proof for existence in general.²³ Axelrod (2005) referred to the so-called *'Conway's Game of Life'* published by Poundstone (1985), which demonstrates that extremely complex behaviour can result from very simple rules.²⁴

(7) *Discovery:* Simulation can support the validation or improvement of the model it is based on. The simulation helps discover relationships between the model's elements. According to Axelrod (2005), it can be assumed that '[...]the simpler the model, the easier it may be to discover and understand the subtle effects of its hypothesized mechanisms' (Axelrod (2005), p. 4).²⁵

As a scientific methodology, a simulation's value lies in the functions of prediction, proof and discovery.²⁶ Thus, for the given research problem a simulation can support the discovery of cause-and-effect relationships between the elements of SC and CT concepts. Furthermore, simulation can be used to predict system behaviour depending on different SC concept adaptations and can thus, support the validation and refinement of the conceptual research model presented in chapter 4.

Restrictions of Simulation Methodology

Despite the advantages of simulation methodology, there are restrictions that have to be considered when applying it.

Often simulation is misunderstood as an optimisation approach.²⁷ Although simulation results can serve as the basis of an iterative optimisation process, this does not necessarily lead to the optimal configuration of a system. Several simulation tools support the searching process for the optimal system configuration and provide enhancement modules for the optimisation of existing models.

Simulation models have subjective characters. They depend on the modelling person. The setup of a simulation can be characterised by the simplification of a given real life situation and is thus, subjective (as classical modelling). Furthermore, the quality of the simulation model depends on the simulation technique, experience, skills and creativity of the modelling person.²⁸ The simulation must also be at a suitable abstraction

²³ cf. Ibid.

²⁴ cf. Poundstone (1985).

²⁵ cf. Axelrod (2005).

 ²⁶ cf. Ibid.
 ²⁷ cf. Ibid.

²⁸ cf. Rabe et al. (2008), pp. 23.

level. It must be avoided that the model is oversimplified and thus, leads to non-resilient results.²⁹

The accomplishment of the simulation study, including data gathering and editing, model building and evaluating the simulation results, can be time consuming and cost-intensive. Thus, benefit and effort must be balanced.³⁰

The results of a simulation study are based on simplified assumptions made to reduce the complexity of the model and to reduce the simulation time. Detailed results can give a false understanding of the accuracy level of the simulation results. Thus, the generalisability of the results is restricted and this must always be considered with regard to the made assumptions.³¹ This simplification of reality negatively affects the accuracy of the simulation results. Thus, the model provides results only for the given context for the transfer to reality. For other parameters, the simulations might not be valid. Thus, the validation of the model is critical for a resilient simulation.

This discussion of the restrictions and disadvantages of the simulation methodology proves the importance of the validation and verification steps as well as sensitivity analyses of the developed model.

The following section introduces specific criteria for the evaluation of simulation studies.

5.1.3 Quality Criteria for Simulation Studies

To derive reliable statements from a simulation the specific quality criteria of the model and development process must be considered. These criteria result from the scientific-theoretical positioning and the application orientation of the thesis in hand (cf. section 1.2).³²

The underlying problem formulation and the purpose of the simulation are central to the design of the choice of simulation approach and tool, level of detail and included variables. However, independent quality criteria are suitable for all types of simulations. These quality criteria correspond to the classical research quality criteria, but extend and

²⁹ cf. Craig (2011).

 $^{^{30}}$ cf. Kuhn et al. (1998).

³¹ cf. Craig (2011). ³² cf. Gilbert et al. (1999).

interpret these for the field of simulation.³³ The simulation study must meet the targets of (1) internal and external validity, (2) usability, (3) extendibility and (4) replication.³⁴

'Validity' means the correct implementation of the model in the simulation tool. 'Inter*nal validity*' stands for the accurateness of the programming of the model itself. This is sometimes difficult to achieve, for example an unexpected simulation result may be caused by a programming mistake or it may be a surprising consequence of the model itself.³⁵ 'External validity' means the correct modelling of reality. This encompasses the correctness of model assumptions, postulated framework conditions and the empirical valid modelling of the simulation object.³⁶ In this context, the '*sensitivity*' addresses the applicability of the simulation model for changing initial situations and parameters.³⁷ This is vital especially if comparisons of different initial parameters are to be analysed. The simulation model should be robust regarding differentiations. This aspect must be proven in a verification and validation phase including a sensitivity analysis. For instance, Rabe et al. (2008) presented a number of different validation and verification techniques for simulation studies.³⁸

'Usability' means that the researcher as well as all other users and readers are able to follow and understand the structure and functioning of the model as well as the programming and interpretation of the simulation results.³⁹

'Extendibility' means that the simulation can be adapted for further use and other users. During the development and documentation of the model, this aspect must be considered. Extendibility is closely related with ability for replication.⁴⁰

'Replication' confirms whether the claimed results of a given simulation are reliable in terms of reproduction. This confirmation is necessary to avoid the publication of results based on programming errors, misrepresentation and analysing errors.⁴¹

³³ See for instance Albers et al. (2007), pp. 33; Lamnek (2005), pp. 118; Lienert et al. (1998), pp. 7.

³⁴ cf. Axelrod (1997a), p. 7.

³⁵ cf. Axelrod (2005), p. 7; Axelrod (1997b). 36

cf. Rabe et al. (2008), pp. 15. cf. Ibid., p. 16; Scharkow (2004), p. 7. 37

³⁸

cf. Rabe et al. (2008), pp. 93. 39

cf. Axelrod (2005); Wenzel et al. (2008). 40

cf. Axelrod (2005); Scharkow (2004), p. 7; Wenzel et al. (2008). 41

cf. Scharkow (2004), p. 7.

The 'kiss' principle' ('Keep it simple, stupid!) is a recommended approach to reach the stated quality criteria.⁴² The principle considers that researchers and the general audience have limited cognitive ability. If there is a surprising result, it is helpful to be confident that everything in the model can be tracked. Simplicity allows other researchers the chance to replicate the model and to extend it in other directions.⁴³

5.1.4 Structure and Classification of Simulation Models

This section introduces different classification approaches for simulations as the basis for the choice of a specific simulation approach and tool in section 5.1.5.

A simulation model usually consists of simulated units as well as objects and transition rules. Depending on the given application field, simulation units can be variables, objects, cells or agents. The transition rules describe how the simulation units change over time. The simulated units and rules can be depicted using different simulation approaches. The choice of suitable simulation approaches depends on the specific target of the simulation and the necessary level of abstraction.⁴⁴

Continuous vs. Discrete Simulation Models

Simulation time can be either continuous by proceeding in discrete time intervals or discrete and event-based.⁴⁵ In a *continuous simulation*, variables change in arbitrary small time steps. The mathematical basis of continuous models are (non-linear) differential equation systems. These can be solved by numerical integration. *Discrete models* are restricted to discrete time steps with discrete value changes of the state variables. A phase-based discrete simulation time proceeds in a defined step width. This is suitable for online observations of the simulation since simulation time usually has a defined relation to actual time. Modelling is comparably easy, but simulation time might be wasted if many simulated variables are inactive for a longer time. An *event-based simulation* has flexible time steps based on the appearance of events.⁴⁶ Simulated units are only active if they are activated by an event. After the reaction of an event and the passing of the effect, the current time is set on the time point of the next event. The

⁴² cf. Axelrod (1997a).

⁴³ cf. Ibid.; Axelrod (1997b); Axelrod (2000).

⁴⁴ cf. Herrler (2007).

⁴⁵ cf. Ibid.; Troitzsch (2004).

⁴⁶ cf. Brade (2003); Kuhn et al. (1998); Stefanovic et al. (2009).

main disadvantage is the elaborative handling of the event list since the simulation time includes time jumps.⁴⁷ For the model building for discrete event simulations, standardised elements, such as random numbers, queuing lines and probability distributions, can be used. Furthermore, petri net theory can be applied to the development of discrete event simulation models.⁴⁸

Furthermore, so-called macro and agent-based simulations can be distinguished. Macro simulations are based on sets of differential equations or system dynamics. System dynamics simulations are usually applied to macro policy-type problems.⁴⁹ In an agent-based simulation, the population variables and other state variables are replaced by the simulation of the behaviour and interaction of the single objects.⁵⁰

Deterministic, Stochastic and Interactive Simulation Models

Simulation models can be distinguished with regard to the type of variables used. In a *deterministic simulation model*, all environmental factors and the behaviour of the model must be clearly determined. The real system is modelled without coincidence parameters and according to clear rules. Thus, each simulation run leads to identical results.

However, a simulation model can include *stochastic and interactive elements*. A simulation model is stochastic if its behaviour is determined by one or more random variables. Models are dynamic to the extent that time is an overarching component of the model and its corresponding behaviour.⁵¹

Probability distributions can be understood as abstraction elements. Instead of stable cause-and-effect relationships, probability distributions derived from reality can be used for the simulation model.⁵² If stochastic elements are included in the simulation model, the results of the simulation run are not determined. Thus, it is necessary to accomplish a series of simulation runs to get convincing results.⁵³

⁴⁷ cf. Brade (2003).

⁴⁸ cf. Hung et al. (2006); Biswas et al. (2004)

 ⁴⁹ For details on system dynamics applications in the field of SCM see Forrester (1995); Georgiadis et al. (2005); Holmes (1991); Lane (1999).
 ⁵⁰ Lane data and a simulation in the field of SCM see Chang et al. (2001); Hornlar (2007); Kuhn et al. (1908); Lättilä et al.

For details on agent-based simulation in the field of SCM see Chang et al. (2001); Herrler (2007); Kuhn et al. (1998); Lättilä et al. (2010); Troitzsch (2004).
 of Dealeur (2000)

⁵¹ cf. Dooley (2002).

⁵² cf. Troitzsch (2004). ⁵³ In 2010 Owen publish

⁵³ In 2010 Owen published a literature review on the usage of different simulation methods in the field of SCM.

5.1.5 Application Fields of the Simulation Methodology in Transport and SCM Science

Simulation has become an important approach for decision support in the field of SCM. In practice, especially in the field of production, it has been applied for the validation of new layout choices, ⁵⁴ capacity planning and the evaluation of PPC approaches (shop floor planning, PPC, (dynamic) scheduling of the production orders, capacity plans and labour allocation).⁵⁵ Lately, several other application fields, e.g., SC planning, product development and design and marketing, have risen to prominence.⁵⁶

According to Goldsim (2011), four decision support situations for simulation models in SCM can be distinguished: (1) optimisation problems, (2) decision analysis, (3) diagnostic evaluations and (4) risk management.

A simulation for (1) optimisation purposes aims at finding the optimal guidelines to either maximize or minimize a certain result by means of iterative improvements, e.g., the optimal inventory level, an allocation of DCs and stores or the machine order planning. A simulation for (2) decision analysis is a quantitative comparison of two or more given alternatives, e.g., the simulation-based analysis of the performance effect of the SC caused by different location alternatives for a new factory. (3) Diagnostic evaluations can be used to identify problems with an unknown cause. The SC simulation improves the understanding of cause-and-effect relationships and allows the evaluation of different solution possibilities. Simulation supports (4) SC risk management in order to avoid unexpected disruptive events, e.g., the disruptions of material flow and to evaluate risk impacts such as fire, floods or labour strikes. This can help anticipate risk and develop suitable mitigation plans.⁵⁷ Changes in one segment of a SC can cause short-term and long-term effects on SCP. To avoid unexpected disruptions, e.g., the effects of outsourcing particular processes, building new factories or switching suppliers can be anticipated. Thus, in contrast to decision analysis, simulation is used to support the implementation of the project in order to reduce costs, meet schedules and minimize risks.58

200

⁵⁴ cf. Kuhn et al. (1998); Kühn (2006); Rabe et al. (2008).

⁵⁵ cf. Kosturiak et al. (1999).

⁵⁶ cf. Ibid.

⁵⁷ cf. Goldsim (2007), pp. 6. ⁵⁸ cf. Ibid., p. 7.

Terzi et al. (2004) conducted a literature review on the application fields of the simulation methodology in SCM.⁵⁹ The authors distinguished between the different objectives of simulation studies (e.g., strategic, tactical or operative), different simulation approaches (simulation tools and languages) and different development stages of the models (conceptual models to commercial applications).⁶⁰

According to Terzi et al. (2004), the two main objectives for the use of simulation models in the field of SCM and transport are network design and strategic decision support.⁶¹ Network design includes the determination of nodes in the network as well as the design of the connections between these.⁶² Strategic decision support can mean the evaluation of the suitability of strategies and concepts, such as quick response, CPFR or outsourcing to third parties.⁶³

A simulation methodology can be applied to all central process types in SCM. Following the classification of APS systems, simulations can be applied to demand and sales planning, SC planning, supply and distribution channel configuration, location of stocks, selection of suppliers and partners,⁶⁴ inventory planning over multiple stages, distribution and transportation planning⁶⁵ and production planning and scheduling.⁶⁶

Furthermore, simulation models have been applied to specific problems, such as the investigation of the effects of uncertainty,⁶⁷ different order release mechanisms,⁶⁸ partial shipments⁶⁹ specific handling efforts and different inventory policies, ⁷⁰ and the impact of transhipments on service levels and costs.⁷¹ Disney and Towill (2002), for instance, analysed the configuration of VMI approaches,⁷² including transport processes.⁷³

⁵⁹ cf. Terzi et al. (2004).

⁶⁰ cf. Ibid., p. 10.

 $^{^{61}}$ cf. Owen (2010).

⁶² cf. Terzi et al. (2004), pp. 23. ⁶³ cf. Ibid., pp. 25.

⁶⁴ cf. Kosturiak et al. (1999); Schunk (2000); Stadtler (2005); Stadtler et al. (2000).

⁶⁵ cf. Bagchi et al. (1998).

 ⁶⁶ cf. Ibid.; Biswas et al. (2004); Chan et al. (2011); Chatfield et al. (2009); Goldsim (2007); Kot et al. (2009); Owen (2010); Rodriguez-Rodriguez et al. (2011); Stefanovic et al. (2009); Sen et al. (2004); Schunk (2000); Thron et al. (2006); Thron et al. (2007); Yang et al. (2009).

⁶⁷ cf. Petrovic (2001).

 ⁶⁸ cf. Chan et al. (2001); Chan et al. (2002).
 ⁶⁹ cf. Baneria et al. (2001)

⁶⁹ cf. Banerjee et al. (2001).

⁷⁰ cf. Towill (1996); Disney et al. (1997). ⁷¹ af Papariae at al. (2003); Hang Minh at

⁷¹ cf. Banerjee et al. (2003); Hong-Minh et al. (2000).

⁷² cf. Disney et al. (2002).

⁷³ cf. Disney et al. (2003)
Furthermore, there is a number of applications of simulation methodology in the field of CT. This includes the evaluation of transport and terminal logistics,⁷⁴ the usage of optimisation methods, vehicle planning, stowage planning, capacity planning for cranes and overheads, storage and stacking logistics, transport optimisation, crane transport optimisation and rail terminals.⁷⁵

In general, simulations support strategic, tactical and operational activities. There are several publications on the choice of different terminal layouts and used handling equipment regarding efficiency and cost (strategic).⁷⁶ Terminal logistics and optimisation methods for CT are another application field for simulation. In particular, for large terminals the simulation technique has gained much interest (operational).⁷⁷ The analysis of process alternatives, for instance in the case of disturbances, is a rather tactical application field.⁷⁸

According to the findings on the advantages and application fields of the simulation method, the following section addresses the applicability of the simulation methodology to the given research problem.

5.1.6 Discrete Event Simulation as the Research Methodology for the Analysis of Performance-oriented Integration of CT into SC Concepts

Simulation is chosen as the research methodology for the problem of the performanceoriented integration of CT into SC concepts. The simulation methodology supports the validation and concretisation of the conceptual model on the integration of CT into SC concepts. Since the technical and organisational elements of SC and CT concepts are central for the thesis in hand, an object-oriented simulation approach is chosen.

This simulation methodology allows analysing the performance effect of CT integration. The simulation methodology is able to deal with:

(1) The high complexity and multiplicity of variables and

(2) Non-linear and time-delayed cause-and-effect relationships between variables.

⁷⁴ cf. Ballis et al. (2004); Dessouky et al. (1995); Ferreira et al. (1993); Hartmann (2004); Klima et al. (1996); Liu et al. (2002); Longo (2010); Ramirez-Marquez et al. (2008).

⁷⁵ cf. Longo (2010); Rizzoli et al. (2002); Ramirez-Marquez et al. (2008); Steenken et al. (2004).

⁷⁶ cf. Abacoumkin et al. (2004). Ballis et al. (2004); Ballis et al. (2002); Longo (2010); Ramirez-Marquez et al. (2008); Rizzoli et al. (2002).77 cf. Gambardella et al. (1998), Rizzoli et al. (2002); Liu et al. (2002); Nam et al. (2002).

⁷⁸

cf. Hartmann (2004); Shabayek et al. (2002).

Furthermore, real experiments in the SC context can be avoided. Thus, the simulation methodology allows the validation and specification of the conceptual research framework introduced in the previous section. In particular, the simulation can predict, prove and discover (cf. section 5.1.2), which is central for the identification of additional relevant cause-and-effect relationships between the elements of SC and CT concepts. It can also analyse the performance effects of the integration of CT into SC concepts and exam the impact of adaptations the SC concept to improve the fit between SC and CT concepts.

Following Chang and Makatsoris (2001), the application of a discrete event simulation approach allows the evaluation of SCP before and after the integration of CT.⁷⁹ It allows in particular what-if analysis to improve the planning basis and the comparison of operational alternatives without the interruption of the running SC and thus, contracts the time for decision processes.

There is a variety of tools offered by different providers and available for the purpose of simulation, e.g., eMplant, Arena, Witness or Quest. Furthermore, object-oriented programming languages such as C++, UML, swarm or Java can be applied.⁸⁰ For the choice of the simulation tool in the thesis in hand, the following two aspects are critical:

(1) *The possibility to address discrete events:* SCs are significantly influenced by random, discrete events (e.g., the disruptions in supply caused by strikes or congestion). The ability of the simulation software to represent these events is important for the evaluation of flexibility, robustness and overall performance.

(2) *The capability to link an external database:* The performance-oriented simulation of CT integration into SC concepts requires information on current inventories, order rates and arrival times. Owing to the large amount of data on the orders, route plans and transport schedules, transport and handling times, transport distances, store opening times and delivery windows, the entering of data by hand into the simulation tool is time-consuming. Thus, the simulation tool must provide an interface to a database system including the relevant input data as well as to document the history data of the simulation runs.

⁷⁹ cf. Chang et al. (2001).

⁸⁰ cf. Scholz-Reiter et al. (2008), p. 121.

The simulation tool used for the thesis in hand is the software '*Flexsim*'.⁸¹ Flexsim is an object-oriented simulation tool based on simulation modules that can be customised according to the specific application field. Flexsim provides a clearly arranged graphical user interface. Furthermore, an integrated 2D and 3D virtual reality animation is included that can display processes in real time.

Flexsim is based on a C++ compiler and FlexScript (a C++ function library) for model development. Thus, all C++ libraries and functions can be utilised.⁸² It is possible to link Flexsim with any ODBC⁸³ database (e.g., Oracle, Access) and data structures (e.g., text files, MS Excel and MS Word files). Furthermore, there are specific applications offered by Flexsim for manufacturing and material handling systems, as well as the simulation of semiconductor fabs and tools, marine container terminals, hospitals and network storage systems.⁸⁴

As a result, Flexsim is chosen as a suitable instrument for the modelling and simulation of the integration of CT into a lean SC. In particular, the object and discrete event orientation and the possibility to link the tool with external databases for input and output data are central decision parameters.

5.2 Process Model for the Simulation Model Development

This section introduces the procedure for the development of the simulation model and the accomplishment and analyses of simulation runs.

The simulation study in the thesis in hand follows a five-step approach following the considerations of Rabe et al. $(2008)^{85}$ and Wenzel et al. $(2008)^{.86}$ In the following section, the five steps are briefly introduced.

Step 1: Problem Formulation and Target Definition

The first step includes the specification of the problem formulation and the definition of targets for the simulation study. The phase includes the definition of the quality criteria for the simulation and of the used terminology.⁸⁷

⁸¹ For further details on the simulation software see http://www.flexsim.de/.

⁸² Additional applications such as ExpertFit®, OptQuest ® and Visio ® can be compiled into the application. cf. Flexsim (2011)

⁸³ ODBC = open database connectivity

cf. Diamond et al. (2002). The as-is analysis is independent of the simulation tool or language.

⁸⁵ cf. Rabe et al. (2008)

⁸⁶ cf. Wenzel et al. (2008), p. 109.

⁸⁷ cf. Ibid., pp. 114.

Step 2: Data Gathering and Editing for Model Configuration and Validation

The second step encompasses the gathering of all relevant data for the simulation study as well as data editing for further processing in the case of data shortages or abundance. Furthermore, the second step includes the testing and definition of probability distributions for the stochastic variables in the simulation model.⁸⁸

Step 3: Development, Validation and Verification of the Simulation Model

The third step involves building the conceptual model, formalised model and simulation model.⁸⁹ Therefore, the SC and CT concepts are analysed, modelled, formalised and finally implemented. Several verification and validation steps ensure the compliance of the developed models with the as-is situation (external validity), the functioning of the model (internal validity) and suitability for the defined research target. A sensitivity analysis makes sure that the model is also valid for a defined range of different input and output parameters.

Step 4: Experimental Plan and Setup of Experiments

The quality of a simulation can be increased by the usage of a predefined experimental plan. It avoids that only the simulation results guide the study, but all interesting aspects are analysed in a structured way.⁹⁰ The simulation scenarios are the basis for the definition of specific experiments. Simulation runs are accomplished according to the experimental plan. The length of a simulation run and, if necessary, of a stabilisation phase, the initial values for the variables and objects as well as the number of simulation runs are determined.

Step 5:Description and Analysis of Simulation Results

In the fifths step, simulation results are analysed, depicted and interpreted.⁹¹ The analysis is completed by a discussion of scope, assumptions and simplifications of the simulation study and the transferability and generalisability of findings.

This classification into five steps for the process of a simulation study serves as a structure for the following sections (section 5.3 to section 5.7).

⁸⁸ cf. Ibid., pp. 119.

This conceptual model must be distinguished from the conceptual research model introduced in section 4.

⁹⁰ cf. Rabe et al. (2008); Wenzel et al. (2008), p. 139.

⁹¹ cf. Wenzel et al. (2008), pp. 139.

5.3 Step 1: Problem Formulation and Target Definition

According to Chang et al. (2011), in the first phase of a SC simulation project an understanding of the overall SC is necessary.⁹² This includes the analyses of industryspecific characteristics (e.g., SC strategy, SC environment, SC actors) and companyspecific characteristics, and the description of the empirical database. Furthermore, the choice of specific performance indicators,⁹³ the development of simulation scenarios and the formulation of the expected simulation results are part of the first simulation phase.⁹⁴

5.3.1 Scope and Target Definition

In a first step, the specific problem and target of the simulation study is defined. To specify and validate the conceptual model presented in chapter 4 one aspect of the framework is selected for the simulation study. The simulation focuses on the analysis of a lean SC concept (type II - continuous replenishment) and the performance-oriented integration of a line transport with regular booking. The analysis is concentrated on the integration of CT offers between the manufacturer and the DC/store. Table 5 - 1 visualises this specification.

SC strategy	I - supplier - manufacturer	II - manufac- turer- store	III - manufac- turer – DC	IV DC - store
lean (type I - plan and optimize)	A, C	(A), (B), C	A, C	A, (B)
lean (type II - continuous replenishment)	А,	-	А	А
leagile (type I - manufacturing postponement)	А	-	B, C	-
leagile (type II - logistics post- ponement)	А	-	В	-
agile	(B)	(C)	(B), (C)	-

Table 5 - 1: Focus of simulation study. (A – line transport (regular booking), B – line transport (flexible booking), C – CT network; A – suitable, (A) restricted suitable)

⁹² cf. Chang et al. (2001).

⁹³ cf. Ibid.

⁹⁴ cf. Arnold (2010).

The targets of the simulation study are:

- To compare the performance effect of CT integration with unimodal road transport,
- To discuss the performance effect of single adaptations to the SC concept, namely adaptations to the production and distribution concepts and
- To uncover unknown cause-and-effect relationships between the elements of the CT and SC.

Thus, in the following section the empirical database and the specific problem formulation are introduced.

5.3.2 Problem Formulation - Understanding the Initial SC Concept

The empirical data for the simulation study are mainly provided by *Manor*, the biggest Swiss department store owner. *Manor* AG owns 70 (2010) department stores and belongs to the *Manor* Group.⁹⁵ Recently, *Manor* experienced an extension and instability of transport times and thus, a reducing adherence to schedules for store deliveries. These delays were caused by congestion related to infrastructure shortages or weather. Transport political regulations (e.g., night, weekend and inner city bans) restricted transport flexibility as well as the enhancement of transport times. In particular, the sustainability strategy of *Manor* urges the transport division to reduce emissions and to improve *Manor*'s 'green image'.⁹⁶

The simulation model is based on a part of *Manor*'s food and non-food distribution network in Western Switzerland. This part of the model was chosen since *Manor* actually considers CT integration in this region and has already accomplished a pilot project for the delivery of the department store in Geneva. *Manor* provides empirical data and experience about the project for the validation of the simulation study in the thesis in hand. Thus, the simulation results can be complemented by qualitative data.

There are eleven department stores in Western Switzerland. They are located in a circle of approx. 80 km from the CT terminal of *Swiss Post* in Daillens. Non-food products are sent to the stores from the two main DCs in central Switzerland (Möhlin and

⁹⁵ The Manor Group also owns FLY and Athleticum, two stores for furniture and sports equipment and fashion. The Manor Group belongs to the Maus Frères Holding in Geneva. cf. Manor (2011b); Maus (2011). Manor has an annual turnover of 2.987 Bio. CHF (2010) and approximately 11,000 employees. cf. Manor (2011a). Manor is the largest department store group in Switzerland and has in this segment a market share of approximately 59%. The biggest competitors are Migros (Globus), Coop City department stores, Jelmoli and Loeb

⁹⁶ cf. Manor (2011c).

Hochdorf). For some deliveries, the shipments of the two DCs are bundled. In this case, the products are transported from Möhlin to Hochdorf by road. Both DCs are close to the CT terminal of *Swiss Post* in Härkingen.

The stores report their sales on a daily basis. They order goods via either next day or same day deliveries. Next day delivery means that the orders must be made by 4:00 PM to be delivered on the next day. The delivery time point depends on the specific opening times of the delivery platform / loading ramp as well as the external restrictions, such as inner city, weekend and night time driving bans.

Manor uses a classification of products according to priority classes. There are priority goods for same or next day delivery. These goods are always transported to the destination, even if an additional transport vehicle is necessary. Non-prioritized goods are seasonal or action goods, which are provided some time in advance to balance transport capacities. This means that the trucks load all prioritized products before loading non-prioritized goods. For non-prioritized products, no additional transport processes are initiated.

Manor accomplishes transport and route planning itself. For transport, *Manor* assigns different road carriers. On a daily basis, these road carriers announce the specific truck characteristics (e.g., ID number, capacity) as the basis for transport and route planning. Transport and route planning are made once for the period January to September and there is a second plan for October to December, reflecting the increase in demand for goods at Christmas. The basis for the thesis in hand is the basic transport and route plan. From October to December, transport quantities are significantly higher. Thus, in accordance with *Manor*, the CT integration for the basic service is evaluated. In practice this can complemented by additional road or CT for the *'high season'*^{*P*7}.

Depending on the product type and quantities, road carriers either deliver the goods in milk runs or via direct transport (cf. section 3.3.3). Furthermore, there is a regional platform in Sion where products from different DCs can be recombined. Figure 5 - 1 provides a schematic overview of the most important sites in the simulation study.

⁹⁷ For the resulting restrictions and generalisability see section 5.7.6.



Figure 5 - 1: Schematic of the as-is situation – Manor distribution network (Western Switzerland).⁹⁸

Recently, *Manor* has considered the integration of CT for store deliveries in Western Switzerland. A one-week test run was completed in May 2011 for delivery to stores in Geneva. The simulation study supports the analysis of the CT into SC concepts and allows the firm to identify suitable adaptation approaches to increase SCP.

5.3.3 CT Integration Scenarios

The integration of CT into the SC of *Manor* is supposed to be accomplished over the terminal network of *Swiss Post*.

*Swiss Post*⁹⁹ operates three CT terminals in central (Härkingen), Eastern (Frauenfeld) and Western Switzerland (Daillens) at the three package distribution centres. They are connected by regular line transport.¹⁰⁰ *Swiss Post* offers as CT operator-free capacities on the trains on a regular and flexible basis to shippers. Owing to the advantageous locations of the terminals, the offer of CT services attracts shippers with distributed sites in central Switzerland as well as in Eastern or Western Switzerland. These termi-

⁹⁸ Own illustration.

Swiss Post has its headquarters in Berne is with approximately 42,000 employees (2010). It is thus the second largest employer in Switzerland with a revenue of 7.895 Bio. CHF (2010). Swiss Post runs about 2600 stores and transports approximately 3 Bio. shipments, 100 Mio. packages and approximately 100 Mio. passengers. Swiss Post runs for the transport of packages three main package DCs. Furthermore, there is quantity of letter DCs and regional letter DCs. Post (2011c). The central letter distribution centres are connected by regular train transports and all their own siding tracks. cf. Post (2011a); Post (2011b).

¹⁰⁰ cf. Post (2011a); Post (2011b).

nals are connected by daily trains by *SBB Cargo*¹⁰¹ and *Railcare*¹⁰². The road carriers *Hugelshofer* and *Dreier* accomplish pre- and ongoing road transport by offering vehicles and driving personnel.¹⁰³ These five companies founded a cooperation to offer CT services to different customers.¹⁰⁴ Each member of the cooperation can act as the central marketing, planning and control company to provide the overhead services to different customers.

The simulation scenario includes modelling the bundling of transport flows for the two DCs and the usage of regular line CT between the CT terminal in Härkingen and Daillens. *Swiss Post* offers line transport for regular and flexible bookings eight times a day in both directions.

The simulation study aims at the analysis of the shift from unimodal road transport to CT for store delivery. The effects on SCP and on the adaptations of the production and distribution concepts to improve integration are examined. Therefore, it is assumed that the transport routes connect the same stores as in the initial situation. Figure 5 - 2 gives a schematic overview of the simulation scenario including CT.

¹⁰¹ SBB Cargo is the biggest rail carrier in Switzerland with approximately a 72% market share. For further information see www.sbbcargo.ch.

Railcare is a comparably small rail carrier focusing on so-called 'pendel trains' for CT. In 2010 Coop bought Railcare. For further information see http://www.Railcare.ch/ and the presented mini cause in section 2.15.

¹⁰³ Hugelshofer and Dreier are Swiss road carriers and exclusive partners for Swiss Post for package deliveries. Furthermore, they are partners for the CT service offered by Swiss Post. Hugelshofer is located in Frauenfeld and accomplishes the pre- and ongoing haulage in Eastern Switzerland. For detailed information on Hugelshofer see Hugelshofer (2011). Dreier, located in Suhr, focuses on the transports in Western Switzerland. For detailed information on Dreier see Dreier (2011).

¹⁰⁴ cf. Post (2011b). enhanced by expert interviews.



Figure 5 - 2: Schematic scenario – Manor distribution network including CT (Western Switzerland).¹⁰⁵

The expected results are the stabilisation of transport times and improved adherence to schedules for store delivery. Furthermore, total emissions can be reduced because of the better emission balance of rail transport in comparison with unimodal road transport. Additionally, a positive impact on the utilisation and number of transport vehicles is assumed.

Furthermore, in this scenario adaptations to the SC concept, namely the production and distribution concepts, are analysed.

For the discussed scenario of CT integration, the expected results are summarised in the form of hypotheses. However, because simulation does not allow the testing of hypotheses in a classical quantitative way, in the following section the term '*scenario*' is applied to the formulation of expected results. These statements correspond to the different simulation setups and give important hints for the setup of the experimental plan (cf. section 5.6).

¹⁰⁵ Own illustration.

5.3.4 Performance Indicators for the Evaluation of Simulation Scenarios

For the analysis of the developed scenarios, the key figures or performance indicators derived in section 2.3.5 are applied. Performance indicators with a focus on the single order and with a focus on the entire SC concept can be distinguished.

Figure 5 - 3 visualises the performance indicators applied to the simulation study.

Lead time and *adherence to schedules* are evaluated separately for each order and thus, information about the average values of these performance indicators can be gained. The *utilisation of capacities* as well as the *inventory level* are indicators for the evaluation of the performance of the entire system (the specific transport system, production system or the different transport vehicles or production).

Total *lead time* can be calculated by the different lead time parts, such as transport time, handling time, production time, processing time, storage time and waiting time. Adherence to schedules means the meeting of delivery windows for loading and unloading of goods. In the centre of interest is here the meeting of the delivery window and the destination store. For the simulation study, 'adherence' is interpreted in two different ways. (1) Adherence to schedules regarding deviations means that the delivery neither happens too early nor too late. (2) Adherence to schedules regarding delays means that only too late transports are interpreted as non on time. This differentiation was made for two reasons. First, the experts prove that in retail practice temporal buffers are planned to balance unexpected delays. If the transport is too early, the truck driver waits near the destination until the beginning of the delivery window. Second, to improve the generalisability of findings, deviation is analyzed, too. For instance in manufacturing companies or in high frequent transport concepts (e.g., JIS) neither too early nor too late deliveries are acceptable. According to the expert interviews, the delivery windows for store delivery has typically a time span of +/-15 minutes before or after a defined time point. To improve generalisability of findings, again an alternate delivery window size of +/- 30 minutes is analyzed. Additionally, the inventory in the SC is evaluated according to the inventory levels at different sites as well as the amount of goods on transport vehicles. Furthermore, emissions can be calculated for both the order level and the system perspective. Emissions are calculated based on the database provided by Infras (2011) on CO₂, NO_X and particle emissions for different

transport distances, routes and inclination profiles as well as transport vehicle types.¹⁰⁶ A calculation of transport costs is not possible for the simulation study. Therefore, encompassing data on transport cost rates for different transport distances, rates for capital commitment caused by different inventory levels, transhipment costs and overhead costs are necessary. In particular, for the operators and carriers these data are competition-relevant and thus, not available for the simulation study.



Figure 5 - 3: Performance indicators for the evaluation of simulation scenarios.¹⁰⁷

Initially, the performance effect of the CT integration in general is analysed. It is assumed that the integration of the regular line CT into the given lean SC concept leads to the general improvement in SCP. Following the argumentation line of the conceptual framework, the improvement can affect the different dimensions of SCP.

¹⁰⁶ cf. Infras (2011b).

¹⁰⁷ Illustration based on findings in section 2.3.4.

5.3.5 Development of Statements on Expected Simulation Results

Impact of CT Integration on SC Efficiency

Following the considerations in section 2.3 and 4.1, it must be assumed that additional processes generally cause longer lead times (*SCP in terms of lead times*).¹⁰⁸ Furthermore, it can be assumed that the integration of CT leads to more stable lead times and a higher adherence to schedules because in comparison with unimodal road transport on main haulage there are no delays (*SCP in terms of adherence to schedules*). Shorter lead times, reduced transport distances on the road and bundled transport volumes lead to a lower number of required trucks. Thus, the integration of CT may lead to a reduction in overall SC costs (*SCP in terms of costs*). Finally, the shorter transport distances on roads, the bundling effect on main haulage and the better emissions balance of rail transport in comparison with road transport lead to a decrease in emissions levels (*SCP in terms of emissions*).

These considerations are summarised in a first and a second statement on the expected results for the simulation study:

 E_1 : The integration of CT impacts SCP in terms of (a) enhancement of lead time, (b) adherence to schedules and (c) reduction of emissions (in comparison with a unimodal road transport concept).

 E_2 : The integration of CT allows the (a) reduction of road transport vehicles and the (b) improved utilisation of road transport vehicles.

Impact of CT Integration on SC Effectiveness

Furthermore, it is assumed that SC effectiveness is influenced by the integration of CT. Following the consideration in sections 2.3 and 4.1, it can be assumed that the total inventory in the SC increases because of the longer lead times. In comparison with unimodal road transport, time flexibility, e.g., expressed by short-term changes of transport quantity or transport time, decreases since the CT concept depends on defined train schedules with restricted capacity and a comparably high level of fixed costs. These considerations are summarised in a third statement on the expected results of the simulation study.

¹⁰⁸ The CT concept offered by *Swiss Post* is based on comparably short lead times for the rail transport. Owing to the preferred usage of 'pendel trains' the trains have an even higher speed than trucks have. For some specific configurations it can even be assumed that despite the additional handling and transhipment times CT is even faster than is unimodal road transport.

 E_3 : CT integration leads to an (a) increase in the total inventory level and (b) a reduction of time flexibility (in comparison with a unimodal road transport concept).

Although the last statement describes a decline in SCP in comparison with unimodal road transport, the integration of CT must not be refused. This statements can serve as the basis to discuss the trade-off relationships between the 'classical' SCP dimensions and the rather 'innovative' dimensions of adherence to schedules and emissions (cf. sections 2.1 and 2.3).

These three statements serve to explain the general influence of CT integration. The asis simulation model (basic scenario) and CT scenario I serve as benchmarks for the analysis of the 'fit' between CT and SC concepts.

SCP Influence of the Alignment of the SC Concept to the CT Concept

To make statements on the impact on SCP of the goodness of the '*fit*' between CT and SC concepts different adaptations to the SC concept are analysed (CT scenario II, III and IIIa). For each scenario only one parameter is varied while all others are kept constant.

Adaptations of the Production Concept

The adaptation of arrival and departure times in the SC leads to a reduction in waiting times. Thus, it can be assumed that the harmonisation of production arrival times with departure times in CT reduces total lead times as well as adherence to schedules. Furthermore, this synchronisation of processes increases time flexibility; because of the aligned arrival and departure schedules, the planning horizon is improved and enhanced for all SC members. Thus, the shifting of transport quantities or transport times can easily be accomplished. According to the considerations in section 3.3.3, production arrival times can be influenced by changes in the production control logic as well as changes in the production and the stocking type. The considerations are summarised in the fourth statement on the expected results:

 E_4 : The harmonisation of production arrival times with the CT concept leads to a (a) reduction in lead times, (b) increased time flexibility and (c) improved adherence to schedules.

Adaptations of the Distribution Concept

Comparable considerations are valid for the adaptation of the distribution concept (cf. section 3.3.4). As explained above, delivery windows are gaining increasing interest in

practice. In particular, for store deliveries in urban areas, the delivery windows are small and adherence to these schedules is critical. On one hand, regulations, especially in city centres, restrict the length of the delivery windows. On the other hand, there are specific personnel for the placement of goods in the store. Usually, there are no storage capacities, only buffers for the delivered goods. Thus, the goods must be placed on the shelves as soon as they arrive. The expert interviews showed that the store manager tends to be comparably inflexible in the adaptation of delivery time windows and the planning of personnel for unloading. However, the simulation is supposed to show the advantages of an enhancement and the harmonisation of delivery windows with the CT concept. The enhancement leads to an improvement in adherence to schedules.¹⁰⁹ This can positively influence time flexibility, too. The enhancement and harmonisation of delivery windows can also reduce overall lead times because waiting times can be avoided. These considerations are summarised in the fifth statement on the expected results:

 E_5 : The (a) enhancement and the (b) harmonisation of store delivery windows with regards to the CT concept leads to (1) a reduction of lead times and (2) an improvement in adherence to schedules.

As discussed for production concepts, the enhancement of delivery times and thus, total lead time improves the applicability of the CT concept. In retail, there are often short delivery times, for instance same day deliveries, with morning orders and deliveries in the afternoon. Usually, this planning horizon is not sufficient for the integration of CT concepts. Thus, in a sixth statement on expected results aims at the enhancement of delivery time between stores and DCs and manufacturers. It is expected that the enhancement of delivery times causes an increased lead time and thus, increases total inventory, but positively influences adherence to schedules and time flexibility because of the longer planning horizon and thus, the utilisation of transport vehicles.

 E_6 : The enhancement of delivery times leads to (a) increased lead times, (b) increased inventory levels, (c) improved adherence to schedules, (d) time flexibility and the (e) utilisation of transport vehicles.

The following section discusses the data gathering and editing process as the basis for the development of a simulation model.

¹⁰⁹ However, it may cause increased personnel costs.

5.4 Step 2: Data Gathering and Editing for Model Configuration and Validation

This section describes the process of data gathering and editing for the purpose of model configuration and validation. Different data sources were the bases for the simulation model.¹¹⁰ Expert interviews with different SC actors were conducted. Additionally, documents (e.g., time schedules, process descriptions and databases, as well as the results of Internet research) were combined to reach internal and external validity.

Expert Interviews

Manor provided encompassing information on the recent SC concept and ideas on the integration of CT into SC concepts. These expert interviews generated a general understanding of the SC concept, SC structure, partners and locations, as well as productand market-specific characteristics. Furthermore, the specific production and distribution concepts (including the applied I&C systems) were specified, e.g., the applied planning and control mechanisms, logistics and transport concepts, specific target systems and transport quantities, as well as the infrastructural characteristics of different SC sites.

The expert interviews with *Swiss Post* focused on the process-oriented, organisational and technical aspects of the transhipment and handling of the CT terminal as well as for rail transport, e.g., transport and handling times, planning advances, flexibility regarding opening times, transport quantities, short-term changes and train schedules.

Document Analysis

Manor provided a number of different documents on delivery windows, ramp opening times, loading, unloading, order picking and handling times, transport times for the different DCs and stores, delivery schedules (time and day), number of deliveries per week and the allocation of specific transport routes, transport routes, transport modes, road carriers and capacity usage per transport route and day of the week, as well as order lead times. The experts stated that for the loading and unloading of one pallet, 1 minute is necessary. Additionally, an access database was provided including all orders

¹¹⁰ cf. Bortz et al. (2006), pp. 296.

by load carrier from January 2010 to October 2010. It includes information on demanded quantities, type of load carrier, type of goods (priority and non-priority), corresponding transport and dates for all the stores from the two DCs in Möhlin and Hochdorf.

Swiss Post provided the train schedule for package transport between the three terminals in Härkingen, Frauenfeld and Daillens. This schedule includes the transport times, handling times and earliest and latest delivery times according to regulation policy and terminal opening times. Furthermore, a documentation of all technical transport vehicles, locomotives, wagons, cranes, containers and load carriers used for CT was made available. The terminals are specified in terms of track length, track capacity and crane scope. According to the experts, the loading and unloading processes at the terminals take 30 minutes.

I	No.	Site	No.	Site	No.	Site
I	1	DC Hochdorf (DC HOV)	7	Monthey (MON)	12	Vésenaz (VES)
ſ	2	DC Möhlin (DC MOE)	8	Morges (MOR)	13	Vevey (VEV)
	3	PF Sion (PF SIO)	9	Nyon (NYO)	14	Yverdon (YVE)
	4	Chavanne (CHA)	10	Sierre (SIE)	15	Terminal Härkingen (T HAE)
	5	Geneva (GEN)	11	Sion (SIO)	16	Terminal Daillens (T DAI)
ſ	6	Lausanne (LAU)				

Table 5 - 2 provides an overview on the sites included in the simulation model.

 Table 5 - 2: Sites modelled in the simulation.

Literature Review on Transport Emissions

For the evaluation of transport emissions, four key figures were analysed: fuel consumption as well as CO_2 , NO_x and particle emissions.¹¹¹ There are significant differences regarding these emission factors for the transport modes. Figure 5 - 4 illustrates these differences in reference to emissions per ton kilometers (tons multiplied by kilometres). The illustration shows that rail and inland vessels gain advantage over the truck, but especially over air cargo transport.¹¹² Nevertheless, the thesis in hand focuses on road and rail transport.

¹¹¹ cf. Infras (2011a).

¹¹² cf. Umweltbundesamt (2009), p.14.



Figure 5 - 4: Comparison of CO₂, NO_X and particle emissions for different transport modes (under consideration of all climate-related effects of air cargo).¹¹³

Although political debates primarily focus on CO₂ emissions, these are not significant in the differentiation of transport vehicles according to the exhaust emission standards. Emission factors are provided by Infras (2011) in the Handbook of Emission Factors for Road Transport (HBEFA).¹¹⁴ This database contains data on the emissions for different road types, inclinations, traffic situations, vehicle types, utilisation levels and emission categories. For the simulation study, standard emission factors for the traffic compilation of emission categories on the basis of 2010 and for an average utilisation of transport vehicles were used. Since the simulation model does not include transport quantities with weight information, average values were applied on a vehicle basis.¹¹⁵

Reh (2010) provides data on average emissions for trains.¹¹⁶ However, it must be kept in mind that the emission factors depend on the type of locomotive as well as the train

¹¹³ cf. Ibid., p.14.

¹¹⁴ cf. Infras (2011a). ¹¹⁵ cf. Infras (2011)

¹¹⁵ cf. Infras (2011). ¹¹⁶ cf. Reh (2005), p. 136.

weight. Following Reh (2010), it is assumed that there is a share of diesel and electric locomotives on the train routes. Furthermore, for the fuel consumption of rail transport processes, LUBW (2011) presents values referring to the *DB AG* and *Aviso*¹¹⁷ in terms of different fuel consumption factors depending on the locomotive type for service type (line vs. shunt services).¹¹⁸ According to LUBW (2011), freight transports on line connections have an average fuel consumption of 3.3 L/train-km. For the simulation study, it was assumed that an average of 30 shipments is transported by one train.

average emission for train transport	per train-km
NO _X	131.619 g
particle	3.462 g
CO ₂	7117.5 g
fuel consumption	3.3 L

 Table 5 - 3: Emission factors for rail transport.¹¹⁹

Further emissions, for instance those caused by shunting services and terminal processes, were excluded from consideration.

Table 5 - 1 provides an overview of the different data gathered for the simulation study and the specific data sources.

category	simulation object / variable	data sources
SC network	 production sites DCs transhipment points stores 	Manor Swiss Post
transport routes	 route segments transport times route profiles inclination profiles transport schedules (rail and road) 	Manor Swiss Post
products	 production schedule / quantities product types (priority/non-priority) delivery times volume per product type 	Manor
performance requirements for delivery	 demand per day / store delivery windows vendor dock opening times 	Manor
transport vehicles / terminals	 transport capacity terminal and overhead capacities loading / unloading / processing time 	Swiss Post Internet and literature review
external restrictions	 weekend driving bans night driving bans inner-city driving bans emissions profiles (road and rail) 	Swiss Post Internet and literature review

Table 5 - 1: Overview of data sources and data provided.¹²⁰

¹¹⁷ Land register of Baden-Wuerttemberg, Germany.

¹¹⁸ cf. LUBW (2002), p. 17.

¹¹⁹ cf. Reh (2005), p. 136.

¹²⁰ For a list of conducted interviews, see Appendix A-II.

In an iterative process, the data gathered was enhanced and specified using additional telephone interviews and email contact.

Data Editing

To use the provided data, a comprehensive editing process was necessary. This editing process made the information suitable for processing in the simulation tool and the underlying database. Namely, data was edited and summarised in table form and saved as text files.¹²¹ In total, 21 files were created as databases for the simulation study. These text files were the bases for the Oracle database, which was linked with the simulation tool (cf. Appendix D).

Transport Routes, Distances and Speed

The *transport routes* were enhanced by additional information on the different route segments, the required transport time and the allocated transport vehicle. The distances between the different SC sites and the street profiles (share of highway and non-highway distances) were added based on an online route planning instrument.¹²² In response to the expert interviews confirmed by literature, an average transport speed of 50 km/h was determined for all street profiles.¹²³ An average transport time was added based on this information for each transport route. Following a study by the University of Vienna, an inclination profile for all transport was applied.¹²⁴ Table 5 - 4 shows the assumed inclination profile for the calculation of transport emissions.

	inclination						
	+6%	+4%	+2%	+0%	-2%	-4%	-6%
share of entire transport distance	5%	10%	20%	30%	20%	10%	5%

Table 5 - 4: Standardised inclination profile.¹²⁵

¹²¹ These text files were used for the storage and exchange of simply structured data. '*csv*' stands for 'comma-separated values'.

¹²² www.maps.google.com. Documented in Appendix D.

 ¹²³ cf. LUBW (2011); ASTRA (2010); Bühler (2005), pp. 145. Speeds were confirmed by expert interviews. For further information on average transport speeds for different carriers, see Kille et al. (2008); Bühler (2005), pp. 145.

 $^{^{124}}$ cf. Sihn et al. (2010).

¹²⁵ cf. Ibid.

route segment	distance [km]	start	destination	highway [%]	non-highway [%]	transport time [sec.]		
route segment 30	207	DC HOV	СНА	94.69	5.31	10 800		
route segment 31	260	DC HOV	GEN	94.23	5.77	13 620		
route segment 32	209	DC HOV	LAU	93.78	6.22	10 980		
route segment 33	213	DC HOV	MON	94.37	5.63	11 160		
route segment 34	209	DC HOV	MOR	94.74	5.26	10 920		
route segment 35	235	DC HOV	NYO	94.89	5.11	12 240		
route segment 36	272	DC HOV	SIE	95.22	4.78	14 100		
route segment 37	257	DC HOV	SIO	95.33	4.67	13 320		
route segment 38	190	DC HOV	VES	90.53	9.47	10 200		
route segment 39	188	DC HOV	VEV	93.09	6.91	9 900		
route segment 40	176	DC HOV	YVE	93.18	6.82	9 300		
extract Full list documented in Annendix D								

Table 5 - 5 shows an extract from the table of defined route segments.

 Table 5 - 5: Extract of route segments defined for the simulation with distance, street profile and transport time.

The different transport routes were compiled using the different route segments. Furthermore, a primary and specific transport vehicle and start time were defined for each route.¹²⁷ Table 5 - 6 lists the primary transport routes. In the simulation model, additional transport routes with additional transport resources were defined for cases where the demand exceeds the capacity of the primary transport vehicle.¹²⁸

¹²⁶ Full list Appendix D.

¹²⁷ In an additional table, t_vehicle, the transport capacity in pallet equivalents is included.

¹²⁸ There is an additional table (t_starttime) that defines on which days the transport is accomplished.

route	route seg- ment	start	destination	transport resource	transport time [sec.]	start time
Н 60	30	DC HOV	СНА	VI	10 800	11:30 AM
	125	СНА	MOR		900	
	175	MOR	NYO		1 500	
Н 70	31	DC HOV	GEN	VII	13,200	11:00 AM
	141	GE	VES		1,200	
Н 63	32	DC HOV	LAU	V III	10 980	11:00 AM
	154	LAU	VEV		1 500	
Н 75	33	DC HOV	MON	V IV	11 160	12:30 AM
Н 220	28	DC HOV	PF SIO	ΤΙ	13 320	4:30 PM
	111	PF SIO	SIO		1 020	
	201	SIO	SIE		1 080	
Н 240	342	DC HOV	PAY	VV	8 220	4:00 PM
	339	PAY	YVE		2 700	
M 3	52	DC MOE	СНА	V VI	12 480	11:00 AM
	125	СНА	MOR		2 700	
	175	MOR	NYO		1 500	
M 1	53	DC MOE	GEN	V VII	12 780	10:00 AM
	141	GEN	VES		1 200	
M 2	54	DC MOE	LAU	V VIII	10 140	10:45 AM
	148	LAU	MON		2 700	
	167	MON	VEV		4 500	
M 24	343	DC MOE	PAY	V IX	7 320	2:15 PM
	339	PAY	YVE		3 600	
MS 1	46	DC MOE	DC HOV	TII	4 920	2:30 PM

Table 5 - 6: Primary transport routes, transport resources and start times.¹²⁹

Demand and Transport Quantities

Manor provided data about the *daily demand* per store and per DC for all considered stores and the two DCs in between. From this demand, transport quantities were calculated. The database contained the demand per store and for the specific product types, as well as the specific load carriers for each day from January 2010 to September 2010. The information on the different load carriers allowed the calculation of so-called pallet equivalents. This means that according to a specific calculation key provided by *Manor*, demand quantities were standardised. For the analysis of the CT integration, only the differentiation into priority and non-priority goods was modelled.¹³⁰ Thus, based on this standardisation of demand quantities to pallet equivalents, the average demand per store, per DC and per priority class were calculated.

To model the fluctuations in demand and transport quantity, a comprehensive analysis of this demand data was necessary. As a first step, a visual analysis was conducted. As

¹²⁹ The values for the additional routes are identical, but different transport vehicles are used.

¹³⁰ Priority goods must be delivered to the store in the next delivery window. Non-priority goods can have longer lead times and can be delivered to the store whenever there is transport capacity available.



a second step, a univariate analysis of variance was used to analyse the impacts of different parameters on the demand per store, per DC and per priority class.

Figure 5 - 5: Example demand history for delivery to the store in Geneva from DC Hochdorf for prioritised goods from January – September 2010 in pallet equivalents.

Table 5 - 6 shows the demanded pallets of one type of goods, one store and one DC. The illustration shows that there were no obvious regular fluctuations over the analysed period.

As a result of the high fluctuations in demand, it was not possible to identify one specific demand value for the simulation. Therefore, a univariate analysis of variance was conducted to analyse the influences that affect the variability of demand. Figure 5-8 indicates that time-specific aspects strongly influence these fluctuations. Three different influence factors were analysed by ANOVA.¹³¹ The three different predictor variables *month, month phase* (1st - 4th quarter of the month) and *weekday* were investigated in three separate settings.¹³² In each analysis, the effect on the dependent variable *demand quantity* was also provided (partial η^2). By means of the example results for the store in Yverdon, the findings of the ANOVA are introduced. The ANOVA showed that monthly effects were not found to be significant and thus did not need to be considered in the simulation model (F=0.396; partial $\eta^2 = 0.023$). Also, the impact of the *month phase* was found to not be significant (F=138.189; partial $\eta^2 = 0.749$). In

 $^{^{131}}$ ANOVA = with SPSS 18.

¹³² The H_0 -hypothesis states that there is no relationship. SPSS shows the probability p for the H_0 -hypothesis, as well as the proof of statistic F.

conclusion, the only predictor that showed a significant effect on *demand quantity* was the *weekday* variable. Thus, the weekday served as a basis for the calculation of different demand quantities for the simulation model.

Figure 5 - 6 supports this finding. The illustration shows that in each month, the number of demanded pallet equivalents obviously differs for the different weekdays. On Thursdays, the store in Yverdon receives no orders.



Figure 5 - 6: Demand in pallet equivalents per weekday for prioritised goods for the store in Yverdon from DC Möhlin.

Table 5 - 7 shows the average demand and the standard deviation per weekday for the transport of prioritised goods from DC Hochdorf and DC Möhlin to the stores. This data was the basis for product generation in the simulation model. According to a specific schedule, the defined amount of pallet equivalents is generated for transport to the destination store.

start	destination	Monday	Tuesday	Wednesday	Thursday	Friday	generation time
DC HOV	СНА	15	18	10	12	15	11:00 AM
DC HOV	GEN	40	31	27	24	28	10:00 AM
DC HOV	LAU	18	16	15	13	13	10:20 AM
DC HOV	MON	17	0	20	11	10	12:00 AM
DC HOV	MOR	6	9	0	9	7	11:00 AM
DC HOV	NYO	7	1	0	14	2	11:00 AM
DC HOV	SIE	20	12	0	4	31	9:00 AM
DC HOV	SIO	15	10	0	3	31	9:00 AM
DC HOV	VEV	19	14	15	13	13	10:20 AM
DC HOV	VES	8	5	8	0	8	8:00 AM
DC HOV	YVE	11	6	1	0	20	3:10 PM
DC MOE	СНА	8	10	9	9	9	9:15 AM
DC MOE	GEN	20	19	24	18	19	7:45 AM
DC MOE	LAU	9	12	12	10	11	8:15 AM
DC MOE	MON	9	0	14	8	8	8:10 AM
DC MOE	MOR	5	6	0	9	5	9:15 AM
DC MOE	NYO	4	1	0	6	2	7:05 AM
DC MOE	SIE	2	1	0	4	2	7:20 AM
DC MOE	SIO	2	2	0	5	3	7:25 AM
DC MOE	VEV	9	9	11	8	8	8:20 AM
DC MOE	VES	3	4	5	0	8	4:30 PM
DC MOE	YVE	6	4	1	0	12	11:45 AM

Table 5 - 7: Average demand for prioritised goods per weekday and store.¹³³

After this rather general explanation on the data gathering and editing process, the following section introduces the development, validation and verification of the simulation model.

5.5 Step 3: Development, Validation and Verification of the Simulation Model

This section addresses the development, validation and verification of the simulation model. Therefore, the SC concept and the CT concept were analysed, simplified, modelled, formalised and, finally, implemented. Several verification and validation steps ensured the compliance of the developed model with the as-is situation (external validity), the functioning of the model (internal validity) and suitability for the defined research target. A sensitivity analysis ensured that the model was valid for a defined range of different input and output parameters, in particular for different demand situations.¹³⁴

¹³³ The demand for non-prioritised goods can be found in Appendix B.

¹³⁴ The members of the V-Research group in Vorarlberg were the partners supporting the development of the formalised and the executable model.

5.5.1 Conceptual Model

A conceptual model serves to abstract the real or the scenario system. This step can be equated with a simplification of reality.¹³⁵ The formalisation of the conceptual model means the development of the implementation in the executable simulation model.¹³⁶ Different instruments are therefore available for this purpose, such as pseudo-code, structograms, entity-relationship diagrams or UML notation.¹³⁷

Conceptual model building means transferring the problem formulation and model requirements to a determination of what is modelled in what way.¹³⁸ Thereby, the conceptual model is independent of the simulation tool used.¹³⁹ The target and level of accuracy must be defined in this phase of the modelling process.¹⁴⁰ Assumptions are necessary if there are uncertainties about the real world behaviour. Simplifications are necessary to accelerate the simulation process in order to reduce data needs and to improve the transparency of the simulation model.¹⁴¹ In brief, the conceptual model describes the targets, input¹⁴² and output data,¹⁴³ scope, level of detail, assumptions and simplifications relevant to the simulation model.¹⁴⁴

System Limits and Level of Detail of Simulation Study

Figure 5 - 7 illustrates the simplified model of the central simulation objects. The simulation model includes two DCs, two terminals, one regional cross-docking plat-form and eleven stores in Western Switzerland. The flow of goods between the sites is modelled by the different transport vehicles that travel via defined transport routes. Temporal and spatial information on all events (e.g., generation of goods, loading, unloading, transhipment, transport and waiting) was generated for each object.

¹³⁵ cf. Zeigler (1984).

 $^{^{136}}$ cf. Rabe et al. (2008), p.49. The step of formalisation is often skipped due to a project's financial restrictions. cf. Ibid., p.128.

¹³⁷ cf. Ibid., p.48. ¹³⁸ cf. Pidd (2003).

¹³⁹ cf. Fishwick (1995).

¹⁴⁰ cf. Robinson (2008).

¹⁴¹ cf. Ibid., p.279.

The *inputs* (or experimental factors) are those elements of the model that can be altered to effect an improvement in, or provide a better understanding of, the problem situation. They are determined by the objectives. cf. Robinson (2004), p.66.
 The automatic (or experimental factors) are the simplet form a range of the simpletion model. These hears the surgest of the determined by the objectives. cf. Robinson (2004), p.66.

¹⁴³ The *outputs* (or responses) report the results from a run of the simulation model. These have two purposes: first, to determine whether the modelling objectives have been achieved; second, to point to reasons why the objectives are not being achieved, if they are not. cf. Ibid., p.66.

¹⁴⁴ cf. Ibid., pp. 63.



Figure 5 - 7: Conceptual model simulation scenario with CT.

Level of Detail

To analyse the influence of CT integration in the SCP, it was not necessary to model partial processes at the different sites.¹⁴⁵ This means that each site was modelled with only one process. For instance, only one process with a certain duration was specified to model all processes in the CT terminals (e.g., entry, administration, waiting and transhipment) because the lengths of the different partial processes were not relevant for the given research target. This reduction of detail reduces the complexity of the model, but does not restrict the resilience of the simulation results.

Input Parameters

This section gives an overview of the central input data. Figure 5 - 8 shows the different input parameters, the dependencies and connections in a simplified UML class diagram.¹⁴⁶ For object-oriented modelling, a class defines a common structure and a common behaviour of objects (classification). It is visualised by a rectangle with a name (bold) and, optionally, attributes, operations and characteristics separated by horizontal lines.

¹⁴⁵ cf.Rabe et al. (2008), p.69.

¹⁴⁶ A class diagram serves to the visualisation between different classes, interfaces and relations.

As described above, the data was stored in text files.¹⁴⁷ The adaptation of this input data was the central factor to setting up the different simulation scenarios. All tables can be found in Appendix D, ordered by the different scenarios.



Figure 5 - 8: Data model for the simulation study (simplified UML notation).

The input parameters for the simulation study were:

- A list with the different sites in the SC, such as the stores, the CT terminals and the DCs (position_ID).
- A list with the operating times of all the stores (factory_ID).
- A list with the generation time and quantity of goods in the DCs per store, priority class and the weekday (schedule_ID).
- A list with the transport schedules (transport_ID; time_ID) for different transport routes (route_ID) with defined route segments (segment_ID). Furthermore, the specific transport vehicle (vehicle_ID) and the transport mode (mode_ID) were determined.
- A list with the route segments (segment_ID). Each route segment was specified by an inclination profile (inclination_ID) and a street profile (profile_ID). The street

¹⁴⁷ The different files are included in Appendix D.

profile determined the share of highway and non-highway transport distance.

- A list with the handling, transhipment and additional processes and the required times (processes_ID).
- A list with the emission and fuel consumption factors per vehicle and transport distance (emission_ID; consumption_ID).

Basic Assumptions

Some general assumptions and simplifications were necessary for the modelling of the SC concept. In particular, assumptions regarding the (1) demand and transport quantities and the (2) duration and fluctuation of handling and transport processes were necessary. The calculation of emissions included the factor of (3) inclination profiles. Since the simulation model does not include production facilities, it is assumed that adaptations of the production concept could be modelled by adapting the (4) generation of goods in the DCs.

(1) Fluctuations in Demand

Demand and transport quantities in retail underlie different temporal fluctuations. The univariate analysis of variance showed that up to 75% of the fluctuations in demand can be explained by the impact of weekdays.

However, the analysis of the average demands and the standard deviations, as well as the discussions with the interview partners, encouraged the analysis of different demand situations. Thus, the simulation study considered four different demand situations.

(1) Normal demand situation. The mean demand value per weekday was applied (cf. Table 5 - 7 and Appendix B).

(2) Reduced demand situation. Since standard deviation was not consistent for all stores, based on the expert interviews a reduced demand situation was analysed using a reduction in the mean demand of 20% (cf. Appendix D).

(3) Increased demand situation I. A 20% increase in demand equals the average value of standard deviation over the different stores. The expert interviews confirmed that a 20% increase in demand is commonly experienced and thus this increase was chosen for the sensitivity analysis (cf. Appendix D)

(4) Increased demand situation II. A 50% increase in demand equals the peak values of demand for the different stores. The expert interviews confirmed that a 50% fluctuation in demand is experienced less commonly but still regularly. Thus, the interviews encouraged the choice of a demand increase of 50% for the sensitivity analysis (cf. Appendix D).

Extraordinary demand situations were not considered. Usually, in freight transport the

capacity dimensioning is not always oriented to maximum demand situations.¹⁴⁸ In accordance with the views of the interview partners, the analysis focused on the normal demand situation. Furthermore, the train capacities were used to the maximum for the 50% increased demand situation. A further increase would negatively affect the generalisability of the results.

(2) Duration and Fluctuation of Handling and Transport Processes

Based on the expert interviews, it was assumed that *transport, loading and transhipment times* have mean values depending on the transport distance, transport mode and shipment size. For the loading, the unloading rate was assumed to be an average of 1 minute per pallet equivalent; for the transhipment processes, an average time of 30 minutes per truck shipment was assumed. The transport times for road transport were calculated based on the transport distance and an average speed of 50 km/h.¹⁴⁹ For rail transport, the transport time results were derived from the rail schedules provided. Delays and disruptions caused by congestion or different average velocities for different times or weekdays were – to a certain extent – integrated in the comparably low average speed of the trucks, as modelling explicit disruptions in road or rail transport was not possible.

(3) Emissions

The standardised inclination profile provided by Infras (2011) was used for the calculation of fuel consumption and emissions. Furthermore, the transport distance and the share of highway and non-highway distance were the bases for calculation. Values for the average loading and for the average share of emissions categories for trucks on the basis of data from 2010 were used. There was no available information on further influencing parameters, such as driving styles and the specific emissions categories of the transport vehicles, as well as the effects of different average speeds for different times and weekdays.

(4) Generation of Goods

Since no *production facilities* were included in the simulation model, the goods were generated in the DCs. This generation process can be understood as the goods' production. Thus, adaptations to the temporal generation of goods, e.g., arrival times, frequencies and quantities, were used to simulate changes to the production concept (cf. section 3.3.3).

¹⁴⁸ In retail, these situations can occur, for example, ahead and after a holiday period or because of specific marketing activities. The demands and the transport quantities can increase many times over for certain or peak time periods of a few days.

¹⁴⁹ On the basis of the expert interviews confirmed by literature. cf. LUBW (2011); ASTRA (2010); Bühler (2005), pp. 145. For rail transport, fixed transport times based on the transport schedule provided by the *Swiss Post* were used.

(5) Non-Priority Goods

To keep the complexity of the simulation model low, the sequence of order processing was defined by the two different product types: prioritised and non-prioritised products. There was no allocation of due dates to the specific orders. The configuration of the model made sure that priority goods were always transported to the store on the same day as they were generated. If the quantity of prioritised goods exceeded the capacity of the primary transport vehicle, a second vehicle was initialised. These additional transport vehicles were not implemented for non-prioritised goods. These goods stayed at the DC until the next primary transport had free capacity. In practice, the non-priority goods have a due date, too. Usually, the date is some time in advance (usually at least one week). The sensitivity analysis showed that for the simulation, all non-priority goods were taken to the destination in a timely manner.

5.5.2 Simulation Model

Based on the conceptual model, the executable simulation model was developed. The simulation tool Flexsim is used for implementation (cf. section 5.1.6). Activity diagrams were used to provide structured descriptions of the sequence of model configuration for each simulation run.¹⁵⁰ Appendix C-I illustrates the process for the configuration of a simulation model. It determines how and in which order the different simulation modules are generated. Appendix C-II visualises the formal structure of each simulation run, starting from the product creation to the final delivery.

The model generation in this thesis was based on the modular structure of Flexsim. Predefined routes can be added to the model by connection with an underlying database. The different sites (CT terminals, DCs and stores) and transport resources were generated and parameterised.¹⁵¹ At a defined time point, products were generated at the DC, loaded into a defined transport vehicle and taken on a defined transport route to their final destination. Flexsim stored the generated data as a text file.

Figure 5 - 9 shows a screenshot from Flexsim from an example SC.

¹⁵⁰ An activity diagram in UML notation allows an object-oriented description of programs and systems.

¹⁵¹ On the basis of an OBDC connection.



Figure 5 - 9: Screenshot from the simulation model in Flexsim.

Output Data as the Basis for the Calculation of Performance Indicators

The results of each simulation were stored in a text file. Data on all accomplished processes and events per object was saved (cf. history_ID in the UML-diagram in Figure 5 - 8; documented in Appendix D).¹⁵²

The '*lead time*' per order (pallet) resulted from the difference between the arrival and start time. The start time equals the product generation time. The arrival time equals the time point at the completion of unloading at the destination store.

The different '*lead time elements*', namely handling time, transport time and waiting time, were calculated per order. The *handling time* was the sum of all loading and unloading times of a certain type of goods. It was calculated by summing up the time span between the beginning and the end of the loading and unloading respectively transhipment process of the specific truck and/or train. The *transport time* resulted from the difference between the beginning and the end of all transport processes experienced by a certain pallet equivalent. The *waiting time* resulted from the difference between the order lead time and the transport and handling time.

¹⁵² Oracle Express 12.

The *'inventory level'* was calculated as the integral of product generation at a certain time point and the product being removed from the system after the unloading at the store.¹⁵³ The inventory level in the transport process depended on two main factors: the demand quantity and the order lead time. The average inventory was calculated as the integral of the inventory over time per store. Figure 5 - 10 shows an example inventory course for delivery to the Chavannes store for products coming from DC Hochdorf for the as-is situation with normal demand. The curve shape shows two points for each product generation: one point for the prioritised goods and one point for the non-prioritised goods. The steep vertical decline of the curve with several data points visualises the storage of products and the discrete removal of pallet equivalents from the system. The area below this inventory course equals the integral of the curve and thus the average inventory (marked as a grey area). The curve shape is representative of all the stores and DCs.¹⁵⁴ For the given example, the results show an average inventory of 4.4 pallet equivalents for the transport, starting from the provision at the DC and terminating with the unloading in the store.



Figure 5 - 10: Example inventory course for delivery to the Chavannes store with goods from DC Hochdorf (grey = integral of inventory).

The '*adherence to schedules*' was evaluated by the comparison of the actual arrival time at the store and the designated delivery window. The adherence to schedules can be calculated on the basis of four different key figures: (1) the deviation for a \pm -15 minute time window, (2) the deviation for a \pm -30 minute time window, (3) the delay for a \pm -15 minute time

¹⁵⁴ See Appendix D.

¹⁵³ The inventory level is the basis for the evaluation of capital commitment costs.

ute time window and (4) the delay for a +/-30 minute time window. The deviation number includes all the deliveries that were either too early or too late to meet the defined delivery window. The number of delays only includes the deliveries that are late. The ratio of deliveries on time and deviating or delayed transport vehicles equals the adherence to schedules as a percentage. Additionally, the average deviation and average delay from the target arrival time was calculated in minutes.

The '*utilisation of the vehicles*' can be calculated by the division of the maximum capacity and the actual capacity of each transport process.¹⁵⁵ Capacity utilisation in the first route segment was used for evaluation. The effect of declining capacity utilisation because of the unloading of partial deliveries (cf. section 3.3.1) was not considered.

The *'emissions'* can be calculated on the basis of the output data for CO_2 , NO_x and particle emissions in grams for each route segment. This data can be converted to emissions per vehicle, weekday and transport route.

There was no specific data created by the simulation to measure or to evaluate '*flexibility*' and '*product quality*'. Thus, the impact of CT integration on flexibility and product quality in the SC concept is discussed based on the qualitative evaluation of the other simulation results.

The following section introduces the validation and verification steps that were performed to ensure the internal and external validity of the simulation model.

5.5.3 Validation and Verification

To ensure internal and external validity, the conceptual and the formalised models must pass the test of consistency and functionality. The executable model must compare with the behaviour of the real world system. Furthermore, verification between the implementation of the conceptual model in the form of the executable model must be verified. At last, the executable model must be validated in terms of the real world problem. Figure 5 - 11 illustrates the relationships between the real world problem, the conceptual model and the executable simulation model.

¹⁵⁵ A transport process here means the accomplishment of a total transport route.



Figure 5 - 11: Conceptual model validation and verification.¹⁵⁶

Rabe et al. (2008) recommend that different validation and verification techniques are used for the different phases of a simulation study.¹⁵⁷ For the conceptual model, they recommend validation by means of a 'review'¹⁵⁸ of the model with the experts from *Manor* and *Swiss Post*. Furthermore, validation and verification were ensured by so-called 'face validity': a dialogue with experts in the field of SCM and CT. These discussions served to quickly identify the mistakes and irregularities. Additionally, a 'desk-checking' process was accomplished.¹⁵⁹ This included the self-inspection of integrity, consistency and definiteness.

The verification of the model was based on several simulation runs to uncover any mistakes and abnormalities in the model. Initially, the impact of single changes in different input data was examined to ensure the correct functioning of the model. For instance, an increased demand in one certain store was modelled and the correct influence on the generated output data, e.g., the utilisation of the specific transport vehicle, was checked.

For validation purposes in particular, the comparison with the real system regarding transport times, waiting times and adherence to schedules allowed faults caused by typographical errors in the input data or logical errors to be revealed, for instance the errors caused by incorrectly connected transport routes were examined. Further abnormalities, for instance regarding the number, capacity and utilisation of vehicles, transport schedules and demand, were revealed by the discussions with the experts. Via several iterative processes, the simulation model was adapted to the real system behaviour.

The validation process is illustrated by an example. The utilisation and amount of trans-

¹⁵⁶ Illustration with adaptations according to Robinson et al. (2010), p.386.

¹⁵⁷ cf. Rabe et al. (2008), p.113. ¹⁵⁸ cf Ibid. p.97

¹⁵⁸ cf. Ibid., p.97.

¹⁵⁹ cf. Ibid., p.112, Balci (1998), p.356.

port showed significant differences between the simulation results and reality. The discussions with the experts showed that the so-called *'squeeze'* factor was the reason for the deviation. At this point, the capacity of transport vehicles in the simulation was set to a certain default maximum number of pallet spaces.¹⁶⁰ As soon as 100% capacity utilisation was reached at the loading, the simulation model initiated a second 'additional' transport. However, in reality, a certain number of pallets can be piled. The experts proved that for internal transport planning, 130% capacity is assumed because of this squeeze factor. The simulation was adapted to this finding. This significantly reduced the number of additional transport processes. Furthermore, the capacity utilisation is reported up to a maximum level of 130%.

The accomplished triangulation of validation and verification methods balanced the specific disadvantages of the individual validation and verification techniques. They were enhanced by a sensitivity analysis.

Sensitivity Analysis

A sensitivity analysis was performed to prove the functionality of the developed simulation model for extreme high and low demand situations. The analysis shows that the simulation model worked well with very low demand quantities. However, for very high demand situations with more than 50% of the 'normal' demand, the simulation reached its limits. The reasons for this were the capacity restrictions that resulted from the model design.

In accordance with the real system, primary and additional transport vehicles were modelled for each transport route. For extremely high demand situations, not only the primary, but also some additional transport vehicles are used to the full. As a result, prioritised goods were not picked up for delivery to the store on the day of generation. In reality, this would not be acceptable: in these 'extreme' situations (see previous chapter), a third transport resource would be activated. However, in the simulation model, this system behaviour was not designated.¹⁶¹

¹⁶⁰ Typically 33 pallets per truck.

¹⁶¹ One reason for this decision was that in freight transport, capacities are not usually defined according to peak loads. See, for instance, Bendul (2009), p.177;
5.6 Step 4: Experimental Plan and Setup of Experiments

This section describes the experimental plan and the setup of the simulation runs. The initial values for the simulation objects, the simulation duration and the experimental plan are defined. The initial values and the framework conditions of a simulation run significantly influence the simulation results. They were held as constant for all the simulation runs conducted to allow mutual comparability.

Initial Values and Stabilisation Period

For the simulation study, it was assumed that at the beginning of the simulation all the resources (vehicles, ramps and stores) were idle and all queues were empty. This corresponds to the real system because at the weekends there are usually no waiting orders. The simulation started with the generation of defined amounts of goods at defined time points and DCs. Since all resources were idle at the beginning of the simulation run, no stabilisation period was required.¹⁶²

Simulation Step and Simulation Duration

The time step was one second to ensure an adequate level of accuracy for the measurement of the adherence to schedules and the lead times.

There were different demand values for the five weekdays. Since no further temporal fluctuation was considered in the model, the accomplishment of a one-week simulation run was sufficient.

Each simulation run included 1 week (604800 seconds). Bottlenecks or overcapacities were able to be observed in this time span. Four different demand situations were modelled to gain sufficient knowledge on the system behaviour for fluctuating demands.

Experimental Plan

The simulation study was accomplished on the basis of a predefined experimental plan. This plan avoided a research process that was guided by specific results and may have excluded important factor combinations from the analysis.

The scenarios presented in section 5.3 were the bases of the experimental plan. The basic scenario modelled the as-is state of the SC concept. Four simulation runs were accom-

¹⁶² This means that there were no relevant fluctuations of values at the beginning of the simulation run.

plished for the four scenarios, each for a different demand situation: (1) -20% demand quantity, (2) normal demand, (3) +20% demand quantity and (4) +50% demand quantity. In total, 16 simulation runs were accomplished. CT scenario I included rail transport for the main haulage between the DC and the first store without any adaptations of the SC concept. For each adaptation of the SC concept, a separate simulation model needed to be configured.¹⁶³ CT scenario II was equal to CT scenario I, but contained adaptations to the production concept. CT scenario III was also equal to CT I but contained adaptations to the distribution concept. The specific adaptations were derived from the analysis of the basic and CT scenario I. The details are discussed directly before the discussion of simulation results (cf. section 5.7.3 and section 5.7.4). Table 5 - 8 summarises the experimental plan

simulation scenario	description
1 - basic scenario	as-is state
2 – CT scenario I	as-is SC configuration with integrated CT
3 – CT scenario II	CT scenario I with adaptations to the production concept
4 - CT scenario III	CT scenario I with adaptations to the distribution concept
T-11. 5 0 E	

 Table 5 - 8: Experimental plan for the simulation study.

Based on the experimental plan, 16 simulation runs were accomplished. The specific setup and results of the different simulation runs are discussed in the following section.

5.7 Step 5: Description and Analysis of Simulation Results

Introduction This section specifies the setup and discusses the results of the different simulation runs. The section is ordered according to the four different simulation scenarios introduced in the previous section. The full documentation of simulation results can be found in Appendix D.

5.7.1 Basic Scenario – Modelling the 'as-is' Situation

Introduction The basic scenario modelled the as-is situation of *Manor*'s distribution network as introduced above. It includes eleven stores and two DCs. Sites are connected by 12 different transport routes. The transport routes are operated by primary and – if the capacity is used to the full –additional vehicles. The results of the basic scenario can be understood as a *'benchmark'* for the evaluation of the following CT scenarios.

¹⁶³ Axelrod (2005), p. 8.

Utilisation of Transport Vehicles – Basic Scenario

Figure 5 - 12 and Figure 5 - 13 show the results of the transport vehicle utilisation analyses in the basic scenario. The illustrations show the average utilisation per weekday and transport route. The average utilisation of primary transport vehicles was 73% for the normal demand situation; for the additional transport vehicles, it was 61%.

The illustration shows that the utilisation strongly varied from store to store. Missing values mean that there was no transport scheduled for that specific weekday. The analysis shows that the transport routes H 220, H 70 and M 2 were regularly used to the full (here 130%). The reason for the strongly differing utilisation levels was the different demand levels for the different stores. For instance, route M 1 is used to deliver to the stores in Geneva and Vésenaz. Geneva is the biggest *Manor* store in Switzerland. Thus, the demand in pallet equivalents is much bigger in comparison with the amount of pallet equivalents on the transport route H 75, which only delivers goods to the Monthey store, a medium-sized store with comparably low demand. Furthermore, the maximum capacity influences the utilisation. For instance, transport route M 24 relies on a truck with a maximum capacity of 17 pallets (+30% squeeze factor), whereas most other routes are served by trucks with 30 and more pallet spaces (+30% squeeze factor).



Figure 5 - 12: Mean utilisation per primary transport route and weekday for the normal demand situation.

As a result of the high capacity utilisation of the primary transport, additional transport was accomplished. The utilisation of the additional transport vehicles was comparably low. Figure 5 - 13illustrates the mean utilisation of the additional transport processes. Transport route H 240 is not included, since for this transport no additional transports were accomplished.



Figure 5 - 13: Mean utilisation per additional transport route and weekday for the normal demand situation.

In general, non-scheduled transport causes non-linear additional costs. They require additional planning efforts and road carriers charge an increased price for holding the necessary capacities available.¹⁶⁴ Thus, in for *Manor* a small number of additional transport vehicles is seeked. Nevertheless, high utilisation is worthwhile.

Although the thesis in hand did not aim to optimise the as-is distribution concept, this comparably high utilisation of primary transport indicates that there are no capacities to intercept fluctuations in demand without additional transport. Here, a general adaptation of transport routes may be sensible to improve the utilisation considering the number and utilisation of additional transport processes, as well as the required flexibility to deal with fluctuating demands.

The simulation runs with the increased demand situations (illustrated in Figure 5 - 14) prove this. For the 50% increased demand situation, additional transport was required on all transport routes.

¹⁶⁴ Corsten et al. (1992), pp. 180; Stuhlmann (2000).



Figure 5 - 14: Average utilisation of primary transport vehicles for different transport routes and demand situations.

Figure 5 - 15 shows the average utilisation of the additional transport routes for the different demand situations. The illustration shows that for a 50% increase in demand quantity, the transport routes H 70 and M 2 were again used to the full. This means that eventually a third transport vehicle would be required if no adaptations to the transport route planning were made.¹⁶⁵



Figure 5 - 15: Average utilisation of primary transport vehicles for different transport routes and demand situations.

Order Lead Time – Basic Scenario

The order lead time is critical in a lean SC with continuous replenishment processes. A short lead time allows a high level of product availability in the stores and directly impacts

¹⁶⁵ For example the change from a milk-run concept to direct concept (cf. section 3.3.1, for different types of transport concepts).

the capital commitment costs. Thus, a reduction of lead time is generally desirable as long as the product quality and delivery service are not jeopardised.

Figure 5 - 16 and Figure 5 - 17 illustrate the lead time elements per store and DC. The illustrations show that the share of waiting time was comparably high for the stores in Sion and Sierre. The reason was that deliveries to these have recently been made via the regional cross-docking platform in Sion. Thus, the goods are taken from DC Hochdorf via single wagon freight transport to the PF Sion and then by truck to the stores. The goods arrive at stores comparably late in the evening. For all other stores, road transport is the main contributor to the lead time. Depending on the average number of pallets per transport, the handling time varied between 0.58 h for the Yverdon store with a small average number of transported goods and 1.39 h for the Vésenaz store with a very high average number of transported goods.



Figure 5 - 16: Transport time elements per store for DC Hochdorf.

Figure 5 - 17 shows that the share of waiting times was distinctively higher for DC Möhlin. The goods for the stores in Lausanne, Sierre, Sion, Vésenaz and Vevey were all characterised by early generation times. These are partly significantly prior to loading and departure and indicate temporal quota for CT integration. Thus, the lead times were comparably long. In particular, the waiting times elements indicate that there might be temporal quota to be opened for the CT integration.



Figure 5 - 17: Transport time elements per store for DC Möhlin.

Figure 5 - 18 shows the variation of lead time elements for the four different demand situations. The illustration shows that the transport time was not significantly affected by the different transport quantities. The differences resulted from additional waiting times caused by longer waiting times prior to loading and congestion of several trucks at the unloading (only one truck at a time can be handled). Expectedly, the handling and waiting times behaved linearly to the change of demand. However, the lead times behaved in a less than proportionate way to the changed demand situation. The effect of the shortened or increased waiting times affected the total lead time only moderately.



Figure 5 - 18: Average lead time elements for different demand situations.

The lead time directly influences the inventory in a system: the longer the order lead time, the higher the inventory level in the total system.

Inventory – Basic Scenario

Figure 5 - 19 visualises the average inventory level per store and DC. The inventory level depends on two main aspects: the average order lead time and the demand quantity per store. Thus, the different stores showed significantly varied inventory levels. In particular, the comparably big stores in Geneva, Lausanne and Vevey with high demands per day showed high inventory levels. Furthermore, the stores in Sierre, Sion and Vevey were characterised by high inventory levels because of their comparably long lead times (cf. basic scenario).



Figure 5 - 19: Average inventory in the transport between DC and the stores.

Figure 5 - 20 illustrates the average inventory level over all stores for the four different demand situations. The illustration shows that there was a less than proportionate connection between the demand quantity and the inventory level for a reduction in demand and a more than proportionate connection for an increase in demand. For the 20% reduced demand situation, an inventory reduction of 14% can be observed. For a 20% and a 50% increased demand, inventory increases of 23% and 92% can be observed. The reason for this non-linear relationship was the simultaneous change of both influencing factors. The demand and the lead time were affected by the different demand situations.



Figure 5 - 20: Average inventory for different demand situations.¹⁶⁶

Emissions – Basic Scenario

Emissions are affected by the transport distance, the street profile and the inclination profile. The example in Table 5 - 9 shows the road distance, the fuel consumption and the emissions per route segment for transport route H 60.

transport route	route segment start	route seg- ment destination	distance [km]	fuel consumption [g]	CO ₂ [g]	NO _X [g]	particle [g]
H 60	DC HOV	CHA	207	47 610	172 892	1 758	43
	CHA	MOR	11	2 530	8 983	92	2
	MOR	NYO	27	6 210	22 325	227	6
		total	245	56 350	204 200	2 076	51

 Table 5 - 9: Average emissions for transport route H 60.¹⁶⁷

Figure 5 - 21 visualises the CO_2 emissions for the different transport routes. The longer the transport routes were, the higher the CO_2 emissions were. Reductions were the smallest for transport routes that already contained rail transport (H 220 and MS 1). Here, the replacement of the road by rail distance in the simulation scenario was comparably small. The curves of fuel consumption, NO_x and particle emissions are similar to the shape of the CO_2 emissions.¹⁶⁸

¹⁶⁶ Complete table in Appendix D.

¹⁶⁷ Full analysis in Appendix D.

¹⁶⁸ Full tables in Appendix D.



Figure 5 - 21: CO₂ emissions per transport route.

Table 5 - 10 gives an overview of the further emissions factors for one completion of the different transport routes.

route ID	distance rail /road [km]	fuel consumption [l]	CO ₂ [t]	$NO_{X}[t]$	particle [g]
H 60		67.9	0.204	2.076	13.1
H 70	0/245	74.5	0.204	2 070	55.5
H 63	0/234	64.8	0.194	1 978	48.4
Н 75	0 / 213	59.0	0.178	1 808	44.3
H 220	213 / 34	37.7	0.106	1 582	42.3
H 240	0 / 185	51.3	0.154	1 564	38.3
M 3	0 / 284	78.7	0.238	2 420	59.5
M 1	0 / 258	71.5	0.215	2 186	53.6
M 2	0 / 280	77.6	0.234	2 376	58.3
M 24	0 / 173	47.9	0.145	1 471	36.1
M 1	88 / 0	24.4	0.072	736	17.9

Table 5 - 10: Emissions per transport route.

The emissions in the one-week simulation run depended on the emissions per transport route and the number of accomplished primary and additional transport routes. The bases were the emission profiles for each transport route, as well as the number of primary and additional transport.

Table 5 - 11 gives an overview of the emissions caused by all the transportation accomplished in the one-week simulation run.

total per year	emission	unit
fuel consumption	4.73	in 1000 L
CO ₂	13.35	in t
NO _X	140 353	in g
particles	3 473	in g
distance on road	17.06	in 1000 km

Table 5 - 11: Emissions for a one-week simulation run.

Adherence to Schedules – Basic Scenario

Adherence to schedules was measured by comparing the actual arrival time at the store and the designated delivery window.

Figure 5 - 22 shows that there was a significant difference between orders that arrived at the stores too early and those that arrived too late. In reality, a comparably low average speed of transport vehicles is assumed in route planning; in particular, for long transport distances. Delays caused by unexpected disturbances can be balanced in this manner. In cases where no delays occur, the truck driver waits near to the store until the defined delivery window is open. These deliveries can be assumed to be on time. The same principle is valid for the stores in Yverdon, Sion and Sierre. They are not included in the illustration since they all have next day delivery. Therefore, it is assumed that they are all on time.



Figure 5 - 22: Average deviation and delay for different stores and DCs.

These findings are reflected in the analysis of the adherence to schedules, which set the number of too early and too late deliveries as a ratio against the total number of deliveries. Table 5 - 12 summarises the results of the adherence to schedules regarding average deviation and average delay for the different demand situations. The table shows that the adherence to schedules was best in the case of the normal demand situation. For the 20% reduced demand situation, the adherence to schedules regarding deviation was decreased. The reason is that handling times are linearly connected to demand quantity. As a result, for the reduced demand situation, the lead time was shortened and deliveries arrived prior to the defined delivery windows. The decreased adherence to schedules regarding the delay was caused by the reduction in additional transport. The amount of transport with low utilisation and short handling times was reduced and thus the share of non-delayed transport was reduced. For the increased demand situation, the adherence to schedules regarding the delay was reduced to 82%, 76% and 88% for the +/-30 minute delivery windows. The

reason for this decrease was the enhancement of lead times. However, the effect of demand increases on the adherence to schedules appears to be less than proportionate. Obviously, the enhanced delivery windows positively influenced this, too. This indicates that enhanced delivery windows positively impact adherence to schedules in general and thus, may also support the CT integration, which causes enhanced lead times.

demand situation	-20% der	mand	normal d	emand	+20% demand		+50% demand	
delivery window size	+/-15	+/- 30	+/-15	+/- 30	+/-15	+/- 30	+/-15	+/- 30
	min	min	min	min	min	min	min	min
adherence to schedules								
regarding deviations								
(too late and too early								
deliveries)	41%	54%	42%	62%	36%	56%	40%	40%
adherence to schedules								
regarding delays								
(too early deliveries)	88%	94%	95%	99%	82%	88%	76%	88%
mean deviation [h]	0.54		0.49		0.50		0.45	
mean delay [h]	0.09		0.03		0.12		0.17	

Table 5 - 12: Overview of key figures regarding the adherence to schedules for the basic scenario.

Transition The presented results served as the bases for the evaluations of the CT scenarios introduced in the following sections.

5.7.2 CT Scenario I - Integration without Adaptation to the SC Concept

Introduction CT scenario I included identical sites to the basic scenario: eleven stores and two DCs, enhanced by two terminals (cf. section 5.3.3). Accordingly, the transport routes were enhanced. Namely, all *'long-distance'* relations between DCs and the first store on the transport route were altered.¹⁶⁹ The original first route segment (transport from DC to the first store) was replaced by three different route segments: (1) the transport from the DC to the terminal in Härkingen, (2) the rail transport from Härkingen terminal to Daillens terminal and (3) the transport from Daillens terminal to the first store on the transport route. The subsequent route segments remained identical to the basic scenario. According to this, all the transport routes presented in Table 5 - 6 were replaced and renamed. Table 5 - 13 shows an extract of the list (for the full see Appendix D).

¹⁶⁹ Transport routes MS-1 and H 220 were excepted because they are accomplished by single wagon freight transport.

route ID	route	start	destination	transport	transport time [sec]	start time
	segment	DOMON	T T T T	Tesource		11.00.105
H 60 CT	30	DC HOV	THAE	VI	3 120	11:30 AM
		T HAE	T DAI		*	
		T DAI	СНА		780	
	125	СНА	MOR		900	
	175	MOR	NYO		1 500	
H 70 CT	31	DC HOV	T HAE	VII	3 120	11:00 AM
		T HAE	T DAI		*	
		T DAI	GEN		960	
	141	GEN	VES		1 200	
H 63 CT	32	DC HOV	T HAE	V III	3 120	11:00 AM
		T HAE	T DAI		*	
		T DAI	LAU		960	
	154	LAU	VEV		27 000	
H 75 CT	33	DC HOV	THAE	V IV	3 120	12:30 AM
		THAE	T DAI		*	
		T DAI	MON		3 300	
				*For t	rain transport times	soo Tablo 5 14

Table 5 - 13: Extract of route plan for the CT scenario I (complete information in Appendix D).¹⁷⁰

Furthermore, the train schedule between Härkingen terminal and Daillens terminal was integrated into the simulation model. It determines the rail transport times, which differ for the distinct trains. In total, there are eight trains a day from Härkingen terminal to Daillens terminal. Table 5 - 17 shows the different trains' start and transport times. According to the expert interviews, the maximum capacities of the trains are set to ten truck shipments (independent from the utilisation of the truck). The trains are termed with either a sequence of numbers or with specific names.

train ID	name	transport time [sec]	start time	arrival time	capacity [truck shipments]
T 1	Mistral	5 460	11:38 AM	1:43PM	10
Т 2	50727	6 840	2:41 PM	4:31PM	10
Т 3	50735	6 540	7:34 PM	9:33PM	10
T 4	50740	6 660	8:40 PM	10:27PM	10
T 5	50738	6 300	11:14 PM	01:04 AM	10
T 6	50392	8 220	00:52 AM	02:38 AM	10
Τ7	Bluehope	7 260	02:36 AM	04:09 AM	10
T 8	50708	6 420	03:36 AM	04:39 AM	10

 Table 5 - 14: Train schedule between Härkingen terminal and Daillens terminal.

Depending on the specific arrival times at the terminal, shipments are conveyed by the next available train. Therefore, trucks must arrive at the terminal at least 30 minutes prior departure. After the train arrives at the destination terminal, 30 minutes are required for transhipment and overhead before the trucks can leave the terminal for store delivery. To

¹⁷⁰ For complete information on transport routes, transport times and start times, see Appendix D.

simplify the simulation model, it was assumed that the entire truck was transported. Thus, no additional trucks for the ongoing haulage had to be modelled. For the interpretation of results, it was taken into consideration that, in reality, two trucks are necessary, one for the pre-haulage (DC to start terminal) and one for the ongoing haulage (destination terminal to store(s)). Since truck loading is identical for pre- and ongoing-haulage, this simplification did not affect the validity of the simulation results.

Utilisation of Transport Vehicles - CT Scenario I

There was no difference between the utilisation of trucks in the basic scenario and in CT scenario I. The reason was that the transport routes and demand quantity were not been altered. Thus, no additional analysis of utilisation was performed in this section. However, the eventual discussion in the following section contains qualitative considerations of the impact of CT integration, which could not be modelled by the simulation model used.

Nevertheless, the utilisation of the different trains was analysed. Figure 5 - 23 shows that for all the transport, only three different trains were used: T1 - the so-called Mistral train that leaves Härkingen terminal at 11:38 AM; T2 - train no. 50727 that leaves Härkingen terminal at 2:41 PM; and T3 - train no. 50735 that leaves Härkingen terminal at 7:34 PM. The reason for this clustering in the daytime is that all the transport vehicles leave the DCs in the morning or afternoon and reach the trains during the day and in the early evening. The different demand situations did not significantly impact the utilisation of train T1. For train T2, increased demand was reflected by increased utilisation. The number of shipments on train T3 was not increased for the increased demand situations because the capacity of the earlier train T2 did not reach its maximum. Thus, only a single shipment was made, which in most cases reached the terminal after the departure of train T3.

Figure 5 - 23 shows that on no weekday nor for any demand situation were the train capacities maximised. This means that additional waiting times at the Härkingen terminal caused by occupied trains can be excluded from CT scenario I.



Figure 5 - 23: Utilisation of trains for CT scenario I.

Order Lead Time – CT Scenario I

The introduction of additional transhipment processes, rail transport and the resulting changes to the transport routes significantly affected the order lead time and the share of transport, handling and waiting times. Figure 5 - 24 and Figure 5 - 25 illustrate the lead time elements for CT scenario I for the different stores and DCs. The illustration shows that the handling times in particular were affected by the integration of CT. The additional transhipment process caused an increase in the handling time per transport route of one hour.



Figure 5 - 24: Transport time elements per store for DC Hochdorf for CT scenario I and the normal demand situation.

As for the basic scenario the share of lead time elements is slightly different for goods coming from DC Möhlin caused by different product generation times and departure times, different transport durations as wella as different delivery windows and demand quantities.



Figure 5 - 25: Transport time elements per store for DC Möhlin for CT scenario I and the normal demand situation.

Figure 5 - 26 contrasts lead time elements for CT scenario I and the basic scenario. The main part of the overall lead time increase of 24% can be explained by the increase in handling time of 89% and the transport time by 51%. However, waiting time was decreased by 24% with CT integration. The increased level of handling times was a result of the additional transhipment processes at the terminals. The reduction in waiting times can be explained by the replacement of additional transport and handling times. Nevertheless, the share of waiting times was still 25% of the entire lead time. Thus, the analysis indicates that waiting times can be used as initial points to improve the time-related aspects of CT integration into the SC concept.



Figure 5 - 26: Comparison between average lead time elements between basic and CT scenario I.

Figure 5 - 27 contrasts the average lead times of the CT scenario I and the basic scenario for different demand situations. The comparison shows that the impact on lead times is most significant for the normal and the 20% increased demand situations. This corresponds with the findings on the non-linear relationship between demand and lead time.

This finding indicates that CT integration for higher demand levels does not necessarily affect lead times in a linear way.



Figure 5 - 27: Lead time comparison of basic and CT scenario I for different demand levels.

Inventory – CT Scenario I

The inventory levels in CT scenario I were significantly increased in comparison with the basic scenario. Since the input data of the average demand per store and weekday was not altered, the reason for this increase was the enhancement of order lead times and the changes in the durations of certain inventory levels.

Figure 5 - 28 illustrates the different inventory levels per store for CT scenario I in comparison with the basic scenario for the normal demand situation. The total average inventory for all stores was increased by 21%. There is a difference between the increase in order lead time and the increase of inventory, because the relation is – due to the calculation of the integral of the temporal inventory course – not linear (cf. section 5.3.4).



Figure 5 - 28: Comparison of inventory levels per store for CT scenario I and the basic scenario.¹⁷¹

Figure 5 - 29 illustrates the inventory times for the different demand levels. The analysis shows that the non-linear connection between demand and inventory level is also valid for CT scenario I. The reduction in demand of 20% caused a reduction in inventory of 18%. The demand increases of 20% and 50% increased the inventory levels by 28% and 87%. The reason for this was the effect of changed demand quantities reinforced by the result-ing longer lead times.



Figure 5 - 29: Comparison of inventory levels for CT scenario I for the different demand situations.

¹⁷¹ The inventory levels for the DC Möhlin change correspondingly. Full data can be found in Appendix D.

Emissions – CT Scenario I

In comparison with the basic scenario, the introduction of CT caused a significant reduction in emissions. The reason for the reduction was the replacement of road transport distances with rail transport. Table 5 - 15 illustrates this effect with the example of transport route H 60. The first route segment from DC Hochdorf to the first store was replaced by two shorter road route segments and one rail route segment. As a result, the road distance was reduced from 245 km to 105 km and 124 km by rail was integrated into the route. In total, the emissions per transport route completion were significantly reduced.

route	start position	target posi- tion	distance [km] road / rail	fuel consumption [g]	CO ₂ [g]	NO _X [g]	particle [g]
H 60	DC HOV	T HAE	53	12 190	43 087	439	11
	T HAE	T DAI	124	28 520	29 419	544	15
	T DAI	CHA	14	3 220	11 512	117	3
	СНА	MOR	11	2 530	8 983	92	2
	MOR	NYO	27	6 210	22 325	227	6
total			105/ 124	52 670	115 327	1 419	36

 Table 5 - 15: Comparison of emissions for transport route H 60 for CT scenario I.¹⁷²

Figure 5 - 30 visualises the reductions in CO₂ emissions for different transport routes. Emissions were reduced for all transport routes except MS 1 and H 220, where the reduction had been already accomplished by rail transport in the basic scenario and so were thus not altered. The smallest reduction can be observed for H 240. The reason was that on this comparably short transport route, the total distance by road could only be decreased from 185 km in total to 112 km in CT scenario I. Additionally, 124 km by rail transport needed to be considered in the emissions calculation, which counteracted the positive effect. The comparison shows that savings were biggest for the transport routes H 70 (-60%) and M 1 (-63%). On these routes, the highest share of road kilometres was saved by CT integration.

¹⁷² Full analysis in Appendix D.



Figure 5 - 30: CO₂ emissions per transport route – comparison of basic scenario and CT scenarios.¹⁷³

Table 5 - 16 summarises the key emissions figures in comparison with the basic scenario for a one-week simulation run. The table shows the absolute and the relative improvement of emissions with the CT integration.

emission factor	as-is	СТ	possible sav- ings per week	unit	improvement [%]
fuel consumption	4.73	3.02	-1.71	in 1000 L	-36%
CO ₂	13.35	7.29	-6.06	in t	-45%
NO _X	140,353	82,835	-57 518	in g	-41%
particle	3,473	2,074	-1 399	in g	-40%
distance road	17.06	9.85	-7.21	in 1000 km	-42%

Table 5 - 16: Emissions for CT scenario I in comparison with the basic scenario - calculation of possible savings for a one-week simulation run.

Adherence to Schedules – CT Scenario I

The integration of CT into the SC concept significantly increased the order lead time. As a result, the adherence to schedules was negatively affected. Figure 5 - 31 visualises adherence to schedules regarding the deviation and the delay for CT scenario I. In contradiction to the basic scenario, only a small difference between deviation and delay can be observed. The reason was that the CT integration caused a significant enhancement of lead times and thus delays in the majority of the deliveries. Thus, there is only a small differ-

¹⁷³ The curves for the futher emission factors show similar curves. cf. Appendix D.

ence between the key figures measuring deviated (too early and too late) and delayed (too late) deliveries.¹⁷⁴

Different product generation and departure times were the reasons that deliveries from DC Möhlin were delayed much less than deliveries from DC Hochdorf. In DC Möhlin, product generation takes place early in the morning (cf. Table 5 - 7). For the CT integration, this means that most of the transport coming from DC Möhlin catches the train leaving Härkingen in the morning (Mistral, 11:38 AM). Thus, adherence to the delivery windows in the afternoon is better in comparison with DC Hochdorf. The product generation and departure of trucks in DC Hochdorf occur later in the morning and in the afternoon. Trucks leave DC Hochdorf distinctively later and thus catch trains leaving Härkingen terminal in the afternoon. For the basic scenario, the late departure was sufficient to meet the designated delivery windows. However, in CT scenario I, additional transhipment times, waiting times at the terminal and additional transport times caused the deliveries to be significantly delayed.



Figure 5 - 31: Deviation and delay of deliveries per store and DC for CT scenario I.

¹⁷⁴ For DC Hochdorf, the average deviation was 3.6 h and the average delay was 3.59 h. For DC Möhlin, the average deviation was 0.56 h and the average delay was 0.55 h.

Figure 5 - 32 transfers these findings for the calculation of the adherence to schedules for different demand situations. The illustration shows that the adherence to schedules for the normal demand situation was reduced from 95% and 99% to 28% and 39% for the two different delivery window sizes. The analysis shows that for the increased demand levels, the adherence to schedules was reduced even more significantly. For the 20% increased demand situation, the adherence to schedules was reduced to 28% and 32%; for the 50% increased demand situation, it was reduced to 0% and 55%. The level of 55% adherence for the +/-30 minute delivery window can be explained by the increased number of additional transport vehicles. This additional transport was characterised by comparably low utilisation and short lead times. This effect positively influenced the average adherence to schedules. This explanation is also valid for the reduced demand situation. Here, the adherence to schedules was reduced to 41% and 58%.



Figure 5 - 32: Adherence to schedules regarding delays - comparison between basic and CT scenario I.

Transition

In brief, the analysis of CT scenario I shows that the integration of CT negatively influenced adherence to schedules and positively influenced the emissions of the analysed SC concept. The lower level of adherence to schedules mainly resulted from the enhanced lead times. Furthermore, the analysis of the lead times shows that there was still a significant share of waiting times. These waiting times could be interpreted as the initial points to open up the temporal quota to support CT integration. The discussions with the experts proved that an increase in the order lead time and the resulting inventory increase would be reasonable.¹⁷⁵ However, the significant reduction of reliability was not acceptable to the experts. Thus, based on the considerations in chapter 3 and 4, in the following sections specific adaptations to production (cf. section 3.3.3.3) and distribution concepts (cf. section 3.3.2.3) are introduced that initially aim to improve the SCP in terms of adherence to schedules.

5.7.3 CT Scenario II – Production Concept Adaptation

CT scenario II was based on CT scenario I. It differed from CT scenario I in terms of production concept configuration. Section 3.3.3.3 identified the production control logic, type of order release, secondary order release and the stocking type as adaptation points for the production concept configuration to improve CT integration. In the simulation model, the generation of products in the DC was used to model changes to the production concept. Namely, the product generation times were adapted to model the effect of changes in the *production control logic* and *order release*. Based on the analysis of the delivery windows and the train schedules, new order picking (product generation times), loading and DC departure times were determined.

The specific adaptations to the production concept are explained by means of an example. Based on the SCP analysis of CT scenario I, the improvement of adherence to schedules was of paramount interest. Furthermore, it was assumed that the improved alignment of material flows would reduce waiting times at the terminals, thus resulting in lead time and inventory reductions.

Figure 5 - 33 visualises the desired effect. The figure's upper half shows the as-is situation with unimodal road transport for transport route M 3 (basic scenario). On transport route M 3, the vehicles load goods at the DC Möhlin in the morning and deliver them in the afternoon to the Morges and Chavannes stores and in the following morning to the Nyon store.¹⁷⁶ The SCP analysis of CT scenario I shows that without adaptations to the production concept, deliveries were late by an average of 0.98 hours in Morges and by 0.46 hours in Chavannes. Since the store in Nyon receives the delivery the following morning, there was no delay.

¹⁷⁵ Here especially the product value must be considered. For instance, for different products with higher mean values, e.g. electronics or automotive parts, an increase of order lead time and the resulting capital commitment could prevent CT integration.

¹⁷⁶ The driver stays overnight with the truck near the last store on the transport route.

To meet the designated delivery windows, on transport route M 3 goods must be transported by the train leaving the Härkingen terminal at 11:38 AM. Goods can be transported to the stores 30 minutes after the train's arrival at the Daillens terminal. Thus, catching the train at 11:38 AM ensures meeting the delivery windows in Morges and Chavannes.



Figure 5 - 33: Effect of production concept adaptation on train schedule and delivery windows.

Following the presented example, new product generation and loading times were defined for all transport routes. The transport routes that contain only stores with next day deliveries (H 220 and M 24) showed no delays in the SCP analysis of CT scenario I. Hence, no adaptations were necessary in this case. Table 5 - 17 summarises all adaptations to the production concept.

route	product generation (end of or- der picking)	loading	departure DC	store 1	store 2	store 3
M 1		-0:30 h	-0:30 h	GEN	VES	
M 2		-1:15 h	-1:15 h	MON	LAU	VEV
M 3	-0:45 h	-1:30 h	-1:30 h	CHA	MOR	NYO
M 24	no adaptations no	ecessary				
H 60	-2:00 h	-2:00 h	-2:00 h		CHA	
H 63	-1:20 h	-1:20 h	-1:20 h	VEV	LAU	
H 70	-1:00 h	-1:30 h	-1:30 h	GEN	VES	
Н 75	-3:00 h	-3:00 h	-3:00 h	MON		
H 240	no adaptations no	ecessary				

Table 5 - 17: Adaptations to the production concept.

It is assumed that the adaptations of the production concept led to an improvement in the adherence to schedules. However, for an encompassing evaluation of the adaptations' effect on SCP, the impact on the other performance indicators must be examined. Thus, in the following section the impact on order lead times and inventories are evaluated. Utilisa-

tion of road transport vehicles and emissions were not analysed for this scenario because the values did not differ from CT scenario I. There was no impact on the utilisation and emission performance indicators either, since the demand quantities, available transport resources and routes were not altered. However, in reality, the adaptations made may have caused changes that could not be modelled in the simulation model. A qualitative evaluation of the possible effects can be found in the concluding discussion (cf. section 5.7.5)

Utilisation of Trains – CT Scenario II

Figure 5 - 34 shows the capacity utilisation of the three trains used for the transport of goods in CT scenario II. This illustration shows that even for the reduced demand situation of -20%, the capacity of train T1 at 11:38 AM was at the maximum. This caused additional waiting times and thus an enhancement in lead times and a decline in adherence to schedules. For the normal and increased demand situations, the effect on waiting times at the terminal was even stronger.



Figure 5 - 34: Utilisation of trains for CT scenario II.

Order Lead Time – CT Scenario II

The effect of the adaptations to the end of order picking, loading and departure to train schedule and delivery windows was reflected in the order lead times. Figure 5 - 35 contrasts the order lead time elements of CT scenario II and CT scenario I. The illustration shows that the average lead time was reduced by 9% because of the adaptations to the production concept for the normal demand situation. The improvement can mainly be affiliated with the reduction in waiting times of 32%. Also, transport time was slightly reduced. The reason for the transport time reduction was the balancing of arrivals at the stores. This means that trucks did not have to wait for unloading because the vendor dock was occupied.



Figure 5 - 35: Transport time elements per store for CT scenario II for the normal demand situation.

The adaptations to the production concept were defined according to the conditions of the normal demand situation. Hence, lead time improvement was the biggest in this case. Figure 5 - 36 contrasts lead times for the different demand situations. In the basic scenario and CT scenario I, lead time was connected disproportionately with the increase in demand. The reason was that the demand increase primarily influenced the handling times, which is only one element of lead times. Furthermore, lead time enhancements caused by waiting times in the terminals because of missed train capacities reinforced the effect.



Figure 5 - 36: Comparison of lead time for different demand situations between CT scenario I and CT scenario II.

Inventory Level - CT Scenario II

In comparison with CT scenario I, the inventory in CT scenario II was reduced. The main reason was the shortening of order lead times.

Figure 5 - 37 visualises the average inventory level per store and DC. For most stores, a slight reduction in inventory levels was achieved by the adaption of the production concept.



Figure 5 - 37: Average inventory in the transport between DC and stores for CT scenario II in comparison with CT scenario I.

Figure 5 - 38 contrasts lead times for different demand situations for CT scenario II. As for the basic scenario and CT scenario I, inventory appears to be connected in a more than proportionate way with the demand. An increase in demand of 20% caused an increase in inventory of 34%. However, for the increase in demand of 50%, only a 46% rise in inventory can be observed. This system behaviour can be attributed to the higher number of additional road transport and the resulting decrease in average lead times. Thus, in comparison with CT scenario I, in this analysis the effect of the reduced waiting times with full capacities was even stronger; thus the inventory level rise was not that significant.



Figure 5 - 38: Comparison of lead time elements for CT scenario II for different demand situations.

Adherence to Schedules – CT Scenario II

The main focus of the production concept adaptation was the improvement of the adherence to schedules, which was strongly negatively affected by CT integration in CT scenario I.

Figure 5 - 39 illustrates the reduction in delays for the different stores in comparison with CT scenario I. To improve the clarity of the diagram, the analysis in CT scenario II initially focuses on adherence to schedules regarding delayed deliveries.¹⁷⁷ As for CT scenario I, the stores with next day delivery are not included in the diagram. In total, the adaptation of product generation times reduced the mean delay of transport from 2.07 hours to 0.68 hours. The effect strongly varied for the different stores. The reason for this strong variance can be found in the number of stores per route and the average demand. For the transport routes with three delivery points, the utilisation of vehicles and the number of pallet equivalents was comparably high; thus handling times varied more intensively. Consequently, the risk of missed trains and delayed arrivals was increased. The illustration shows that this argument is especially valid for the big stores with high demands, for instance in Vevey and Geneva, and for the stores that are last on the transport routes, such as Morges, Vésenaz and Vevey.



Figure 5 - 39: Comparison of delay between CT scenario I and CT scenario II.

Figure 5 - 40 shows that the adherence to schedule in CT scenario II was significantly improved over that in CT scenario I. For the normal demand situation, an improvement

¹⁷⁷ This focus corresponds to reality: in cases where they reach the site too early, truck drivers wait nearby the destination store until the delivery window.

from 28% to 39% (+/- 15 minute delivery window) and from 39% to 54% (+/-20 minute delivery window) was achieved. It is evident that for higher demand situations, the effect on the adherence to schedules is nearly constant for all demand situations.



Figure 5 - 40: Comparison of adherence to schedules for different demand situations.

	-20% dei	nand	normal demand		+20% de	mand	+50% de	mand
delivery window	+/-15	+/- 30	+/-15	+/- 30	+/-15	+/- 30	+/-15	+/- 30
	min	min	min	min	min	min	min	min
adherence to schedules re-								
garding deviations (too								
early and too late deliveries)	40%	55%	39%	54%	39%	56%	38%	53%
adherence to schedules re-								
garding delays								
(too late deliveries)	41%	56%	41%	55%	41%	57%	40%	54%
mean deviation [h]	0.73		0.73		0.80		0.81	
mean delay [h]	0.73		0.73		0.78		0.79	

Table 5 - 18 summarises the key figures for different demand situations.

Table 5 - 18: Overview of key figures regarding the adherence to schedules for CT scenario II.

The reason that CT scenario II did not reach the same adherence to schedules rate as the basic scenario was due to the risk of missing the defined train departure times. In a unimodal road concept, a few minutes delay to a particular transport vehicle (e.g., caused by longer loading times due to additional transport quantities) causes a few minutes delay to the delivery. In a SC concept including CT, the effect of a few minutes delay at the loading can cause several hours delay for the delivery if the truck misses the designated train. Furthermore, the analysis of train utilisation has shown that in CT scenario II, the train capacities were not sufficient. Thus, shipments were delayed and the adherence to schedules decreases.

Since the adaptations to the production concept were based on normal and average demand quantities, the adaptations need to be small to reach an acceptable level of adherence to schedules; larger temporal shifts may require further operational adaptations. Nevertheless, the analysis generally proves that adaptations to the production concept with regards to a specific CT concept can be utilised to improve SCP.

The following section analyses the SCP effect of adaptations to the distribution concept with regards to a specific CT concept.

5.7.4 CT Scenario III – Distribution Concept Adaptation

CT Scenario III was also based on CT scenario I. It differed from CT scenario I in terms of the distribution concept configuration. Section 3.3.4.3 identifies the regional structure, echelon structure, delivery service and the delivery time point as adaptation points to positively influence the performance-oriented CT integration.

In accordance with the views of the experts from *Manor*, an adaptation of the regional structure or echelon structure was not possible. Thus, the simulation study was used to model adaptations to the *delivery service* and *delivery time point*.

The discussions with the experts showed that alterations to the delivery windows in the stores require operational changes in the store processes. Personnel-intensive storage processes would have to be temporally shifted. In several stores, products are ordered in the morning and delivered in the afternoon. However, in many stores, the storage of products on the shelves does not take place until the next morning. These considerations indicated the need to accomplish a two-step analysis.

First, the SCP impact of shifting delivery windows from the initial time point in the afternoon to a later time point was examined. The temporal shift was accomplished according to the given product generation, loading, departure times and train schedule (CT scenario III). Second, the SCP impact of shifting delivery times from same day to next day delivery was examined (CT scenario IIIa).

Figure 5 - 41 concretises the changes in the simulation scenario. The figure's upper half shows the as-is situation with unimodal road transport for transport route H 75. Assuming that the product generation, loading and departure times were not alterable, the truck arrives at Härkingen terminal at approximately 12:30 AM. The shipment is conveyed by the train leaving the terminal at 2:41 PM.¹⁷⁸ After arrival at 4:31 PM and transhipment (30 mininutes), the truck leaves the terminal at 5:01 PM. The transport from Daillens terminal

¹⁷⁸ It is also possible that the transport meets the train at 11:38 AM, but there is only a small temporal buffer of 8 minutes that can be affected by the duration of loading, as well as fluctuation in transport and loading times.

to the Morges store (44 km) takes approximately 53 minutes. Thus, a shift in the delivery window by 1 h to 6:00 PM was suggested.



Figure 5 - 41: Effect of distribution concept adaptation to train schedule for transport route H 75 (CT scenario III).

In accordance with the example presented, Table 5 - 19 summarises all the adaptations to the distribution concept for CT scenario III.

route	product generation	loading	departure DC	arrival store 1	arrival store 2	arrival store 3	
M 1	7:45 AM	10:00 AM	10:30 AM	GEN +2:30 h	VES + 2:30 h		
M 2	8:15 AM	10:45 AM	11:15 AM	MON +1:30 h	VEV +1:30 h	LAU +1:30 h	
M 3	9:15 AM	11:00 AM	11.30 AM	CHA +2:00 h	MOR +2:00 h	NYO +2:00 h	
M 24	11:45 AM	2:15 PM	2:45 PM	no adaptations necessary			
H 60	11:00 AM	11:30 AM	12:00 AM	MOR +1:45 h	CHA +1:45 h	NYO +1:45 h	
H 63	10:20 AM	11:00 AM	11:30 AM	VEV +2:00 h	LAU +2:00 h		
H 70	10:00 AM	11:00 AM	11:30 AM	GEN +1:00 h	VES +1:00 h		
Н 75	12:00 AM	1:00 PM	1:30 PM	MON +1:00 h			
H 240	3:10 PM	4:00 PM	4:30 PM	no adaptations n	ecessary		

Table 5 - 19: Adaptations to the distribution concept for CT scenario III.

In the following section, the effect of the adaptations to the distribution concept on SCP is analysed.

Order Lead Time – CT Scenario III – Distribution Concept Adaptation

In comparison with CT scenario I, the adaptation of the distribution concept was reflected by enhanced order lead times in CT scenario III. Figure 5 - 42 illustrates the lead times for CT scenario III in comparison to the results of CT scenario I for different demand levels. The illustration shows that for CT scenario III, the effect of lead time enhancement was less significant for higher demand levels (only 6% and 9% for the 20% and 50% increased demand situation in contrast with 12% each for the normal and reduced demand situation).



Figure 5 - 42: Comparison of lead times of CT scenario I and CT scenario III for different demand situations.

The enhancement of lead times was caused by the shift of delivery windows to a later time point. As a result, the products were removed from the simulation at a later time point. As shown previously, the demand situation influences the lead time in a disproportionate way. For CT scenario III, the reduced demand situation caused a reduction in lead times of 2%. In the case of 20% and 50% increased demand situations, the increases in lead times were 0.3% and 12%. The reason for the disproportionate connection was that only the handling times were directly affected by the changes in demand.

Figure 5 - 43 visualises the changes in lead time elements for the normal demand situation. The illustration shows that the adaptation of the distribution concept mainly increased the waiting time ($\pm 27\%$). The increase in waiting time resulted from the shift of delivery windows to a later time point. All deliveries arriving prior to these shifted delivery windows thus experienced additional waiting times. In total, the adaptations to the distribution concept raised the order lead time by 7%.



Figure 5 - 43: Comparison of lead time elements between CT scenario I and CT scenario III.

Inventory Level - CT Scenario III - Distribution Concept Adaptation

In comparison with CT scenario I, the shift of delivery windows enhanced the total lead times and thus the inventory levels. Figure 5 - 44 contrasts inventory levels for all stores and DCs for CT scenario III and CT scenario I. The illustration shows that the influence on the different stores was between 0% and 16%. There were no changes for the Nyon, Sierre, Sion and Yverdon stores because these have next day delivery and no adaptations were made. In total, the increase in inventory levels was proportionate to the increase in lead time of 7%.



Figure 5 - 44: Average inventory in the transport between DC and stores coming from DC Hochdorf for CT scenario III.

Figure 5 - 45 compares the inventory levels for the different demand situations. As explained for the previous scenarios, the reason for this disproportionate increase in inventories was the combination of effects regarding increased lead times and higher transport quantities because of the higher demand.



Figure 5 - 45: Inventory levels for different demand situations - CT scenario III.

Adherence to Schedules – CT Scenario III - Adaptations to the Distribution Concept

Figure 5 - 46 visualises the improvement in the adherence to schedules regarding the too late deliveries by the adaptation of the distribution concept.



Figure 5 - 46: Comparison of delay between CT scenario I and CT scenario III.

Figure 5 - 47 visualises the improvement in adherence to schedules regarding too late deliveries. For the normal demand situation, the adherence was increased to 84% for a delivery window of +/- 15 minutes and to 92% for a delivery window of +/- 30 minutes. For the reduced demand situation, the adherence to schedules was even better: 85% and 96%. For the 20% increased demand situation, the adherence to schedules reached 80% and 92%. Only for the 50% increased demand situation did the adherence to schedules decrease, to 64% and 75%.

The illustration shows that the effect of the distribution concept adaptation regarding the CT concept was best for the normal and the reduced demand situation, since the adapta-

tions were defined according to the normal demand situation. Thus, it can be assumed that when shifting the delivery windows to a later time point, the adherence to schedules would be increased for higher demand levels.



Figure 5 - 47: Comparison of adherence to schedules regarding delays for different demand situations.

Delivery Service Shift to Next day Delivery

Finally, the shift in delivery time from same day to next day delivery was analysed. The simulation results and the discussion with the experts showed that this adaptation of the distribution concept would mean a number of significant advantages for implementation in practice.

The first reason is that shifting delivery windows to the evening hours (as accomplished in CT scenario III) would require relevant operational changes, as stores may already be closed. If the goods can be deloaded and buffered in the evening, there will be no personnel for storing goods on the shelves. Even if moving the shifts of the replenishment personnel were possible, there would be no customers to profit from the availability of goods. The second reason is that goods could be transported on rail at night time. During the night time driving ban, the terminal operator could autonomously decide on which train to put the shipments. The enhanced time span would improve the terminal operator's capacity planning. This is an important aspect, particularly when the capacity of the CT trains is used to the full (as shown for CT scenario II and III). Thus, by incrementing the waiting times at the terminals, the resulting enhancement of lead times and delivery delays could be avoided. Also, the charged road carriers would have additional freedom in terms of loading and delivery times, since there are no specific delivery windows at the *Swiss Post*

terminals.179

However, no separated analyses of the adherence to schedules were accomplished for shifting the delivery times to the next morning. It is assumed that the adherence to schedules in the next morning could reach at least the level of the basic scenario, since from Daillens terminal there are only comparably short road distances to travel to serve the stores. The risk of congestion and delays is thus comparably small.

There were no changes in the utilisation of transport vehicles and the emission levels that could be modelled by means of the simulation study. A qualitative evaluation is accomplished in the concluding discussion (cf. section 5.7.5).

Lead Time - CT Scenario IIIa - Shift to Next Day Delivery

In the simulation, the shift of delivery times significantly increased the waiting times. The additional waiting time overnight caused an increase in the overall waiting time of 277% and in total lead time of 90%. The additional time span was sufficient to deliver all goods, even for the increased demand situations. Thus, lead time was identical for all demand situations. Figure 5 - 48 visualizes the considerations.



Figure 5 - 48: Comparison of lead time elements for CT scenario III and CT scenario IIIa.

Inventory - CT Scenario IIIa

The adaptation of the delivery times to a next day delivery for all stores also caused an increase in inventories. Figure 5 - 49 shows that the shift of delivery windows approxi-

¹⁷⁹ From the shipper's point of view, it could be assumed that the terminal operator and even the road carrier offer improved prices if they are not restricted to use one certain train for the shipment's transport.
mately caused a doubling of inventories. The effect was not linear to the lead time enhancement because the average inventory was calculated as the integral of the inventory course over the time.



Figure 5 - 49: Comparison of inventory levels for CT scenario III and IIIa for different demand situations

Finally, the results of all the simulation scenarios are compared and discussed. In particular, the findings are compared with the expected results (E_1 to E_6). In response to the findings in the thesis in hand, practical recommendations and limitations of the research are derived and explained.

5.7.5 Discussion of Simulation Results

The simulation study shows that SCP can be increased when the SC concept, namely the production and distribution concepts, are aligned with the embedded CT concept. Section 5.3.5 presents the discussion of the SCP effects and the six statements of expected results (E_1 - E_6). Furthermore, unexpected results, such as non-linear cause-and-effect relationships are discussed. Finally, the generalisability and transferability of the findings are evaluated.

SCP Effect of CT Integration into a Lean SC Concept

As a first step, the performance-effect of the CT integration in comparison to the SC concept based on unimodal road transport is evaluated in the following section. Therefore, SCP of the basic scenario is contrasted with the SCP of the analysed CT scenarios.

 E_1 : The integration of CT impacts SCP in terms of (a) enhancement of lead time, (b) adherence to schedules and (c) reduction of emissions (in comparison with a unimodal road transport concept). a) The simulation study shows that CT integration causes a significant enhancement of lead times. The effect is less than proportionate for all scenarios and demand situations. CT integration without adaptations to the SC concept led to an increase in lead time of 24%. The increased lead times resulted from increased transport times (because of rail transport), handling times (because of additional transhipment processes) and waiting times for the next train departure at the terminal in the case that a train is missed.

Adaptations to the production concept, namely changes to the product generation time, reduced this increase in lead times by 13%. Adaptations to the distribution concept, namely shifting delivery windows by a few hours, caused an increase of 31% in terms of improved lead times; a shift to next morning delivery caused a rise of 162%.

Adaptations to the SC concept were made with the target to show cause-and-effect relationships between the elements of the CT, the SC concepts and the resulting SCP effect, rather than to optimise the modelled SC. Thus, it can be assumed that further improvements could be achieved for a more detailed analysis of the initial situation and the accomplishment of more specific adaptations (for instance, different adaptations for different weekdays). For instance, the analysis of CT scenario III shows that there were contradictions in the SCP system. A smaller shift in the delivery windows would have positively influenced total lead time, but negatively influenced adherence to schedules.

Thus, for the generalisation and transfer of results, company-specific SCP understanding and weighting of targets must be considered.



Figure 5 - 50: Comparison of lead time for different simulation scenarios.

b) The discussions with the experts proved the importance of adherence to schedules in the real system. The simulation shows that CT integration initially negatively affected adherence to schedules. However, adaptations to the production and distribution concept with regards to the specific CT can effectively improve SCP in terms of adherence to schedules. Figure 5 - 51 contrasts the adherence to schedules for different demand situations regarding delayed deliveries. The figure shows that although improvements were achieved, only for CT IIIa was an acceptable level of adherence to schedules (more than 92%) for a +/-30 minute delivery window achieved. This indicates that an in-depth analysis of the SC concept and the CT concepts, as well as the specific framework conditions, is necessary to benefit from the findings presented in the thesis in hand. Furthermore, it can be assumed that a combination of aligned adaptations in different SC sub-concepts is even more effective than isolated adaptations.



Figure 5 - 51: Comparison of adherence to schedules regarding delayed deliveries.

Adherence to schedules is interpreted in two different ways. In the chosen example of a lean retail SC, adherence to schedules regarding delays (namely the share of notdelayed deliveries) is central. However, in other industries, the share of deviations, (namely the share of too early *and* delayed deliveries) is significant. For instance, in highly frequent transport concepts with high-value goods (for instance in JISdeliveries), a too early arrival must be as negatively evaluated as a delayed one. However, the analysis shows that the number of both delayed and deviated deliveries can be reduced by adaptations to the production and distribution concepts.

c) The analysis shows that CT integration significantly influenced SCP in terms of emissions levels. Table 5 - 20 projects the results from the one-week simulation run to calculate the possible savings in one year. However, discussions with the experts proved that the projection of results for a period of one year is necessary to compare

activities with other companies, but also for purposes within the corporate company, e.g., for emissions reduction activities. In total, *Manor* could save more than 300 tons of CO_2 per year through CT integration into the SC concept for the eleven stores in Western Switzerland alone.

total per year	basic scenario	CT scenarios	savings per year	unit
fuel consumption [l]	245.9	156.8	-88.97	in 1000 L
CO ₂ [t]	694.5	379.3	-315.19	in t
NO _X	7.3	4.3	-2.99	in t
particle [g]	0.2	0.1	-0.07	in t
distance road [km]	793.7	391.8	401.86	in 1000 km

Table 5 - 20: Comparison of projected emission factors and possible savings for one year.

 E_2 : The integration of CT allows the (a) reduction of road transport vehicles and the (b) improved utilisation of road transport vehicles.

The impact on the number and utilisation of transport vehicles could not be entirely answered by means of the simulation study. A number of different aspects must be considered to evaluate the impact of CT integration on the number of road transport vehicles. After CT integration for each transport route, two trucks are necessary: one truck for pre-haulage and one truck for ongoing-haulage.¹⁸⁰

For the calculation of utilisation levels, the loading on the first route segment after leaving the DC serves as the basis because here, utilisation reaches its maximum.¹⁸¹ On long transport distances, drivers stay with the goods overnight before store delivery the next morning. The simulation results indicate that in the case of CT integration, it is possible that each truck accomplishes two or more transport routes per day: either between a DC and a terminal or between a terminal and the stores. This balances the number of trucks to the level in the basic scenario. In particular, for trucks between terminal and stores, the share of idle driving distance can be reduced. Furthermore, no overnight stays of trucks and drivers are necessary. Depending on the specific framework conditions and concentrating single trucks to one region, this can mean that when a driver's shift is terminated, the truck can be further utilised. This increases utilisation in a wider sense. However, for quantification purposes, an in-depth analysis would be necessary.

In summary, the expected result of a reduced number of road transport vehicles could

¹⁸⁰ In the basic scenario one truck accomplishes the entire transport route.

¹⁸¹ With the first unloading process, utilisation decreases for each store delivery. Transport back to the DC was excluded from consideration.

not be confirmed but is presumable. The discussions with the experts indicated that a reduction is possible. The simulation results indicate that, especially for the transport vehicle accomplishing the store delivery, improved utilisation can be assumed.

 E_3 : CT integration leads to an (a) increase in the total inventory level and (b) a reduction of time flexibility (in comparison with a unimodal road transport concept)."

a) The simulation study shows that CT integration leads to an increased inventory level. Depending on order lead times and the log course over time, the inventory level was increased by 26% with CT integration, without adaptations to the SC concept in comparison with the basic scenario. The adaptations to the production concept reduced the inventory increase to 18%; adaptations to the distribution concept raised it to 35%. The shift of all delivery times to next day led to an increase in the total inventory level of 78%.

Figure 5 - 52 shows that the increase in inventory was less than proportionate to the lead time enhancement but more than proportionate to the changes in demand.



Figure 5 - 52: Comparison of inventory levels for all scenarios and for different demand situations.

b) Time flexibility could not be directly evaluated by means of the simulation study. However, the simulation results provide initial points for an argument regarding time flexibility. As shown in the discussion of the simulation results, catching specific trains is critical for SCP. For closely coupled material flow processes, missing designated trains does inevitably mean that the lead time is enhanced by at least the amount of waiting time.¹⁸²

¹⁸² This goes along with a raise of the average inventory level.

As shown in 3.1.4.1 lean SC concepts (type I - continuous replenishment) aim at the reduction of temporal buffers. Thus, delays at the terminal also mean that adherence to schedules is negatively affected.¹⁸³ Thus, CT integration causes a high deviation in lead times and adherence to schedules, which means restrictions are required for the predictability, 'plannability' and control of the SC concept. To avoid this effect, actors must focus their activities on meeting the designated trains. As a result, time flexibility is reduced, for instance expressed in short-term changes in departures or transport quantities. However, time flexibility can be ensured by an enhancement of delivery times (here modelled by the shift from same day to next day delivery), as shown for CT scenario IIIa. In this case, the terminal and CT operators gain flexibility and can balance capacities. In brief, a target conflict, between time flexibility and adherence to schedules, can be identified.

Performance Effect of Adaptations to the SC Concept

As a second step, the performance effect of single adaptations to the SC concept, namely adaptations to the production (E_4) and distribution concept (E_5) , is discussed.

 E_4 : The harmonisation of production arrival times with the CT concept leads to a (a) reduction in lead times, (b) increased time flexibility and (c) improved adherence to schedules.

a) The analysis of CT scenario II shows that the adaptation of the product generation time reduced lead times on average by 9% in comparison with CT scenario I. The main part of the reduction was achieved by eliminating waiting times in the distribution concept. After the adaptation, the increase in lead times in comparison with the basic scenario was only 13%. The discussions with the experts showed that this increase in lead times and the resulting increase in inventory levels would be acceptable. Furthermore, the discussions proved that the suggested adaptations to the production concept are applicable in practice. Only the changes to the order picking process to allow an earlier departure for transport route H 70 (DC Hochdorf to Geneva) would require significant operational adaptations. Since the demanded quantities of the stores on this route are comparably high, the shift in order picking would mean that the DC must shift its opening times.

¹⁸³ In the field production it has be shown that a strong distribution of lead times negatively affects the planning basis and thus, further performance parameters of a production system.

b) Time flexibility was qualitatively evaluated. The analysis shows that adaptations to the production concept influence time flexibility in an indifferent way according to the interpretation of the term 'harmonisation'.

Shifting product generation times to enhance the time span to train departure increases the time flexibility of road carriers and DCs. Shifting product generation times according to train departure times and in the sense of reducing temporal buffers decreased time flexibility. Thus, the expected result should be reformulated to: *The enhancement of the time quota between production arrival times with the CT concept regarding train departures increases time flexibility*.

c) CT scenario II shows that in comparison with CT scenario I, the adaptation of the production concept improved SCP in terms of adherence to schedules. The analysis shows that the bigger the effect of adaptations, the longer the delivery windows. Although the average delays could be reduced from 2.03 hours to 0.73 hours, adherence to schedules was only improved from 28% to 41% (+/-15 minute delivery window) and from 39% to 55% (+/- 30 minute delivery window).

The discussions with the experts proved that this adherence to schedules is not sufficient for implementation in practice. For implementation, an adherence to schedules of at least 90% is required. These findings indicate that isolated adaptations of production concepts may not be sufficient for performance-oriented CT integration. Presumably, the integrated application of adaptations to different SC sub-concepts would be promising.

 E_5 : The (a) enhancement and the (b) harmonisation of store delivery windows with regards to the CT concept leads to (1) a reduction of lead times and (2) an improvement in adherence to schedules.

a) The impact of the enhancement of delivery windows was examined for all simulation scenarios. As expected, the simulation shows in all scenarios that the enhancement from a +/- 15minute to a 30 +/-minute delivery window positively influenced the adherence to schedules in a more than proportionate way. However, the enhancement of delivery windows means that capacities (e.g., for personnel and handling equipment for unloading, buffering and replenishment) must be held available, which may cause additional costs. Thus, this adaptation might be restricted to larger stores with a higher number of deliveries and personnel flexibility.

b) The harmonisation of store delivery windows with the CT concept, namely the train departure times, increased order lead times and improved adherence to schedules. The

effect was stronger the further the delivery windows were shifted backwards. This finding confirms a conflict in the SCP target system between lead time and adherence to schedules. The simulation study shows that shifting delivery windows improved adherence to schedules regarding too late deliveries nearly to the level of the basic scenario. For the enhanced delivery windows, 92% adherence to schedules was reached. Presumably to the disadvantage of lead times, the adherence to schedules could be further increased to reach the required level of practice implementation. Thus, the first part of the expected result cannot be confirmed. The simulation shows that the harmonisation of delivery windows with the CT caused an enhancement of lead times (because of an increase in order lead times).

 E_6 : The enhancement of delivery times leads to (a) increased lead times, (b) increased inventory levels, (c) improved adherence to schedules, (d) time flexibility and the (e) utilisation of transport vehicles.

a) + b) The analysis of CT scenario IIIa shows that the enhancement of delivery times increases lead times and thus inventory levels. The relationship between the increases in lead times and inventory was less than proportional.

c) Since the shift in target arrivals could not be directly modelled, it was assumed that the adherence to schedules could reach at least the level of the basic scenario. The ongoing transport route starts the next morning from the Daillens terminal. This transport route is compiled of comparably short route segments. It can therefore be assumed that there are only few possible delays and thus all deliveries are on time.¹⁸⁴

d) + e) As shown for E_1 , time flexibility for all SC actors was increased by enhanced time buffers between scheduled events. Road carriers, terminal operators and CT operators in particular could improve their planning and thus their capacity utilisation.¹⁸⁵ As long as the product generation times are not adapted, it can be assumed that road carriers can improve utilisation capacities. However, this effect could not be shown by the simulation study. With regards to train capacities, the simulation shows that for the shifted product generation times, the capacities on single trains were used to the full. As explained before, this indicates that shifting deliveries to the next morning allows more balanced capacity utilisation.

¹⁸⁴ The effect on adherence to schedules could not be modelled.

¹⁸⁵ cf. Bendul (2009), pp. 170.

5.7.6 Limitations, Validity and Generalisability of Findings

The simulation model provides realistic data for the basic scenario, as well as for the integration of CT into the SC concept, as the discussions with the experts and the comparison with the as-is situation show. In the following section, the discussion of the restrictions and findings of the simulation study is accomplished in two steps. First, the restrictions in a narrower sense - focusing on the simulation model and the application scope - are discussed. Second, the findings are discussed in a wider sense - considering the previous findings of the thesis in hand, as well as the transferability and generalisability of the findings.

Internal and External Validity - Applicability of Suggested Adaptations to Practice

Several validation and verification steps were performed to improve the simulation's level of internal and external validity (cf. section 5.5.3). However, to reduce the complexity of the simulation model, several assumptions and simplifications were necessary. These aspects are discussed in relation to the background of the applicability and generalisability of results in practice. The restrictions discussed in this section do not influence the comparability of SCP between the different scenarios and thus the validity of identified cause-and-effect relationships. However, the restrictions must be considered to evaluate the generalisability and applicability of the findings for practical use.

The quality of a simulation model depends on the quality of the input data. The database provided by *Manor* is comprehensive and has a high level of detail. The iterative discussions with the experts allowed intensive validation and verification steps. The comparison of simulation results for the basic scenario showed a high level of similarity regarding SCP indicators. Thus, it can be stated that the simulation reached a high level of validity.

Emissions levels were calculated only for the transport processes. For a more detailed analysis, further considerations (e.g., on the necessary shunting and handling processes, as well as the specific transport vehicles and their current utilisation) would be required. The specific emission categories of the trucks as well as the propulsion modes (diesel or electricity) of the locomotives could be included to make the calculation more specific.

Lead transport times and loading times were calculated on the basis of transport distance and demand quantity. On the basis of the expert interviews, a conservative transport speed of 50 km/h was assumed. Influencing parameters, such as congestion, technical disturbances, driving styles and break times were not modelled. Depending on the situation- and company-specific SCP requirements regarding adherence to schedules and accuracy of emissions levels, it would be necessary to assess the impacts of different transport routes (e.g., specific traffic situations and route-specific inclination profiles) and transport times (weekday and time).

SCP in terms of *flexibility* could only be evaluated on a qualitative basis. Here, the focus was on time flexibility that can be expressed by the possibility of short-term changes regarding times, quantities and destinations. For an encompassing evaluation of flexibility, further aspects, such as restrictions to advance announcements for the terminal and CT services, would be required. Furthermore, the SC concept's flexibility regarding disturbances in rail or road transport could be analysed in depth. For this, comprehensive further analysis would be necessary.

Also, the impact of CT integration on SCP regarding the *utilisation* of transport vehicles was evaluated qualitatively. The calculation of utilisation for the basic scenario and CT scenario I served as the basis. The SCP effect of transport route adaptations (in terms of the combinations of delivery stores per route) was not a central aspect of the research question. Although CT integration may encourage company- and situation-specific changes to transport routes, this aspect was not included in the simulation study. Further information on transport capacity planning and control of specific carriers would have been required to evaluate how many and which transport routes in the CT scenarios could be accomplished by a certain transport vehicle.

For the evaluation of utilisation, the utilisation on the first route segment served as the basis. However, in practice, the utilisation of a certain vehicle changes at each loading and unloading process. However, the discussion with the experts showed that this approach is acceptable and refers to practice. Nevertheless, the consideration of partial utilisation would open up the possibility to further improve capacity utilisation by changes to transport routes or by additional sites.

The calculation of *inventory levels* was performed only for the moment of product generation in the DC and the moment of unloading at the destination store. This provides restricted statements on the total SC inventory. Since production sites were not directly included in the simulation model, for a correct evaluation of the inventory levels, the production itself, the transport from production to DC and transport within the DC would need to be considered.

Exclusion of Stochastic Elements

The presented simulation model was setup without stochastic elements. Different demand quantities were modelled by four simulation runs with different demand levels per scenario. Additionally, different demand quantities per weekday were included. The exclusion of further stochastic elements and fluctuations offers the advantage that SCP effects can be directly referred to a certain adaptation of the SC concept. If stochastic elements were included, the researcher had to clarify, whether a certain effect results from the stochastic input data or from the characteristics of modelled SC concept. Furthermore, a high number of simulation runs was necessary to achieve a desired level of significance (cf. section 5.1.4). However, for a simulation study which is setup as decision support for CT integration, an inclusion including stochastic elements might be advisable depending on the specific framework conditions.

Despite these restrictions, the findings of the simulation study show a high level of internal and external validity. However, to apply the suggested adaptations and the calculated key figures in practice, limitations regarding generalisability must be considered. Several findings can be transferred to other industries, product groups and countries, but there are also limitations.

Generalisability and Transferability of Findings

The simulation study was set up with a limited scope of a lean SC concept and regular line transport on the basis of a retail distribution network. In the following section, the relevant differences between the chosen scenarios and other SC concept types, CT concept types, industries, product groups and countries are worked out to evaluate the generalisability and transferability of the findings.

Fluctuations in demand were modelled by four different demand situations. The discussion with the experts showed that in a retail SC, these situations are representative of standard delivery. However, extraordinary demand situations were consciously excluded. Thus, transferability to *agile and leagile* (downstream of the customer decoupling point) SC concepts, with highly fluctuating demands, is limited. The simulated SC concept was characterised by comparably short lead times. All priority goods were handled with a lead time below one day. The simulation shows that the enhancement of delivery times improves SCP-oriented integration. Thus, a transfer to other *lean SC concept* types (type I - plan and optimise and type II - continuous replenishment) would be, in general, possible.

This means that the findings are transferable to lean SCs in other industries and other

product types. Relations between a certain source and a certain destination with predictable and stable transport quantities and frequencies on a daily basis are suitable for transferring the findings.¹⁸⁶ Transport vehicles should be organised according to defined time schedules. This is the prerequisite for the regular booking of capacities on certain trains.

In the presented example, the gained advantages regarding emission levels caused the accepted lead time and inventory increases. However, for transferring results to other application fields, two main aspects must be considered. (1) The situation-specific *spatial configuration* of the SC concept is critical for CT integration. The availability of terminals and suitable CT services in the required start and destination region(s) must be given. (2) The specific product value affects the capital commitment due to inventories. In retail, the average value is comparably low (e.g., in comparison with the automotive and electronics industries). To transfer findings to other products and industries, the *relative value of goods* is central. For instance, in JIS deliveries, a lead time and inventory increase may not be acceptable, as the high-value goods may cause a significant increase in capital commitment.

Furthermore, the findings are transferable *to other CT concepts*. The suggested adaptations are also valid, but the resulting extra effort required for the adaptations must be weighed against the resulting advantages.

The simulation model was limited to a certain aspect of the conceptual model presented in chapter 4. The procurement concept was not modelled at all. Thus, no statements on the developed propositions can be made with regards to procurement. Also, it was limited to material flow processes. Information and financial flows were not modelled in the simulation. Thus, no statements can be derived on the intensity of the coordination effort required for an SC concept with unimodal road transport or for SC concepts with a CT concept. As a result, the impact of integrative measures and instruments could not be examined in the simulation model. These rather general limitations are comprehensively discussed in the concluding sixth chapter.

¹⁸⁶ Therefore, a network with several sources and destinations (as in the example SC) is not necessary.

5.8 Intermediate Findings

This section briefly summarises the main findings of the simulation study in the context of all the findings in the thesis in hand.

First, the simulation study sustains the findings of the previous chapters in the present thesis (under consideration of the presented limitations). It supports the main assumptions of the conceptual research framework. The simulation study proves that situation-specific adaptations to SC concepts open up time requirements for the CT integration, thus positively improving SCP in the case of a lean SC concept and for a regular line transport concept.

As a result of the number of validation and verification steps, the developed simulation model was suitable for the analysis of cause-and-effect relationships between elements of the CT and SC concepts, although several assumptions and simplifications were necessary to keep the model's complexity low.

The comparison of the four different scenarios and for different demand situations pointed out target conflicts in the key figures of the SCP system. In particular, the connection between the targets for lead times and inventory level reductions and the improvement of adherence to schedules and flexibility turned out to be central for CT integration.

It could be demonstrated that CT integration leads, as expected, to an enhancement of lead times. However, a disproportional effect on inventory levels, adherence to schedules and time flexibility was evident in the simulation study. The reasons for this connection were the defined and comparably seldom train departure times. In contradiction with unimodal road transport concepts, in which a few minutes delay at loading or transport usually causes a few minutes delay at delivery, in a CT concept, a few minutes delay in loading or transport may cause several hours delay at delivery, since missing a designated train increases delay times.

The same explanation is valid for the unexpected results for the harmonisation of delivery windows and production times with the CT concept. The synchronisation, namely a reduction of time quotas between events in the SC (e.g., between production generation, order picking, loading and delivery windows), caused a reduction in adherence to schedules. Whereas a reduction of average lead times could be observed, the expected positive effect on adherence to schedules and time flexibility could not be confirmed. This finding encourages discussion of the appropriate level of harmonisation and integration. Referring to the findings on vulnerability in integrated SCs (cf. section 2.1.4), the simulation study indicates that an overly strong temporal integration of material flow endangers SCP regarding the rather innovative key figures of adherence to schedules and flexibility. In general, the simulation indicates that *'deceleration'*, rather than a focus on decreasing lead times, reduces the vulnerability of the SC and thus supports the integration of performance-oriented CT into SC concepts.

As a result, it can be stated that adaptations to the SC concept regarding the CT concept positively influence SCP, but that a situation- and company-specific choice and weighting of key figures is the prerequisite for success.

Finally, the analysis indicates that integrated adaptations to different SC sub-concepts are promising with regards to improving the performance level of the SC concept including unimodal road transport. Thus, performance-oriented CT integration requires cross-functional cooperation. However, this again indicates that for applicability in practice, also the managerial and behavioural aspects of the integration problem (cf. section 2.1.5) are both highly significant for CT integration.

A discussion of the results with regards to the previous findings of the thesis in hand, in particular with reference to the conceptual research framework and the propositions, follows in the concluding chapter.

6 Implications for Science and Practice on the Performanceoriented Integration of CT into SC Concepts

This thesis addresses the question, of how CT can be integrated into SC concepts with regard to the target system of SCP (main research question Q_0). The thesis was able to develop an approach for the structured description of SCP requirements regarding the embedded and linking transport concepts (research question Q_1). Additionally, a conceptual framework for the description of cause-and-effect relationships between the elements of SC and CT concepts was developed (research question Q_2). The impact of integrative measures and instruments on the fit between SC and CT concepts was discussed with regard to the target system of SCP and practical recommendations on the application of the measures were derived (research question Q_3). This section addresses the central managerial (section 6.1) and practical implications (section 6.2). Furthermore, the most important limitations and future research perspectives (section 6.3) are highlighted.

6.1 Management Implications

The thesis in hand gives practical recommendations on the performance-oriented integration of CT into SC concepts. It highlights the performance-oriented applicability of CT in general for all SC actors. First, the thesis outlines the need to find alternatives to unimodal road transport and to reduce the related vulnerability of SCs. Second, the thesis discusses the main challenges of integrating material and information flow processes in SCs including CT. Third, it introduces solutions for the configuration of SCs including CT with regard to SCP. Fourth, by means of a simulation study the thesis shows the applicability of findings for practice and quantifies the SCP effect of SC concept adaptations for the analyzed SC concept.

Meaning of SCP as an Innovative Target System for all SC Actors

The thesis discusses SCP as a target system for all SC actors that can deal with the internal and external challenges of SC concepts. Therefore, a target system and corresponding performance indicators are developed, which allow for the evaluation of strategic and operative aspects of the integration problem. The developed target system comprises emissions, flexibility and reliability aspects next to 'classical' key figures such as lead time, capacity utilisation and inventory level. For managers, the thesis reveals that the changing framework conditions of SCs will prospectively require an alternative weighting of these target dimensions. Thus, the thesis presents an innovative evaluation system for integration activities in the field of transport. Furthermore, it shows that CT can be integrated to increase the reliability and sustainability of a SC and to accept an increase in lead times, inventories and costs instead.

The simulation study indicated that the cause-effect-relationships between the elements of CT and SC concepts are often non-linear. For managers two main findings are central: (1) Different cause-and-effect relationships have to be considered for a SC concept including a CT concept and a SC concept including a unimodal road transport concept. (2) In a SC concept including CT, the reduction of time quotas between the events in the SC affect the adherence to schedules and flexibility of the SC concept in a more than proportionate way. Thus, the aspect of 'deceleration' might be central for the discussion regarding the weighting of target dimensions.

Changing Shippers Perspectives - Transport as a Guiding Factor for SC Concept Configuration

The thesis in hand highlights the meaning of transport concepts as linking and embedded elements in SC concepts. The impact of transport concepts for the close integration of material and information flow processes and thus, for SCP is accentuated. This research suggests a perspective change regarding the meaning of transport concepts in SCs in general. The thesis is based on the idea that the transport concept is not a 'servant' to the SC concept, but a central element. Therefore, the thesis presents an approach to an innovative perspective on the integration problem. It suggests treating CT with its several inherent and external complexity factors as the guiding factors for SC concept configuration. This means that the procurement, production and distribution concepts have to be adapted to the CT concept to improve SCP.

A simulation study on a lean SC shows that the adaptation of the SC concept configuration with regard to a given and non-adaptable CT concept allows increasing SCP not only with regard to reliability and sustainability, but also – depending on the situationspecific framework conditions – reaching the performance level of unimodal road transport, too.

Different Implications for Different SC Actors

The thesis in hand takes a rational and non-actor-specific research perspective. Thus, different implications for different SC actors can be derived.

The thesis highlights the performance capabilities of CT services for *shippers*. Many shippers still do not consider CT as an alternative to unimodal road transport. The

290

promising results of the thesis should encourage shippers to consider CT as a competitive alternative for meeting the requirements of a new performance orientation in corporate and SC target systems. Additionally, shippers are advised to change their understandings of transport concepts as the 'servants' of the procurement, production and distribution concepts. Therefore, the thesis presents a set of practical recommendations on the choice of a suitable CT concept. In a first step, it recommends orientating to the underlying SC and the specific integration point in the SC. In particular, lean and leagile SC concepts are suitable for CT integration because of the underlying high level of demand predictability or comparably long lead times. The higher, more frequent and more stable transport quantities are the more line transport with regular booking can be applied. However, for rather fluctuating transport quantities with long lead times, line transport concepts with flexible bookings are suitable. Packaged goods with comparably long lead times can be handled by CT networks. Despite these rather general fits between the SCP requirements of SC concepts and performance characteristics of CT concepts there can be significant differences in certain SCP requirements. In a second step, based on an empirically developed classification of SCP requirements, the selection of practical recommendations on adaptation points in the procurement, production and distribution concepts are proposed to improve the fit and allow performanceoriented integration. In a third step, the thesis introduces a selection of concrete integrative measures and instruments to improve the integration of material and information flow processes to increase SCP and to reach competitiveness with unimodal road transport, even regarding classical key figures such as lead time and costs.

The thesis encourages the communication of the benefits for all SC actors by showing the potential for improved SCP from CT integration. *CT actors*, such as rail carriers, terminal operators or CT operators, are encouraged to communicate that CT is a competitive alternative to unimodal road transport. CT actors must use ongoing changes in the SC environment, such as increasing energy prices, SC vulnerability and sustainability orientation, to explain that CT may prospectively lead to a higher level of SCP. This means that CT actors have to change their own understandings in terms of integrated SC actors, too. Thus, the thesis in hand invites CT actors to free the transport concept from its role as 'servant' to SC concepts. CT actors must focus their activities on the main SCM target, namely the end customer. This requires that CT actors improve their understandings of shippers' operative challenges. The results of the thesis in hand allow CT actors to show shippers not only performance potential, but also the concrete adaptation points and instruments to reach it. In other words, it invites CT actors to advise shippers regarding the choice of suitable CT concepts, the necessary adaptations to the SC concept and the usage of integrative measures and instruments to allow performance-oriented CT integration.

For *transport policy*, the thesis highlights the impact of political regulations and support for transport concepts in general. In particular, the thesis accentuates the political influence on the competitive situation with unimodal road transport. Thus, the thesis in hand encourages transport policy to create a stable and long-term planning basis for all companies. The suggested perspective change emphasises the need for a long planning horizon since the thesis shows the strategic dimension of transport mode choice as well as material and information flow integration. Since SCs including CT are often international, the thesis also encourages a European and international aligned transport policy to provide suitable infrastructure capacities of road, train paths and terminals.

Structured Description of SCP Requirements on Transport Concepts

The thesis has shown that SC and CT actors have a limited mutual understanding of operative challenges, problems and target systems. Therefore, it presents a structured approach to describe shippers' SCP requirements regarding transport concepts with seven performance dimensions. The approach is not restricted to the field of CT and can be applied to any type of transport concept. Both SCP requirements of a SC concept and SCP capabilities of a transport concept can be described. It allows the comparison of requirements of a certain SC concept with the capabilities of a specific CT concept and supports the identification of specific adaptation points and integrative measures. Up to now, no structuring approach for transport concept integration has been available. Thus, the structuring approach serves to increase the mutual knowledge of the requirements on the transport concept to reach a close integration of material and information flow processes.

The thesis in hand shows the need for action by all SC actors by showing recent and prospective challenges of SCs and approaches.

6.2 Scientific Implications

The thesis in hand presents the deductive and theory-guided development of a conceptual research framework on the performance-oriented integration of CT into SC concepts.

Enhancement of SCM Conception - Transport Actors as Integrated SC Actors

In this context, the research proves that the conception of SCM is still in need of improvement. CT actors are not perceived as integrated SC actors. Hence, this research closes the identified research gap at the interface between transport and SCM. The thesis generally highlights the meaning of transport concepts for SCP and provides an approach for the structured description of SCP requirements.

Development of a Conceptual Research Framework

Based on this understanding, the research in hand develops a complete conceptual research framework on the performance-oriented integration of CT into SC concepts. With regard to the novelty and complexity of the given research problem, the thesis takes a rational and non-actor-specific research perspective. Configuration theory reflects the complexity and chosen research perspective. It serves as the basis for the conceptual research framework. There are already publications on configuration theory in the context of SCM, but the thesis in hand transfers configuration theory to the new field of CT. A complete conceptual research framework describes all relevant aspects and the main cause-and-effect relationships for the research problem (cf. section 4.2.3). The three-fold research framework breaks down the research problem into partial problems and transfers rather abstract constructs to measureable and observable variables.

The developed research framework states that the alignment of SC concepts with regard to the CT concept improves SCP. Therefore, the thesis develops material and information flow-oriented typologies for SC and CT concepts. Furthermore, a description structure for SCP requirements and performance capabilities as well as a set of suitable performance indicators is presented. These performance indicators regard the comparably new performance orientation and consider reliability and sustainability aspects. Thus, from a scientific point of view, the thesis contributes to transferring the comparably new research field of SCP management to the field of transport, especially CT.

The framework demonstrates the way to deal with the research problem and with the SC concept as the configuration variable as well as the CT as the framework variable. The developed framework serves as a visualisation of the innovative perspective on the research problem of the thesis in hand. It is enhanced by integrative measures and instruments as the 'moderating variable'. The basis was a comprehensive and structured literature review for integrative measures in the field of SCM. The literature provides an innovative overview of publications in the field of integration research with a focus

on technical and organisational aspects.

The conceptual research framework is implicitly based on rather general propositions on the effect of CT integration into SCs. These propositions serve as initial points for further research. Applying configuration theory, the propositions are developed to an encompassing and complete set of cause-and-effect relationships for the integration of CT into SC concepts.

Next to these managerial and scientific implications, different limitations of the thesis in hand must be considered.

6.3 Limitations and Further Research

The discussion of the contributions of the thesis in hand must be accompanied by its limitations. These limitations relate to the research perspective, simplifications of the conceptual research framework and methodological limitations regarding the need for a qualitative and quantitative verification of findings in general and the limitations regarding the applied simulation methodology in particular.

Limitations to Technical and Organisational Aspects of the Integration Problem

Sustainability and reliability are recent key figures in the target system of SCP to which companies align their activities. Rising transport costs, caused by increasing energy prices and political regulations, may prospectively further shift the competitive situation between unimodal road transport and CT in favour of CT.

Owing to the novelty of the research field, the thesis in hand aims to highlight the potential of CT integration. This potential orientation is reflected by the neutral and nonactor-specific research perspective. This perspective corresponds to the central idea of SCM conception. Thus, it focuses on the technical and organisational aspects of the integration problem, rather than on the behaviour- and management-specific aspects. This restriction reduces the problem complexity, but strongly simplifies the problem. This means that the thesis in hand does not discuss actor-specific considerations supporting or against a close integration of material and information flow processes. Furthermore, the thesis does not include recommendations on the management of such an integrated SC.

Thus, for the application in practice, management methods, the power and relationship structure, the risk structure as well as cultural aspects have to be considered next to the rather technical considerations presented in the thesis in hand. According to Cooper (1997), these aspects have the same relevance for reaching SCP as the technical as-

pects do.¹ This reduction of the integration problem to material and information flows and thus, technical and organisational aspects is the main limitation of the thesis in hand. For further research, the managerial and behavioural aspects must be analysed in depth and quantified.

Scope of SC Integration

The thesis in hand assumes that the integration of all CT and SC actors is beneficial. However, the comprehensive integration of all business partners is not always reasonable from an economic point of view, because some relationships are only of shortterm character. Furthermore, transaction costs must be taken into consideration for the development of all cooperative relationships. The variety of publications on dyadic cooperation proves that the management of one-to-one relationships is already a demanding task. Thus, for future research, the scope of SC integration must be challenged. Therefore, an actor-specific perspective is necessary. In particular, the application of transaction cost theory is advisable for this task. Taking an actor-specific research perspective, a number of further theoretical approaches can be analysed regarding their contributions to the problem of CT integration.

Simplifications for the Development of the Conceptual Research Framework

The development of the conceptual research framework required several simplifications. Based on configuration theory, the typologies of SC and CT concepts are analysed regarding their mutual 'fit'. It is assumed that a SC concept can be described by the configuration of the embedded procurement, production and distribution concepts. To reduce the complexity of the analysed cause-and-effect relationships process manufacturing² was excluded from consideration. Thus, the transferability of the findings must be analysed beginning with a comprehensive description of the characteristics of process production. Following the restriction to the technical and organisational aspects of the research problem, it was assumed that this fit could be described by the description approach with seven performance dimensions. The consideration of the behavioural and management aspects of the integration problem would require the enhancement of the developed description approach.

The thesis points out a number of framework conditions that influence and often exac-

¹ cf. Cooper et al. (1997).

² Production of goods in bulk quantities, rather than discrete units.

erbate the integration of CT into SC concepts. The framework summarises these framework conditions as 'SC strategy' and 'SC environment'. Whereas the thesis in hand deeply considers the meaning of SC strategy, further framework conditions are only touched. The meaning of changing shippers' requirements and transport policy for CT have been discussed in general, but a comprehensive determination of the influence of different conditions is missing. Further framework conditions, such as general aspects related to country and specific aspects such as the characteristics of other SC actors, competitors or customers, are excluded. An exploratory analysis, for instance using case studies, is recommended to identify and quantify the impact of different environmental factors. Thus, the conceptual research framework leaves room for further research, which should aim at the identification of the additional relevant elements for performance-oriented CT integration and a validation of the propositions and cause-and-effect relationships.

Methodological Limitations

For the thesis in hand, a combined inductive-deductive research approach was chosen. The predominant conceptual part of the thesis is enhanced by a simulation study that delivers information on inductive interpretation. In the field of SCM, experiments are often not possible because material and information flow processes cannot be interrupted. Furthermore, the novelty and complexity of the given research problem prevent the accomplishment of a quantitative empirical analysis of the research problem. Thus, a simulation approach was chosen to validate and enhance the selection of the deduced findings. However, a simulation cannot replace the empirical testing of deductively gained hypotheses. Thus, for further research, the quantitative review and empirical testing of the conceptual research framework, propositions and cause-and-effect relationships are necessary. The empirical analysis should build on the developed framework and should aim at the explorative generation of empirical knowledge. In particular, the suggested integrative measures and instruments as well as the framework conditions and the chosen performance indicators must be challenged and enhanced.

Limitations of the Simulation Study

The conducted simulation study is based on empirical data provided by *Manor*, a big Swiss retailer. The object-oriented and discrete event simulation encompassed two DCs in central Switzerland and eleven stores. The target of the simulation was the evaluation of SCP using a defined set of deductively developed performance indicators. The effect of CT integration in general and the effect of specific SC concept adaptations in the production and distribution concept were analysed. These adaptations refer to the propositions and cause-and-effect relationships presented in the conceptual research framework.

Scope of the Simulation Study

The setup of the simulation study does not cover all aspects of the conceptual research framework, namely the simulation study is limited to lean SCs (type I - continuous replenishment) and the CT concept of line transport with regular bookings. The influence of the integrative measures and instruments is not included at all. Furthermore, only adaptations of the production and distribution concept were be analysed in the simulation in order to reduce the complexity of the simulation model. Thus, the results derived from the simulation study can be generalized only to a certain extent.

The simulation study presented in the thesis in hand is based on the terminal network of *Swiss Post*. This means, for a generalisation of results the influence of central framework conditions must be discussed. Despite the growing importance of sustainability and reliability, reality proves that costs are still a central factor for transport mode choice. *Swiss Post* offers CT services with a comparable cost level to unimodal road transport. The reason is the *Swiss Post*'s effort to increase the capacity utilisation of existing terminals, overheads and trains. Thus, in this example the competitive situation with unimodal road transport is shifted in favour of CT.

Assumptions and Simplifications

The simulation is based on several assumptions and simplifications. These aspects must be taken into consideration when generalising the results. These simplifications encompass for instance the speed of road transports, duration of handling times, emission factors as well as demand quantities. Furthermore, a number of assumptions were necessary for the calculation of performance indicators (cf. section 5.7.6 for a comprehensive review of the impact of simplifications and assumptions on simulation results). However, a number of validation and verification steps ensures a high level of validity of the simulation's findings. Limitations of the simulation study are shortly discussed regarding their impact on the transferability and generalisability of simulation results.

The applied target system for the simulation study does not include the consideration of transport costs. The calculation of transport costs depends on a number of companyspecific and competition-relevant information. Furthermore, the calculation of costs must be understood as a snapshot of the recent framework conditions since volatile energy prices and tolls are included in the calculation. A profitability analysis is central for the decision making process on CT integration in practice.

To reduce the complexity, the model does not encompass the material flow processes in the stores. It is assumed that the goods leave the system as soon as the transport reaches the store and there is a open delivery window. Furthermore, no production sites are included in the model, but the goods' generation is modelled to take place in the DC. Although this simplification allows analyzing different adaptations of production concepts, for an evaluation of the inventory in the entire SC concept, inventories in production, on the the transport between production and in the DC must be included in the calculation of SC inventory. The order lead time, too, is abbreviated in comparison with the real system. However, for a comparison of the unimodal road transport scenario with different CT scenarios these simplifications are acceptable, since they do not influence the relations of the performance indicators.

Transferarbiltiy and Generalisability of Simulation Results

The findings of the simulation study are transferable to lean SCs in other industries and other product types. Relations between a certain source and a certain destination with predictable and stable transport quantities and frequencies on a daily basis are suitable for transferring the findings. However, for transferring results to other SC concept, two further aspects must be considered. (1) The situation-specific *spatial configuration* of the SC concept is critical for CT integration. The availability of terminals and suitable CT services in the required start and destination region(s) must be given. (2) The specific product value affects the capital commitment due to inventories. In retail, the average value is comparably low (e.g., in comparison with the automotive and electronics industries). To transfer findings to other products and industries, the *relative value of goods* is central. Furthermore, the findings are transferable *to other CT concepts*. The suggested adaptations are also valid, but the resulting extra effort required for the adaptations must be weighed against the resulting advantages.

In total, the simulation study supports the findings of the conceptual research framework. Furthermore, it reveals several non-linear cause-and-effect relationships between the elements of CT and SC concepts. Addionally, the simulation study indicates that the scope of SC concept adaptations to the CT concept is critical. It could be shown, that temporal buffers can be exploited for CT integration, but have also an important role for the stability of SC concepts. Thus, it can be summarized that for the reduction of temporal quotas and the harmoniazation of SC and CT concepts the situationspecific framework conditions and the resulting situation-specific cause-and-and-effect relationships must be considered.

Outlook and Limitations of Configuration Theory

Configuration theory was applied as a framework for the problem solution process.

CT integration influences other company and SC fields. These impacts must be considered for the integration decision. Usually, a company has more than one SC. The adaptation of one transport concept to CT can influence other SCs. The impact on other corporate departments, such as marketing, should be taken into consideration. The integration of CT and the suggested adaptation of SC concepts may significantly influence the existing relationships with suppliers and service providers. Trust and experience are central to these relationships. These aspects may compensate for the positive performance effect of CT integration.

Configuration theory implies working with typologies. This causes the danger that the defined variable settings are rather ideal than real. Furthermore, the usage of typologies risks that the research does have a comparably low level of detail. This means for prospective research that the typology of the CT concept must be reviewed and enhanced in more detail. However, one central criticism of configuration theory is that a falsification of typologies is not possible.³ For instance, a classification according to regional character such as domestic, hinterland and international transport would be sensible. One disadvantage of configuration theory is that it does not make recommendations on the type or number of suitable variables for the definition of typologies.⁴

Configuration theory suggests using longitudinal analyses to identify the structure and direction of cause-and-effect relationships. This means that the evaluation of CT integration using a simulation study was suitable to analyse the effect on SCP for a longer period. Thus, prospective research might focus on exploratory and longitudinal case studies. The danger of this approach is the allocation of cause and effects since recent events are usually the results of previous activities.⁵

³ cf. Wolf (2008), p. 481.

⁴ cf. Ibid., p. 482.

⁵ cf. Ibid., p. 480.

References

1815.ch (2011): Weichen gestellt. URL: [http://www.1815.ch/wallis/aktuell/weichen-gestellt-11659.html] last access: July 22, 2011 (published November 22, 2010).

Abacoumkin, C. / Ballis, A. (2004): Development of an expert system for the evaluation of conventional and innovative technologies in the intermodal transport area. In: European Journal of Operational Research. 152(2004)2, pp. 410-419.

Abshire, R. D. / Premeaux, S. R. (1991): Motor Carrier Selection Criteria: Perceptual Differences Between Shippers and Carriers. In: Transportation Journal. 31(1991)1, pp. 31-35.

Accenture / INSEAD (2003): A Global Study of Supply Chain Leadership and its Impact on Business Performance. Whitepaper 2003.

Afshin Mansouri, S. / Gallear, D. / Askariazad, M. H. (2010): Decision support for build-to-order supply chain management through multiobjective optimization. International Journal of Production Economics. In Press, Corrected Proof 2010.

Agarwal, A. / Shankar, R. / Tiwari, M. K. (2006): Modeling the metrics of lean, agile and leagile supply chain: an ANP-based approach. In: European Journal of Operational Research. 173(2006)1, pp. 211-225.

Agersekov, H. / Lungren, H. / Sorensen, T. (1983): Planning and design of ports and marine terminals. In: Transportation Research Part A: General. 18(1983)5-6, pp. 477-477.

Aggarwal, R. M. (2007): Role of Risk Sharing and Transaction Costs in Contract Choice: Theory and Evidence from groundwater Contracts. In: Journal of Economic Behavior & Organization. 63(2007)3, pp. 475-496.

Ahn, H. (2003): Effektivitäts- und Effizienzsicherung. Controlling-Konzept und Balanced Scorecard. Frankfurt a. M. 2003.

Aigbedo, H. (2007): An assessment of the effect of mass customization on suppliers' inventory levels in a JIT supply chain. In: European Journal of Operational Research. 181(2007)2, pp. 704-715.

Akkermans, H. A. / Bogerd, P. / Yücesan, E. / van Wassenhove, L. N. (2003): The impact of ERP on supply chain management: Exploratory findings from a European Delphi study. In: European Journal of Operational Research. 146(2003)2, pp. 284-301.

Akkermans, H. A. / van der Horst, H. (2002): Managing IT infrastructure standardisation in the networked manufacturing firm. In: International Journal of Production Economics. 75(2002)1-2, pp. 213-228.

Albers, S. / Klapper, D. / Konradt, U. / Walter, A. / Wolf, J. (2007): Methodik der empirischen Forschung. Wiesbaden 2007.

Alicke, K. (2003): Planung und Betrieb von Logistiknetzwerken – Unternehmensübergreifendes Supply Chain Management. Berlin 2003.

Almehdawe, E. / Mantin, B. (2010): Vendor managed inventory with a capacitated manufacturer and multiple retailers: Retailer versus manufacturer leadership. In: International Journal of Production Economics. 128(2010)1, pp. 292-302.

Amador, C. / Emond, J. P. (2010): Evaluation of sensor readability and thermal relevance for RFID temperature tracking. In: Computers and Electronics in Agriculture. 73(2010)1, pp. 84-90.

Amaral, J. / Kuettner, D. (2008): Analyzing Supply Chains at HP Using Spreadsheet Models. In: Interfaces. 38(2008)4, pp. 228-240.

Anand, G. / Ward, P. T. / Tatikonda, M. V. / Schilling, D. A. (2009): Dynamic capabilities through continuous improvement infrastructure. In: Journal of Operations Management. 27(2009)6, pp. 444-461.

Anckar, B. / Walden, P. / Jelassi, T. (2002): Creating customer value in online grocery shopping. In: International Journal of Retail & Distribution Management, 30(2002)4, pp. 211-220.

Anderson, J. / Hong, L. (2008): Sensor Resource Management driven by threat projection and priorities. In: Information Sciences. 178(2008)8, pp. 2007-2021.

Andrews, D. C. / Stalick, S. K. (1994): Business Reengineering: The Survival Guide. Prentice Hall 1994.

Arnold, D. / Kuhn, A. / Furmanns, K. / Isermann, H. / Tempelmeier, H. (2008): Handbuch Logistik. 3rd updated Edition 2008.

Arnold, E. (2010): Wissenschaftstheorie als angewandte

Wissenschaft:Anwendungsszenarien, Forschungsdesigns und Qualitätssicherung von Computersimulationen. Whitepaper. Stuttgart. URL:

[http://www.eckhartarnold.de/papers/Projektskizze_ausfuehrlich.pdf] last access: July 22, 2011.

Arnold, P. / Peeters, D. / Thomas, I. (2004): Modelling a rail/road intermodal transportation system. In: Transportation Research Part E: Logistics and Transportation Review. 40(2004)3, pp. 255-270.

Arnold, U. (1997): Beschaffungsmanagement. 2nd Edition. Stuttgart 1997.

Arnold, U. (2002): Global Sourcing: Strategiedimensionen und Strukturanalyse. Handbuch Industrielles Beschaffungsmanagement. Wiesbaden 2002.

Arshinder / Kanda, A. / Deshmukh, S. G. (2008): Supply chain coordination: Perspectives, empirical studies and research directions. In: International Journal of Production Economics. 115(2008)2, pp. 316-335.

ASTRA (2010): Geschwindigkeiten in Steigungen und Gefällen. Bern 2010.

Aviv, Y. (2001): The effect of collaborative forecasting on supply chain performance. In: Management Science. 47(2001)10, pp. 1326-1343.

Axelrod, R. (2005): Advancing the Art of Simulation in the Social Sciences. In: Rennard, J.-P.: Handbook of Research on Nature Inspired Computing for Economy and Management. Hersey 2005, pp. 1-13.

Axelrod, R. (1997a): Advancing the art of simulation in the social sciences. In: Conte, R. et al.: Simulating Social Phenomena. Berlin 1997a, pp. 21-40.

Axelrod, R. (1997b): The dissemination of culture: a model with local convergence and global polarization. In: Journal of conflict resolution. 1(1997b)41, pp. 203-226.

Axelrod, R. (2000): On Six Advances in Cooperation Theory. In: Analyse & Kritik. 1(2000)22, pp. 130–151.

Bacher, A. (2004): Instrumente des Supply Chain Controlling – Theoretische Herleitung und Überprüfung der Anwendbarkeit in der Unternehmenspraxis. Wiesbaden 2004.

Bachmann, H. (2008): Post Merger-Integration von Logistikdienstleistern. Konzeptionelle und empirische Analyse branchenspezifischer Integrationskompetenzen. Diss. Universität St.Gallen 2008.

Baen, J. S. (2001): The Effects of Technology on Retail Sales, Commercial Property Values and Percentage Rents In: Journal Journal of Real Estate Portfolio Management 6(2001)2, pp. 185-201.

BAFU (2010): Luftschadstoff-Emissionen des Strassenverkehrs 1990–2035 - Aktualisierung 2010. Bundesamt für Umwelt (BAFU). . Bern 2010.

Bagchi, P. K. / Skjoett-Larsen, T. (2002): Integration of Information Technology and Organizations in a Supply Chain. In: The international Journal of Logistics Management. 14(2002)3, pp. 89-108.

Bagchi, S. / Buckley, S. / Ettl, M. / Lin, G. (1998): Experience using the IBM supply chain simulator. Proceedings of the Winter Simulation Conference 1998.

Bahnonline (2011): MGBahn vor Vertragsunterzeichnung: Kooperation mit Logistikpartner Planzer im Güterverkehr. URL:

[http://www.bahnonline.ch/wp/34739/mgbahn-vertragsunterzeichnung-logistikpartnerplanzer-gueterverkehr.htm] last access: July 22, 2011 (published June 17, 2011).

Bahrami, K. (2001): Horizontale Transportlogistik-Konzentrationen. Synergiepotential für die Herstellung kurzlebiger Konsumgüter. Diss. Universitätt Wien 2001.

Balakrishnan, A. / Geunes, J. (2004): Collaboration and coordination in supply chain management and E-commerce. In: Production and Operations Management. 13(2004)1, pp. 1-2.

Balci, O. (1998): Verification, Validation and Accreditation. Proceedings of Winter Simulation Conference 1998.

Ballis, A. / Golias, J. (2002): Comparative evaluation of existing and innovative railroad freight transport terminals. In: Transportation Research Part A: Policy and Practice. 36(2002)7, pp. 593-611.

Ballis, A. / Golias, J. (2004): Towards the improvement of a combined transport chain performance. In: European Journal of Operational Research. 152(2004)2, pp. 420-436.

Banerjee, A. / Burton, J. / Banerjee, S. (2003): A simulation study of lateral shipments in a single supplier, multiple buyers supply chain networks. In: International Journal of Production Economics. 81-82(2003)1, pp. 103-114.

Banerjee, A. / Burton, J. / Bistline, W. (2001): Controlled partial shipments in twoechelon supply chain networks: a simulation study. In: International Journal of Production Economics 1(2001)71, pp. 91-100.

Banerjee, A. / Kim, S.-L. / Burton, J. (2007): Supply chain coordination through effective multi-stage inventory linkages in a JIT environment. In: International Journal of Production Economics. 108(2007)1-2, pp. 271-280.

Barash, M. M. (1990): Poka-yoke: Improving product quality by preventing defects. In: Journal of Manufacturing Systems. 9(1990)2, pp. 178-179.

Bardi, E. J. / Bagchi, P. K. / Raghunathan, T. S. (1989): Motor Carrier Selection in a Deregulated Environment. In: Transportation Journal. 29(1989)1, pp. 4-11.

Barker, R. C. (1995): Financial Performance Measurement: Not a Total Solution. In: Management Decision Science. 33(1995)2, pp. 31-39.

Barua, A. / Whinston, A. B. / Yin, F. (2000): Value and productivity in the Internet economy. In: IEEE Computer. 5(2000)2, pp. 102-105.

Basu, R. (2008): Tools for improvement. In: Implementing Six Sigma and Lean. Oxford 2008, pp. 112-132.

Basu, R. / Wright, J. N. (2008): Resource and capacity management. In: Total Supply Chain Management. Oxford 2008, pp. 63-77.

Baumgarten, H. / Kasiske, F. (2004): Transportkostengerechte Gestaltung von Prozessen im Supply Chain Management. In: Internationales Verkehrswesen. 56(2004)1+2, pp. 11-16.

BAV (2011a): Offertverfahren alpenquerender Güterverkehr 2011. Bundesamt für Verkehr (BAV). Bern. URL:

[http://www.bav.admin.ch/verlagerung/index.html?lang=it&download=M3wBPgDB/8 ull6Du36WenojQ1NTTjaXZnqWfVqHahmfhnapmmc7Zi6rZnqCkkIN4fnl7bKbXrZ6l huDZz8mMps2gpKfo&.pdf] last access: July 22, 2011.

BAV (2011b): Verlagerungspolitik – von der Strasse auf die Schiene. Bundesamt für Verkehr (BAV). Bern. URL:

[http://www.bav.admin.ch/dokumentation/publikationen/00475/00476/00700/index.ht ml?lang=de] last access: July 15, 2011.

Beco (2005): Massnahmenplan zur Luftreinhaltung 2000 / 2015. Berner Wirtschaft. URL: [http://www.vol.be.ch/site/beco-publ-imm-massnahmenplan-standort.pdf] last access: July 22, 2011.

Bendul, J. (2011): Coping with Intermodality in global Supply Chains - Balancing the Roles of Companies and Pan-European Transport Policy In: Tanner, A.-C. et al.: Schweiz und Europa - Auswirkungen auf Wirtschaft, Recht und Gesellschaft. Bern 2011, pp.

Bendul, J. (2009): Kapazitätsmanagement in Güterverkehr. In: Stölzle, W.Faganini, H.P.: Güterverkehr kompakt. München 2009, pp. 166-179.

Bendul, J. / Bobel, T. (2009): Gewinner von morgen - Kombinierter Verkehr als Strategieoption für Logistikdienstleister. In: LOG.Punkt. 2(2009)1, pp. 32-34.

Benkenstein, M. / Steiner, S. / Spiegel, T. (2007): Die Wertkette in Dienstleistungsunternehmen. In: Wertschöpfungsprozesse bei Dienstleistungen. Wiesbaden 2007, pp. 51-70.

Bergström, M. / Stehn, L. (2005): Matching industrialised timber frame housing needs and enterprise resource planning: A change process. In: International Journal of Production Economics. 97(2005)2, pp. 172-184.

Berry, W. L. / Cooper, M. C. (1999): Manufacturing flexibility: methods for measuring the impact of product variety on performance in process industries. In: Journal of Operations. 17(1999)pp. 163-178.

Bessant, J. / Levy, P. / Sang, B. / Lamming, R. (1994): Managing successful total quality relationships in the supply chain. In: European Journal of Purchasing & Supply Management. 1(1994)1, pp. 7-17.

BfS (2005): Externe Kosten. Bundesamt für Statistik (BfS). URL: [http://www.bfs.admin.ch/bfs/portal/de/index/themen/11/02/blank/key/externe_kosten. html] last access: July 22, 2011.

Biethahn, J. / Hummeltenberg, W. / Schmitdt, B. / Stähly, P. / Witte, T. (1999): Simulation als betriebliche Entscheidungshilfe - State of the Art und neuere Entwicklungen. Heidelberg 1999.

Binder, B. (2003): Prozessorientiertes Performance Measurement – Einführung und Anwendung in der Telekommunikationsbranche. Wiesbaden 2003.

Biswas, S. / Narahari, Y. (2004): Object oriented modeling and decision support for supply chains. In: European Journal of Operational Research. 153(2004)3, pp. 704-726.

BMU (2011): Verkehr und Umwelt – Herausforderungen. Bundesministerium für Umwelt, Naturschutz und Reaktorsicherheit. Berlin. Berlin 2011.

BMVBS (2009): Europäische Verkehrspolitik. Bundesbehörde für Verkehr, Bauwesen, Städtebau und Raumordnung sowie das Wohnungswesen. URL: [http://www.bmvbs.de/DE/VerkehrUndMobilitaet/Verkehrspolitik/EuropaeischeVerke hrspolitik/europaeische-verkehrspolitik_node] last access, July 27, 2011.

Bobel, T. / Bendul, J. (2009): Rechtzeitig miteinander reden - Kontraktlogistik im Kombinierten Verkehr. In: DVZ. 69(2009)1, pp. 12.

Bobel, T. / Schönberger, R. (2011): Kontraktlogistik zwischen globalen und regionalen Supply Chains - Der Logistikdienstleister als geostrategischer Intermediär. In: Schönberger, R.Elbert, R.: Dimensionen der Logistik - Funktionen, Institutionen und Handlungsebenen. 2011, pp.

Bonomyoung, R. / Beresford, A. K. C. (2001): Multimodal transport: the case of Laotian garment exporters. In: International Journal of Physical Distribution & Logistics Management. 31(2001)9, pp. 663-685.

Bönte, W. (2008): Inter-Firm Trust in Buyer-Supplier Relations: are Knowledge Spillovers and geographical Proximity relevant? In: Journal of Economic Behavior & Organization. 67(2008)3-4, pp. 855-870.

Bortz, J. (2005): Statistik für Human- und Sozialwissenschaftler. Heidelberg 2005.

Bortz, J. / Döring, N. (2006): Forschungsmethoden und Evaluation für Human- und Sozialwissenschafter. 4th Edition. Heidelberg 2006.

Bose, I. / Pal, R. / Ye, A. (2008): ERP and SCM systems integration: The case of a valve manufacturer in China. In: Information & Management. 45(2008)4, pp. 233-241.

Böse, J. W. / Voss, S. (2000): Informationsmanagement im Kombinierten Verkehr In: Dadurna, J.R.Voss, S.: Informationsmanagement im Verkehr. Heidelberg 2000, pp. 269-322.

Bowersox, D. J. / Closs, D. J. (1996): Logistical Management: the integrated supply chain process. New York 1996.

Bowersox, D. J. / Closs, D. J. / Cooper, M. C. (2002): Supply Chain Logistics Management. New York 2002.

Bowersox, D. J. / Closs, D. J. / Stank, T. P. (1999): 21st Century Logistics: Making Supply Chain Integration a Reality. Oak Brook 1999.

Bowman, E. H. / Helfat, C. E. (2001): Does corporate strategy matter. In: Strategic Management Journal. 22(2001)1, pp. 1-23.

Bracht, U. / Geckler, D. / Wenzel, S. (2011): Digitale Fabrik: Methoden und Praxisbeispiele Heidelberg 2011.

Brade, D. (2003): A generalized process for the verification and validation of models and simulation results. München 2003.

Brandenburg, H. / Gutermuth, J. / Oelfke, D. (2008): Güterverkehr-Spedition-Logistik. Leistungserstellung in Spedition und Logistik. 37th Edition Berlin 2008.

Bretthauer, K. M. / Mahar, S. / Venakataramanan, M. A. (2005): Inventory and distribution strategies for retail/e-tail organizations. In: Computers & Industrial Engineering. 58(2005)1, pp. 119-132.

Bretzke, W. / Barkawi, K. (2010): Nachhaltige Logistik - Antworten auf eine globale Herausforderung. Berlin, Heidelberg 2010.

Brown, S. / Blackmon, K. (2005): Aligning manufacturing strategy and business-level competitive strategy in new competitive environments: the case for strategic resonance. In: Journal of Management Studies. 42 (2005)4, pp. 793-815.

Brun, A. / Caridi, M. / Fahmy Salama, K. / Ravelli, I. (2006): Value and risk assessment of supply chain management improvement projects. In: International Journal of Production Economics. 99(2006)1-2, pp. 186-201.

Brun, A. / Zorzini, M. (2009): Evaluation of product customization strategies through modularization and postponement. In: International Journal of Production Economics. 120(2009)1, pp. 205-220.

Brunner, M. (2008): Resource-Dependence-Ansatz. In: Schwaiger, M.: Theorien und Methoden der Betriebswirtschaft Handbuch für Wissenschaftler und Studierende. München 2008, pp. 31-38.

Bucklin, L. P. (1965): Postponement, speculation and the structure of distribution channels. In: Journal of Marketing Research. 2(1965)1, pp. 26-32.

Bühler, G. (2005): Verkehrsmittelwahl im Güterverkehr. Freiburg 2005.

Buttermann, G. / Germain, R. / Iyer, K. N. S. (2008): Contingency theory 'fit' as gestalt: An application to supply chain management. In: Transportation Research Part E: Logistics and Transportation Review. 44(2008)6, pp. 955-969.

Cachon, G. P. / Lariviere, M. A. (2001): Contracting to assure supply: how to share demand forecasts in a supply chain. In: Management Science. 47(2001)5, pp. 629-646.

Callioni, G. / Montgros, X. / Slagmulder, R. / Wassenhove, L. N. / van Wright, L. (2005): Inventory-Driven Costs. In: Harvard Business Review 83(2005)3, pp. 135-141.

Campbell, A. J. / Wilson, D. T. (1996): Managed networks: creating strategic advantage. In: Iacobucci, D.: Networks in Marketing, Sage Publications. Thousand Oaks 1996, pp. 1-27.

Campbell, A. M. / Savelsbergh, M. W. P. (2004): Delivery Volume Optimization. In: Transportation Science. 38(2004)2, pp. 210-223.

Canie, M. C. J. / Gelderman, C. J. (2005): Purchasing strategies in the Kraljic matrix—A power and dependence perspective. In: Journal of Purchasing & Supply Management. 11 (2005)2005, pp. 141–155.

Cao, M. / Zhang, Q. (2010): Supply Chain Collaboration: Impact on Collaborative Advantage and Firm Performance. In: Journal of Operations Management. 29(2010)3, pp. 163-180

Caplice, C. / Sheffi, Y. (2003): Optimization-based procurement for transportation services. In: Journal of Business Logistics. 24(2003)1, pp. 109-128.

Caplice, C. / Sheffi, Y. (1994): A Review and Evaluation of Logistics Performance Measurement Systems. In: The international Journal of Logistics Management. 6(1994)1, pp. 61-74.

Caputo, A. C. / Pelagagge, P. M. / Scacchia, F. (2003): Integrating transport systems in supply chain management software tools. In: Industrial Management & Data Systems. 103(2003)7, pp. 503-515.

Carbone, V. / Stone, M. A. (2005): Growth and relational strategies used by the European logistics service providers: Rationale and outcomes. In: Transportation Research Part E: Logistics and Transportation Review. 41(2005)6, pp. 495-510.

Cargobeamer (2011a): Engpässe des Kombinierten Verkehrs URL: [http://www.cargobeamer.com/index.php?article_id=54&clang=0] last access: July 15, 2011.

Cargobeamer (2011b): Zielmarkt für den Cargobeamer. URL: [http://www.cargobeamer.com/index.php?article_id=46&clang=0] last access: July 15, 2011.

Carlson, R. / Erixon, M. / Forsberg, P. / Pålsson, A. C. (2001): System for integrated business environmental information management. In: Advances in Environmental Research. 5(2001)4, pp. 369-375.

Carr, A. / Kaynak, H. (2007): Communication methods, information sharing, supplier development and performance. In: International Journal of Operations & Production Management. 27(2007)1, pp. 346-370.

Carter, C. R. / Rogers, D. S. (2008): A framework of sustainable supply chain management: moving toward new theory. In: International Journal of Physical Distribution & Logistics Management. 38(2008)5, pp. 360-387.

Carter, J. R. / Ferrin, B. G. (1995): The impact of transportation costs on supply chain management. In: Journal of Business Logistics. 16(1995)1, pp. 189-212.

Catalan, M. / Kotzab, H. (2003): Assessing the responsiveness in the Danish mobile phone supply chain. In: International Journal of Physical Distribution & Logistics Management. 33(2003)8, pp. 668-685.

Chan, F. T. S. / Humphreys, P. / Lu, T. H. (2001): Order release mechanisms in supply chain management: a simulation approach. In: International Journal of Physical Distribution and Logistics Management. 2(2001)31, pp. 124–139.

Chan, F. T. S. / Nelson, K. H. T. / Lau, H. C. W. (2002): A simulation approach in supply chain management. In: Integrated Manufacturing Systems 2(2002)13, pp. 117-122.

Chan, F. T. S. / Zhang, T. (2011): The impact of Collaborative Transportation Management on supply chain performance: A simulation approach. In: Expert Systems with Applications. 38(2011)3, pp. 2319-2329.

Chang, I. C. / Hwang, H.-G. / Liaw, H.-C. / Hung, M.-C. / Chen, S.-L. / Yen, D. C. (2008): A neural network evaluation model for ERP performance from SCM perspective to enhance enterprise competitive advantage. In: Expert Systems with Applications. 35(2008)4, pp. 1809-1816.

Chang, Y. / Makatsoris, C. (2001): Supply chain modeling using simulation. In: International Journal of Simulation: Systems, Science & Technology. 2(2001)1, pp. 24-30.

Changxian, L. / Houping, T. / Jianping, S. / Desheng Dash, W. (2009): Incentive Contract Design in competing Distribution Channels. In: Production Planning & Control. 20(2009)4, pp. 295-305.

Chao, C.-C. / Yang, J.-M. / Jen, W.-Y. (2007): Determining technology trends and forecasts of RFID by a historical review and bibliometric analysis from 1991 to 2005. In: Technovation. 27(2007)5, pp. 268-279.

Chatfield, D. C. / Harrison, T. P. / Hayya, J. C. (2009): SCML: An information framework to support supply chain modeling. In: European Journal of Operational Research. 196(2009)2, pp. 651-660.

Chen, C.-J. / Huang, C.-C. (2004): A multiple criteria evaluation of high-tech industries for the science-based industrial park in Taiwan. In: Information & Management. 41(2004)7, pp. 839-851.

Chen, I. J. / Paulraj, A. (2004): Towards a theory of supply chain management: the constructs and measurements. In: Journal of Operations Management. 22(2004)2, pp. 119-150.

Cheng, J.-H. (2010): Inter-organizational relationships and information sharing in supply chains. In: International Journal of Information Management. 314(2010)4, pp. 374-384

Childerhouse, P. / Towill, D. (2000): Engineering supply chains to match customer requirements. In: Logistics Information Management. 13(2000)6, pp. 337 - 346.

Choi, T. (1995): Conceptualizing continuous improvement: Implications for organizational change. In: Omega. 23(1995)6, pp. 607-624.

Choi, T. Y. / Wu, Z. (2009): Triads in supply networks: theorizing buyer-supplier-supplier relationships. In: Journal of Supply Chain Management. 45(2009)1, pp. 8-25.

Choy, K. L. / Lee, W. B. / Lo, V. (2002): Development of a case based intelligent customer-supplier relationship management system. In: Expert Systems with Applications. 23(2002)3, pp. 281-297.

Christopher, M. (2005a): Logistics and Supply Chain Management - Creating Value-Adding Networks. 3rd Edition. Harlow 2005a.

Christopher, M. (2005b): Logistics and Supply Chain Management – Strategies for Reducing Cost and Improving Service. London 2005b.

Christopher, M. / Towill, D. (2001): An integrated model for the design of agile supply chains. In: International Journal of Physical Distribution & Logistics Management. 31(2001)4, pp. 235-246.

Christopher, M. / Towill, D. R. (2002): Developing Market Specific Supply Chain Strategies. In: International Journal of Logistics Management. 13(2002)1, pp. 1-14.

Christopher, M. C. (2000): The agile supply chain: competing in volatile markets. In: Industrial Marketing Management. 29(2000)1, pp. 37-44.

Christopher, M. C. / Peck, H. / Towill, D. R. (2006): A taxonomy for selecting global supply chain strategies. In: International Journal of Logistics Management. 17(2006)2, pp. 277-287.

Christopher, M. C. / Tonwill, D. R. / Aitken, J. / Childerhouse, P. (2008): Value stream classification. In: Journal of Manufacturing Technology Management 20(2008)4, pp. 460-474.

Cigolini, R. / Cozzi, M. / Perona, M. (2004): A new framework for supply chain management: conceptual model and empirical test. In: International Journal of Operations & Production Management. 24(2004)1, pp. 7-41.

Cimino, M. G. C. A. / Marcelloni, F. (2011): Autonomic tracing of production processes with mobile and agent-based computing. In: Information Sciences. 181(2011)5, pp. 935-953.

Clausen, U. / Kochsiek, J. / Kuchenbecker, M. (2006): Bedeutung von Gleisanschlüssen und Einzelwagenverkehr, Logistikfähigkeit und SCM-Integration. In: Logistik Management. 8(2006)3, pp. 19-27.

Consultants, K. P. T. / GmbH, K. (2004): Study on Infrastructure Capacity Reserves for Combined Transport by 2015. 2004.

Cooper, M. C. / Lambert, D. M. / Pagh, J. D. (1997): Supply Chain Management: More than a new name for logistics. In: The International Journal of Logistics Management. 8(1997)1, pp. 172-187.

Cordes, M. (2011): Interview with Karsten Sachsenröder: 'Der KV wird schwarze Zahlen schreiben'. In: Verkehrsrundschau Spezial - Sonderheft für Güterverkehr Strasse-Schiene. 46(2011)1, pp. 8-10.

Corsten, H. (2004): Produktionswirtschaft: Einführung in das industrielle Produktionsmanagement. 10. updated edition. München et al. 2004.

Corsten, H. / Stuhlmann, S. (1992): Kapazitätsplanung in Dienstleistungsunternehmen. In: Corsten, H.: Kapazitätsmessung, Kapazitätsgestaltung, Kapazitätsoptimierung eine betriebswirtschaftliche Kernfrage. Stuttgart 1992, pp. 178-192.

Cousins, P. D. / Handfield, R. B. / Lawson, B. / Petersen, K. J. (2006): Creating supply chain relational capital: The impact of formal and informal socialization processes. In: Journal of Operations Management. 24(2006)6, pp. 851-863.

Craig, D. (2011): Disadvantages of Simulation.

http://web.cs.mun.ca/~donald/msc/node7.html last access: July 18, 2011.

Craig, R. C. / Dale, S. R. (2008): A framework of sustainable supply chain management: moving toward new theory. In: International Journal of Physical Distribution & Logistics Management. 38(2008)5, pp. 360-387.

Credite-Suisse (2007): Branchenanalyse, Gross- und Detailhandel. Aktuelle Lage und Trends für 2008. 2007.

Crum, M. R. / Allen, B. J. (1997): A Longitudinal Assessment of Motor Carrier-Shipper Relationship Trends. In: Transportation Journal. 37(1997)1, pp. 5-17.

CSCMP (2011): Supply Chain Management Definitions. URL: [http://cscmp.org/aboutcscmp/definitions.asp] last access:July 15, 2011.

Dahm, C. (2011): Ran an das Thema externe Kosten - Kommentar zum Weissbuch der Europäische Kommission. In: DVZ. 24(2011)65, pp. 2.

Danielis, R. / Marcucci, E. / Rotaris, L. (2005): Logistics managers' stated preferences for freight service attributes. In: Transportation Research Part E: Logistics and Transportation Review. 41(2005)3, pp. 201-215.

Darwish, M. A. / Odah, O. M. (2010): Vendor managed inventory model for single-vendor multi-retailer supply chains. In: European Journal of Operational Research. 204(2010)3, pp. 473-484.

Das, A. / Narasimhan, R. / Talluri, S. (2006): Supplier integration - finding an optimal configuration. In: Journal of Operations Management. 24(2006)5, pp. 563-582.

Das, K. (2010): Integrating effective flexibility measures into a strategic supply chain planning model. In: European Journal of Operational Research. 211(2010)1, pp. 170-183.

Davenport, T. H. (1993): Process Innovation, Reengineering Work through Information Technology. Boston 1993.

DBAG (2011): Flüsterbremse reduziert Lärmbelastung. URL:

[http://www.deutschebahn.com/site/bahn/de/nachhaltigkeit/umwelt/laermminderung/fl uesterbremse/fluesterbremse.html] last access: July 15, 2011 (published March 22, 2011)

DBAG (2009): Lange Güterzüge - Projekt GZ 1000. 2009.

DBAG (2007): Wettbewerbsbericht 2007. 2007.

De Jong, G. / Ben-Akiva, M. (2007): A micro-simulation model of shipment size and transport chain choice. In: Transportation Research Part B: Methodological. 41(2007)9, pp. 950-965.

de Miroschedji, S. A. (2002): Globale Unternehmens- und Wertschöpfungsnetzwerke. Wiesbaden 2002.

de Treville, S. / Shapiro, R. D. / Hameri, A. P. (2004): From supply chain to demand chain: the role of lead-time reduction in improving demand chain performance. In: Journal of Operations Management. 21(2004)6, pp. 613-623.

Dehning, B. / Richardson, V. J. / Zmud, R. W. (2007): The financial performance effects of IT-based supply chain management systems in manufacturing firms. In: Journal of Operations Management. 25(2007)4, pp. 806-824.

Delen, D. / Hardgrave, B. C. / Sharda, R. (2007): RFID for Better Supply-Chain Management through Enhanced Information Visibility. In: Production & Operations Management. 16(2007)5, pp. 613-624.

Delfmann, W. / Reihlen, M. / Wickinghoff, C. (2003): Prozessorientierte Logistik-Leistungsrechnung. In: Delfmann, W.Reihlen, M.: Controlling von Logistikprozessen – Analyse und Bewertung logistischer Kosten und Leistungen. Stuttgart 2003, pp. 17-56.

Dessouky, M. M. / Leachman, R. C. (1995): Simulation Modeling Methodology for Analyzing Large Complex Rail Networks. In: Simulation Modelling Practice and Theory. 65(1995)2, pp. 131-142.

Devaraj, S. / Krajewski, L. / Wei, J. C. (2007): Impact of eBusiness technologies on operational performance: The role of production information integration in the supply chain In: Journal of Operations Management. 25(2007)6, pp. 1199-1216.

Diabat, A. / Govindan, K. (2011): An analysis of the drivers affecting the implementation of green supply chain management. In: Resources, Conservation and Recycling. 55(2011)6, pp. 659-667.

Diamond, R. / Harrell, C. R. / Henriksen, J. O. / Nordgren, W. B. / Pegden, C. D. / Rohrer, M. W. / Waller, A. P. / Law, A. M. (2002): The current and the future status of simulation software. In: Proceedings of the Winter Simulation Conference Year

Dias, J. C. Q. / Calado, J. M. F. / Mendonça, M. C. (2010): The role of European «ro-ro» port terminals in the automotive supply chain management. In: Journal of Transport Geography. 18(2010)1, pp. 116-124.

Dickmann, P. (2009): Schlanker Materialfluss mit Lean Production, Kanban und Innovationen. Berlin 2009.

Disney, S. M. / Naim, M. M. / Towill, D. R. (1997): Dynamic simulation modelling for lean logistics. In: International Journal of Physical Distribution and Logistics Management. 3–4 (1997)27, pp. 174-196.

Disney, S. M. / Potter, A. T. / Gardner, B. M. (2003): The impact of vendor managed inventory on transport operations. In: Transportation Research Part E. 1(2003)39, pp. 363–380.

Disney, S. M. / Towill, D. R. (2002): A discrete transfer function model to determine the dynamic stability of a vendor managed inventory supply chain. In: International Journal of Production Research. 40(2002)1, pp. 179-204.

Dobie, K. (2005): The Core Shipper Concept: A Proactive Strategy for Motor Freight Carriers. In: Transportation Journal. 44(2005)2, pp. 37-53.

Dooley, K. (2002): Simulation Research Methods. In: Baum, J.: Companion to Organizations. London 2002, pp. 829-848.

Dörflinger, M. / Marxt, C. (2001): Mass Customization – neue Potenziale durch kundenindiviuelle Massenproduktion. In: IO Management 3(2001)1, pp. 228-233.
Dowell, G. / Hart, S. / Yeung, B. (2000): Do Corporate Global Environmental Standards create or destroy Market Value? In: Management Science. 46(2000)8, pp. 1059.

Dreier (2011): Unternehmensportrait Dreier. URL: [www.dreier.lu] last access: July 22, 2011.

Droge, C. / Jayaram, J. / Vickery, S. K. (2004): The effects of internal versus external integration practices on time-based performance and overall firm performance. In: Journal of Operations Management. 22(2004)1, pp. 557-573.

Duclos, L. K. / Vokurka, R. J. / Lummus, R. (2003): A conceptual model of supply chain flexibility. In: Industrial Management & Data Systems. 103(2003)5/6, pp. 446-475.

Duffy, R. / Fearne, F. (2006): The Impact of Supply Chain Partnerships on Supplier Performance. In: International Journal of Logistics Management. 15(2006)1, pp. 57-72.

Dupre, K. / Gruen, T. W. (2004): The use of category management practices to obtain a sustainable competitive advantage in the fast-moving' consumer-goods industry. In: Journal of Business & Industrial Marketing. 19(2004)7, pp. 444-459.

Eisenhardt, K. M. (1988): Agency- and Institutional-Theory Explanations: The Case of Retail Sales Compensation. In: Academy of Management Journal. 31(1988)3, pp. 488-544.

Eisenhardt, K. M. (1989): Agency Theory: An Assessment and Review. In: Academy of Management Review. 14(1989)1, pp. 57-74.

EK (2005): Europäische Kommission: Transeuropäisches Verkehrsnetz - TEN-V – vorrangige Achsen und Projekte. Brüssel 2005.

Ekwall, D. (2007): Antagonistic gateways in the transport network in a supply chain perspective. Göteborg 2007.

Ellram, L. M. / Cooper, M. C. (1990): Supply Chain Management, Partnership, and the Shipper - Third Party Relationship. In: The International Journal of Logistics Management. 1(1990)2, pp. 1-10.

Elmahdi, A. / Malano, H. / Khan, S. (2006): Using a System Dynamics Approach to Model Sustainability Indicators for Irrigation Systems in Australia. In: Natural Resource Modeling. 19(2006)4, pp. 465-481.

Engelke, M. (1997): Qualität logistischer Dienstleistungen: Operationalisierung von Qualitätsmerkmalen, Qualitätsmanagement, Umweltgerichtigkeit. Berlin 1997.

Erdmann, M.-K. (2003): Supply Chain Performance Measurement - Operative und Strategische Management- und Controllingansätze. Lohmar 2003.

Erera, A. L. / Morales, J. C. / Savelsbergh, M. (2005): Global intermodal tank container management for the chemical industry. In: Transportation Research Part E: Logistics and Transportation Review. 41(2005)6, pp. 551-566.

Ernst, E. (2011): Schnell auf der Schiene. In: Verkehrsrundschau Spezial - Sonderheft für Güterverkehr Strasse-Schiene. 1(2011)1, pp. 14-15.

ERSRail (2011): Timetables. URL: [http://www.ersrail.com/timetables.html] last access: July 20, 2011.

Escudero, L. F. / Galindo, E. / García, G. / Gómez, E. / Sabau, V. (1999): Schumann, a modeling framework for supply chain management under uncertainty. In: European Journal of Operational Research. 119(1999)1, pp. 14-34.

Esper, T. L. / Williams, L. R. (2003): The Value of Collaborative Transportation Management (CTM): Its Relationship to CPFR and Information Technology. In: Transportation Journal. 42(2003)4, pp. 55-65.

EU (2011): Intermodaler Verkehr: Das Programm MARCO POLO.

Zusammenfassungen der EU Gesetzgebung URL:

[http://europa.eu/legislation_summaries/transport/intermodality_transeuropean_networ ks/l24159_de.htm] last access: July 15, 2011.

Evers, P. T. / Harper, D. V. / Needham, P. M. (1996): The Determinants of Shipper Perceptions of Modes. In: Transportation Journal. 36(1996)2, pp. 13-25.

Evers, P. T. / Johnson, C. J. (2000): Performance Perceptions, Satisfaction, and Intention: The Intermodal Shipper's Perspective. In: Transportation Journal. 40(2000)2, pp. 27-39.

Fabbe-Costes, N. / Jahre, M. (2008): Supply chain integration and performance: a review of the evidence. In: International Journal of Logistics Management. 19(2008)2, pp. 130-154.

Fabbe-Costes, N. / Jahre, M. (2007): Supply chain integration improves performance: the Emperor's new suit? In: International Journal of Physical Distribution & Logistics Management. 37(2007)10, pp. 835-855.

Fabbe-Costes, N. / Jahre, M. / Roussat, C. (2009): Supply chain integration: the role of logistics service providers. In: International Journal of Productivity and Performance Management. 58(2009)1, pp. 71-91.

Fearne, A. / Fowler, N. (2006): Efficiency versus effectiveness in construction supply chains: the dangers of 'lean' thinking in isolation. In: Supply Chain Management: An International Journal. 11(2006)4, pp. 283-287.

Feitzinger, E. / Lee, H. L. (1997): Mass Customization at Hewlett-Packard: The Power of Postponement. In: Harvard Business Review. 75(1997)1, pp. 116-121.

Ferdows, K. / De Meyer, A. (1990): Lasting improvements in manufacturing performance: in search of a new theory. In: Journal of Operations Management. 9(1990)2, pp. 168-184.

Fernie, J. (1995): International comparisons of supply chain management in grocery retailing. In: Service Industries Journal. 15(1995)4, pp. 134-147.

Ferreira, L. / Sigut, J. (1993): Measuring the Performance of Intermodal Freight Terminals. In: Transportation Planning and

Technology in Society. 17(1993)3, pp. 269-275.

Fiala, P. (2005): Information sharing in supply chains. In: Omega. 33(2005)5, pp. 419-423.

Fiedler, J. (2005): Bahnwesen 2005.

FinanceContent (2011): BRIC und Next Eleven. URL: [http://www.finanzcontent.de/bric-und-next-eleven-19915.html] last access July 10, 2011.

Fisher, M. L. (1997): What is the right supply chain for your product? In: Harvard Business Review. 75(1997)2, pp. 105-116.

Fisher, M. L. / Hammond, J. H. / Obermeyer, W. R. / Raman, A. (1994): Making supply meet demand in an uncertain world. In: Harvard Business Review. 72(1994)pp. 83-93.

Fishwick, P. A. (1995): Simulation Model Design and Execution: Building Digital Worlds. Prentice-Hall 1995.

Fleischmann, B. (1993): Designing Distribution Systems with Transport Economies of Scale. In: European Journal of Operational Research. 70(1993)1, pp. 31-42.

Fleischmann, B. / Meyr, H. / Graves, S. C. / Kok, A. G. d. (2003): Planning Hierarchy, Modeling and Advanced Planning Systems. In: Handbooks in Operations Research and Management Science. 2003, pp. 455-523.

Flexsim (2011): Product Overview. URL: [

http://www.flexsim.de/index.php?option=com_content&view=category&layout=blog &id=42&Itemid=71&lang=de] last access: July 22, 2011.

Forrester, J. W. (1995): The Beginning of System Dynamics. In: McKinsey Quarterly. 1995)4, pp. 4-16.

Forslund, H. (2009): Logistics service performance contracts: design, contents and effects. In: International Journal of Physical Distribution & Logistics Management. 39(2009)2, pp. 131.

Foster Jr, S. T. (2008): Towards an understanding of supply chain quality management. In: Journal of Operations Management. 26(2008)4, pp. 461-467.

Fraport (2011): Neue Landebahn Nordwest. URL:

[http://www.ausbau.fraport.de/cms/default/rubrik/5/5856.htm] last access: July 15, 2011.

Frazier, G. L. (1999): Organizing and Managing Channels of Distribution. In: Journal of the Academy of Marketing Science. 27(1999)2, pp. 226-240.

Fricker, D. (2008): Aktuelle Probleme des Supplier Relationship Management, Befunde und Massnahmen. Köln 2008.

Frindik, R. (2004): Innovationen beim Horizontalumschlag im KV. In: Internationales Verkehrswesen. 56(2004)3, pp. 93-96.

Frohlich, M. T. / Westbrook, R. (2001): Arcs of integration: an international study of supply chain strategies. In: Journal of Operations Management. 19(2001)2, pp. 185-200.

Frohlich, M. T. / Westbrook, R. (2002): Demand chain management in manufacturing and services: web-based integration, drivers and performance. In: Journal of Operations Management. 20(2002)6, pp. 729-745.

Fugate, B. / Sahin, F. / Mentzer, J. T. (2006): SCM coordination mechanisms In: Journal of Business Logistics. 27(2006)2, pp. 129-161.

Gabler (2011): Incoterms. URL:

[http://wirtschaftslexikon.gabler.de/Definition/incoterms.html] last access:July 15th, 2011.

Galgenmüller, F. / Kieninger, M. / Jochen, M. (2004): Einflüsse von Kooperationsentscheidungen auf die Werttreiber eines Unternehmens. In: Gleich, R.: Network Value Added - Planung und Steuerung von Netzwerken in der Automobilindustrie. St. Gallen 2004, pp. 79-96.

Gambardella, L. M. / Rizzoli, A. / M., Z. (1998): Simulation and planning of an intermodal container terminal. In: Simulation Modelling Practice and Theory. 71(1998)2, pp. 107-116.

Gareis, K. (2002): Das Konzept Industriepark aus dynamischer Sicht. Theoretische Fundierung - empirische Ergebnisse - Gestaltungsempfehlungen. Wiesbaden 2002.

Gehrken, M. (2011): Vorschlag für ein neues Trassenpreissystem. URL: [http://www.astag.ch/?rub=1572&id=7533] last access: July 15, 2011.

Geir, G. / Marianne, J. / Gøran, P. (2006): Supply chain management - back to the future? In: International Journal of Physical Distribution & Logistics Management. 36(2006)8, pp. 643.

Georgiadis, P. / Vlachos, D. / Iakovou, E. (2005): A system dynamics modeling framework for the strategic supply chain management of food chains. In: Journal of Food Engineering. 70(2005)3, pp. 351-364.

Gerbens-Leenes, P. W. / Moll, H. C. / Schoot Uiterkamp, A. J. M. (2003): Design and development of a measuring method for environmental sustainability in food production systems. In: Ecological Economics. 46(2003)2, pp. 231.

Gibson, B. J. / Mentzer, J. T. / Cook, R. L. (2005): Supply Chain Management: The puruit of consesus definition. In: Journal of Business Logistics. 26(2005)2, pp. 17-25.

Gibson, B. J. / Rutner, S. M. / Keller, S. B. (2002): Shipper-carrier partnership issues, rankings and satisfaction. In: International Journal of Physical Distribution & Logistics Management. 32(2002)8, pp. 669-681.

Gibson, B. J. / Sink, H. L. (1993): Shipper-carrier relationships and carrier selection criteria. In: Logistics & Transportation Review. 29(1993)4, pp. 371-374.

Gilbert, N. / Troitzsch, K. G. (1999): Simulation for the Social Scientist. London 1999.

Gimenez, C. / Lourenco, H. R. (2008): E-SCM: Internet's Impact on Supply Chain Processes. In: International Journal of Logistics Management. 19(2008)3, pp. 309-343.

Giunipero, L. / Hooker, R. / Joseph-Matthews, S. / Yoon, T. / Brudvig, S. (2008): A Decade of SCM Literature: past, present and future Implications. In: Journal of Supply Chain Management. 44(2008)4, pp. 66.

Gladen, W. (2003): Kennzahlen- und Berichtssysteme - Grundlagen zum Performance Measurement. Wiesbaden 2003.

Gleich, R. (2001): Das System des Performance Measurement - Theoretisches Grundkonzept, Entwicklungs- und Anwendungsstand. München 2001.

Goldsby, T. J. / Griffis, S. E. / Roath, A. S. (2006): Modeling lean, agile, and leagile supply chain strategies. In: Journal of Business Logistics. 27(2006)1, pp. 57-80.

Goldsby, T. J. / Stank, T. P. (2000): World Class Logistics Performance and environmentally responsible Logistics Practices. In: Journal of Business Logistics. 21(2000)2, pp. 187-209.

Goldsim (2007): Dynamic Simulation and Supply Chain Management. Whitepaper. GoldSim Technology Group. URL:

[http://www.goldsim.com/downloads/whitepapers/scmpaper.pdf] last access: July 22, 2011.

González-Benito, J. / Martínez-Lorente, A. R. / Dale, B. G. (2003): A study of the purchasing management system with respect to total quality management. In: Industrial Marketing Management. 32(2003)6, pp. 443-454.

Grant, R. M. (2002): Contemporary Strategy Analysis: Concepts, Techniques, Applications. Oxford. 2002.

Grant, R. M. (1991): The Resource-Based Theory of Competitive Advantage: Implications for Strategy Formulation. In: California Management Review. 33(1991)3, pp. 114.

Grawe, S. J. / Chen, H. / Daugherty, P. J. (2009): The relationship between strategic orientation, service innovation, and performance. In: International Journal of Physical Distribution & Logistics Management. 39(2009)4, pp. 282.

Greeff, G. / Ghoshal, R. / Mackay, S. / Gerhard, G. / Ranjan, G. / Steve, M. (2004): Product and plant knowledge management. In: Practical E-Manufacturing and Supply Chain Management. Oxford 2004, pp. 185-213.

Griffiths, J. / Margetts, D. (2000): Variation in production schedules – implications for both the company and its suppliers. In: Journal of Materials Processing Technology. 103(2000)3, pp. 155-159.

Grimm, J. H. / Hinz, A. / Skorna, A. C. H. (2010): Risikomanagement zur Sicherstellung nachhaltiger Wertschöpfungsketten: Die Herausforderung für Unternehmen, ökologischen und sozialen Nachhaltigkeitsproblematiken in der Wertschöpfungskette erfolgreich zu begegnen. In: Logistics Innovation. 1(2010)4, pp. 20-23.

Grössler, A. / Schieritz, N. (2005): Of stocks, flows, agents and rules - 'Strategic' Simulation in Supply Chain Research. In: Kotzab, H. et al.: Research Methodologies in Supply Chain Management. Heidelberg 2005, pp. 445-460.

Gudehus, T. (2005): Logistik - Grundlagen - Strategien - Anwendungen. 3rd Edition. Berlin, Heidelberg, New York 2005.

Gunasekaran, A. (2004): Supply chain management: Theory and applications. In: European Journal of Operational Research. 159(2004)2, pp. 265-268.

Gunasekaran, A. / Ngai, E. W. T. (2004): Information systems in supply chain integration and management. In: European Journal of Operational Research. 159(2004)2, pp. 269-295.

Gundlach, G. T. / Bolumole, Y. A. / Eltantawy, R. A. / Frankel, R. (2006): The changing landscape of supply chain management, marketing channels of distribution, logistics and purchasing. In: Journal of Business & Industrial Marketing. 21(2006)7, pp. 428-438.

Güngör, Z. / Delice, E. K. / Kesen, S. E. (2011): New product design using FDMS and FANP under fuzzy environment. In: Applied Soft Computing. In Press, Accepted Manuscript 2011.

Gupta, M. C. (1995): Environmental Management and its Impact on the Operations Function. In: International Journal of Operations & Production Management. 15(1995)8, pp. 34.

Hackstein, R. (1989): Produktionsplanung und -steuerung (PPS). Ein Handbuch für die Betriebspraxis. . Düsseldorf 1989.

Halevi, G. (2001): 110 manufacturing methods. In: Handbook of Production Management Methods. Oxford 2001, pp. 59-310.

Hambrick, D. C. (1980): Operationalising the concept of business-level strategy in research. In: Academy of Management Review. 5(1980)4, pp. 567-576.

Hamm, V. (1997): Informationstechnikbasierte Referenzprozesse - Prozeßorientierte Gestaltung des industriellen Einkaufs. Wiesbaden 1997.

Hammer, R. M. (2003): Zur Weiterentwicklung des Performance Measurement-Ansatzes. In: Matzler, K. et al.: Werte schaffen – Perspektiven einer stakeholderorientierten Unternehmensführung. . Wiesbaden 2003, pp. 513-534.

Handfield, R. B. / Bechtel, C. (2002): The role of trust and relationship structure in improving supply chain responsiveness. In: Industrial Marketing Management,. 31(2002)3, pp. 367-382.

Hannerman, R. A. (1995): Simulation Modeling and Theoretical Analysis in Sociology. In: Sociological Perspectives. 38(1995)4, pp. 457-462.

Hannerman, R. A. / Patrick, S. (1997): On the Uses of Computer-assisted Simulation Modeling in the Social Sciences. In: Sociological Research online. 2(1997)2, pp. 1-10.

Hao, Q. / Shen, W. (2008): Implementing a hybrid simulation model for a Kanbanbased material handling system. In: Robotics and Computer-Integrated Manufacturing. 24(2008)5, pp. 635-646.

Harland, C. (1996): Supply chain management: relationships, chains and networks. In: British Journal of Management Decision. 7(1996)1, pp. 63-80.

Harland, C. / Knight, L. A. (2001): Supply network strategy: role and competence requirements. In: International Journal of Operations & Production Management. 21(2001)4, pp. 476-489.

Harland, C. / Lamming, R. C. / Cousins, P. D. (1999): Developing the concept of supply strategy. In: International Journal of Operations & Production Management, . 19(1999)7, pp. 650-673.

Harland, C. / Zheng, J. / Johnsen, T. / Lamming, R. (2004): A conceptual model for researching the creation and operation of supply networks. In: British Journal of Management. 15(2004)1, pp. 1-21.

Harrison, A. (1996): An investigation of the impact of schedule stability on supplier responsiveness. In: International Journal of Logistics Management. 7(1996)2, pp. 83-91.

Hartmann, S. (2004): Generating scenarios for simulation and optimization of container terminal logistics. In: OR Spectrum. 26(2004)2, pp. 171-192.

Harvey, J. (1998): Service quality: a tutorial. In: Journal of Operations Management. 16(1998)5, pp. 583-597.

Hatteland, M. J. a. C. J. (2004): Packages and physical distribution Implications for integration and standardisation. In: International Journal of Physical Distribution & Logistics Management. 34(2004)2, pp. 123-139.

Häusler, P. (2001): Integration der Logistik in Unternehmensnetzwerke. Frankfurt a.M. 2001.

Hayes, R. H. / Schmenner, R. W. (1978): How should you organize manufacturing. In: Harvard Business Review. 56(1978)1, pp. 105-118.

Hayes, R. H. / Wheelwright, S. C. (1979): Link manufacturing process and product life cycles. In: Harvard Business Review. 57(1979)1, pp. 133-140.

Heinrici, T. (2011): Kombi kontra Konventionell In: DVZ. 56(2011)10.05.2011, pp. 33.

Heister, G. (2005): Eisenbahnbetriebstechnologie. Heidelberg et al. 2005.

Henning, G. P. / Alves, R. M. d. B. / Augusto, C. / Nascimento, O. d. / Biscaia, E. C. (2009): Production Scheduling in the Process Industries: Current Trends, Emerging Challenges and Opportunities. In: Computer Aided Chemical Engineering Year

Henselek, H. F. (1996): Das Management von Unternehmenskonfigurationen Wiesbaden 1996.

Herrler, R. (2007): Agentenbasierte Simulation zur Ablaufoptimierung in Krankenhäusern und anderen verteilten, dynamischen Umgebungen. Würzburg 2007.

Heuermann, C. (2002): Internationalisierung und Logistikstrategien. Arbeitsbericht Nr. 4 des Seminars für Allgemeine Betriebswirtschaftslehre Köln 2002.

Heusler, K. F. (2004): Implementierung von SCM - Kompetenzorientierte Analyse aus Perspektive eines Netzwerkakteures. Wiesbaden 2004.

Hewitt, F. (1994): Supply Chain Redesign. In: The International Journal of Logistics Management. 5(1994)2, pp. 1-10.

Hill, C. A. / Scudder, G. D. (2002): The use of electronic data interchange for supply chain coordination in the food industry. In: Journal of Operations Management. 20(2002)4, pp. 375-387.

Hilletofth, P. (2009): How to develop a differentiated supply chain strategy. In: Industrial Management & Data Systems. 109(2009)1, pp. 16-33.

Ho, C.-F. / Chi, Y.-P. / Tai, Y.-M. (2005): A structural approach to measuring uncertainty in supply chains. In: International Journal of Electronic Commerce. 9(2005)3, pp. 91-114.

Hochhold, S. / Rudolph, B. (2009): Principal-Agent-Theorie. In: Schwaiger, M.Meyer, A.: Theorien und Methoden der Betriebswirtschaft Handbuch für Wissenschaftler und Studierende. München 2009, pp. 131-142.

Hoffmann, A. (2007): Unternehmensübergreifendes Kostenmanagement in intermodalen Prozessketten - Theoretische Fundierung und erste empirische Ergebnisse. Köln 2007.

Hoffmann, O. (2000): Performance Management – Systeme und Implementierungsansätze, 2nd Edition. Bern et al. 2000.

Hofmann, E. (2011): Lecture Notes - Supply Chain Management II. Spring semester 2011. Universität St.Gallen 2011.

Hofmann, E. (2010): Linking corporate strategy and supply chain management. In: International Journal of Physical Distribution & Logistics Management. 40(2010)4, pp. 256-276.

Hofmann, E. / Elbert, R. (2004): Collaborative Cash Flow Management: Financial Supply Chain Management als Herausforderung der Netzkompetenz. In: Pfohl, H.-C.: Netzkompetenz in Supply Chains. Wiesbaden 2004, pp. 93-117.

Hofmann, E. / Locker, A. (2009): Value-based performance measurement in supply chains: a case study from the packaging industry. In: Production Planning & Control. 20(2009)1, pp. 68-81.

Holderied, C. (2005): Güterverkehr, Spedition und Logistik München 2005.

Höll, A. (2008): Property-Rights-Theorie. In: Schwaiger, M.Anton, M.: Theorien und Methoden der Betriebswirtschaft Handbuch für Wissenschaftler und Studierende. München 2008, pp. 149-158.

Holmes, R. K. (1991): System Enquiry: A System Dynamics Approach. In: European Journal of Operational Research. 52(1991)3, pp. 386-386.

Homeyer, D. (2004): Bekanntgabe 1 zur Konzernrichtlinie 408. In: Deine Bahn. 1(2004)2, pp. 97-102.

Hong-Minh, S. M. / Disney, S. M. / Naim, M. M. (2000): The dynamics of emergency transshipment supply chains. In: International Journal of Physical Distribution and Logistics Management 9(2000)30, pp. 788–815.

Hoppe, N. / Conzen, F. (2002): Europäische Distributionsnetzwerke. Voraussetzungen, Projektablauf, Fallbeispiele. Wiesbaden 2002.

Horváth, P. / Minnning, F. (2001): Wertorientiertes Management in Deutschland, Grossbritannien, Italien und Frankreich - Eine empirische Analyse. In: Controlling. 13(2001)6, pp. 273-282.

Hsu, C.-C. / Kannan, V. R. / Keah-Choon, T. / Keong Leong, G. (2007): Information sharing, buyer-supplier relationships and firm performance - A multi-region analysis. In: International Journal of Physical Distribution & Logistics Management. 28(2007)4, pp. 296-310.

Hua, B. / Yuan, J. / Hui, D. C. W. / Bingzhen, C. / Arthur, W. W. (2003): The coevolutionary supply chain. In: Computer Aided Chemical Engineering. 2003, pp. 487-492.

Hua, Z. / Zhang, X. / Xu, X. (2011): Product design strategies in a manufacturer-retailer distribution channel. In: Omega. 39(2011)1, pp. 23-32.

Huang, S. H. / Uppal, M. / Shi, J. (2002): A productdriven approach to manufacturing supply chain selection. In: Supply Chain Management: An International Journal. 7(2002)3/4, pp. 189-199.

Huang, Y.-Y. / Li, S.-J. (2008): Suitable application situations of different postponement approaches: Standardization vs. modularization. In: Journal of Manufacturing Systems. 27(2008)3, pp. 111-122.

Hugelshofer (2011): Die Hugelshofer Gruppe. URL:

[http://www.hugelshofer.ch/gruppe/zahlen/] last access: July 20, 2011.

Hülsmann, M. / Windt, K. (2007): Approaches to Methods of Autonomous Cooperation and Control for the Management-, Information-, and Communication-Layer of Logistics. In: Hülsmann, M.Windt, K.: Understanding Autonomous Cooperation & Control - The Impact of Autonomy on Management, Information, Communication, and Material Flow. 2007, pp. 163-167.

Hung, W. Y. / Samsatli, N. J. / Shah, N. (2006): Object-oriented dynamic supply-chain modelling incorporated with production scheduling. In: European Journal of Operational Research. 169(2006)3, pp. 1064-1076.

Hvolby, H.-H. / Steger-Jensen, K. (2010): Technical and industrial issues of Advanced Planning and Scheduling (APS) systems. In: Computers in Industry. 61(2010)9, pp. 845-851.

Iakovou, E. / Vlachos, D. (2008): Sustainable Logistics and green Supply Chain Management: new strategic Priorities. In: Balkantrieb. 2008)pp.

ICC (2010a): The 11 Incoterms® rules (2010). URL:

[http://www.iccwbo.org/incotermsrules/] last access: July 18, 2011.

ICC (2010b): Incoterms® 2010 Rules by the International Chamber of Commerce (ICC). URL: [http://www.iccwbo.org/incoterms/] last access: July 18, 2011.

Ihde, G. B. (2001): Transport, Verkehr, Logistik gesamtwirtschaftliche Aspekte und einzelwirtschaftliche Handhabung. Vahlens Handbücher der Wirtschafts- und Sozialwissenschaften. 3rd updated Edition. München 2001.

Ihme, J. (2006): Logistik im Automobilbau: Logistikkomponenten und Logistiksysteme im Fahrzeugbau. München 2006.

Ilgin, M. A. / Gupta, S. M. (2010): Performance improvement potential of sensor embedded products in environmental supply chains. In: Resources, Conservation and Recycling. 55(2010)6, pp. 580-559.

Infras (2011a): Handbook emission factors for road transport (HBEFA) 3.1. Bern 2011a.

Infras (2011b): Handbuch für Emissionsfaktoren des Strassenverkehrs 3.1 (HBEFA). URL: [http://www.hbefa.net/d/index.html] last access: July 15, 2011.

Inman, R. R. / Gonzalves, J. A. (1997): Measuring and analysing supply chain schedule stability; a case study in the automotive industry. In: Production Planning & Control. 8(1997)3, pp. 194-204.

Investorwords (2011): Defintion: Original Equipment Manufacturer (OEM). URL: [http://www.investorwords.com/5689/OEM.html] last access: July 15th, 2011.

Iskan, S. (2009): Verkehrsmittelwahl für internationale Lieferbeziehungen. Hamburg 2009.

Jaber, M. Y. / Bonney, M. / Guiffrida, A. L. (2010): Coordinating a three-level supply chain with learning-based continuous improvement. In: International Journal of Production Economics. 127(2010)1, pp. 27-38.

Jagdev, H. S. / Thoben, K. D. (2001): Anatomy of enterprise collaborations. In: Production Planning & Control. 12(2001)pp. 437-451.

Janz, O. (2003): Integriertes Transpormanagement - Angebots- und nachfrageorientierte Planung und Steuerung komplexer Transportsysteme. Köln 2003.

Jayanth, J. / Keah-Choon, T. (2010): Supply chain integration with third-party logistics providers. In: International Journal of Production Economics. 125(2010)2, pp. 262-271.

Jayaram, J. / Tan, K.-C. (2011): Supply chain integration with third-party logistics providers. In: International Journal of Production Economics. 125(2011)2, pp. 262-271.

Jayaraman, V. / Srivastava, R. / Benton, W. C. (1999): Supplier Selection and Order Quantity Allocation: A Comprehensive Model. In: Journal of Supply Chain Management: A Global Review of Purchasing & Supply. 35(1999)2, pp. 50-58.

Jehle, M. (2004): Wertorientiertes Supply Chain Management und Supply Chain Controlling. Dortmund 2004.

Johnson, P. F. / Klassen, R. D. / Leenders, M. R. / Awaysheh, A. (2007): Utilizing ebusiness technologies in supply chains: the impact of firm characteristics and teams. In: Journal of Operations Management. 25(2007)3, pp. 1255-1274.

Jonsson, P. / Kjellsdotter, L. / Rudberg, M. (2007): Applying advanced planning systems for supply chain planning: three case studies. In: International Journal of Physical Distribution & Logistics Management. 37(2007)10, pp. 816-834.

Jörgl, T. (2011): Binnenschifffahrt: Probleme im Hinterland. URL: [http://www.logistik-heute.de/node/8534] last access: July 15, 2011 (published May 11, 2011).

Jünemann, R. (1989): Materialfluß und Logistik - Systemtechnische Grundlagen mit Praxisbeispielen. Berlin 1989.

Jung, J. W. / Lee, Y. H. (2010): Heuristic algorithms for production and transportation planning through synchronization of a serial supply chain. In: International Journal of Production Economics. 124(2010)2, pp. 433-447.

Kaijie, Z. / Zhang, R. Q. / Fugee, T. (2007): Pushing Quality Improvement Along Supply Chains. In: Management Science. 53(2007)3, pp. 421-436.

Kandemir, D. / Hult, G. T. M. (2005): A conceptualization of an organizational learning culture in international joint ventures. In: Industrial Marketing Management. 34(2005)5, pp. 430-439.

Kannan, V. R. / Tan, K. C. (2005): Just in time, total quality management, and supply chain management: understanding their linkages and impact on business performance. In: Omega. 33(2005)2, pp. 153-162.

Kaplan, R. S. / Norton, D. P. (1992): The Balanced Scorecard – Measures that Drive Performance. In: Harvard Business Review. 70(1992)1, pp. 71-79.

Kaplan, R. S. / Norton, D. P. (2004): Measuring the Strategic Readiness of Intangible Assets. In: Harvard Business Review. 82(2004)2, pp. 52-63.

Karrer, M. (2006): Supply Chain Performance Measurement - Entwicklung aus Ausgestaltung einer unternehmensübergreifenden Steuerkonzeption. Duisburg-Essen 2006.

Kasiske, F. (2004): Supply Chain Management im Kontext der europäischen Verkehrspolitik - Szenariobasierte Ansätze zur Gestaltung und Optimierung von Transportnetzen. Berlin 2004.

Kaynak, H. / Hartley, J. L. (2008): A replication and extension of quality management into the supply chain. In: Journal of Operations Management. 26(2008)4, pp. 468-489.

Ke, W. / Wei, K. K. (2007): Factors affecting trading partners' knowledge sharing: Using the lens of transaction cost economics and socio-political theories. In: Electronic Commerce Research and Applications. 6(2007)3, pp. 297-308.

Kearns, G. S. / Lederer, A. L. (2003): A resource-based view of strategic IT alignment: how knowledge sharing creates competitive advantage. In: Decision Science. 34 (2003)pp. 1-29.

Keebler, J. S. (2001): Measuring Performance in the Supply Chain. In: Mentzer, J.T.: Supply Chain Management. Thousand Oaks et al. 2001, pp. 411-435.

Keil, T. / Eloranta, E. / Holmström, J. / Järvenpää, E. / Takala, M. / Autio, E. / Hawk, D. (2001): Information and communication technology driven business transformation -- a call for research. In: Computers in Industry. 44(2001)3, pp. 263-282.

Kempainen, K. / Vepsalainen, A. P. J. (2003): Trends in Industrial Supply Chains and Networks. In: International Journal of Physical Distribution & Logistics Management. 33(2003)8, pp. 701.

Kent, J. L. / Parker, R. S. / Luke, R. H. (2001): An Emperical Examination of Shipper Perceptions of Service-Selection Attributes in Five Truckload Industry Segments. In: Transportation Journal. 41(2001)1, pp. 27-36.

Kern, W. (1979): Produktionswirtschaft. In: Kern, W.: Handwörterbuch der Produktionswirtschaft. Stuttgart 1979, pp. 1647-1660.

Kersten, W. / Hohrath, P. (2007): Risiko-Management in internationalen Supply Chains. In: Wimmer, T.: 24. Deutscher Logistik-Kongress. Berlin 2007, pp. 165-190.

Kieser, A. (1999): Der Situative Ansatz. In: Kieser, A.: Organisationstheorien. Stuttgart et al. 1999, pp. 169-198.

Kieser, A. / Kubicek, H. (1992): Organisation Berlin, New York 1992.

Kille, C. / Schmitdt, N. (2008): Wirtschaftliche Rahmenbedingungen des Güterverkehrs - Studie zum Vergleich der Verkehrsträger im Rahmen des Logistikprozesses in Deutschland. Nürnberg 2008.

Kim, B. / Leung, J. M. Y. / Park, K. T. / Zhang, G. Q. / Lee, S. (2002): Configuring a manufacturing firm's supply network with multiple suppliers. In: IIE Transactions. 34(2002)8, pp. 663-677.

Kim, K. / Choi, I. / Park, C. (2011): A rule-based approach to proactive exception handling in business processes. In: Expert Systems with Applications. 38(2011)1, pp. 394-409.

Kim, S. W. (2009): Quality Management Strategy in Supply Chain for Performance Improvement. In: Asian Journal on Quality. 10(2009)3, pp. 43-64.

Kirsch, W. (1973): Betriebswirtschaftliche Logistik. Systeme, Entscheidungen, Methoden. Wiesbaden 1973.

Kisperska-Moron, D. / Swierczek, A. (2010): The selected determinants of manufacturing postponement within supply chain context: An international study. In: International Journal of Production Economics. 2010)pp.

Klaas-Wissing, T. / Bendul, J. (2010): Logistics Innovation in Combined Transport. In: Wimmer, T.Grosche, T.: Structural Changes in Logistics Networks. Hamburg 2010, pp. 119-132.

Klaas, T. (2009): Konfigurationsansatz in der Logistikforschung. In: Albers, S.Reihlen, M.: Management integrierter Wertschöpfungsnetzwerke. Köln 2009, pp. 49-72.

Klaas, T. (2002): Logistik-Organisation - Ein konfigurationstheoretischer Ansatz zur logistikorientierten Organisationsgestaltung. Integrierte Logistik und Unternehmensführung. Wiesbaden 2002.

Klein, T. / Thomas, A. (2009): Opportunities to reconsider decision making processes due to Auto-ID. In: International Journal of Production Economics. 121(2009)1, pp. 99-111.

Klima, V. / Kavicka, A. (1996): RBSIM - Simulation Model of Marshalling Yard Operation. In: Proceedings of the International Conference on Computer Aided Design, Manufacture and Operation in the Railway and other Mass Transit Systems. Berlin 1996, pp. 493-500.

Klotz, H. (2011): Kombinierter Verkehr muss Qualität verbessern. In: DVZ. 56(2011)65, pp. 1.

Knorr, A. (2005): Verkehrspolitik: Die Entwicklung im (Nachkriegs-)Deutschland. In: Leschke, M.Pies, I.: Wissenschaftliche Politikberatung - Theorien, Konzepte, Institutionen Stuttgart 2005, pp. 74-94.

Ko, J. M. / Kwak, C. / Cho, Y. / Kim, C. O. (2011): Adaptive product tracking in RFID-enabled large-scale supply chain. In: Expert Systems with Applications. 38(2011)3, pp. 1583-1590.

Kobayashi, T. / Tamaki, M. / Komoda, N. (2003): Business process integration as a solution to the implementation of supply chain management systems. In: Information & Management. 40(2003)8, pp. 769-780.

Koch, S. / Strahringer, S. (2008): Customer & Supplier Relationship Management. Heidelberg 2008.

Kombiverkehr (2011): Beispielrechnung Kombinierter Verkehr URL: [http://www.kombiverkehr.de/neptun/neptun.php/oktopus/page/1/184] last access: July 15, 2011. *Kong, S. C. W. / Li, H. / Hung, T. P. L. / Shi, J. W. Z. / Castro-Lacouture, D. / Skibniewski, M. (2004):* Enabling information sharing between E-commerce systems for construction material procurement. In: Automation in Construction. 13(2004)2, pp. 261-276.

Koppelmann, U. (2002): Zur Bedeutung der Beschaffungsfunktion in deutschen Unternehmen. Köln 2002.

Kosturiak, J. / Gregor, M. (1999): Simulation in production system life cycle. In: Computers in Industry 38(1999)1, pp. 159–172.

Kot, S. / Ślusarczyk, B. (2009): Process Simulation in Supply Chain using Logware Software. In: Annales Universitatis Apulensis Series Oeconomica, 11(2), 2009. 11(2009)2, pp. 932-937.

Kotzab, H. (1997): Neue Konzepte der Distributionslogistik von Handelsunternehmen. Wiesbaden 1997.

Krajewski, L. / Wei, J. C. (2001): The Value of Production Schedule Integration in Supply Chains. In: Decision Sciences. 32(2001)4, pp. 601-634.

Krajewski, L. / Wei, J. C. / Tang, L. L. (2005): Responding to schedule changes in build-to-order supply chains. In: Journal of Operations Management. 23(2005)2, pp. 452-469.

Kreutzberger, E. / Liedtke, G. / Schepperle, H. (1999): Innovative networks and newgeneration terminals for intermodal transport. Proceedings 5th TRAIL Annual Congress 1999.

Kroes, J. R. / Ghosh, S. (2010): Outsourcing congruence with competitive priorities: Impact on supply chain and firm performance. In: Journal of Operations Management. 28(2010)2, pp. 124-143.

Ku, K.-C. / Gurumurthy, C. K. / Kao, H.-P. (2007): Inter-firms collaboration of joint venture in IC foundry business. In: Technovation. 27(2007)5, pp. 296-305.

Kubicek, H. (1977): Heuristische Bezugsrahmen und heuristisch angelegte Forschungsdesigns als Elemente einer Konstruktionsstrategie empirischer Forschung. In: Köhler, R.: Empirische und handlungstheoretische Forschungskonzeptionen in der Betriebswirtschaftslehre. Stuttgart 1977, pp. 5-36.

Kudla, N. / Klaas-Wissing, T. (2011): Incentives for Sustainable Logistics: Proactive Logistics Service Providers vs. Shipper's Requirements? : A Tentative Taxonomy based on 10 Case Studies. 73. Wissenschaftliche Jahrestagung des Verbandes der Hochschullehrer für Betriebswirtschaft e.V. 2011.

Kuhn, A. / Reinhardt, A. / Wiendahl, H.-P. (1998): Simulationsanwendungen in Produktion und Logistik. Braunschweig 1998.

Kühn, W. (2006): Digitale Fabrik: Fabriksimulationen für Produktion und Logistik München, Wien 2006.

Küppler, H.-U. / Helber, K. (2004): Ablauforganisation in Produktion und Logistik. 3rd updated edition. Stuttgart 2004.

Kwak, C. / Choi, J. S. / Kim, C. O. / Kwon, I.-H. (2009): Situation reactive approach to Vendor Managed Inventory problem. In: Expert Systems with Applications. 36(2009)5, pp. 9039-9045.

La Londe, B. J. / Masters, J. M. (1994): Emerging Logistics Strategies: Blueprints for the Next Century. In: International Journal of Physical Distribution & Logistics Management. 24(1994)7, pp. 35-47.

Lambert, D. M. / Cooper, M. C. (2000): Issues in Supply Chain Management. In: Industrial Marketing Management. 29(2000)1, pp. 65-83.

Lambert, D. M. / Cooper, M. C. / Pagh, J. D. (1998): Supply chain management: implementation issues and research opportunities. In: International Journal of Logistics Management Decision. 9(1998)2, pp. 1-20.

Lambert, D. M. / Pohlen, T. L. (2001): Supply chain metrics. In: The international Journal of Logistics Management. 12(2001)1, pp. 1–19.

Lamming, R. C. / Johnsen, T., Zheng, J. / Harland, C. (2000): An initial classification of supply networks. In: International Journal of Operations & Production Management Decision. 20(2000)6, pp. 675-691.

Lamnek, S. (2005): Qualitative Sozialforschung. 4th updated Edition. Weinheim 2005.

Lancioni, R. A. / Smith, M. F. / Oliva, T. A. (2000): The Role of the Internet in Supply Chain Management. In: Industrial Marketing Management. 29(2000)1, pp. 45-56.

Lane, D. C. (2000): Should System Dynamics be described as a 'hard' or 'deterministic' Systems Approach? In: Systems Research and Behavioral Science. 17(2000)1, pp. 3.

Lane, D. C. (1999): Social Theory and System Dynamics Practice. In: European Journal of Operational Research. 113(1999)3, pp. 501-527.

Lassar, W. M. / Kerr, J. L. (1996): Strategy and Control in Supplier-Distributor Relationships: an Agency Perspective. In: Strategic Management Journal. 17(1996)8, pp. 613-628.

Lättilä, L. / Hilletofth, P. / Lin, B. (2010): Hybrid simulation models - When, Why, How? In: Expert Systems with Applications. 37(2010)12, pp. 7969-7975.

Lebas, M. J. / Euske, K. (2002): A Conceptual and Operational Delineation of Performance. In: Neely, A.: Business Performance Measurement. Cambridge 2002, pp. 65-79.

Lechner, K. / Egger, A. / Schauer, R. (2008): Einführung in die Allgemeine Betriebswirtschaftslehre. Wien 2008.

Lee-Kelley, L. / Sankey, T. (2008): Global virtual teams for value creation and project success: A case study. In: International Journal of Project Management. 26(2008)1, pp. 51-62.

Lee, C. K. M. / Ho, W. / Ho, G. T. S. / Lau, H. C. W. (2011): Design and development of logistics workflow systems for demand management with RFID. In: Expert Systems with Applications. 38(2011)5, pp. 5428-5437.

Lee, H. L. (2002): Aligning supply chain strategies with

product uncertainties. In: California Management Review. 44(2002)3, pp. 105-119.

Lee, H. L. / Padmanabhan, V. / Whang, S. (1997): Information distortion in a supply chain: the bullwhip effect. In: Management Science. 43(1997)4, pp. 546-558.

Lee, S. / Lim, G. G. (2005): The impact of partnership attributes on EDI implementation success. In: Information & Management. 42(2005)4, pp. 503-516.

Lenzen, M. / Murray, J. / Sack, F. / Wiedmann, T. (2007): Shared Producer and Consumer Responsibility -- Theory and Practice. In: Ecological Economics. 61(2007)1, pp. 27-42.

Li-Ling, H. / Chao-Min, C. / Chen, J. C. H. / Chin-Cheng, L. (2009): The impacts of supply chain management systems on information sharing and integrated-performance. In: Human Systems Management. 28(2009)3, pp. 101-121.

Li, J. / Sikora, R. / Shaw, M. J. / Woo Tan, G. (2006a): A strategic analysis of inter organizational information sharing. In: Decision Support Systems. 42(2006a)1, pp. 251-266.

Li, S. / Lin, B. (2006): Accessing information sharing and information quality in supply chain management. In: Decision Support Systems. 42(2006)3, pp. 1641-1656.

Li, S. / Ragu-Nathan, B. / Ragu-Nathan, T. S. / Subba Rao, S. (2006b): The impact of supply chain management practices on competitive advantage and organizational performance. In: Omega. 34(2006b)2, pp. 107-124.

Liao, Z. / Rittscher, J. (2007): Integration of supplier selection, procurement lot sizing and carrier selection under dynamic demand conditions. In: International Journal of Production Economics. 107(2007)pp. 502-510.

Lienert, G. / Raatz, U. (1998): Testaufbau und Testanalyse. Weinheim 1998.

Lin, L. C. (2009): An integrated framework for the development of radio frequency identification technology in the logistics and supply chain management. In: Computers & Industrial Engineering. 57(2009)3, pp. 832-842.

Lin, P.-C. / Wang, J. / Huang, S.-H. / Wang, Y.-T. (2010): Dispatching ready mixed concrete trucks under demand postponement and weight limit regulation. In: Automation in Construction. 19(2010)6, pp. 798-807.

Linton, J. D. / Klassen, R. / Jayaraman, V. (2007): Sustainable supply chains: An introduction. In: Journal of Operations Management. 25(2007)6, pp. 1075-1082.

Liu, C. / Jula, H. / Ioannou, P. A. (2002): Design, simulation, and evaluation of automated container terminals. In: IEEE Transactions on Intelligent Transportation Systems. 3(2002)1, pp. 12-26.

Liu, H.-T. (2011): Product design and selection using fuzzy QFD and fuzzy MCDM approaches. In: Applied Mathematical Modelling. 35(2011)1, pp. 482-496.

Liu, J. / Zhang, S. / Hu, J. (2005): A case study of an inter-enterprise workflow-supported supply chain management system. In: Information & Management. 42(2005)3, pp. 441-454.

Lo, C. K. Y. / Yeung, A. C. L. / Cheng, T. C. E. (2009): ISO 9000 and supply chain efficiency: Empirical evidence on inventory and account receivable days. In: International Journal of Production Economics. 118(2009)2, pp. 367-374.

Lo, S. / Power, D. (2010): An empirical investigation of the relationship between product nature and supply chain strategy. In: Supply Chain Management: An International Journal. 15(2010)2, pp. 139-153.

Lödding, H. (2001): Dezentrale Bestandsorientierte Fertigungsregelung. Düsseldorf 2001.

Longo, F. (2010): Design and integration of the containers inspection activities in the container terminal operations. In: International Journal of Production Economics. 125(2010)2, pp. 272-283.

Loo, R. (2003): Assessing 'team climate' in project teams. In: International Journal of Project Management. 21(2003)7, pp. 511-517.

López, T. S. (2011): RFID and sensor integration standards: State and future prospects. In: Computer Standards & Interfaces. 33(2011)3, pp. 207-213.

Lu, C.-S. (2003): The impact of carrier service attributes on shipper-carrier partnering relationships: a shipper's perspective. In: Transportation Research: Part E. 39(2003)5, pp. 399-415.

Lu, T.-T. / Chen, J.-C. (2010): Incremental or radical? A study of organizational innovation: An artificial world approach. In: Expert Systems with Applications. 37(2010)12, pp. 8193-8200.

LUBW (2002): Möglichkeiten zur Minderung von Diesellok-Emissionen. online available under URL: [http://www.lubw.baden-

wuerttemberg.de/servlet/is/23231/diesellok-bericht-

2002.pdf?command=downloadContent&filename=diesellok-bericht-2002.pdf] last access: July 22, 2011. Karlsruhe 2002.

LUBW (2011): Verkehr. Landesanstalt für Umwelt, Messungen und Naturschutz Baden-Württemberg. URL: [http://www.lubw.baden-

wuerttemberg.de/servlet/is/41578/02_verkehr.pdf?command=downloadContent&filen ame=02_verkehr.pdf] last access: July 20, 2011.

Luczak, H. / Eversheim, W. (1999): Produktionsplanung und -steuerung, Grundlagen, Gestaltung und Konzepte. Berlin, Heidelberg, New York 1999.

Luhmann, N. (1999): Funktionen und Folgen formaler Organisation. Berlin 1999.

Lummus, R. / Duclos, L. K. / Vokurka, R. J. (2003): Supply chain flexibility: building a new model. In: Global Journal of Flexible Systems Management. 4(2003)4, pp. 1-14.

Lyu, J. / Ding, J.-H. / Chen, P.-S. (2010): Coordinating replenishment mechanisms in supply chain: From the collaborative supplier and store-level retailer perspective. In: International Journal of Production Economics. 123(2010)1, pp. 221-234.

Ma, S. / Wang, W. / Liu, L. (2002): Commonality and postponement in multistage assembly systems. In: European Journal of Operational Research. 142(2002)3, pp. 523-538.

Mabert, V. A. / Soni, A. / Venkataramanan, M. A. (2003): Enterprise resource planning: Managing the implementation process. In: European Journal of Operational Research. 146(2003)2, pp. 302-314.

MacDuffie, J. P. / Sethuraman, K. / Fisher, M. L. (1996): Product variety and manufacturing performance: evidence from the international assembly plant study. In: Management Science. 42(1996)3, pp. 350-369.

Macharis, C. / Pekin, E. (2009): Assessing policy measures for the stimulation of intermodal transport: a GIS-based policy analysis. In: Journal of Transport Geography. 17(2009)6, pp. 500-508.

Macharzina, K. / Engelhard, J. (1991): Paradigm Shift in International Business Research – From Partist and Eclectic Approaches to the GAINS Paradigm. In: Management International Review. 31(1991)Special Issue, pp. 23-43.

Macharzina, K. / Wolf, J. (2005): Unternehmensführung - Das internationale Managementwissen. 5th Edition. Wiesbaden 2005.

Mahnam, M. / Yadollahpour, M. R. / Famil-Dardashti, V. / Hejazi, S. R. (2009): Supply chain modeling in uncertain environment with bi-objective approach. In: Computers & Industrial Engineering. 56(2009)4, pp. 1535-1544.

Maier, G. / Bergman, E. M. / Lehner, P. (2002): Modelling preferences and stability among transport alternatives. In: Transportation Research: Part E. 38(2002)5, pp. 319.

Makinwa, K. A. A. (2010): Smart temperature sensors in standard CMOS. In: Procedia Engineering. 5(2010)1, pp. 930-939.

Manor (2011a): Facts - Kennzahlen URL: [http://www.manor.ch/de/zahlen/facts-kennzahlen.html] last access: July 18, 2011.

Manor (2011b): Mehr als hunder Geschäftsjahre. URL: [http://www.manor.ch/de/mehr+als+hundert+gesch%C3%A4ftsjahre/facts-geschichte.html] last access: July 18, 2011.

Manor (2011c): Nachhaltigkeit - Worauf wir wert legen. URL: [http://www.manor.ch/de/werte/nachhaltigkeit-vorrauf-wir-wert-legen.html] last access: July 18, 2011.

Manuj, I. / Mentzer, J. T. (2008): Global supply chain risk management strategies. In: International Journal of Physical Distribution & Logistics Management. 38(2008)3, pp. 192-233.

Marbacher, A. (2001): Demand & Supply Chain Management. Bern 2001.

Markley, M. J. / Davis, L. (2007): Exploring future Competitive Advantage through sustainable Supply Chains. In: International Journal of Physical Distribution & Logistics Management. 37(2007)9, pp. 763-774.

Mason-Jones, R. / Naylor, B. / Towill, D. R. (2000): Lean, agile or leagile? Matching your supply chain to the marketplace. In: International Journal of Production Research. 38(2000)17, pp. 4061-4070.

Mason-Jones, R. / Towill, D. R. (1997): Information enrichment: designing the supply chain for competitive advantage. In: 2. 4(1997)137-148, pp.

Maus (2011): Landmark Brands. Maus Frères Holding. URL: [http://www.maus.ch/en/#p5]

McCutcheon, D. M. / Raturi, A. S. / Meredith, J. R. (1994): The customizationresponsiveness squeze. In: Sloan Management Review. 35(1994)3, pp. 89-99.

McKelvey, B. (1999): Theory in Organization Science: Seizing the Promise or Becoming a Fad? In: Emergence. 1(1999)1, pp. 5-32.

McKinnon, A. (2009): The present and future land requirements of logistical activities. In: Land Use Policy. 26(2009)Supplement 1, pp. S293-S301.

McKinsey & Company, I. (2007): Kosten und Potenziale der Vermeidung von Treibhausgasemissionen in Deutschland. 2007.

Mearsk (2011): Emma Mearsk. URL: [http://www.emma-maersk.com/specification/] last access: July 10, 2011.

Meisner Rosen, C. / Bercovitz, J. / Beckman, S. (2000): Environmental Supply-Chain Management in the Computer Industry: a Transaction Cost Economics Perspective. In: Journal of Industrial Ecology. 4(2000)4, pp. 83-103.

Meixell, M. J. / Norbis, M. (2008): A review of the transportation mode choice and carrier selection literature. In: The International Journal of Logistics Management. 19(2008)2, pp. 183-211.

Mele, F. D. / Méndez, C. A. / Espuña, A. / Puigjaner, L. / Marquardt, W. / Pantelides, C. (2006): A novel combined approach for supply chain modeling and analysis. In: Computer Aided Chemical Engineering. 2006, pp. 2207-2212.

Mentzer, J. T. / DeWitt, W. / Keebler, J. S. / Min, S. (2001a): Defining Supply Chain Management. In: Journal of Business Logistics. 22(2001a)2, pp. 1-26.

Mentzer, J. T. / DeWitt, W. / Keebler, J. S. / Soonhoong, M. / Nix, N. W. / Smith, C. D. / Zacharia, Z. G. (2001b): Defining SCM In: Journal of Business Logistics. 22(2001b)2, pp. 1-25.

Mentzer, J. T. / Konrad, B. (1991): An Efficiency/Effectiveness Approach to Logistics Performance Analysis. In: Journal of Business Logistics. 12(1991)1, pp. 33-61.

Mert, T. / Richey, R. G. / Louis, D. M. / Weaver, K. M. (2007): Exploration, Exploitation and Satisfaction in Supply Chain Portfolio Strategy. In: Journal of Business Logistics. 28(2007)1, pp. 25.

Meyer, G. G. / Främling, K. / Holmström, J. (2009): Intelligent Products: A survey. In: Computers in Industry. 60(2009)3, pp. 137-148.

Meyr, H. / Stadtler, H. (2005): Types of Supply Chains. In: Stadtler, H.Kilger, C.: Supply Chain Management and Advanced Planning: Concepts, Models, Software and Case Studies. Berlin 2005, pp. 65 – 80.

Michaelraj, L. A. / Shahabudeen, P. (2009): Replenishment policies for sustainable business development in a continuous credit based vendor managed inventory distribution system. In: Computers & Industrial Engineering. 56(2009)1, pp. 260-266.

Miller, D. (1981): Toward a New Contingency Approach: The Search for Organizational Gestalts. In: Journal of Management Studies. 18(1981)1, pp. 1-26.

Miller, D. / Friesen, P. H. (1984): Organizations - A Quantum View. Englewood Cliffs 1984.

Miller, T. / De Matta, R. (2003): Integrating Production and transportation scheduling decisions between two geographically separated plants. In: Journal of Business Logistics. 24(2003)1, pp. 111-146.

Min, H. / Zhou, G. (2002): Supply chain modeling: past, present and future. In: Computers & Industrial Engineering. 43(2002)1-2, pp. 231-249.

Min, S. / Mentzer, J. / Ladd, R. (2007): A market orientation in supply chain management. In: Journal of the Academy of Marketing Science. 35(2007)4, pp. 507-522.

Mintzberg, H. (1981): Organization Design: Fahion or Fit? In: Harvard Business Revew. 59(1981)1, pp. 103-116.

Mintzberg, H. (1983): Power in and around Organizations. Englewood Cliffs 1983.

Mintzberg, H. (1979): The Structuring of Organizations - A Synthesis of the Research. Englewood Cliffs 1979.

Mizgier, K. J. / Wagner, S. M. / Holyst, J. A. (2010): Modeling defaults of companies in multi-stage supply chain networks. International Journal of Production Economics. In Press, Corrected Proof. 2010.

Moffat, L. K. (1998): Tools and teams: competing models of integrated product development project performance. In: Journal of Engineering and Technology Management. 15(1998)1, pp. 55-85.

Mohammed, I. R. / Shankar, R. / Banwet, D. K. (2008): Creating flex-lean-agile value chain by outsourcing: an ISM-based interventional roadmap. In: Business Process Management Journal. 14(2008)3, pp. 338-389.

Möller, K. (2003): Supply Chain Valuation - Wertschöpfung in und durch Supply Chain Networks. In: Supply Chain Controlling in Theorie und Praxis - Aktuelle Konzepte und Unternehmensbeispiele. Wiesbaden 2003, pp. 49-82.

Moore, D. M. / Antill, P. D. (2001): Integrated Project Teams: the way forward for UK defence procurement. In: European Journal of Purchasing & Supply Management. 7(2001)3, pp. 179-185.

Mortensen, M. H. / Freytag, P. V. / Arlbjørn, J. S. (2008): Attractiveness in supply chains: a process and matureness perspective. In: International Journal of Physical Distribution & Logistics Management. 38(2008)10, pp. 799.

Mukhopadhyay, T. / Kekre, S. (2002): Strategic and operational benefits of electronic integration in B2B procurement processes. In: Management Science. 48(2002)10, pp. 1301-1313.

Müller-Stewens, G. / Lechner, C. (2005): Strategisches Management: Wie strategische Initiativen zum Wandel führen: Der St.Galler General Management Navigator. 3. updated Edition. 2005.

Müller, I. / Klaus, P. (2009): Die Zukunft des Ladungsverkehrs in Europa. Hamburg 2009.

Müller, J. F. (2011): Schüttgüter erobern den Kombitransport In: Verkehrsrundschau Spezial - Sonderheft für Güterverkehr Strasse-Schiene. 56(2011)1, pp. 12-13.

Murphy, D. J. / Farris, M. T. (1993): Time-Based strategy and carrier selection. In: Journal of Business Logistics. 14(1993)2, pp. 25-40.

N.N. (2009): Bundesministerium stoppt Start der Elbvertiefung. URL:

[http://www.abendblatt.de/hamburg/article1136880/Bundesministerium-stoppt-Start-der-Elbvertiefung.html] last access: July 15, 2011 (published August 12, 2009).

N.N. (2004a): DB AG fordert mehr europäischen Wettbewerb im Güterverkehr durch effiziente Zulassungsverfahren. In: Eisenbahn-Revue international 1(2004a)5, pp. 205-209

N.N. (2011a): Der Rheinpegel sinkt stetig. Frankfurter Rundschau. URL: [http://www.fr-online.de/wirtschaft/der-rheinpegel-sinkt-stetig/-/1472780/8445034/-/] last access: July 15, 2011.

N.N. (2011b): Flüsterbremsen für alte Güterwaggons gegen den Lärm. URL: [http://www.stuttgarter-zeitung.de/inhalt.fluesterbremsen-fuer-alte-gueterwaggonsgegen-den-laerm.e9e75929-025e-4cb2-8480-8a885dda2029.html] last access:July 20, 2011 (published July 14, 2011).

N.N. (2011c): Piratenangriff auf deutschen Supertanker abgewehrt. Hamburger Abendblatt. URL:

[http://www.abendblatt.de/politik/ausland/article1894359/Piratenangriff-aufdeutschen-Supertanker-abgewehrt.html] last access: July 2015 (published May 18, 2011).

N.N. (2011d): Schiffsunfall in Nähe der Loreley. URL:

[http://www.focus.de/panorama/vermischtes/rhein-schiffsunfall-in-naehe-der-loreley-zwei-besatzungsmitglieder-vermisst_aid_589835.html] last access: July 15, 2011 (published: January 13, 2011).

N.N. (2004b): Technische Übersetzung aktueller Nachfrageprognosen für den Schienengüterverkehr auf die Trassenkapazitäten 2010 - 2020. Zürich 2004b.

N.N. (2011e): Wettbewerb auf der Schiene EU verklagt Deutschland vor EuGH In: DVZ. 70(2011e)64, pp. 1.

Nagel, C. (2008): The integration of rail transport into freight transport logistics chains of logistics service providers. Berlin 2008.

Naim, M. M. / Potter, A. T. / Mason, R. J. / Bateman, N. (2006): The role of transport flexibility in logistics provision. In: International Journal of Logistics Management. 17(2006)3, pp. 297-311.

Nam, K. / Kwak, K. / Yu, M. (2002): Simulation study of container terminal performance. In: Journal of Waterway. 128(2002)3, pp. 126-132.

Narasimhan, R. / Swink, M. / Kim, S. W. (2006): Disentangling leanness and agility: an empirical investigation. In: Journal of Operations Management. 24(2006)5, pp. 440.

Naylor, J. B. / Naim, M. M. / Berry, D. (1999): Leagility: Integrating the lean and agile manufacturing paradigm in the total supply chain. In: International Journal of Production Economics. 62(1999)1, pp. 107-118.

Neely, A. / Gregory, M. / K., P. (1995): Performance Measurement System Design. In: International Journal of Operations and Production Management. 19(1995)4, pp. 80-116.

Neher, A. (2003): Wertorientierung im Supply Chain Controlling. In: Stölzle, W.Otto, A.: Supply Chain Controlling in Theorie und Praxis: Aktuelle Konzepte und Unternehmensbeispiele. Wiesbaden 2003, pp. 27-47.

Neiro, S. M. S. / Pinto, J. M. (2004): A general modeling framework for the operational planning of petroleum supply chains. In: Computers & Chemical Engineering. 28(2004)6-7, pp. 871-896.

Newell, S. / Huang, J. C. / Galliers, R. D. / Pan, S. L. (2003): Implementing enterprise resource planning and knowledge management systems in tandem: fostering efficiency and innovation complementarity. In: Information and Organization. 13(2003)1, pp. 25-52.

Ngai, E. / Riggins, F. (2008): RFID: Technology, applications, and impact on business operations. In: International Journal of Production Economics. 112(2008)2, pp. 507-509.

Nielsen, N. P. H. / Holmström, J. (1995): Design for speed: a supply chain perspective on design for manufacturability. In: Computer Integrated Manufacturing Systems. 8(1995)3, pp. 223-228.

Norman, A. / Jansson, U. (2004): Ericsson's proactive supply chain risk management approach after a serious sub-supplier accident. In: International Journal of Physical Distribution & Logistics Management. 34(2004)2, pp. 434-456.

Nyhuis, P. / Wiendahl, H.-P. (2008): Logistische Kennlinien. Berlin 2008.

Ohno, T. (1988): The Toyota Production System - Beyond Large Scale Production,. Portland, Oregon 1988.

Ojasalo, J. (2004): Key network management. In: Industrial Marketing Management. 33(2004)3, pp. 195-205.

Ojeda, R. d. / Maria, J. (2002): Optimierung der Interoperabilität der RENFE auf europäischer Ebene. In: rail international - Schienen der Welt, pp. 35-41.

Olhager, J. (2003): Strategic positioning of the order penetration point. In: International Journal of Production Economics 85(2003)1, pp. 319–329.

Otto, A. / Kotzab, H. (2002): Ziel erreicht? Sechs Perspektiven zur Ermittlung des Erfolgsbeitrags des Supply Chain Management. In: Hahn, D.Kaufmann, L.: Handbuch Industrielles Beschaffungsmanagement. Wiesbaden 2002, pp. 125-150.

Owen, C. (2010): Simulation in the Supply Chain Context: Matching the Simulation Tool to the Problem. Proceedings of the Operational Research Society Simulation Workshop. Birmingham 2010.

Özbayrak, M. / Papadopoulou, T. C. / Akgun, M. (2007): Systems dynamics modelling of a manufacturing supply chain system. In: Simulation Modelling Practice and Theory. 15(2007)10, pp. 1338-1355.

Pagell, M. / Krause, D. / Klassen, R. (2008): Sustainable Supply Chain Management: Theory and Practice. In: Journal of Supply Chain Management: A Global Review of Purchasing & Supply. 44(2008)pp. 85.

Pagell, M. / Wu, Z. (2009): Building a more complete Theory of sustainable Supply Chain Management using Case Studies of 10 Exemplars. In: Journal of Supply Chain Management: A Global Review of Purchasing & Supply. 45(2009)2, pp. 37-56.

Pagh, J. / Cooper, M. (1998): Supply chain postponement and speculation strategy: how to choose the right strategy. In: Journal Business Logistics. 19(1998)2, pp. 13-33.

Panayides, P. M. / Song, D.-W. (2008): Evaluating the integration of seaport container terminals in supply chains. In: International Journal of Physical Distribution & Logistics Management. 38(2008)7, pp. 562-584.

Panayides, P. M. / Venus Lun, Y. H. (2009): The impact of trust on innovativeness and supply chain performance. In: International Journal of Production Economics. 122(2009)1, pp. 35-46.

Papageorgiou, L. G. (2009): Supply chain optimisation for the process industries: Advances and opportunities. In: Computers & Chemical Engineering. 33(2009)12, pp. 1931-1938.

Parast, M. M. (2011): The effect of Six Sigma projects on innovation and firm performance. In: International Journal of Project Management. 29(2011)1, pp. 45-55.

Parment, A. (2008): Distribution strategies for volume and premium brands in highly competitive consumer markets. In: Journal of Retailing and Consumer Services. 15(2008)4, pp. 250-265.

Parsons, A. L. (2002): What Determines Buyer-Seller Relationship Quality? An Investigation from the Buyer's Perspective. In: Journal of Supply Chain Management: A Global Review of Purchasing & Supply. 38(2002)2, pp. 4-12.

Patil, R. / Avittathur, B. / Shah, J. Supply chain strategies based on recourse model for very short life cycle products. In: International Journal of Production Economics. 128(1, pp. 3-10.

Perona, M. / Saccani, N. (2004): Integration techniques in customer-supplier relationships: An empirical research in the Italian industry of household appliances. In: International Journal of Production Economics. 89(2004)pp. 189-205.

Persona, A. / Regattieri, A. / Pham, H. / Battini, D. (2007): Remote control and maintenance outsourcing networks and its applications in supply chain management. In: Journal of Operations Management. 25(2007)6, pp. 1275-1291.

Peters, N. (2010): Inter-organisational Design of Voluntary Sustainability Initiatives: Increasing the Legitimacy of Sustainability Strategies for Supply Chains. Wiesbaden 2010.

Petersen, K. J. / Handfield, R. B. / Ragatz, G. L. (2005): Supplier integration into new product development: coordinating product, process and supply chain design. In: Journal of Operations Management. 23(2005)3-4, pp. 371-388.

Petrovic, D. (2001): Simulation of supply chain behaviour and performance in an uncertain environment. In: International Journal of Production Economics 1(2001)71, pp. 429–438.

Pflughoeft, K. A. / Ramamurthy, K. / Soofi, E. S. / Yasai-Ardekani, M. / Zahedi, F. (2003): Multiple conceptualizations of small business web use and benefit. In: Decision Science. 34(2003)3, pp. 467-512.

Pfohl (2010): Logistiksysteme - Betriebswirtschaftliche Grundlagen. Heidelberg 2010.

Pfohl, H.-C. (2003): Entwicklungstendenzen auf dem Markt logistischer Dienstleistungen. Güterverkehr - Eine Integrationsaufgabe für die Logisitik Darmstadt 2003.

Pfohl, H.-C. (2006): Logistikmangement - Funktionen und Instrumente. Heidelberg et al. 2006.

Pfohl, H.-C. (2002): Risiko- und Chancenmanagement. Berlin 2002.

Pfohl, H.-C. / Gomm, M. / Hofman, E. (2003): Netzwerke in der Transportlogistik. Darmstadt 2003.

Pfohl, H.-C. / Stölzle, W. (1997): Planung und Kontrolle. München 1997.

Pibernik, R. / Zhang, Y. / Kerschbaum, F. / Schröpfer, A. (2007): Secure collaborative supply chain planning and inverse optimization - The JELS model. In: European Journal of Operational Research. 208(2007)1, pp. 75-85.

Pidd, M. (2003): Tools for Thinking: Modeling in Management Science. 2nd Edition. Chichester 2003.

Piller, F. T. (2006): Mass Customization: Ein wettbewerbs-strategisches Konzept im Informationszeitalter. Wiesbaden 2006.

Piontek, J. (1994): Beschaffungscontrolling. München 1994.

Planco (2007): Verkehrswirtschaftlicher und ökologischer Vergleich der Verkehrsträger Straße, Schiene und Wasserstraße. Koblenz 2007.

Plenert, G. (2007): What are the technical tools of lean management? In: Reinventing Lean. Burlington 2007, pp. 233-269.

Poirier, C. C. / Quinn, F. J. (2004): How are we doing? A survey of Supply Chain Progress. In: Supply Chain Management Review. 8(2004)8, pp. 24-31.

Polo-Redondo, Y. / Cambra-Fierro, J. (2008): Influence of the standardization of a firm's productive process on the long-term orientation of its supply relationships: An empirical study. In: Industrial Marketing Management. 37(2008)4, pp. 407-420.

Polzin, D. W. (1999): Multimodale Unternehmensnetzwerke im Güterverkehr. München 1999.

Porter, M. E. (2004): Competitive Advantage: Creating and Sustaining Superior Performance. New York 2004.

Porter, M. E. (1980): Competitive Strategy: Techniques for analyzing industries and competitors: with a new introduction. New York 1980.

Post (2011a): Brief- und Paketzentren. URL: [http://www.post.ch/post-startseite/post-konzern/post-konzern-uebersicht/post-kundenkontakt/post-briefzentrum/post-briefzentrum-zentren-der-briefverarbeitung/post-briefzentrum-zentren-gross.jpgw] last access: July 21, 2011.

Post (2011b): Die Schweizerische Post. Kombiverkehr: der Umwelt zuliebe. URL: [http://www.post.ch/post-startseite/post-konzern/post-engagement/post-engagementklimaschutz/post-engagement-bahntransporte/log-download-kombiverkehr.pdf] last access: July 22, 2011.

Post (2011c): Wichtigste Zahlen 2010. URL: [http://www.post.ch/post-startseite/post-berichterstattung/post-berichterstattung-das-wichtigste/post-berichterstattung-kennzahlen/post-berichterstattung-vergleich.htm] last access: July 22, 2011.

Poundstone, W. (1985): The recursive universe. Chicago 1985.

Power, D. (2005): Supply chain management integration and implementation: a literature review. In: Supply Chain Management: An International Journal. 10(2005)4, pp. 252-263.

Premeaux, S. R. (2002): Motor Carrier Selection Criteria: Perceptual Differences between Shippers ad Motor Carriers. In: Transportation Journal. 42(2002)2, pp. 28-38.

Progtrans (2007): Kannibalisierungs-Effekt Wagenladungsverkehr – unbegleiteter Kombinierter Verkehr. Bundesamt für Verkehr (BAV). Bern 2007.

Punch, K. F. (2005): Introduction to Social Research: Quantitative and Qualitative Approaches. 2nd Edition. London 2005.

Qiu, R. G. / Tang, Y. / Xu, Q. (2006): Integration design of material flow management in an e-business manufacturing environment. In: Decision Support Systems. 42(2006)2, pp. 1104-1115.

Quak, H. J. / de Koster, M. B. M. (2009): Delivering Goods in Urban Areas: How to Deal with Urban Policy Restrictions and the Environment. In: Transportation Science. 43(2009)2, pp. 211-227.

Rabe, M. / Spieckermann, S. / Wenzel, S. (2008): Verifikation und Validierung für die Simulation in Produktion und Logistik. Berlin 2008.

Raffée, H. (1989): Gegenstand, Methoden und Konzepte der Betriebswirtschaftslehre. In: Bitz, M. et al.: Vahlens Kompendium der Betriebswirtschaftslehre. München 1989, pp. 3-46.

Railcare (2011): Modulare Transportkette von Tür zu Tür. URL: [http://railcare.ch/index.php?id=2.] last access: July 20, 2011.

Rajagopal, P. (2002): An innovation--diffusion view of implementation of enterprise resource planning (ERP) systems and development of a research model. In: Information & Management. 40(2002)2, pp. 87-114.

Ramanathan, U. / Muyldermans, L. (2011): Identifying the underlying structure of demand during promotions: A structural equation modelling approach. In: Expert Systems with Applications. 38(2011)5, pp. 5544-5552.

Ramdas, K. (2003): Managing product variety: an integrative review and research directions. In: Production and Operations Management Decision. 12(2003)1, pp. 79-101.

Ramirez-Marquez, J. E. / Rocco, C. M. (2008): All-terminal network reliability optimization via probabilistic solution discovery. In: Reliability Engineering & System Safety. 93(2008)11, pp. 1689-1697.

Randall, T. / Ulrich, K. (2001): Product variety, supply chain structure, and firm performance: analysis of the US bicycle industry. In: Management Science. 47(2001)3, pp. 1588-1604.

Ranganathan, C. / Dhaliwal, J. S. / Teo, T. S. H. (2004): Assimilation and diffusion of Web technologies in supply-chain management: an examination of key drivers and performance impacts. In: International Journal of Electronic Commerce. 9(2004)1, pp. 127-161.

Rapptrans (2005): Bewertung von Qualitätsmerkmalen im Güterverkehr. Bern 2005.

Reckenfelderbäumer, M. (1995): Marketing-Accounting im Dienstleistungsbereich. Konzeption eines prozesskostengestützten Instrumentariums. Wiesbaden 1995.

Reh, F. (2005): Gleisanschlüsse im Schienenverkehr. Köln 2005.

Reichhart, A. / Holweg, M. (2007): Creating the customer-responsive supply chain: a reconciliation of concepts. In: International Journal of Operations & Production Management. 27(2007)11, pp. 1144-1172.

Reiner, G. (2004): Supply Chain Performance Measurement with Customer Satisfaction and Uncertainties. In: Logistik Management - Prozesse, Systeme, Ausbildung. Heidelberg 2004, pp. 217-233.

Resch, B. (2009): Preismanagement im Kombinierten Verkehr - Theoretische Konzeption und erste empirische Ergebnisse. München 2009.

Richardson, G. P. (1991): Feedback Thought in Social Science and Systems Theory. Philadelphia, PA 1991.

Riedl, A. (2011): Die Freiheit ist nicht grenzenlos. In: DVZ. 56(2011)20.5.2011, pp. 46.

Riezebos, J. / Klingenberg, W. / Hicks, C. (2009): Lean Production and information technology: Connection or contradiction? In: Computers in Industry. 60(2009)4, pp. 237-247.

Rizzoli, A. E. / Fornara, N. / Gambardella, L. M. (2002): A simulation tool for combined rail/road transport in intermodal terminals. In: Mathematics and Computers in Simulation. 59(2002)1-3, pp. 57-71.

Robinson, S. (2008): Conceptual Modeling for Simulation Part I: Definition and Requirements. In: Journal of the Operational Research Society. 59 (2008)3, pp. 278-290.

Robinson, S. (2004): Simulation: The Practice of Model Development and Use. Wiley 2004.

Robinson, S. / Brooks, R. / Kotiadis, K. / Zee, D.-J. V. D. (2010): Conceptual Modeling for Discrete-Event Simulation. Lacaster 2010.

Röder, A. / Tibken, B. (2006): A methodology for modeling inter-company supply chains and for evaluating a method of integrated product and process documentation. In: European Journal of Operational Research. 169(2006)3, pp. 1010-1029.

Rodriguez-Rodriguez, R. / Gonzalez, P. P. / Leisten, R. (2011): From competitive to collaborative supply networks: A simulation study. In: Applied Mathematical Modelling. 35(2011)3, pp. 1054-1064.

Rosenzweig, E. D. / Roth, A. V. (2004): Towards a theory of competitive progression: evidence from high-tech manufacturing. In: Production & Operations Management. 13(2004)4, pp. 354-368.

Rosenzweig, E. D. / Roth, A. V. / Dean, J. W. (2003): The influence of an integration strategy on competitive capabilities and business performance: An exploratory study of consumer products manufacturers. In: Journal of Operations Management. 21(2003)4, pp. 437-456.

Rössl, D. (1990): Die Entwicklung eines Bezugsrahmens und seine Stellung im Forschungsprozess. In: Journal für Betriebswirtschaft. 40(1990)2, pp. 99-110.

Rostamzadeh, R. / Sofian, S. (2011): Prioritizing effective 7Ms to improve production systems performance using fuzzy AHP and fuzzy TOPSIS (case study). In: Expert Systems with Applications. 38(2011)5, pp. 5166-5177.

Rothfuss, F. (2005): Industrieökonomik und ihre Anwendung in der Betriebswirtschaft. In: Schwaiger: 2005, pp. 41-60.

Rudberg, M. / Olhager, J. (2003): Manufacturing networks and supply chains: an operations strategy perspective. In: Omega. 31(2003)1, pp. 29-39.

Rutner, S. M. / Gibson, B. J. / Williams, S. R. (2003): The impacts of the integrated logistics systems on electronic commerce and enterprise resource planning systems. In: Transportation Research Part E: Logistics and Transportation Review. 39(2003)2, pp. 83-93.

Sahin, F. / Robinson, E. P. J. (2005): Information sharing and coordination in maketo-order supply chains. In: Journal of Operations Management. 23(2005)6, pp. 579-598.

Sako, M. / Helper, S. (1998): Determinants of trust in supplier relations: Evidence from the automotive industry in Japan and the United States. In: Journal of Economic Behavior & Organization. 34(1998)pp. 387-417.

Salema, M. I. G. / Barbosa-Povoa, A. P. / Novais, A. Q. (2010): Simultaneous design and planning of supply chains with reverse flows: A generic modelling framework. In: European Journal of Operational Research. 203(2010)2, pp. 336-349.

Sames, G. / Bündenbender, W. (1997): Aachener PPS-Modell - Das morphologische Merkmalsschema. Sonderdruck, Forschungsinstitut für Rationalisierung. Aachen 1997.

Sanders, N. R. (2007): An empirical study of the impact of e-business technologies on organizational collaboration and performance. In: Journal of Operations Management. 25(2007)pp. 1332-1347.

Sanders, N. R. (2008): Pattern of information technology use: The impact on buyersuppler coordination and performance. In: Journal of Operations Management. 26(2008)3, pp. 349-367.

Sarac, A. / Absi, N. / Dauzère-Pérès, S. (2010): A literature review on the impact of RFID technologies on supply chain management. In: International Journal of Production Economics. 128(2010)1, pp. 77-95.

Sarantakos, S. (2004): Social Research. 4th Edition. Sourth Yarra 2004.

Sari, K. (2008): On the benefits of CPFR and VMI: A comparative simulation study. In: International Journal of Production Economics. 113(2008)2, pp. 575-586.

Sarimveis, H. / Patrinos, P. / Tarantilis, C. D. / Kiranoudis, C. T. (2008): Dynamic modeling and control of supply chain systems: A review. In: Computers & Operations Research. 35(2008)11, pp. 3530-3561.

Sarkis, J. / Zhu, Q. / Lai, K.-h. (2011): An organizational theoretic review of green supply chain management literature. In: International Journal of Production Economics. 130(2011)1, pp. 1-15.

Saunders, M. / Lewis, P. / Thornhill, A. (2007): Research Methods for Business Students. London 2007.

SBBCargo (2011): Produkt: Cargo Domino. URL: [www.sbbcargo.com/produkt_domino.pdf] last access: August 10, 2011. Scharkow, M. (2004): Simulation sozialer Netze in der Wissenskommunikation. URL: [http://underused.org/fileadmin/docs/simsocnets_02.pdf] last access: July 15, 2011.

Schary, P. B. / Skjøtt-Larsen, T. (2001): Managing the Global Supply Chain. 2. Edition. Kopenhagen 2001.

Scherer, A. / Beyer, R. (1998): Der Konfigurationsansatz im Strategischen Management - Rekonstruktion und Kritik. In: Die Betriebswirtschaft. 58(1998)3, pp. 332-347.

Schippers, W. A. J. (2001): An integrated approach to process control. In: International Journal of Production Economics. 69(2001)1, pp. 93-105.

Schmitz, J. / Platts, K. W. (2004): Roles of Supplier Performance Measurement: Indication from a Study in the Automotive Industry. In: Management Decision. 41(2004)8, pp. 711-721.

Scholz-Reiter, B. / De Beer, C. / Freitag, M. / Hamann, T. / Rekersbrink, H. / Tervo, J. T. (2008): Dynamik logistischer Systeme. In: Nyhuis, P.: Beiträge zu einer Theorie der Logistik. Berlin 2008, pp. 109-138.

Schönsleben, P. (2002): Integrales Logistikmanagement. Planung und Steuerung von umfassenden Geschäftsprozessen. Berlin et al. 2002.

Schönsleben, P. / Nienhaus, J. / Schnetzler, M. / Sennheiser, A. / Weidemann, M. (2003): SCM – Stand und Entwicklungstendenzen in Europa. In: Supply Chain Management. 3(2003)1, pp. 19-27.

Schorb, U. / Halsband, E. / Frank, A. (2007): Kontraktlogistik in der Automobilindustrie. In: Stölzle, W. et al.: Handbuch Kontraktlogistik: Management komplexer Logistikdienstleistungen. Weinheim 2007, pp. 619-637.

Schrank, R. (2002): Neukonzeption des Performance Measurements – Der GOPE Ansatz. Sternenfels 2002.

Schunk, D. (2000): Using simulation to analyze supply chain. Proceedings of the Winter Simulation Conference 2000.

Schwaninger, M. / Grösser, S. (2008): System Dynamics as Model-Based Theory Building. In: Systems Research and Behavioral Science. 25(2008)4, pp. 447-465.

SECO (2011): Freihandelsabkommen.Staatsekretariat für Wirtschaft (SECO). URL: [http://www.seco.admin.ch/themen/00513/00515/01330/index.html?lang=de] last access: July 10, 2011.

Seidelmann, C. (2010): 40 Jahre Kombinierter Verkehr Strasse-Schiene in Europa. 2010.

Seidelmann, C. (1997): Der Kombinierte Verkehr - ein Überblick. In: Bloech, J.Ihde, G.B.: Vahlens grosses Logistiklexikon. München 1997, pp. 431-434.

Seidl, J. (2002): Business Process Performance - Modellbezoge Beurteilung und Ansätze zur Optimierung. In: HMD - Praxis der Wirtschaftsinformatik 39(2002)227, pp. 37-35.

Sen, W. / Pokharel, S. / YuLei, W. (2004): Supply chain positioning strategy integration, evaluation, simulation, and optimization. In: Computers & Industrial Engineering. 46(2004)4, pp. 781-792.

Sennheiser, A. / Schnetzler, M. J. (2008): Wertorientiertes Supply Chain Management - Strategien zur Mehrung und Messung des Unternehmenswertes durch SCM. Berlin 2008.

Seuring, S. / Müller, M. (2008a): Core Issues in sustainable Supply Chain Management - a Delphi Study. In: Business Strategy and the Environment. 17(2008a)8, pp. 455-466.

Seuring, S. / Müller, M. (2008b): From a Literature Review to a conceptual Framework for Sustainable Supply Chain Management. In: Journal of Cleaner Production. 16(2008b)15, pp. 1699-1710.

Shabayek, A. A. / Yeung, W. W. (2002): A simulation for the Kwai Chung container terminal in Hong Kong. In: European Journal of Operational Research 1(2002)40, pp. 1-11.

Shang, K. (2009): Integration and Organizational Learning Capabilities in Third Party Logistics Providers. In: The Service Industries Journal. 29(2009)3, pp. 331-343.

Sharland, A. / Eltantawy, R. A. / Giunipero, L. C. (2003): The impact of cycle time on supplier selection and subsequent performance outcomes. In: Journal of Supply Chain Management. 39(2003)3, pp. 4-12.

Sharma, S. / Pablo, A. L. / Vredenburg, H. (1999): Corporate Environmental Responsiveness Strategies. In: Journal of Applied Behavioral Science. 35(1999)1, pp. 87.

Sheffi, Y. (2005): The resilient enterprise: overcoming vulnerability for competitive advantage. Cambridge 2005.

Sheffi, Y. (1990): Third Party Logistics: present and future Prospects. In: Journal of Business Logistics. 11(1990)2, pp. 27.

Sheu, J.-B. (2002): A sequential group decision-making approach to strategic planning for the development of commercial vehicle operations systems. In: Transportation Research Part A: Policy and Practice. 36(2002)4, pp. 335-350.

Shewchuck, P. (1998): Agile manufacturing: one size does not fit all. In: International Conference on Manufacturing Value Chains Troon Year

Shin, K. / Leem, C. S. (2002): A reference system for internet based inter-enterprise electronic commerce. In: Journal of Systems and Software. 60(2002)3, pp. 195-209.

Shin, T.-H. / Chin, S. / Yoon, S.-W. / Kwon, S.-W. (2011): A service-oriented integrated information framework for RFID/WSN-based intelligent construction supply chain management. Automation in Construction. In Press, Corrected Proof 2011.

Shinghal, N. / Fowkes, T. (2002): Freight mode choice and adaptive stated preferences. In: Transportation Research: Part E. 38(2002)5, pp. 367-378.

Shobrys, D. E. / White, D. C. (2002): Planning, scheduling and control systems: why cannot they work together. In: Computers & Chemical Engineering. 26(2002)2, pp. 149-160.

Siebel, L. (2002): Trends im Handel - ECR - SCM - CPFR, was macht die Praxis? Dortmund 2002.

Siegmann, J. (2011a): Lichtraumprofil und Fahrzeugbegrenzung im europäischen Schienenverkehr. URL: [http://fis.server.de/servlet/is/325031/] last access: July 16, 2011.

Siegmann, J. (2011b): Maximale Zuglängen im europäischen Schienenverkehr. URL: [http://fis.server.de/servlet/is/325137/] last access: July 15, 2011 (published October 10, 2010).

Siemieniuch, C. E. / Sinclair, M. A. (2002): On complexity, process ownership and organisational learning in manufacturing organisations, from an ergonomics perspective. In: Applied Ergonomics. 33(2002)5, pp. 449-462.

Sihn, W. / Hillbrand, C. / Meizer, F. / Leitner, R. / Prochazka, M. (2010): Development of a Simulation Model for Multimodal, Cross-Company Logistics Networks. In: Proceedings of Heinz Nixdorf Symposium. 2010, pp. 26-36.

Siller, U. (2011): Optimierung globaler Distributionsnetzwerke. Grundlagen, Methodik, praktische Anwendung. Wiesbaden 2011.

Simatupang, T. M. / Wright, A. C. / Sridharan, R. (2002): The knowledge of coordination for supply chain integration. In: Business Process Management Journal. 8(2002)3, pp. 289-308.

Simon, H. A. / Wegener, M. / Button, K. / Nijkamp, P. (2007): A Behavioral Model of Rational Choice. In: Planning History and Methodology. 2007, pp. 12-31.

Sinha, A. K. / Aditya, H. K. / Tiwari, M. K. / Chan, F. T. S. (2011): Agent oriented petroleum supply chain coordination: Co-evolutionary Particle Swarm Optimization based approach. In: Expert Systems with Applications. 38(2011)5, pp. 6132-6145.

Skjoett-Larsen, T. (2000): Third Party Logistics - from an interorganizational Point of View. In: International Journal of Physical Distribution & Logistics Management. 30(2000)2, pp. 112.

Skorna, A. / Bode, C. / Wagner, S. (2009): Technology-Enabled Risk Management along the transport logistics chain. In: Managing Risk and Security. Bern 2009, pp. 197-220.

Smith, S. / Yen, C.-C. (2010): Green product design through product modularization using atomic theory. In: Robotics and Computer-Integrated Manufacturing. 26(2010)6, pp. 790-798.

Soller, G. (2011): Wo man Glas im Griff hat. In: Verkehrsrundschau Spezial - Sonderheft für Güterverkehr Strasse-Schiene. 1(2011)1, pp. 18-19.

Soltani, R. / Sadjadi, S. J. (2010): Scheduling trucks in cross-docking systems: A robust meta-heuristics approach. In: Transportation Research Part E: Logistics and Transportation Review. 46(2010)5, pp. 650-666.

Son, J. Y. / Sheu, C. (2008): The impact of replenishment policy deviations in a decentralized supply chain. In: International Journal of Production Economics. 113(2008)2, pp. 785-804.

Sroufe, R. / Curkovic, S. (2008): An examination of ISO 9000:2000 and supply chain quality assurance. In: Journal of Operations Management. 26(2008)4, pp. 503-520.

Staber, U. (2000): Steuerung von Unternetzewerken: Organisationstheoretische Perspektiven und soziale Mechanismen. In: Sydow, J.Windeler, A.: Steuerung von Netzwerken - Konzepte und Praktiken. Opladen 2000, pp. 58-87.

Stadtler, H. (2005): Supply chain management and advanced planning--basics, overview and challenges. In: European Journal of Operational Research. 163(2005)3, pp. 575-588.

Stadtler, H. / Kilger, C. (2000): Supply Chain Management and Advanced Planning. Berlin 2000.

Staehle, W. (1976): Der situative Ansatz in der Betriebswirtschaftslehre. In: Ulrich, H.: Zum Praxisbezug der Betriebswirtschaftslehre in wissenschaftstheoretischer Sicht. Stuttgart, Bern 1976, pp. 33-50.

Staehle, W. (1994): Management. Eine verhaltenswissenschaftliche Perspektive. München 1994.

Stanley, L. L. / Wisner, J. D. (2001): Service quality along the supply chain: implications for purchasing. In: Journal of Operations Management. 19(2001)3, pp. 287-306.

Startton, R. / Warburton, R. D. H. (2003): The strategic integration of agile and lean supply. In: International Journal of Production Economics. 85(2003)2, pp. 183-198.

Stavrulaki, E. / Mark, D. (2010): Aligning products with supply chain processes and strategy. In: International Journal of Logistics Management, The. 21(2010)1, pp. 127-151.

Steenken, D. / Voß, S. / Stahlbock, R. (2004): Container terminal operation and operations research – a classification and literature review. In: OR Spectrum 1(2004)26, pp. 3-49.

Stefanovic, D. / Stefanovic, N. / Radenkovic, B. (2009): Supply network modelling and simulation methodology. In: Simulation Modelling Practice and Theory. 17(2009)4, pp. 743-766.

Stefansson, G. (2002): Business-to-business data sharing: A source for integration of supply chains. In: International Journal of Production Economics. 75(2002)1-2, pp. 135-146.

Stefansson, G. / Tilanus, B. (2000): Tracking&Tracing – Principles and Practice. In: International Journal of Technology Management 20(2000)3/4, pp. 252-271.

Sterman, J. D. (2001): System Dynamics Modeling: Tools for learning in a complex world. In: California Management Review. 43(2001)4, pp. 8-25.

Stevenson, W. B. / Greenberg, D. (2000): Agency and Social Networks: Strategies of Action in a social Structure of Position, Opposition, and Opportunity. In: Administrative Science Quarterly. 45(2000)4, pp. 651-666.

Stölzle, W. (1999): Industrial Relationships. München 1999.

Stölzle, W. / Bendul, J. (2010): Combined Subsequent Transports Driving Global Supply Chains. In: Festschrift zur Emeritierung von Hans-Christian Pfohl. In: Elbert, R.Schönberger, R.: Dimensionen der Logistik - Funktionen, Institutionen und Handlungsebenen Darmstadt 2010, pp. 155-190.

Stölzle, W. / Bendul, J. (2008): Wettbewerbssituation der Rollenden Landstrasse. St.Gallen 2008.

Stölzle, W. / Bendul, J. C. K., Nicole (2009): Intermodal Transport and Supply Chains: Moving the Global Economy. International Transport Forum. 2009.

Stölzle, W. / Heusler, K. F. / Karrer, M. (2001): Die Integration der Balanced Scorecard in das Supply Chain Management-Konzept. In: Logistik Management. 3(2001)2/3, pp. 73-85.

Stölzle, W. / Hofman, E. / Gebert, K. (2011): Logistikmarktstudie 2011. 2011.

Stölzle, W. / Karrer, M. (2004a): Finanzielle Performance von Logistikkooperationen – Anforderungen und Messkonzepte. In: Pfohl, H.-C.: Erfolgsfaktor Kooperation in der Logistik: Outsourcing – Beziehungsmanagement – Finanzielle Performance. Berlin 2004a, pp. 167-194.

Stölzle, W. / Karrer, M. (2002): Performance Management in der Supply Chain – Potentiale durch die Balanced Scorecard. In: BVL: Wissenschaftssymposium der BVL München 2002, pp. 57-81.

Stölzle, W. / Karrer, M. (2004b): Von der Unternehmens- zur Supply Chain Performance - ein konzeptioneller Beirtag für das Management von Supply Chains. In: Logistik Management - Prozesse, Systeme, Ausbildung. Heidelberg 2004b, pp. 167-194.

Stölzle, W. / Moll, K. / Müller, A. / Bendul, J. (2009): Anschlussgleise – Überprüfung von Zweckmässigkeit und Effizienz der staatlichen Förderung. Bern 2009.

Storey, J. / Emberson, C. / Reade, D. (2005): The barriers to customer responsive supply chain management. In: International Journal of Operations & Production Management. 25(2005)3/4, pp. 242-260.

Strassner, M. / Fleisch, E. (2005): Innovationspotential von RFID für das Supply-Chain-Management. In: Wirtschaftsinformatik. 47(2005)1, pp. 45-54.

Straube, F. / Pfohl, H.-C. / Bundesvereinigung Logistik (Deutschland) (2008): Trends und Strategien in der Logistik - globale Netzwerke im Wandel Umwelt, Sicherheit, Internationalisierung, Menschen. Hamburg 2008.

Stuhlmann, S. (2000): Kapazitätsgestaltung in Dienstleistungsunternehmen. Eine Analyse aus Sicht des Externen Faktors. Wiesbaden 2000.

Su, *Y.-f. / Yang*, *C. (2010):* Why are enterprise resource planning systems indispensable to supply chain management? In: European Journal of Operational Research. 203(2010)1, pp. 81-94.

Sundaram, D. / Zhou, W. / Piramuthu, S. / Pienaar, S. (2010): Knowledge-based RFID enabled Web Service architecture for supply chain management. In: Expert Systems with Applications. 37(2010)12, pp. 7937-7946.

Svensson, G. (2007): Aspects of sustainable Supply Chain Management (SSCM): conceptual Framework and empirical Example. In: Supply Chain Management: An International Journal. 12(2007)4, pp. 262-266.

Svensson, G. (2004): Key areas, causes and contingency planning of corporate vulnerability in supply chains: A qualitative approach. In: International Journal of Physical Distribution & Logistics Management. 34(2004)2, pp. 728-747.

Swafford, P. M. / Ghosh, S. / Murthy, N. (2006): A framework for assessing value chain agility. In: International Journal of Operations & Production Management. 26(2006)1/2, pp. 118-140.

Swaminathan, J. M. / Lee, H. L. / Graves, S. C. / Kok, A. G. d. (2003): Design for Postponement. In: Handbooks in Operations Research and Management Science. 2003, pp. 199-226.

Tajima, M. (2007): Strategic value of RFID in supply chain management. In: Journal of Purchasing and Supply Management. 13(2007)4, pp. 261-273.

Tan, K. C. (2001): A framework of supply chain management literature. In: European Journal of Purchasing and Supply Management. 7(2001)1, pp. 39-48.

Tang, C. / Tomlin, B. (2008): The power of flexibility for mitigating supply chain risks. In: International Journal of Production Economics. 116(2008)1, pp. 12-27.

Taplin, J., R. D. / Bent, D. / Aeron-Thomas, D. (2006): Developing a Sustainability Accounting Framework to inform strategic Business Decisions: a Case Study from the Chemicals Industry. In: Business Strategy and the Environment. 15(2006)5, pp. 347-360.

Taylor, J. C. / Jackson, G. C. (2000): Conflict, Power and Evolution in the Intermoal Transportation Inudstry's Channel of Distribution In: Transportation Jounal. 39(2000)3, pp. 5-17.

Terzi, S. / Cavalieri, S. (2004): Simulation in the supply chain context: a survey. In: Computers in Industry Volume. 53(2004)1, pp. 3-16.

Teunis, G. (2003): Engpassorientierte Werkstattsteuerung für teilautonome kooperative Fertigungen. Hannover 2003.

Thamhain, H. J. (2004): Linkages of project environment to performance: lessons for team leadership. In: International Journal of Project Management. 22(2004)7, pp. 533-544.

Thillmann, H. (2006): Beförderung von außergewöhnlichen Sendungen. In: Deine Bahn. 1(2006)8, pp. 30-37.

Thomas, D. J. / Griffin, P. M. (1996): Coordinated supply chain management. In: European Journal of Operational Research. 94(1996)1, pp. 1-15.

Thompson, A. A. / Strickland, A. J. / Gamble, J. E. (2005): Crafting and Executing Strategy:The Quest for Competitive Advantage - Concepts & Cases. New York 2005.

Thron, T. / Nagy, G. b. / Wassan, N. (2007): Evaluating alternative supply chain structures for perishable products. In: International Journal of Logistics Management. 18(2007)3, pp. 364-384.

Thron, T. / Nagy, G. b. / Wassan, N. (2006): The impact of various levels of collaborative engagement on global and individual supply chain performance. In: International Journal of Physical Distribution & Logistics Management. 36(2006)8, pp. 596-620.

Towill, D. R. (1996): Industrial dynamics modeling of supply chains. In: International Journal of Physical Distribution and Logistics Management. 2(1996)23, pp. 23-42.

Transwaggon (2011): The Transwaggon group. URL:

[http://www.transwaggon.com/Company.205.0.html] last access: July 15, 2011.

Trappey, C. V. / Lin, G. Y. P. / Trappey, A. J. C. / Liu, C. S. / Lee, W. T. (2010): Deriving industrial logistics hub reference models for manufacturing based economies. In: Expert Systems with Applications. 38(2010)2, pp. 1223-1232.

Trkman, P. / Štemberger, M. I. / Jaklic, J. / Groznik, A. (2007): Process approach to supply chain integration. In: Transportation Science. 12(2007)2, pp. 116-128.

Troitzsch, K. G. (2004): Validating Simulation Models. In: Networked Simulations and Simulated Networks. 1(2004)1, pp. 265–270.

Trost, D. G. (1999): Vernetzung im Güterverkehr: ökonomische Analyse von Zielen, Ansatzpunkten und Massnahmen zur Implementierung integrierter Verkehrssysteme unter Berücksichtigung logistischer Ansprüche verschiedener Marktsegmente. Hamburg 1999.

Trumpfheller, M. / Hofmann, E. (2004): Supply Chain Relationship Management – Beziehungsmanagement als konstitutives Element der Netzkompetenz in Supply Chains. In: Pfohl, H.-C.: Netzkompetenz in Supply Chains – Grundlagen und Umsetzung. Wiesbaden 2004, pp. 67-91.

Truscott, W. T. (2003): How does Six Sigma compare with other improvement initiatives? In: Six Sigma. Oxford 2003, pp. 38-76.

Tsou, J.-C. / Chen, W.-J. (2008): The impact of preventive activities on the economics of production systems: Modeling and application. In: Applied Mathematical Modelling. 32(2008)6, pp. 1056-1065.

UIC (2010): 2010 Report on Combined Transport in Europe. International Union of Railways. Paris 2010.

UIRR (2011): Charakteristika unbegleiteter Verkehr. URL:

[http://www.uirr.com/de/road-rail-ct/ct-offers/unaccompanied-services.html] last access: July 15, 2011

Ulrich, H. (1981): Die Betriebswirtschaftslehre als anwendungsorientierte Sozialwissenschaft. In: 1981, pp. 168-199.

Ulrich, H. (1995): Von der Betriebswirtschaftslehre zur systemorientierten Managementlehre. In: Wunderer, R.: Betriebswirtschaftslehre als Management- und Führungslehre. Stuttagart 1995, pp. 161-178.

Ulrich, H. / Krieg, W. (1974): St.Galler Management-Modell. Bern 1974.

Umweltbundesamt (2009): Daten zum Verkehr - Ausgabe 2009. Dessau-Rosslau 2009.

Ursel, S. / Hassmann, V. (2010): Best in Procurement. In: BME Sonderdruck - BIP 1(2010)3, pp. 2-4.

Vachon, S. / Klassen, R. D. (2006): Extending green practices across the supply chain: the impact of upstream and downstream integration. In: International Journal of Operations & Production Management. 26(2006)7, pp. 795-821.

van der Aalst, W. M. P. / Reijers, H. A. / Weijters, A. J. M. M. / van Dongen, B. F. / Alves de Medeiros, A. K. / Song, M. / Verbeek, H. M. W. (2007): Business process mining: An industrial application. In: Information Systems. 32(2007)5, pp. 713-732.

van der Aalst, W. M. P. / van Dongen, B. F. / Herbst, J. / Maruster, L. / Schimm, G. / Weijters, A. J. M. M. (2003): Workflow mining: A survey of issues and approaches. In: Data & Knowledge Engineering. 47(2003)2, pp. 237-267.

van Hoek, R. I. (2001): The rediscovery of postponement a literature review and directions for research. In: Journal of Operations Management. 19(2001)2, pp. 161-184.

Vanpoucke, E. / Boyer, K. K. / Vereecke, A. (2009): Supply chain information flow strategies: an empirical taxonomy. In: International Journal of Operations & Production Management. 29(2009)12, pp. 1213-1241.

VAP (2010): Wagenladungsverkehr als Haupterlös von SBB Cargo. Media release October 29, 2010. Zürich 2010.

Varma, V. A. / Reklaitis, G. V. / Blau, G. E. / Pekny, J. F. (2007): Enterprise-wide modeling & optimization--An overview of emerging research challenges and opportunities. In: Computers & Chemical Engineering. 31(2007)5-6, pp. 692-711.

VDA (1997): Begriffsbestimmung im Transport- und Lieferprozess der Automobilindustrie: VDA-Empfehlung 5002. Verband der Automobilindustrie (VDA). Frankfurt 1997.

VDV (2000): Schienengüterverkehr: Konzeption der DB Cargo AG ist eine Chance für alle Eisenbahnen. URL:

[http://www.vdv.de/medienservice/pressemitteilungen_entry.html?nd_ref=138] last access: July 15, 2011 (published: November 22, 2000).

Vekehr (2011): XPressNet: Eilige Sendungen fahren im Kombi-Verkehr. URL: [www.verkehr.co.at/XPressNet-Eilige-Sendungen-fahren-im-Kombi-Verkehr.211.0.html] last access: July 15, 2011 (published April 8, 2011)

Veliyath, R. / Srinivasan, T. C. (1995): Gestalt Approaches to Assessing Strategic Coalignment - A Conceptual Integration. In: British Journal of Management. 6(1995)3, pp. 205-219.

Verdouw, C. N. / Beulens, A. J. M. / Trienekens, J. H. / Wolfert, J. (2010): Process modelling in demand-driven supply chains: A reference model for the fruit industry. In: Computers and Electronics in Agriculture. 73(2010)2, pp. 174-187.

Vereecke, A. / Muylle, S. (2006): Performance improvement through supply chain collaboration in Europe. In: International Journal of Operations & Production Management. 26(2006)11, pp. 1176-1198.

Véronneau, S. / Roy, J. (2009): RFID benefits, costs, and possibilities: The economical analysis of RFID deployment in a cruise corporation global service supply chain. In: International Journal of Production Economics. 122(2009)2, pp. 692-702.

Verwijmeren, M. (2004): Software component architecture in supply chain management. In: Computers in Industry. 53(2004)2, pp. 165-178.

Vickery, S. K. / Jayaram, J. / Droge, C. / Calantone, R. (2003): The effects of an integrative supply chain strategy on customer service and financial performance: an analysis of direct versus indirect relationships. In: Journal of Operations Management. 21(2003)5, pp. 523-539.

Vitasek, K. L. / Manrodt, K. B. / Kelly, M. (2003): Solving the supply-demand mismatch. In: Supply Chain Management Review. 1(2003)1, pp. 58-64.

VÖV (2008): Manual Schienengüterverkehr Schwqeiz - Eine Einführung für die Praxis, Politk und Medien. Bern 2008.

Vrijhoef, R. / Koskela, L. (2000): The four roles of supply chain management in construction. In: European Journal of Purchasing & Supply Management. 6(2000)3-4, pp. 169-178.

VTG (2011): The Fascination Of The Rails. URL: [http://www.vtg-rail.com/] last access: July 15, 2011.

Wada, H. / Boonma, P. / Suzuki, J. (2011): Chronus: A Spatiotemporal Macroprogramming Language for Autonomic Wireless Sensor Networks. In: Autonomic Network Management Principles. Oxford 2011, pp. 167-203.

Wagner, S. / Güller, P. / Pillet, G. (1999): Europäischer Güterverkehr - Einbindung der Schweiz. Bern 1999.

Wahl, C. (1999): Bestandsmanagement in Distributionssystemen mit dezentraler Disposition. St.Gallen 1999.

Walker, H. / Di Sisto, L. / McBain, D. (2008): Drivers and barriers to environmental supply chain management practices: Lessons from the public and private sectors. In: Journal of Purchasing and Supply Management. 14(2008)1, pp. 69-85.

Wang, E. / Chou, H.-W. / Jiang, J. (2005): The impacts of charismatic leadership style on team cohesiveness and overall performance during ERP implementation. In: International Journal of Project Management. 23(2005)3, pp. 173-180.

Wang, H. / Zhang, H. (2010): A distributed and interactive system to integrated design and simulation for collaborative product development. In: Robotics and Computer-Integrated Manufacturing. 26(2010)6, pp. 778-789.

Wang, T. / Toktay, B. (2008): Inventory Management with Advance Demand Information and Flexible Delivery. In: 2008)pp.

Wang, T. F. / Cullinane, K. (2006): The efficiency of European container terminals and implications for supply chain management. In: Maritime Economics and Logistics. 8(2006)82, pp. 99-106.

Wang, X. / Meng, Q. (2010): The impact of landbridge on the market shares of Asian ports. In: Transportation Research Part E. 1(2010)47, pp. 190-203.

Webb, L. (1994): Green Purchasing: forging a new Link in the Supply Chain. In: PPI. 36(1994)6, pp. 52.

Weber, J. (2002): Logistikkostenrechnung. Kosten-, Leistungs- und Erlösinformationen zur erfolgsorientierten Steuerung der Logistik. Berlin 2002.

Weber, J. / Blum, H. (2001): Logistik-Controlling - Konzept und empirischer Stand. In: krp - Kostenrechnungspraxis: Zeitschrift für Controlling, Accounting & System-Anwendungen. 45(2001)5, pp. 275-282.

Weber, J. / Dehler, M. (2000): Entwicklungsstand der Logistik. In: Pfohl, H.-C.: Supply Chain Management: Logistik plus? Logistikkette - Marketingkette - Finanzkette. Berlin 2000, pp. 45-68.

Webster, M. (2002): Supply system structure, management and performance: a conceptual model. In: International Journal of Management Reviews. 4(2002)4, pp. 353-369.

Weisbuch, G. (2000): Environment and Institutions: a complex dynamical Systems Approach. In: Ecological Economics. 35(2000)3, pp. 381-391.

Welker, G. A. / van der Vaart, T. / Pieter van Donk, D. (2008): The influence of business conditions on supply chain information-sharing mechanisms: A study among supply chain links of SMEs. In: International Journal of Production Economics. 113(2008)2, pp. 706-720.

Wenzel, S. / Weiss, M. / Collisi-Böhmer, S. / Pitsch, H. / Rose, O. (2008): Qualitätskriterien für die Simulation in Produktion und Logistik - Planung und Durchführung von Simulationsstudien. Berlin, Heidelberg 2008.

Weserbahn (2011): Projekt Eisenbahnnetzwerk Bremen / Niedersachsen URL: [http://www.weserbahn.com/7101.php] last access July 15, 2011.

Weserems (2010): Wegen Wirtschaftskrise Engpässe im Güterverkehr. weserems.business. URL: [http://weser-ems.business-on.de/deutsche-bahngueterverkehr-wirtschaftskrise-gueterloks-bericht-_id10370.html] last access: July 15, 2011 (published April 6, 2011)

Whipple, J. M. / Lynch, D. F. / Nyaga, G. N. (2010): A buyer's perspective on collaborative versus transactional relationships. In: Industrial Marketing Management. 39(2010)3, pp. 507-518.

Wiendahl, H. P. (2009): Betriebsorganisation für Ingenieure. 7th updated Edition. München, Wien 2009.

Wildemann, H. (2004a): Betreibermodelle. 3rd Edition. München 2004a.

Wildemann, H. (2004b): Bewertung logistischer leistungen und Kosten in der Supply Chain - BILOG Forschungsbericht. München 2004b.

Wildemann, H. (2004,): Produktionsorganisation. In: Schreyögg, A.v. Werder, G.: Handwörterbuch Unternehmensführung und Organisation. 2004,, pp. 1182 – 1189.

Wildemann, H. (2000): Von Just-in-Time zu Supply Chain Management. In: Wildemann, H.: Supply Chain Mangement. München 2000, pp. 21-47.

Wildemann, H. / Niemeyer, A. (2011): Das Milkrun-Konzept: Logistikkostensenkung durch auslastungsorientierte Konsolidierungsplanung. URL:

[http://www.tcw.de/uploads/html/publikationen/aufsatz/files/Logistikkostensenkung_ Milkrun_Niemeyer.pdf] last access: July 19, 2011.

Wilke, H. (2001): Systemtheorie III: Steuerungstheorie. Grundzüge einer Theorie der Steuerung komplexer Sozialsysteme. Stuttgart et al. 2001.

Williamson, O. E. (1996): Economics and organization: a primer. In: California Management Review. 38 (1996)2, pp. 131-146.

Win, A. (2008): The value a 4PL provider can contribute to an organisation. In: International Journal of Physical Distribution & Logistics Management. 38(2008)9, pp. 674-684.
Windt, K. / Bendul, J. (2006): Zukunftsweisende Produktionssteuerungsverfahren für innovative Fertigungsverfahren und Maschinenkonzepte - Ein Szenario für das Jahr 2020. In: Industrie Management. 22(2006)1, pp. 31-34.

Windt, K. / Böse, F. / Philipp, T. (2008): Autonomy in production logistics: Identification, characterisation and application. In: International Journal Robotics and Computer-Integrated Manufacturing. 24(2008)4, pp. 572-578.

Wittenbrink, P. (2008): Modal-Split und CO2 – Umfrageergebnisse der BME-Umfrage 2008. 2008.

WKO (2011): Welche Fahrverbote gibt es in Österreich? Welche Auswirkungen haben Sie? Wirtschaftskammer Österreich. URL:

[http://portal.wko.at/wk/format_detail.wk?angid=1&stid=500872&dstid=686] last access: July 15, 2011

Wohlgemuth, O. (2002): Management netzwerkartiger Kooperationen - Instrumente für die unternehmensübergreifende Steuerung. Wiesbaden 2002.

Wolf, J. (2000): Der Gestaltansatz in der Management und Organisationslehre. Wiesbaden 2000.

Wolf, J. (2008): Organisation, Management, Unternehmensführung - Theorien, Praxisbeispiele und Kritik. Wiesbaden 2008.

Womack, J. / Jones, D. T. / Roos, D. (1990): The machine that changed the world. New York 1990.

Womack, J. P. / Jones, D. T. (1996): Lean Thinking: Banish Waste and Create Wealth in Your Corporation. New York 1996.

Wong, C. Y. / Arlbjørn, J. S. / Johansen, J. (2005): Supply chain management practices in toy supply chains. In: Supply Chain Management: An International Journal. 10(2005)5, pp. 367-378.

Wong, H. / Potter, A. / Naim, M. (2010): Evaluation of postponement in the soluble coffee supply chain: A case study. In: International Journal of Production Economics. 131(2010)1, pp. 355-364.

Wu, C. W. / Chen, C. L. (2006): An integrated structural model toward successful continuous improvement activity. In: Technovation. 26(2006)5-6, pp. 697-707.

Xia, D. / Chen, B. (2011): A comprehensive decision-making model for risk management of supply chain. In: Expert Systems with Applications. 38(2011)5, pp. 4957-4966.

Yang, H. / Yang, L. / Yang, S.-H. (2010): Hybrid Zigbee RFID sensor network for humanitarian logistics centre management. In: Journal of Network and Computer Applications. In Press, Corrected Proof(2010)pp.

Yang, T. / Wen, Y.-F. / Wang, F.-F. (2009): Evaluation of robustness of supply chain information-sharing strategies using a hybrid Taguchi and multiple criteria decision-making method. In: International Journal of Production Economics. In Press, Corrected Proof(2009)pp.

Yao, Y. / Dong, Y. / Dresner, M. (2010): Managing supply chain backorders under vendor managed inventory: An incentive approach and empirical analysis. In: European Journal of Operational Research. 203(2010)2, pp. 350-359.

Yeung, A. C. L. (2008): Strategic supply management, quality initiatives, and organizational performance. In: Journal of Operations Management. 26(2008)4, pp. 490-502.

Yeung, J. H. Y. / Selen, W. / Zhou, D. / Zhang, M. (2007): Postponement strategy from a supply chain perspective: cases from China. In: International Journal of Physical Distribution & Logistics Management. 37(2007)4, pp. 331-356.

Zachariassen, F. (2008): Negotiation strategies in supply chain management. In: International Journal of Physical Distribution & Logistics Management. 38(2008)10, pp. 764-781.

Zamponi, R. (2011): Firma bietet Kämpfer gegen Piraten - für 6500 Euro im Monat. Hamburger Abendblatt. URL:

[http://www.abendblatt.de/hamburg/article1890816/Firma-bietet-Kaempfer-gegen-Piraten-fuer-6500-Euro-im-Monat.htm] last access: July 15, 2011 (published May 16, 2011).

Zeigler, B. P. (1984): Theory of Modelling and Simulation. Melbourne 1984.

Zhu, K. / Kraemer, K. L. (2002): E-commerce metrics for net-enhanced organizations: assessing the value of e-commerce to firm performance in the manufacturing sector. In: Information Systems Research. 13(2002)3, pp. 275-295.

Zhu, Q. / Cote, R. P. (2004): Integrating green Supply Chain Management into an embryonic eco-industrial Development: a Case Study of the Guitang Group. In: Journal of Cleaner Production. 12(2004)8-10, pp. 1025-1035.

Zimmermann, K. / Seuring, S. (2009): Two case studies on developing, implementing and evaluating a balanced scorecard in distribution channel dyads. In: International Journal of Logistics Research and Applications. 12(2009)1, pp. 63-81.

Zobel, A. (2011): Und täglich lauert die Rampe. In: DVZ. 56(2011)10.5.2011, pp. 42.

Zoryk-Schalla, A. J. / Fransoo, J. C. / de Kok, T. G. (2004): Modeling the planning process in advanced planning systems. In: Information & Management. 42(2004)1, pp. 75-87.

Appendix

Appendix A	349
A-I List of Conducted Expert Interviews for Development of SCP Requirements Classification	
A-II List of Conducted Interviews for Simulation Study Development	
Appendix B –Simulation Model - Demand per Store and Weekday for Non-prioritized Goods	351
Appendix C – Activity Diagrams – Simulation Study	352
C-I - Sequence of Model Configuration	
CII - Sequence of Simulation Runs	
Appendix D –Simulation StudyData Pool (CD)	354

Appendix A

A-I List of Conducted Expert Interviews for the Development of the SCP Requirements Classification

surname	name	position	company main location		date	
Ackermann	Josef	CEO	Swissmill	Zurich	April 30, 2009	
Andenmatten	Walter	Head of Logistics	Manor	Zurich	November 11, 2009	
Arnold	Josef	CEO	Zuckerfabrik Aar- berg und Frauenfeld AG	Frauenfeld	April 30, 2009	
Beutler	Martin	Executive	Fenaco	Bern	July 2, 2009	
Buchs	Marius	Head of Logistics	Tetrapack	Lausanne	November 9, 2009	
Deutschmann	Rainer	Member of the Management Board	Manor	Zurich	April 28, 2009	
Flück	Marlene	Head of Logistics	Mibelle AG	Buchs	October 2, 2009	
Häfliger	Luigi	Member of the Executive Com- mittee	mber of the Fiege Schweiz Bern ecutive Com- tee		April 29, 2009	
Hartl	Dr. Rolf	CEO	Erdölvereinigung	Zurich	April 29, 2009	
Hungerbühler	Peter	Head of Logistics	KVA Thurgau	Weinfelden	November 2, 2009	
Imhof	Dr. Marcel	CEO	Schmolz & Bicken- bach AG	Dusseldorf	April 8, 2009	
Kerr	Barbara	Head of Logistics	Соор	Basel	October 28, 2009	
Krattiger	Moreno	CEO	Panlog	Emmenbrücke	April 8, 2009	
Lindenblat	Dr. Dieter	CEO	Thyssenkrupp Verkehr GmbH	Duisburg	May4, 2009	
Menzi	Urs	General Manager	Eternit AG	Niederurnen	June 29, 2009	
Metzger	Bernhard	Head of Logistics	Migros	Zurich	November 30, 2009	
Runge	Lars	Engineering	SF-Chem AG	Pratteln	June 30, 2009	
Schwendener	Edwin	Head of Logistics	Alcan Airex AG	Altenrhein	June 29, 2009	
Thaler	Hugo	Head of Logistics	Bischofszell Na- hrungsmittel	Bischofszell November 30, 20		
von Rotz	Josef	Head of Procure- ment	Swissmill	Zurich	November 6, 2009	
Walt	Ulrich	Head of Logistics	Holcim	Zurich	October 16, 2009	
Zingg	René	Business unit Manager Logistics	Geiser agro.com AG	Langenthal	July 1, 2009	
Zurflüh	Christian	Head of Opera- tions	Manor	Zurich	November 3, 2009	

surname	name	position	company	location	date	
Andermatten	Walter	Head of Import and Supply Chain	Manor	Basel	August 11, 2011	
Bürge	Rene	Head of Site Logistics	Swiss Post, Post Logistics	Frauenfeld	i May 16, 2011	
Deutschmann	Rainer	Member of the Management Board	Manor	Olten	April 5, 2011	
				Basel	August 11, 2011	
Ernst	Thomas	Head of System	Swiss Post, Post Logistics	Zurich	March 28, 2011	
Fiechter	Stefan	Head of Combi- nedTransort	Swiss Post, Post Logisitcs	Härkingen	April 26, 2011	
Koch	Oliver	Head of DC Hochdorf	Manor	Basel	August 11, 2011	
Schürmann	Markus	Head of DC Möhlin	Manor	Basel	August 11, 2011	
Zurflüh	Christian	Head of Opera- tions	Manor	Olten	April 14, 2011	
				Basel	April 5, 2011	
				Basel	August 11, 20011	

A-II List of Conducted Interviews for Simulation Study Development

		Monday	Tuesday	Wednesday	Thursday	Friday	generation time
DC HOV	СНА	3	3	4	3	3	11:00 AM
DC HOV	GEN	20	19	24	18	19	10:00 AM
DC HOV	LAU	2	4	4	3	4	10:20 AM
DC HOV	MON	2	0	6	3	2	12:00 AM
DC HOV	MOR	1	1	0	3	1	11:00 AM
DC HOV	NYO	1	0	0	3	1	11:00 AM
DC HOV	SIE	3	1	0	14	6	9:00 AM
DC HOV	SIO	3	1	0	12	5	9:00 AM
DC HOV	VEV	3	6	6	5	5	10:20 AM
DC HOV	VES	1	1	1	0	2	8:00 AM
DC HOV	YVE	1	2	0	0	4	3:10 PM
DC MOE	CHA	7	10	8	9	8	9:15 AM
DC MOE	GEN	22	28	25	28	23	7:45 AM
DC MOE	LAU	9	13	10	12	11	8:15 AM
DC MOE	MON	6	0	14	6	6	8:10 AM
DC MOE	MOR	6	6	0	14	6	9:15 AM
DC MOE	NYO	2	3	0	5	4	7:05 AM
DC MOE	SIE	2	3	0	4	6	7:20 AM
DC MOE	SIO	2	3	0	4	7	7:25 AM
DC MOE	VEV	7	7	9	8	8	8:20 AM
DC MOE	VES	2	4	3	0	5	4:30 PM
DC MOE	YVE	3	6	3	0	9	11:45 AM

Appendix B –Simulation Model - Demand per Store and Weekday for Non-prioritized Goods





C-I - Sequence of Model Configuration

Figure Appendix - 1: Activity diagram in UML notation for model configuration.





C--II - Sequence of Simulation Runs

Figure Appendix - 2: Activity diagram in UML notation for simulation run.

Appendix D – Simulation Study Data Pool (CD)

The attached CD includes all relevant input data, output data as well as the developed simulation model.